LINEARIZATIONS OF RATIONAL MATRICES

Autor: María del Carmen Quintana Ponce

Tesis depositada en cumplimiento parcial de los requisitos para el grado de Doctor en

Ingeniería Matemática

Universidad Carlos III de Madrid

Directores: Froilán Martínez Dopico Silvia Marcaida Bengoechea

Tutor: Froilán Martínez Dopico

Septiembre, 2021

Esta tesis se distribuye bajo licencia "Creative Commons Reconocimiento -No Comercial - Sin Obra Derivada"



AGRADECIMIENTOS

"Knowing is not enough, we must apply. Willing is not enough, we must do." – Bruce Lee

Una vez acabada mi Tesis Doctoral es imposible no pararse a pensar y agradecer a todas aquellas personas que han hecho posible, de una forma u otra, haber culminado este trabajo.

En primer lugar, quiero agradecer a mi director de tesis, Froilán Dopico, el haberme dado la oportunidad de ser su estudiante de doctorado. Gracias por todo lo que me has enseñado, por tu paciencia, rigurosidad, entusiasmo y por tus buenos consejos, no solo en el ámbito de la investigación. Es un placer trabajar a tu lado.

Gracias también a mi codirectora de tesis, Silvia Marcaida, que aún en la distancia siempre ha estado dispuesta a ayudarme. Gracias por haber mantenido un contacto continuado conmigo, por tu dedicación, apoyo y los recibimientos durante mis visitas a la UPV.

A Paul Van Dooren, que aunque no sea uno de mis directores de tesis lo considero como tal. Gracias por tu inmenso talento profesional y darme la oportunidad de colaborar contigo.

A Javier Pérez, gracias por compartir conmigo tus ideas y guiarme durante mi estancia en Montana, siempre con una sonrisa.

A mis compañeros de despacho en la UC3M, Javi, Luismi y Carla, a los que considero amigos. Gracias por todos los buenos momentos compartidos y por toda vuestra ayuda.

A todo el Departamento de Matemáticas de la UC3M y, en particular, al imprescindible Alberto Calvo.

A Vanni Noferini, por darme la oportunidad de empezar a trabajar con él en la Universidad de Aalto, Finlandia, durante la última etapa de mi tesis doctoral.

A mi profesor de matemáticas en el instituto, Genaro, por haber sido mi referente para dar el paso a adentrarme en el fascinante mundo de las matemáticas.

A mi directora de trabajos fin de grado y máster en la UMA, M^aÁngeles, por apoyarme y animarme a que continuara mi formación con la realización de un doctorado.

Gracias a mi familia, a mis padres y a mis hermanas, por todo vuestro cariño y por confiar en mí.

A Marco Antonio, gracias por aparecer en mi vida y apoyarme siempre.

A Rocío, con la que he convivido durante la mitad de esta andadura. Gracias por escucharme y tener siempre las palabras exactas.

A Tatiana, gracias por todos esos cafés y charlas que me hacen desconectar cuando voy a Málaga.

Y por último, pero el más importante de todos, gracias a mi hermano Jesús. Aunque ya no esté físicamente a mi lado, siempre ha sido y será el pilar más importante y fundamental en mi vida. Pensar en ti siempre me da fuerzas para seguir adelante y te tengo presente en mi mente todos los días de mi vida.

CONTENIDOS PUBLICADOS Y PRESENTADOS

Los resultados originales que aparecen en el Capítulo 3 de esta tesis han sido publicados en

1 – F. M. DOPICO, S. MARCAIDA, M. C. QUINTANA, Strong linearizations of rational matrices with polynomial part expressed in an orthogonal basis, Linear Algebra and its Applications, 570 (2019) 1–45. DOI: 10.1016/j.laa.2019.02.003

La autora de esta tesis es el autor principal de esta publicación. El material de esta fuente incluido en la tesis no está señalado por medios tipográficos ni referencias.

Los resultados originales que aparecen en los Capítulos 4 y 5 de esta tesis han sido publicados o enviados a publicar en

2 – F. M. DOPICO, S. MARCAIDA, M. C. QUINTANA, P. VAN DOOREN, Local linearizations of rational matrices with application to rational approximations of nonlinear eigenvalue problems, Linear Algebra and its Applications, 604 (2020) 441 – 475. DOI: 10.1016/j.laa.2020.07.004

3- F. M. DOPICO, S. MARCAIDA, M. C. QUINTANA, P. VAN DOOREN, *Block full rank linearizations of rational matrices*, submitted. URL: arXiv:2011.00955v1

La autora de esta tesis es el autor principal de estas publicaciones. El material de estas fuentes incluido en la tesis no está señalado por medios tipográficos ni referencias.

Los resultados originales que aparecen en el Capítulo 6 de esta tesis han sido enviados a publicar en

4 – J. PÉREZ, M. C. QUINTANA, *Linearizations of rational matrices from general representations*, submitted. URL: arXiv:2003.02934v1

La autora de esta tesis es el autor principal de esta publicación. El material de esta fuente incluido en la tesis no está señalado por medios tipográficos ni referencias.

Los resultados originales que aparecen en el Capítulo 7 de esta tesis han sido publicados en

5 – F. M. DOPICO, M. C. QUINTANA, P. VAN DOOREN, *Linear system matrices of rational transfer functions*, to appear in "Realization and Model Reduction of Dynamical Systems", A Festschrift to honor the 70th birthday of Thanos Antoulas", Springer-Verlag. URL: arXiv:1903.05016v2

La autora de esta tesis es el autor principal de esta publicación. El material de esta fuente incluido en la tesis no está señalado por medios tipográficos ni referencias.

Los resultados originales que aparecen en el Capítulo 8 de esta tesis se encuentran en avanzado estado de preparación en

6 – F. M. DOPICO, M. C. QUINTANA, P. VAN DOOREN, Strongly minimal self-conjugate linearizations for polynomial and rational matrices.

La autora de esta tesis es el autor principal de esta publicación. El material de esta fuente incluido en la tesis no está señalado por medios tipográficos ni referencias.

El Capítulo 9 está completamente incluido en la publicación

7 – F. M. DOPICO, M. C. QUINTANA, P. VAN DOOREN, Structural backward stability in rational eigenvalue problems solved via block Kronecker linearizations, submitted. URL: arXiv:2103.16395v1

La autora de esta tesis es el autor principal de esta publicación. El material de esta fuente incluido en la tesis no está señalado por medios tipográficos ni referencias. Este trabajo ha sido desarrollado en el Departamento de Matemáticas de la Universidad Carlos III de Madrid (UC3M) bajo la dirección del profesor Froilán Martínez Dopico y codirección de la profesora Silvia Marcaida Bengoechea. Se contó durante cuatro años con un contrato predoctoral FPI, referencia BES-2016-076744, asociado al proyecto ALGEBRA LINEAL NUMÉRICA ESTRUCTURADA PARA MATRI-CES CONSTANTES, POLINOMIALES Y RACIONALES, referencia MTM2015-65798-P, del Ministerio de Economía y Competitividad, y cuyo investigador principal fue Froilán Martínez Dopico. Asociado a este contrato, se contó con una ayuda para realizar parte de este trabajo durante dos estancias internacionales de investigación. La primera estancia de investigación se realizó del 30 de enero de 2019 hasta el 1 de marzo de 2019 en el Department of Mathematical Engineering, Université catholique de Louvain (Bélgica), bajo la supervisión del profesor Paul Van Dooren. La segunda estancia de investigación se realizó del 15 de septiembre de 2019 hasta el 19 de noviembre de 2019 en el Department of Mathematical Sciences, University of Montana (EEUU), bajo la supervisión del profesor Javier Pérez Álvaro. Dado que la entidad beneficiaria del contrato predoctoral es la UC3M mientras que el otro codirector de tesis, la profesora Silvia Marcaida Bengoechea, pertenece al Departamento de Matemáticas de la Universidad del País Vasco (UPV/EHU), el trabajo con la profesora Silvia Marcaida se reforzó mediante visitas a la UPV/EHU, financiadas por ayudas de la RED temática de Excelencia ALAMA (Algebra Lineal, Análisis Matricial y Aplicaciones) asociadas al los proyectos MTM2015-68805-REDT y MTM2017-90682-REDT.

Abstract

This PhD thesis belongs to the area of Numerical Linear Algebra. Specifically, to the numerical solution of the Rational Eigenvalue Problem (REP). This is a type of eigenvalue problem associated with rational matrices, which are matrices whose entries are rational functions. REPs appear directly from applications or as approximations to arbitrary Nonlinear Eigenvalue Problems (NLEPs). Rational matrices also appear in linear systems and control theory, among other applications. Nowadays, a competitive method for solving REPs is via linearization. This is due to the fact that there exist backward stable and efficient algorithms to solve the linearized problem, which allows to recover the information of the original rational problem. In particular, linearizations transform the REP into a generalized eigenvalue problem in such a way that the pole and zero information of the corresponding rational matrix is preserved. To recover the pole and zero information of rational matrices, it is fundamental the notion of polynomial system matrix, introduced by Rosenbrock in 1970, and the fact that rational matrices can always be seen as transfer functions of polynomial system matrices.

This thesis addresses different topics regarding the problem of linearizing REPs. On the one hand, one of the main objectives has been to develop a theory of linearizations of rational matrices to study the properties of the linearizations that have appeared so far in the literature in a general framework. For this purpose, a definition of local linearization of rational matrix is introduced, by developing as starting point the extension of Rosenbrock's minimal polynomial system matrices to a local scenario. This new theory of local linearizations captures and explains rigorously the properties of all the different linearizations that have been used from the 1970's for computing zeros, poles and eigenvalues of rational matrices. In particular, this theory has been applied to a number of pencils that have appeared in some influential papers on solving numerically NLEPs through rational approximation.

On the other hand, the work has focused on the construction of linearizations of rational matrices taking into account different aspects. In some cases, we focus on preserving particular structures of the corresponding rational matrix in the linearization. The structures considered are symmetric (Hermitian), skew-symmetric (skew-Hermitian), among others. In other cases, we focus on the direct construction of the linearizations from the original representation of the rational matrix. The representations considered are rational matrices expressed as the sum of their polynomial and strictly proper parts, rational matrices written as general transfer function matrices, and rational matrices expressed by their Laurent expansion around the point at infinity. In addition, we describe the recovery rules of the information of the original rational matrix from the information of the new linearizations, including in some cases not just the zero and pole information but also the information about the minimal indices. Finally, in this dissertation we tackle one of the most important open problems related to linearizations of rational matrices. That is the analysis of the backward stability for solving REPs by running a backward stable algorithm on a linearization. On this subject, a global backward error analysis has been developed by considering the linearizations in the family of "block Kronecker linearizations". An analysis of this type had not been developed before in the literature.

Contents

A	ckno	wledgements	i				
P۱	Publicactions						
A	Abstract						
1	Intr	oduction, motivation and summary of main results	1				
	1.1	Linearizations of rational matrices	3 5				
	1.2	Structure and organization of the thesis	7				
	1.3	Notation	9				
2	Pre	liminaries on rational matrices	11				
	2.1	Zeros and poles: Local Smith–McMillan form	12				
	2.2	The rational eigenvalue problem	15				
	2.3	Minimal bases and minimal indices	16				
	2.4	Polynomial system matrices	18				
		trices	21				
	2.5	Definitions of strong linearizations of polynomial and rational matri-					
		ces in the literature	23				
		2.5.1 Strong linearizations of polynomial matrices	23				
		2.5.2 Strong linearizations of rational matrices	25				
3	Stre	ong linearizations of rational matrices with polynomial part ex-					
	pres	ssed in an orthogonal basis	29				
	3.1	Some preliminaries	29				
	3.2	\mathbb{M}_1 -strong linearizations	32				
	3.3	\mathbb{M}_2 -strong linearizations	36				
	3.4	Recovering eigenvectors from \mathbb{M}_1 - and \mathbb{M}_2 -strong linearizations of ra-	20				
		uonai matrices	38 20				
		3.4.2 Figenvectors from M_strong linearizations	58 44				
		J.4.2 Engenvectors from 1012-Strong intentizations	44				

	0.0	Symmetric realizations of symmetric rational matrices	40
	3.6	Symmetric strong linearizations for symmetric rational matrices	50
	3.7	Hermitian strong linearizations for Hermitian rational matrices	58
	3.8	Strong linearizations of rational matrices with polynomial part ex-	
		pressed in other polynomial bases	60
4	Loc	cal linearizations of rational matrices	63
	4.1	Polynomial system matrices minimal in a set and at infinity	63
		4.1.1 Polynomial system matrices minimal in a set	64
		4.1.2 Polynomial system matrices minimal at infinity	67
	4.2	Local linearizations of rational matrices	70
		4.2.1 Linearizations in a set	70
		4.2.2 Linearizations at infinity and in sets containing infinity	73
	4.3	Block full rank pencils	76
	4.4	Block full rank linearizations of rational matrices	82
		4.4.1 Block full rank linearizations at finite points	82
		4.4.2 Block full rank linearizations at infinity	86
		4.4.3 Strong block minimal bases linearizations as block full rank	
		linearizations	88
		4.4.4 Recovery of eigenvectors	89
5	App	plication of the local linearization theory to linearizations of	
5	App rati	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems	93
5	App rati 5.1	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94
5	App rati 5.1	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94 101
5	App rati 5.1 5.2	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94 101 105
5	App rati 5.1 5.2	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94 101 105 113
5	App rati 5.1 5.2	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94 101 105 113
5	Apprati 5.1 5.2 Line	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure Linearizations of arbitrary rational transfer functions	93 94 101 105 113 L15
5	App rati 5.1 5.2 Line 6.1	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure Linearizations of arbitrary rational transfer functions 1 Linearizations in a set	93 94 101 105 113 L15 116
5	App rati 5.1 5.2 Line 6.1 6.2	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94 101 105 113 115 116 120
5	App rati 5.1 5.2 Line 6.1 6.2 6.3	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure earizations of arbitrary rational transfer functions Linearizations in a set Linearizations at infinity	93 94 101 105 113 115 116 120 1222
5	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure earizations of arbitrary rational transfer functions Linearizations in a set Linearizations at infinity An illustrative example Recovery of eigenvectors, minimal bases and minimal indices	93 94 101 105 113 115 116 120 122 123
5	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure earizations of arbitrary rational transfer functions Linearizations in a set Linearizations at infinity An illustrative example Recovery of eigenvectors, minimal bases and minimal indices	93 94 101 105 113 115 116 120 122 123 123
5	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94 101 105 113 115 116 120 122 123 123 123
5	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils	93 94 101 105 113 115 116 120 122 123 125 130
5	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils Solution (State) 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure Linearizations of arbitrary rational transfer functions Linearizations in a set Linearizations at infinity An illustrative example Recovery of eigenvectors, minimal bases and minimal indices 6.4.1 One-sided factorizations 6.4.3 Recovery of eigenvectors Application to scalar rational equations	93 94 101 105 113 116 120 122 123 123 125 130 132
5 6 7	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4 6.5 Str	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure earizations of arbitrary rational transfer functions 1 Linearizations in a set Linearizations at infinity An illustrative example Recovery of eigenvectors, minimal bases and minimal indices 6.4.1 One-sided factorizations 6.4.3 Recovery of eigenvectors Application to scalar rational equations	93 94 101 105 113 113 120 122 123 123 125 130 132
5 6	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4 6.5 Stree 7.1	plication of the local linearization theory to linearizations of nonlinear eigenvalue problems NLEIGS pencils NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure earizations of arbitrary rational transfer functions 1 Linearizations in a set Linearizations at infinity An illustrative example Recovery of eigenvectors, minimal bases and minimal indices 6.4.1 One-sided factorizations 6.4.2 Recovery of minimal bases and minimal indices 6.4.3 Recovery of eigenvectors Application to scalar rational equations 1 application to scalar rational equations	93 94 101 105 113 116 120 122 123 123 123 123 130 132 137
5 6	App rati 5.1 5.2 Lind 6.1 6.2 6.3 6.4 6.5 Stro 7.1 7.2	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure earizations of arbitrary rational transfer functions 1 Linearizations in a set Linearizations at infinity An illustrative example Recovery of eigenvectors, minimal bases and minimal indices 6.4.1 One-sided factorizations 6.4.2 Recovery of minimal bases and minimal indices 6.4.3 Recovery of eigenvectors Application to scalar rational equations Application to scalar rational equations Some preliminaries Strong irreducibility and minimality	93 94 101 105 113 115 116 120 122 123 125 130 132 135 137
5 6	App rati 5.1 5.2 Line 6.1 6.2 6.3 6.4 6.5 Stro 7.1 7.2	plication of the local linearization theory to linearizations of ional approximations of nonlinear eigenvalue problems NLEIGS pencils 5.1.1 The NLEIGS low rank problem Linearizations for AAA rational approximations of NLEPs 5.2.1 Low-rank structure earizations of arbitrary rational transfer functions Linearizations in a set Linearizations at infinity An illustrative example Recovery of eigenvectors, minimal bases and minimal indices 6.4.1 One-sided factorizations 6.4.2 Recovery of minimal bases and minimal indices 6.4.3 Recovery of eigenvectors Application to scalar rational equations Some preliminaries Strong irreducibility and minimality	93 94 101 105 113 115 120 122 123 123 123 125 130 132 137 137

		7.2.2 Strongly minimal linearizations and their relation with other classes of linearizations	ર
	73	Reduction to a strongly minimal linear system matrix	5
	7.0 7.4	Computational aspects	0
	7.5	Numerical results 15	0
	1.0		U
8	Con	structing strongly minimal linearizations for rational matrices	۲
	0.1	Laurent expansions: preserving structures 15 For arbitrary polynomial matrices 15	о С
	0.1	For arbitrary polynomial matrices	0 0
	0.2 8 3	For self-conjugate polynomial matrices	2
	0.0 8 /	For solf conjugate strictly proper rational matrices	7
	8.5	For arbitrary and self-conjugate rational matrices	1 0
	8.6	Algorithmic aspects	J 1
	0.0		T
9	Stru	ctural backward stability in rational eigenvalue problems solved	•
	via	olock Kronecker linearizations	3
	9.1	Some preliminaries $\dots \dots \dots$	4
	0.0	9.1.1 Norms $\dots \dots \dots$) 0
	9.2	Generalized Sylvester equations	8
	9.3	Singular value bounds	0.0
	9.4	Restoring the structure of the linearization after perturbations \dots 18 0.4.1 Step 1. Destering the block anti-triangular structure 18	ა ი
		9.4.1 Step 1. Restoring the Kronecker block $K(\lambda) = K(\lambda)$ and the	0
		9.4.2 Step 2. Restoring the Kronecker blocks $\Lambda_1(\lambda)$, $\Lambda_2(\lambda)$ and the identity I	2
		$0.4.3 \text{Stop 3: Restoring the constant } B \text{ and } C \text{ matrices} \qquad 10$	5
		9.4.5 Step 5. Restoring the constant <i>D</i> and <i>C</i> matrices $\dots \dots \dots$ 19 0.4.4 Putting it all together 10	7
	95	Sufficient conditions for structural backward stability 20	1
	9.5	Bestoring the structure when the polynomial part of the rational ma-	T
	5.0	trix is linear 20	4
	97	Scaling for obtaining structural backward stability 20	5
	9.8	Numerical experiments	7
	0.0		•
10	Con	clusions, publications and open problems 21	3
	10.1	Conclusions and original results	3
	10.2	Publications	8
	10.3	Contributions to conferences	9
	10.4	Open problems and future research	0
\mathbf{A}	Pro	of of Theorem 6.4.9 22	3
В	Pro	of of Theorem 6.4.11 22	7

C Auxiliary result for Lemma 9.3.4	233
Bibliography	235

Chapter 1

Introduction, motivation and summary of main results

The main objects of study in this PhD thesis are rational matrices. A rational matrix $R(\lambda)$ is a matrix whose entries are quotients of polynomials in the scalar variable λ , i.e., rational functions. Rational matrices have received a lot of attention since the 1950s, as a consequence of their fundamental role in linear systems and control theory [67, 68]. Classical references on rational matrices and their applications to these areas are [55, 78].

Rational matrices can have poles and zeros and have rational right and left null spaces, which can be trivial, i.e., equal to {0}. Via the notion of the Smith-McMillan form, one can associate partial multiplicities to the poles and zeros, and via the notion of minimal polynomial bases for rational vector spaces, one can associate the so called right and left minimal indices to the right and left null spaces, which exist only when the rational matrix is singular, i.e., rectangular or square with identically zero determinant. All these quantities are among the most relevant structural data of a rational matrix [55, 68].

Many classic problems in linear systems and control theory can be posed in terms of rational matrices [55, 78, 90] and are related to the computation of their zeros and poles [84]. For that, it is fundamental the key concept introduced by Rosenbrock [78] in 1970 of polynomial system matrices of rational matrices. This notion allows us, among other things, to include simultaneously all the information about the zeros and the poles of a rational matrix into a polynomial matrix. More precisely, polynomial system matrices $P(\lambda)$ are block partitioned polynomial matrices of the form

$$P(\lambda) := \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix},$$

where $A(\lambda)$ is assumed to be regular. Rosenbrock showed that the finite pole and zero structure of its transfer function matrix

$$R(\lambda) := D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$$

can be retrieved from the polynomial matrices $A(\lambda)$ and $P(\lambda)$, respectively, provided $P(\lambda)$ is minimal, meaning that the polynomial matrices

$$\begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix}$$
 and $\begin{bmatrix} A(\lambda) \\ -C(\lambda) \end{bmatrix}$,

have, respectively, full row and column rank for all finite λ .

Currently, the computation of the zeros of rational matrices is also playing a fundamental role in the very active area of Nonlinear Eigenvalue Problems (NLEPs) [46], either because they appear directly in rational eigenvalue problems (REPs) modeling real-life problems [69] or because other NLEPs are approximated by REPs [46, 47, 60, 61, 35, 80]. Given a rational matrix $R(\lambda)$, the REP consists of finding scalars λ_0 such that λ_0 is not a pole of $R(\lambda)$, i.e., $R(\lambda_0)$ has finite entries, and that there exist nonzero constant vectors x and y satisfying

$$R(\lambda_0)x = 0$$
 and $y^T R(\lambda_0) = 0$,

under the assumption that $R(\lambda)$ is regular, i.e., $R(\lambda)$ is square and its determinant is not identically equal to zero. The scalar λ_0 is said to be an eigenvalue of $R(\lambda)$ and the vectors x and y^T are called, respectively, right and left eigenvectors associated with λ_0 . A non-regular rational matrix $R(\lambda)$ is also called singular. In general, regardless of whether $R(\lambda)$ is regular or singular, a scalar λ_0 is said to be an eigenvalue of $R(\lambda)$ if λ_0 is not a pole of $R(\lambda)$ and

$$\operatorname{rank} R(\lambda_0) < \operatorname{nrank} R(\lambda),$$

where nrank $R(\lambda)$ denotes the normal rank of $R(\lambda)$, i.e., the rank of $R(\lambda)$ over the field of rational functions in λ . Then, the problem of finding the eigenvalues of a rational matrix can also be seen as the problem of finding the zeros of $R(\lambda)$ that are not poles. If λ_0 is a pole of $R(\lambda)$ and there exists a polynomial vector $v(\lambda)$ such that $v(\lambda_0) \neq 0$ and that $\lim_{\lambda \to \lambda_0} R(\lambda)v(\lambda) = 0$ then λ_0 is said to be an eigenpole of $R(\lambda)$ [2]. A couple of examples of REPs and associated rational matrices are:

• The loaded elastic string problem in [13], whose corresponding rational matrix is of the form:

$$R(\lambda) = A - \lambda B + \frac{\lambda}{\lambda - \sigma} E,$$

where $A, B \in \mathbb{R}^{n \times n}$ are symmetric tridiagonal matrices, E has only one nonzero entry in (n, n) position and n is large.

• The damped vibration of a viscoelastic structure problem in [69], whose corresponding rational matrix is of the form:

$$R(\lambda) = \lambda^2 M + K - \sum_{i=1}^k \frac{1}{1 + b_i \lambda} G_i,$$

where $M, K \in \mathbb{R}^{n \times n}$ are positive definite and n is large.

1.1 Linearizations of rational matrices

In the 1970s the first numerical algorithms for computing the structural data of rational matrices were developed, see [84] and the references therein. The most reliable algorithms were based on constructing a matrix pencil, i.e., a matrix polynomial of degree 1, containing the information about the structural data of the considered rational matrix [84, 92]. These pencils are among the first examples of linearizations of rational matrices and are, in fact, particular instances of minimal polynomial system matrices. Then, backward stable algorithms developed also in the 1970s, for computing the eigenvalues and/or other structural data of general pencils [71, 83], were applied to these matrix pencils. Nowadays, given a matrix pencil $L(\lambda)$ linearizing a rational matrix, one can apply to $L(\lambda)$ the backward stable eigenvalue algorithms developed in [71, 83] for problems of moderate size, or Krylov methods adapted to the structure of $L(\lambda)$ in the large-scale setting [24, 47].

As we explained in the previous paragraph, the approach of constructing a linear polynomial matrix containing information about the structural data of rational matrices was first introduced in the 1970s. However, a formal definition of linearization of rational matrices was not given at that time. The term linearization of rational matrix was also used in the reference [79] and in works on NLEPs [47, 60, 35] without referring to a formal definition of linearization. A first formal definition of linearization of a rational matrix was proposed in [2]. Then, a different definition was introduced in [6], together with the first formal definition of strong linearization, i.e., a pencil that allows to recover both the finite and infinite pole and zero structure of $R(\lambda)$. However, the pencils considered for linearizing NLEPs do not satisfy the definitions of linearizations given in [2, 6]. This was our motivation to develop a more general theory of linearizations of rational matrices in Chapter 4.

In addition to formal definitions, some works on linearizations of rational matrices have introduced families of strong linearizations that are constructed from the fact that any rational matrix $R(\lambda)$ can be written as

$$R(\lambda) = D(\lambda) + C(I_n\lambda - A)^{-1}B$$
(1.1)

where $D(\lambda)$ is a polynomial matrix, called the polynomial part of $R(\lambda)$, and $C(I_n\lambda - A)^{-1}B$ is a minimal state-space realization [48] of the strictly proper part of $R(\lambda)$ (see (2.1)). Thanks to this property, strong linearizations of rational matrices are constructed from strong linearizations of the polynomial part $D(\lambda)$ combined with minimal state-space realizations of the strictly proper part. In addition, the study of the recovery properties from these families of linearizations has received considerable attention. References in these lines include [2, 3, 4, 6, 8, 17, 18, 19, 27].

Among the new classes of strong linearizations, we mention the family of "strong block minimal bases linearizations" of rational matrices introduced in [6, Theorem 5.11], as a wide family of strong linearizations constructed by considering "strong block minimal bases pencils" associated to their polynomial parts, see [26]. They include as particular cases the Fiedler-like linearizations (modulo permutations) [8, Section 8] and are valid for general rectangular rational matrices. Chapter 9 is devoted to analyse the backward stability when running a backward stable algorithm for computing the eigenvalues of a particular type of the strong block minimal bases linearization of rational matrices developed in [6], which are called "block Kronecker linearizations". For that, we generalize in a nontrivial way the analysis of backward errors of the "block Kronecker linearizations" of polynomial matrices developed in [26] in combination with the theory of strong linearizations for rational matrices in [6].

The question whether or not other strong linearizations of rational matrices can be constructed based on other kinds of strong linearizations of the polynomial parts arises naturally. For answering this question, we construct in Chapter 3 strong linearizations of a rational matrix by using strong linearizations of its polynomial part $D(\lambda)$ that belong to other important family of strong linearizations of polynomial matrices, i.e., the so-called vector spaces of linearizations, originally introduced in [63], further studied in [21, 72], and extended in [36]. In particular, we consider strong linearizations of $D(\lambda)$ that belong to the ansatz spaces $\mathbb{M}_1(D)$ or $\mathbb{M}_2(D)$, developed by Faßbender and Saltenberger in [36], where polynomial matrices are expressed in terms of polynomial bases other than the monomial basis. Another motivation of the results in Chapter 3 is that, in order to compute the eigenvalues of polynomial matrices from linearizations, the work [57] shows that, for polynomial matrices of large degree, the use of the monomial basis to express the matrix leads to numerical instabilities due to the ill-conditioning of the eigenvalues in certain situations. According to the algorithms in [24, 79, 81], it is expected that this instability appears also while computing eigenvalues of REPs when the polynomial part of the rational matrix has large degree and is expressed in terms of the monomial basis. For that reason, it is of interest to consider rational matrices with polynomial parts expressed in other bases as the Chebyshev basis. As a consequence of the results in Chapter 3, we can conclude that the combination of the results in Chapter 3 and those in [6] allows us to construct very easily infinitely many strong linearizations of rational matrices via the following three-step strategy: (1) express the rational matrix as the sum of its polynomial and strictly proper parts; (2) construct any of the strong linearizations of the polynomial part known so far; and (3) combine adequately that strong linearization with a minimal state-space realization of the strictly proper part.

Despite the intense activity described in the previous paragraph, there are pencils that have been used in influential references as [47, 60] for solving numerically REPs that approximate NLEPs which do not satisfy the definitions of linearization of rational matrices given in [2, 6]. The reason is that these definitions focus on pencils that allow to recover the complete pole and zero structure of rational matrices, while in [47, 60] only the eigenvalue information in a certain subset of the complex plane is necessary. This was our motivation to the development in Chapter 4 of a new theory of linearizations of rational matrices in a local sense. These linearizations are pencils that preserve the structure of zeros and poles of the corresponding rational matrix in a particular subset of the underlying field, in the whole underlying field and/or at infinity. Apart from a new definition, a specific family of local linearizations of rational matrices is also introduced in Chapter 4, that are called block full rank linearizations, as a template that covers many of the pencils, available in the literature, for linearizing rational matrices. In Chapter 5 we study the properties of the linearizations for rational approximations of NLEPs in [47, 60] by using the theory in Chapter 4.

As we explained, there exist different methods for constructing linearizations of rational matrices when the corresponding rational matrix is expressed as in (1.1). Furthermore, if the rational matrix is not in the form (1.1), there exist procedures for obtaining such a representation [55, 78, 90]. However, these procedures are not simple, and may introduce errors that were not present in the original problem. Motivated by this fact, we construct in Chapter 6 linearizations for rational matrices from more general representations. In particular, we will show how to construct linearizations of rational matrices that are written in the general form

$$R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda), \qquad (1.2)$$

where $A(\lambda)$, $B(\lambda)$, $C(\lambda)$ and $D(\lambda)$ are polynomial matrices, possibly non linear and possibly expressed in different bases. Representations of the form (1.2) arise naturally, for example, when solving REPs of the form

$$R(\lambda)x = \left(P(\lambda) + \sum_{i=1}^{m} \frac{n_i(\lambda)}{d_i(\lambda)} (A_i\lambda - B_i)\right)x = 0,$$

where $P(\lambda)$ is a polynomial matrix, $\frac{n_i(\lambda)}{d_i(\lambda)}$ are scalar rational functions, and A_i and B_i are constant matrices.

1.1.1 Strongly minimal linearizations: preserving structures

In Chapter 7 we consider a particular type of the strong linearizations for rational matrices defined in Chapter 4, which we called strongly minimal linearizations. Such strongly minimal linearizations of a rational matrix $R(\lambda)$ are linear polynomial system matrices of the form

$$L(\lambda) := \begin{bmatrix} \lambda A_1 - A_0 & \lambda B_1 - B_0 \\ -\lambda C_1 + C_0 & \lambda D_1 - D_0 \end{bmatrix},$$

where $R(\lambda)$ is the transfer function matrix of $L(\lambda)$ and such that the pencils

$$\begin{bmatrix} \lambda A_1 - A_0 & \lambda B_1 - B_0 \end{bmatrix}$$
 and $\begin{bmatrix} \lambda A_1 - A_0 \\ -\lambda C_1 + C_0 \end{bmatrix}$

6

have no finite or infinite eigenvalues. We will see that the strong minimality conditions imply the strong irreducibility conditions in [91], and the former are easier to test. In addition, we will also show that when the strong minimality conditions are not satisfied, we can reduce the system matrix to one where they are satisfied without modifying the corresponding transfer function matrix $R(\lambda)$.

One important property of strongly minimal linearizations is that they can preserve many different structures of the original rational matrix without imposing any restriction on such a matrix, as we explain in Chapter 8. This result is in stark contrast with previous results existing in the literature on linearizations that preserve structures, which impose conditions on the rational matrix as a consequence of using other definitions of linearizations. This is one of the most important results in this thesis, and for proving it, we will use some ideas developed in [27] (Chapter 3), where we also construct structured linearizations for symmetric and Hermitian rational matrices. In contrast, in Chapter 8 we use the notion of strongly minimal linearization instead of the notion of strong linearization for rational matrices in [6].

It is known that structured polynomial and rational matrices have symmetries in their spectra [40, 51, 59, 64], and these spectral symmetries reflect specific physical properties, as they originate usually from the physical symmetries of the underlying applications [66, 64]. Such special structures occur in numerous applications in engineering, mechanics, control, and linear systems theory. Some of the most common algebraic structures that appear in applications are the (skew-)symmetric [51], and alternating structures [64]. Symmetric (or Hermitian) matrix polynomials arise in the classical problem of vibration analysis [41], and alternating matrix polynomials find applications, for instance, in the study of corner singularities in anisotropic elastic materials [70], in the study of gyroscopic systems [59], in the continuous-time linear-quadratic optimal control problem and in the spectral factorization problem [85, 40].

Because of the numerous applications where structured polynomial and rational matrices occur, there have been several attempts to construct linearizations for them that display the same structure [1, 19, 27, 37, 40, 51, 58]. But these earlier attempts impose certain conditions on the corresponding polynomial or rational matrix for the construction of the linearization to apply, such as regularity, strict properness or invertibility of certain matrix coefficients. In Chapter 8 we give a construction of structured linearizations for structured polynomial and rational matrices without imposing any conditions, by using the notion of strongly minimal linearization. Moreover, the proof used for this construction is different from these earlier papers, and we claim it to be simpler as well.

We give more details on the structure of the thesis in the following section.

1.2 Structure and organization of the thesis

This dissertation is organized as follows:

Chapter 2 presents some preliminaries and basic notions on rational matrices that will be used throughout the dissertation. In particular, we introduce the definitions of the structural data of rational matrices and the key concept of polynomial system matrix of a transfer function matrix introduced by Rosenbrock [78], as well as some important properties related to it.

Chapter 3 is devoted to the construction of strong linearizations of rational matrices by using the definition of strong linearization in [6] and writing the corresponding rational matrix as the sum of its polynomial and strictly proper parts. In particular, in Sections 3.2 and 3.3, we construct strong linearizations of rational matrices with polynomial parts expressed in terms of a basis that satisfies a three term recurrence relation. In addition, in Section 3.8, we briefly discuss how to construct strong linearizations when the polynomial part is expressed in other bases. We emphasize that the construction of these new strong linearizations is a consequence of the theory of strong linearizations developed in [6] together with Lemma 3.1.2. More precisely, given a strong linearization of a rational matrix, Lemma 3.1.2 allows to obtain infinitely many strong linearizations of the rational matrix by using strict equivalence with a certain structure. The rest of this chapter is organized as follows. In Section 3.4, we show how to recover the eigenvectors of the rational matrix from those of its strong linearizations constructed in Sections 3.2 and 3.3. Moreover, given a symmetric rational matrix, in Section 3.6 we construct strong linearizations that preserve its symmetric structure by using symmetric realizations of the strictly proper part, which are introduced in Section 3.5, and strong linearizations in the double ansatz space $\mathbb{DM}(D)$ [36] of the polynomial part. Finally, in Section 3.7, we present analogous results for Hermitian rational matrices.

Chapter 4 presents a rigorous theory of local linearizations of rational matrices. For that, we first extend the concept of Rosenbrock's minimal polynomial system matrices to a local sense. Local minimal polynomial system matrices are defined and studied in Section 4.1. Section 4.2 presents the main definitions and properties of local linearizations of rational matrices. Sections 4.3 and 4.4 introduce the very general families of block full rank pencils and linearizations, as templates that cover many of the pencils available in the literature for linearizing rational matrices. Then, in **Chapter 5**, the theory of local linearizations is applied to a number of pencils that have appeared in the influential papers [47, 60] on solving numerically NLEPs by combining rational approximations and linearizations of the resulting rational matrices. It will be emphasized throughout the chapter that the theory of local linearizations allows us to view these pencils, and to explain their properties, from rather different perspectives. In particular, the pencils introduced in [47] are analysed and studied in Section 5.1, and those in [60] in Section 5.2.

In Chapter 6, by using the theory of local linearizations in Chapter 4, we

construct a new family of linearizations for rational matrices from general representations. These linearizations are given in Sections 6.1 and 6.2. In Section 6.3, we present an example that highlights the difference between our approach and the previous approaches to the problem of linearizing rational matrices. In Section 6.4 we study how to recover minimal bases, minimal indices and eigenvectors of rational matrices from those of their linearizations constructed in Sections 6.1 and 6.2. Finally, we apply the new linearizations to solve scalar rational equations in Section 6.5.

Chapter 7 recalls the notion of strong minimality of polynomial system matrices introduced in Chapter 4 and also the definition of strongly irreducible polynomial system matrix in [91]. These notions are given in Section 7.2 where, in addition, we establish the relation between them for the case of linear polynomial system matrices. In Subsection 7.2.2, we study the relation of strongly minimal linearizations with other classes of linearizations for polynomial and rational matrices in the literature. We then give, in Section 7.3, an algorithm to construct a strongly minimal linear system matrix from an arbitrary one, and we discuss the computational aspects in Section 7.4. Finally, we show some numerical experiments in Section 7.5.

Chapter 8 is devoted to the construction of strongly minimal linearizations for arbitrary and structured rational matrices. For that, in Section 8.1, we first show how to construct strongly minimal linearizations of arbitrary polynomial matrices and, in Section 8.2, we extend this construction to structured strongly minimal linearizations of structured polynomial matrices. Then, in Sections 8.3 and 8.4, we develop the same results for strictly proper rational matrices. That is, we build strongly minimal linearizations for arbitrary and structured strictly proper rational matrices, respectively. In Section 8.5, we combine the results in the previous sections of the chapter to construct strongly minimal linearizations for arbitrary and structured rational matrices. Finally, in Section 8.6, we comment some algorithmic aspects.

Chapter 9 studies the backward stability of running a backward stable algorithm to compute the eigenvalues on a pencil $S(\lambda)$ that is a strong linearization of a rational matrix of block Kronecker type. We describe in Section 9.2 the basic systems of matrix equations we will use, and, in Section 9.3, some bounds for the singular values of certain matrices related to these systems of matrix equations. In Section 9.4 we explain how to restore the structure of block Kronecker linearizations of rational matrices after they suffer sufficiently small perturbations. Then, in Section 9.5, we give sufficient conditions on the pencil $S(\lambda)$ and on the corresponding rational matrix that guarantee structural backward stability for (regular or singular) REPs solved via block Kronecker linearizations. In Section 9.6 we state the results for rational matrices having a linear polynomial part, since those in previous sections are developed for rational matrices with polynomial parts of degree greater than 1. In Section 9.7 we derive a scaling technique that allows to guarantee structural backward stability, taking into account the conditions in Section 9.5. We

conclude the chapter, in Section 9.8, by presenting a number of numerical results illustrating our theoretical bounds.

Finally, **Chapter 10** summarizes the main conclusions of this dissertation (see Section 10.1). In addition, we give a list of papers (published or in progress) where the original contributions of this thesis are contained (Section 10.2) and a list of conferences where many of the results have been presented (Section 10.3). Moreover, in Section 10.4, some open problems related to the results of this dissertation are proposed.

1.3 Notation

Throughout this dissertation, \mathbb{F} denotes an arbitrary field. The algebraic closure of \mathbb{F} is denoted by $\overline{\mathbb{F}}$. In some chapters, \mathbb{F} will be considered to be an algebraically closed field, that is, $\mathbb{F} = \overline{\mathbb{F}}$.

The ring of univariate polynomials in the variable λ with coefficients in \mathbb{F} is denoted by $\mathbb{F}[\lambda]$, whose elements are of the form

$$p(\lambda) := a_n \lambda^n + a_{n-1} \lambda^{n-1} + \dots + a_1 \lambda + a_0$$
 with $a_i \in \mathbb{F}$ for $i = 0, \dots, n$.

If \mathbb{F} is algebraically closed then every non-constant polynomial in $\mathbb{F}[\lambda]$ has a root in \mathbb{F} . If $a_n \neq 0$ then $p(\lambda)$ is said to have degree n and is denoted by $\deg(p(\lambda)) = n$. The field of rational functions over \mathbb{F} , denoted by $\mathbb{F}(\lambda)$, is the quotient field of the polynomial ring $\mathbb{F}[\lambda]$. Namely,

$$\mathbb{F}(\lambda) = \left\{ \frac{p(\lambda)}{q(\lambda)} : p(\lambda), q(\lambda) \in \mathbb{F}[\lambda], q(\lambda) \neq 0 \right\},\$$

with the equivalence relation $\frac{p_1(\lambda)}{q_1(\lambda)} = \frac{p_2(\lambda)}{q_2(\lambda)}$ if $p_1(\lambda)q_2(\lambda) = q_1(\lambda)p_2(\lambda)$.

The sets of $p \times m$ matrices with elements in $\mathbb{F}(\lambda)$, $\mathbb{F}[\lambda]$ and \mathbb{F} are denoted by $\mathbb{F}(\lambda)^{p \times m}$, $\mathbb{F}[\lambda]^{p \times m}$ and $\mathbb{F}^{p \times m}$, respectively. The elements of $\mathbb{F}[\lambda]^{p \times m}$ are called polynomial matrices. The degree of a polynomial matrix $P(\lambda)$ is the maximum degree of its entries and is denoted by $\deg(P(\lambda))$ or $\deg P(\lambda)$. If $\deg(P(\lambda))$ is equal to 1 or 0, then $P(\lambda)$ is said to be a pencil. The elements of $\mathbb{F}(\lambda)^{p \times m}$ are called rational matrices. The normal rank of a polynomial or rational matrix $R(\lambda)$ is the size of its largest non identically zero minor and is denoted by nrank $R(\lambda)$.

Given a constant matrix $A \in \mathbb{F}^{p \times m}$, the rank of A is the size of its largest non zero minor and is denoted by rank A. The transpose matrix of A is denoted by A^T . If \mathbb{F} is the field of complex numbers \mathbb{C} , A^* denotes the conjugate transpose of A. The Kronecker product of two constant matrices A and B is denoted by $A \otimes B$ (see [54, Chapter 4]). Diagonal matrices, with diagonal entries d_1, \ldots, d_m , are often denoted by diag (d_1, \ldots, d_m) .

Chapter 2

Preliminaries on rational matrices

In this chapter, we introduce preliminaries and basic notions on rational matrices. Some results presented here are original contributions of the author. In particular, such results are Proposition 2.1.5, Lemma 2.4.5, Propositions 2.4.7 and 2.4.8 and Lemma 2.4.10.

We consider an arbitrary field \mathbb{F} , e.g., the field of real numbers \mathbb{R} or of complex numbers \mathbb{C} . Recall that rational matrices are matrices whose entries are rational functions, and a rational function $r(\lambda)$ is a ratio such that both numerator $n(\lambda)$ and denominator $d(\lambda)$ are polynomials. Namely,

$$r(\lambda) = \frac{n(\lambda)}{d(\lambda)} \in \mathbb{F}(\lambda).$$

Regarding the degrees of the numerator and the denominator, the rational function $r(\lambda)$ is said to be proper if $\deg(n(\lambda)) \leq \deg(d(\lambda))$, and strictly proper if $\deg(n(\lambda)) < \deg(d(\lambda))$. We denote by $\mathbb{F}_{pr}(\lambda)$ the ring of proper rational functions. The units of $\mathbb{F}_{pr}(\lambda)$ are called *biproper rational functions*, i.e., rational functions having the same degree of numerator and denominator. $\mathbb{F}_{pr}(\lambda)^{p\times m}$ denotes the set of $p \times m$ matrices with entries in $\mathbb{F}_{pr}(\lambda)$, which are called *proper matrices*. A *biproper matrix* is a square proper matrix whose determinant is a biproper rational function.

By the division algorithm for polynomials, any rational function $r(\lambda)$ can be uniquely written as $r(\lambda) = p(\lambda) + r_{sp}(\lambda)$, where $p(\lambda)$ is a polynomial and $r_{sp}(\lambda)$ a strictly proper rational function. Therefore, any rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ can be uniquely written as

$$R(\lambda) = D(\lambda) + R_{sp}(\lambda) \tag{2.1}$$

where $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ is a polynomial matrix and $R_{sp}(\lambda) \in \mathbb{F}_{pr}(\lambda)^{p \times m}$ is a strictly proper rational matrix, i.e., the entries of $R_{sp}(\lambda)$ are strictly proper rational functions. Then, $D(\lambda)$ is called the *polynomial part* of $R(\lambda)$ and $R_{sp}(\lambda)$ is called the strictly proper part of $R(\lambda)$. A rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ is said to be regular or nonsingular if $R(\lambda)$ is square (i.e., p = m) and det $R(\lambda) \not\equiv 0$. Otherwise, $R(\lambda)$ is said to be singular. Square polynomial matrices with nonzero constant determinant are said to be unimodular, i.e., matrices $U(\lambda)$ in $\mathbb{F}[\lambda]^{m \times m}$ such that det $U(\lambda)$ is a nonzero constant. Equivalently, a polynomial matrix $U(\lambda)$ is unimodular if and only if the inverse of $U(\lambda)$ is also polynomial. A diagonal form for rational matrices $R(\lambda)$ is the so-called Smith-McMillan form, introduced by McMillan in 1952 [68], which uses unimodular equivalences. That is, transformations of the form $U_1(\lambda)R(\lambda)U_2(\lambda)$, where both $U_1(\lambda)$ and $U_2(\lambda)$ are unimodular matrices.

Definition 2.0.1 ((Finite or global) Smith–McMillan form). For any rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ there exist unimodular matrices $U_1(\lambda) \in \mathbb{F}[\lambda]^{p \times p}$ and $U_2(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ such that

$$U_1(\lambda)R(\lambda)U_2(\lambda) = \begin{bmatrix} \operatorname{diag}\left(\frac{\epsilon_1(\lambda)}{\psi_1(\lambda)}, \dots, \frac{\epsilon_r(\lambda)}{\psi_r(\lambda)}\right) & 0\\ 0 & 0_{(p-r)\times(m-r)} \end{bmatrix}$$
(2.2)

where $r = \operatorname{nrank} R(\lambda)$ and, for $i = 1, \ldots, r$, $\frac{\epsilon_i(\lambda)}{\psi_i(\lambda)}$ are nonzero irreducible rational functions with $\epsilon_i(\lambda)$ and $\psi_i(\lambda)$ monic polynomials (i.e., with leading coefficient equal to 1) and $\epsilon_1(\lambda) | \cdots | \epsilon_r(\lambda)$ while $\psi_r(\lambda) | \cdots | \psi_1(\lambda)$, where | stands for divisibility. The diagonal matrix on the right hand side in (2.2) is called the (finite or global) Smith-McMillan form of $R(\lambda)$.

Other more recent references for the Smith-McMillan form of a rational matrix are [55, 78, 90]. The rational functions $\frac{\epsilon_1(\lambda)}{\psi_1(\lambda)}, \ldots, \frac{\epsilon_r(\lambda)}{\psi_r(\lambda)}$ in (2.2) are called the *(finite) invariant rational functions* of $R(\lambda)$ and the *finite poles* and *zeros* of $R(\lambda)$ are the roots in $\overline{\mathbb{F}}$ of the denominators and numerators of the invariant rational functions, respectively. We give more details about zeros and poles in the following section. Notice that the Smith-McMillan form of a rational matrix is invariant under unimodular equivalence. If $R(\lambda)$ is polynomial then $\psi_1(\lambda) = \cdots = \psi_r(\lambda) = 1$, $\epsilon_1(\lambda), \ldots, \epsilon_r(\lambda)$ are called the *invariant polynomials* of $R(\lambda)$, and the diagonal matrix in (2.2) is called the *Smith normal form* of $R(\lambda)$.

2.1 Zeros and poles: Local Smith–McMillan form

As we explained, the finite poles and zeros of a rational matrix are the roots in $\overline{\mathbb{F}}$ of the polynomials that appear on the denominators and numerators, respectively, in its (global) Smith-McMillan form. In this section, we introduce a local definition of the Smith-McMillan form of rational matrices, and more notions related to their poles and zeros. For that, we first introduce some definitions and equivalence transformations on rational matrices.

Definition 2.1.1. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$. Let $\lambda_0 \in \mathbb{F}$, and $\Sigma \subseteq \mathbb{F}$ be nonempty.

- (i) $R(\lambda)$ is defined at λ_0 if $R(\lambda_0) \in \mathbb{F}^{p \times m}$.
- (ii) $R(\lambda)$ is defined in Σ if $R(\lambda)$ is defined at each $\lambda_0 \in \Sigma$.
- (iii) $R(\lambda)$ is defined at ∞ if $R(1/\lambda)$ is defined at 0.

Definition 2.1.2. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$. Let $\lambda_0 \in \mathbb{F}$, and $\Sigma \subseteq \mathbb{F}$ be nonempty.

- (i) $R(\lambda)$ is invertible at λ_0 if it is defined at λ_0 and det $R(\lambda_0) \neq 0$.
- (ii) $R(\lambda)$ is invertible in Σ if it is invertible at each $\lambda_0 \in \Sigma$.

(iii) $R(\lambda)$ is invertible at ∞ if $R(1/\lambda)$ is invertible at 0.

Notice that a rational matrix $R(\lambda)$ is defined at a point λ_0 if and only if λ_0 is not a pole of $R(\lambda)$. In addition, a rational matrix is unimodular if and only if it is invertible in $\overline{\mathbb{F}}$, and is biproper if and only if it is invertible at infinity. See [55] and [90] for more information on these and other concepts related to rational matrices.

In regard to the previous definitions, we introduce some equivalence relations defined in the set of rational matrices [9, 10, 42].

Definition 2.1.3 (Equivalences). Let $\Sigma \subseteq \mathbb{F}$ be nonempty. Two rational matrices $R_1(\lambda), R_2(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ are said to be equivalent in Σ if there exist rational matrices $G_1(\lambda) \in \mathbb{F}(\lambda)^{p \times p}$ and $G_2(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ both invertible in Σ such that

$$G_1(\lambda)R_1(\lambda)G_2(\lambda) = R_2(\lambda).$$

This is denoted by

$$R_1(\lambda) \sim_{\Sigma} R_2(\lambda).$$

When $\Sigma = \{\lambda_0\}$, we have local equivalence at λ_0 and is denoted by $R_1(\lambda) \sim_{\lambda_0} R_2(\lambda)$. If $G_1(\lambda)$ and $G_2(\lambda)$ are biproper then $R_1(\lambda)$ and $R_2(\lambda)$ are said to be equivalent at ∞ , which is denoted by $R_1(\lambda) \sim_{\infty} R_2(\lambda)$.

If, in Definition 2.1.3, the rational matrices $G_1(\lambda)$ and $G_2(\lambda)$ are both unimodular, then the standard definition of unimodular equivalence is recovered. In this case, $R_1(\lambda)$ and $R_2(\lambda)$ are said to be unimodularly equivalent.

We can now present the notion of the local Smith-McMillan form (see [10, 88]). Since poles and zeros of a rational matrix belong to the algebraic closure $\overline{\mathbb{F}}$, we consider $\overline{\mathbb{F}}$ instead of \mathbb{F} in order to define the local Smith-McMillan form.

Definition 2.1.4 (Local Smith–McMillan form). Let $\lambda_0 \in \overline{\mathbb{F}}$ and let $R(\lambda) \in \overline{\mathbb{F}}(\lambda)^{p \times m}$ be a rational matrix, with $r = \operatorname{nrank} R(\lambda)$. $R(\lambda)$ admits a representation of the form:

$$R(\lambda) \sim_{\lambda_0} \begin{bmatrix} \operatorname{diag}\left((\lambda - \lambda_0)^{\nu_1}, \dots, (\lambda - \lambda_0)^{\nu_r}\right) & 0\\ 0 & 0_{(p-r) \times (m-r)} \end{bmatrix}$$
(2.3)

for some integers ν_1, \ldots, ν_r with $\nu_1 \leq \ldots \leq \nu_r$. The diagonal matrix on the right hand side in (2.3) is called the local Smith-McMillan form of $R(\lambda)$ at λ_0 . The integers ν_1, \ldots, ν_r are uniquely determined by $R(\lambda)$ and λ_0 , and are called the *invariant orders or structural indices* at λ_0 of $R(\lambda)$. In order to define zeros and poles, together with partial multiplicities, we need to distinguish between positive and negative invariant orders [55, 90]:

- (a) If $\nu_i > 0$, for some i = 1, ..., r, then λ_0 is said to be a zero of $R(\lambda)$ with partial multiplicity ν_i ; and the factor $(\lambda \lambda_0)^{\nu_i}$ is called a zero elementary divisor of $R(\lambda)$ at λ_0 .
- (b) If $\nu_i < 0$, for some i = 1, ..., r, then λ_0 is said to be a *pole* of $R(\lambda)$ with *partial multiplicity* $-\nu_i$; and the factor $(\lambda \lambda_0)^{-\nu_i}$ is called a *pole elementary divisor* of $R(\lambda)$ at λ_0 .

The zero and pole elementary divisors of $R(\lambda)$ in a nonempty subset $\Sigma \subseteq \overline{\mathbb{F}}$ are the zero and pole elementary divisors of $R(\lambda)$ for all $\lambda_0 \in \Sigma$, respectively. If $R(\lambda)$ is a polynomial matrix then the nonzero integers ν_i are all positive and are called the *partial multiplicities* of $R(\lambda)$ at λ_0 , and the factors $(\lambda - \lambda_0)^{\nu_i}$ with $\nu_i \neq 0$ are called *elementary divisors* of $R(\lambda)$ at λ_0 .

The invariant orders at infinity $q_1 \leq \cdots \leq q_r$ of a rational matrix $R(\lambda)$ are defined as the invariant orders at $\lambda_0 = 0$ of $R(1/\lambda)$, and the Smith-McMillan form of $R(\lambda)$ at ∞ is of the form

$$R(\lambda) \sim_{\infty} \begin{bmatrix} \operatorname{diag}\left(\frac{1}{\lambda^{q_1}}, \dots, \frac{1}{\lambda^{q_r}}\right) & 0\\ 0 & 0_{(p-r)\times(m-r)} \end{bmatrix}.$$
 (2.4)

For $q_i < 0$, $-q_i$ are the partial multiplicities of ∞ as pole while, for $q_i > 0$, q_i are the partial multiplicities of ∞ as zero (see [55]).

By using the local Smith–McMillan form, the next result in [28] shows that the equivalence of rational matrices in nonempty sets is a local property.

Proposition 2.1.5. Let $\Sigma \subseteq \overline{\mathbb{F}}$ be nonempty. Two rational matrices of the same size are equivalent in Σ if and only if they are equivalent at each $\lambda_0 \in \Sigma$.

Proof. If two rational matrices are equivalent in Σ then, by Definition 2.1.3, it is straightforward that they are equivalent at each $\lambda_0 \in \Sigma$. For the converse, suppose that $G(\lambda) \sim_{\lambda_0} H(\lambda)$ for all $\lambda_0 \in \Sigma$. Then, $G(\lambda)$ and $H(\lambda)$ have the same local Smith– McMillan forms at each $\lambda_0 \in \Sigma$. Let us consider $M_G(\lambda)$ and $M_H(\lambda)$ as the global Smith–McMillan forms of $G(\lambda)$ and $H(\lambda)$, respectively. Thus, there exist unimodular matrices $U_i^G(\lambda)$, $U_i^H(\lambda)$, for i = 1, 2, such that $G(\lambda) = U_1^G(\lambda)M_G(\lambda)U_2^G(\lambda)$, $H(\lambda) = U_1^H(\lambda)M_H(\lambda)U_2^H(\lambda)$, and we can write

$$M_G(\lambda) = \operatorname{diag}\left(f_1(\lambda)g_1(\lambda), \dots, f_r(\lambda)g_r(\lambda), 0_{(p-r)\times(m-r)}\right), \text{ and} M_H(\lambda) = \operatorname{diag}\left(f_1(\lambda)h_1(\lambda), \dots, f_r(\lambda)h_r(\lambda), 0_{(p-r)\times(m-r)}\right),$$

where $f_i(\lambda)$ are rational functions which are either equal to one or have poles or zeros in Σ , while $g_i(\lambda)$ and $h_i(\lambda)$ are rational functions that do not have any poles or zeros in Σ . Let us define $R(\lambda) := \operatorname{diag}\left(\frac{h_1(\lambda)}{g_1(\lambda)}, \dots, \frac{h_r(\lambda)}{g_r(\lambda)}, I_{m-r}\right)$. Hence, $M_H(\lambda) = M_G(\lambda)R(\lambda)$. Therefore, $H(\lambda) = U_1^H(\lambda)U_1^G(\lambda)^{-1}G(\lambda)U_2^G(\lambda)^{-1}R(\lambda)U_2^H(\lambda)$, and $G(\lambda) \sim_{\Sigma} H(\lambda)$ since the matrices $U_1^H(\lambda)U_1^G(\lambda)^{-1}$ and $U_2^G(\lambda)^{-1}R(\lambda)U_2^H(\lambda)$ are invertible in Σ .

2.2 The rational eigenvalue problem

A finite eigenvalue of a rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ is any $\lambda_0 \in \overline{\mathbb{F}}$ such that

rank $R(\lambda_0) < \operatorname{nrank} R(\lambda)$, with $R(\lambda_0) \in \overline{\mathbb{F}}^{p \times m}$

That is, λ_0 is a finite zero of $R(\lambda)$ but not a pole. The *Rational Eigenvalue Problem* (REP) consists of finding the eigenvalues of $R(\lambda)$. If $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ is regular, which is the most common case in applications of REPs, the REP is equivalent to the problem of finding scalars $\lambda_0 \in \overline{\mathbb{F}}$ such that there exist nonzero constant vectors $x \in \overline{\mathbb{F}}^{m \times 1}$ and $y \in \overline{\mathbb{F}}^{m \times 1}$ satisfying

$$R(\lambda_0)x = 0$$
 and $y^T R(\lambda_0) = 0$,

respectively. The vectors x are called *right eigenvectors associated with* λ_0 , and the vectors y left eigenvectors. Given $\lambda_0 \in \overline{\mathbb{F}}$ and $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$, we define the following vector spaces over $\overline{\mathbb{F}}$:

$$\mathcal{N}_r(R(\lambda_0)) = \{ x \in \overline{\mathbb{F}}^{m \times 1} : R(\lambda_0) x = 0 \}, \text{ and} \\ \mathcal{N}_\ell(R(\lambda_0)) = \{ y^T \in \overline{\mathbb{F}}^{1 \times p} : y^T R(\lambda_0) = 0 \},$$

which are called, respectively, the right and left nullspaces over $\overline{\mathbb{F}}$ of $R(\lambda_0)$. If λ_0 is an eigenvalue of $R(\lambda)$, then $\mathcal{N}_r(R(\lambda_0))$ and $\mathcal{N}_\ell(R(\lambda_0))$ are non trivial and contain, respectively, the right and left eigenvectors of $R(\lambda)$ associated with λ_0 .

Rational matrices may also have infinite eigenvalues. In order to define them, we need the notion of reversal.

Definition 2.2.1 (Reversal of a rational matrix). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form (2.1). We define the reversal of $R(\lambda)$ as the rational matrix

$$\operatorname{rev} R(\lambda) = \lambda^d R\left(\frac{1}{\lambda}\right)$$

where $d = \deg(D(\lambda))$ if $R(\lambda)$ is not strictly proper, and d = 0 otherwise.

This definition of reversal extends the definition of reversal for polynomial matrices (see [22, Definition 2.12] or [63, Definition 2.2]). Following the usual definition in polynomial matrices [63, Definition 2.3], we say that $R(\lambda)$ has an *eigenvalue at* infinity if rev $R(\lambda)$ has an eigenvalue at $\lambda = 0$. If $R(\lambda)$ has an eigenvalue at infinity, we say that z is a right (respectively left) eigenvector associated with infinity if z is a right (respectively left) eigenvector associated with 0 of rev $R(\lambda)$.

Remark 2.2.2. Recall that, for finite points, eigenvalues are defined as those zeros that are not poles. However, if we define eigenvalues at infinity in this way any non-proper $R(\lambda)$ would not have eigenvalues at infinity. This is due to the fact that if $R(\lambda)$ is not proper, i.e., $\deg(D(\lambda)) \ge 1$, $R(\lambda)$ has always a pole at ∞ (see [10]). In particular, this would happen if $R(\lambda)$ is a polynomial matrix. Therefore, as in the polynomial case in [44], we consider rev $R(\lambda)$ in order to define eigenvalues at infinity.

Remark 2.2.3. The standard literature on polynomial matrices [39, 44] uses only the term eigenvalues instead of zeros and poles and the eigenvalue structure at infinity is defined through the notion of the reversal. We discuss the relation between finite and infinite zeros and poles and eigenvalues of polynomial matrices in this remark. Note first that a polynomial matrix $P(\lambda)$ does not have finite poles, i.e., all the invariant orders ν_i in (2.3) are nonnegative for any finite λ_0 . Then, the finite eigenvalues of $P(\lambda)$ and their partial multiplicities [44] are exactly the same as the finite zeros of $P(\lambda)$ and their partial multiplicities. However, a polynomial matrix $P(\lambda)$ of degree d and normal rank r is said to have an eigenvalue at ∞ with partial multiplicities $t_q \leq \cdots \leq t_r$ if the reversal rev $P(\lambda) := \lambda^d P(1/\lambda)$ has an eigenvalue at 0 with partial multiplicities $t_q \leq \cdots \leq t_r$. In this situation the invariant orders of $P(\lambda)$ at ∞ (i.e., the invariant orders of $P(1/\lambda)$ at 0) are

$$(q_1, q_2, \dots, q_r) = (\underbrace{0, \dots, 0}_{q-1}, t_q, \dots, t_r) - (d, d, \dots, d).$$
 (2.5)

Thus, the pole-zero and eigenvalue structures at infinity are different but easily related through (2.5).

2.3 Minimal bases and minimal indices

In this section, we review the notions of minimal bases and minimal indices of rational subspaces [38] and rational matrices.

It is known that every rational vector subspace $\mathcal{V} \subseteq \mathbb{F}(\lambda)^n$ over the field $\mathbb{F}(\lambda)$ has bases consisting of polynomial vectors. We refer to such bases as *polynomial bases*. Among them some are minimal in the following sense introduced by Forney [38]: a *minimal basis* of \mathcal{V} is a polynomial basis of \mathcal{V} consisting of polynomial vectors whose sum of degrees is minimal among all polynomial bases of \mathcal{V} . Minimal bases are not unique, but the ordered list of degrees of the polynomial vectors in any minimal basis of \mathcal{V} is always the same. Hence, these degrees are uniquely determined by \mathcal{V} and are called the *minimal indices* of \mathcal{V} . Let now $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix. We consider the rational vector subspaces:

$$\mathcal{N}_r(R) = \{ x(\lambda) \in \mathbb{F}(\lambda)^{m \times 1} : R(\lambda) x(\lambda) = 0 \}, \text{ and } \\ \mathcal{N}_\ell(R) = \{ y(\lambda)^T \in \mathbb{F}(\lambda)^{1 \times p} : y(\lambda)^T R(\lambda) = 0 \},$$

which are called the *right* and *left null-spaces over* $\mathbb{F}(\lambda)$ of $R(\lambda)$, respectively. If $R(\lambda)$ is singular at least one of these null-spaces is non-trivial. If $\mathcal{N}_r(R)$ (resp. $\mathcal{N}_\ell(R)$) is non trivial, it has minimal bases and minimal indices, which are called the right (resp. left) minimal bases and *right (resp. left) minimal indices of* $R(\lambda)$. By the rank-nullity theorem,

 $\dim \mathcal{N}_{\ell}(R) = p - \operatorname{nrank} R(\lambda), \text{ and } \dim \mathcal{N}_{r}(R) = m - \operatorname{nrank} R(\lambda).$

Therefore, if $r = \operatorname{nrank} R(\lambda)$, then $R(\lambda)$ has p - r left minimal indices and m - r right minimal indices.

Minimal bases appearing in this dissertation are arranged as the columns or rows of polynomial matrices. With a slight abuse of notation, we say that a $p \times m$ polynomial matrix with p > m (resp. p < m) is a *minimal basis* if its columns (resp. rows) form a minimal basis of the rational subspace they span. The following definitions are useful for characterizing minimal bases.

Definition 2.3.1. The *i*th column (resp. row) degree of a matrix polynomial $B(\lambda)$ is the degree of the *i*th column (resp. row) of $B(\lambda)$.

Definition 2.3.2. Let $B(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ be a polynomial matrix with column (resp. row) degrees d_1, d_2, \ldots, d_m (resp. d_1, d_2, \ldots, d_p). The highest column (resp. row) degree coefficient matrix of $B(\lambda)$, denoted by B_{hcd} (resp. B_{hrd}), is the $p \times m$ constant matrix whose jth column (resp. row) is the coefficient of λ^{d_j} in the jth column (resp. row) of $B(\lambda)$. The polynomial matrix $B(\lambda)$ is called column (resp. row) reduced if B_{hcd} (resp. B_{hrd}) has full colum (resp. row) rank.

Theorem 2.3.3 states one of the most useful characterizations of minimal bases (see [38, Main Theorem] or [26, Theorem 2.2]).

Theorem 2.3.3. The columns (resp. rows) of a polynomial matrix $B(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ with p > m (resp. p < m) are a minimal basis of the subspace they span if and only if $B(\lambda)$ is column (resp. row) reduced and $B(\lambda_0)$ has full column (resp. row) rank for all $\lambda_0 \in \overline{\mathbb{F}}$.

Associated with minimal bases, the notion of dual minimal basis [26, Definition 2.5] is also considered.

Definition 2.3.4. Let $K(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ be a minimal basis with p < m. Another minimal basis $N(\lambda) \in \mathbb{F}[\lambda]^{q \times m}$ is said to be dual to $K(\lambda)$ if p + q = m and $K(\lambda)N(\lambda)^T = 0$. Then $N(\lambda)$ is said to be a dual minimal basis of $K(\lambda)$.

2.4 Polynomial system matrices

Polynomial system matrices are a classical tool for studying rational matrices. They were introduced by Rosenbrock and are analyzed in detail in [78]. Among them, minimal polynomial system matrices have been used in many problems dealing with rational matrices because they allow to extract all the information about finite poles and zeros.

Definition 2.4.1 (Polynomial system matrix and transfer function). Any rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ can be written as

$$R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$$
(2.6)

for some polynomial matrices $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$, $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ with $A(\lambda)$ regular if n > 0. Then, a polynomial matrix of the form

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix}$$
(2.7)

is called a polynomial system matrix of $R(\lambda)$. That is, $R(\lambda)$ is the Schur complement of $A(\lambda)$ in $P(\lambda)$ and is called the transfer function matrix of $P(\lambda)$.

If n = 0 in (2.6), we assume that the matrices $A(\lambda)$, $B(\lambda)$ and $C(\lambda)$ are empty, and $R(\lambda) = D(\lambda)$ is a polynomial matrix. But polynomial matrices can also be represented as in (2.6) with n > 0. For instance, this can be obtained if $A(\lambda)$ is unimodular. In general, representations of the form (2.6) always exist for any rational matrix $R(\lambda)$ and are not unique. They are called *realizations* of $R(\lambda)$. We refer to $A(\lambda)$ as the *state matrix* of $P(\lambda)$ and the integer deg(det $A(\lambda)$) is called the *order* of $P(\lambda)$. Moreover, $P(\lambda)$ is said to have *least order*, or to be *minimal*, if its order is the smallest integer for which polynomial matrices $A(\lambda)$, $B(\lambda)$, $C(\lambda)$ and $D(\lambda)$ satisfying (2.6) exist. Another equivalent definition for minimality is the following [78].

Definition 2.4.2 (Minimal polynomial system matrix). The polynomial system matrix $P(\lambda)$ in (2.7), with n > 0, is said to have least order, or to be minimal or irreducible, if the matrices

$$\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix} \quad and \quad \begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix}$$
(2.8)

have no eigenvalues in $\overline{\mathbb{F}}$. In such a case, the realization in (2.6) is also said to be minimal.

The least order is uniquely determined by $R(\lambda)$ and is denoted by $\nu(R(\lambda))$. It is also called the *least order* of $R(\lambda)$ ([78, Chapter 3, Section 5.1] or [90, Section 1.10]). From [78, Chapter 3, Theorem 4.1], it can be deduced that $\nu(R(\lambda))$ is the degree of the polynomial that results by making the product of the denominators in the (finite) Smith–McMillan form of $R(\lambda)$. **Remark 2.4.3.** Notice that the definition of polynomial system matrix $P(\lambda)$ in (2.7) includes a specific partition. However, the state matrix $A(\lambda)$ might be a submatrix of $P(\lambda)$ different from the (1, 1)-block. Throughout this dissertation, polynomial matrices are partitioned in different ways giving rise to different polynomial system matrices of (possibly) different transfer functions. In such cases, we often use expressions as " $P(\lambda)$ is a polynomial system matrix with state matrix $A(\lambda)$ " in order to avoid ambiguities. In the case n = 0, we use the expression " $P(\lambda)$ is a polynomial system matrix".

The main property of a polynomial system matrix $P(\lambda)$ being minimal, or of least order, is that the finite pole and zero information of its transfer function matrix is contained in $P(\lambda)$ [78].

Theorem 2.4.4. Let

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+p) \times (n+m)}$$

be a minimal polynomial system matrix, with state matrix $A(\lambda)$, whose transfer function matrix is $R(\lambda)$. Then the finite elementary divisors of $A(\lambda)$ are the finite pole elementary divisors of $R(\lambda)$, and the finite elementary divisors of $P(\lambda)$ are the finite zero elementary divisors of $R(\lambda)$.

A polynomial system matrix $P(\lambda)$ of $R(\lambda)$ is said to be a polynomial system matrix in *state-space form* if $A(\lambda) = \lambda I_n - A$, $B(\lambda) = B$ and $C(\lambda) = C$ for some constant matrices $A \in \mathbb{F}^{n \times n}$, $B \in \mathbb{F}^{n \times m}$ and $C \in \mathbb{F}^{p \times n}$. It is known that any strictly proper rational matrix admits *state-space realizations* (see [78] or [55]). Thus, if we consider $R(\lambda)$ as in (2.1), this means that for some positive integer *n* there exist constant matrices $A \in \mathbb{F}^{n \times n}$, $B \in \mathbb{F}^{n \times m}$ and $C \in \mathbb{F}^{p \times n}$ such that $R_{sp}(\lambda) =$ $C(\lambda I_n - A)^{-1}B$ and

$$\begin{bmatrix} \lambda I_n - A & B \\ -C & D(\lambda) \end{bmatrix}$$

is a polynomial system matrix of $R(\lambda)$. Therefore $R(\lambda) = D(\lambda) + C(\lambda I_n - A)^{-1}B$. In addition, the state-space realization may always be taken of least order, or minimal (i.e., such that the polynomial system matrix in state-space form is of least order).

Notice that any polynomial system matrix $P(\lambda)$ can be written as the following block LDU factorization:

$$P(\lambda) = \begin{bmatrix} I_n & 0\\ -C(\lambda)A(\lambda)^{-1} & I_p \end{bmatrix} \begin{bmatrix} A(\lambda) & 0\\ 0 & R(\lambda) \end{bmatrix} \begin{bmatrix} I_n & A(\lambda)^{-1}B(\lambda)\\ 0 & I_m \end{bmatrix}$$

Then the relation between the normal ranks of $P(\lambda)$ and its transfer function matrix $R(\lambda)$ is

nrank
$$P(\lambda) = n + \operatorname{nrank} R(\lambda).$$
 (2.9)

Using the factorization above and the rank property in (2.9), the following Lemma 2.4.5 included in [75] establishes a linear map between the right (resp. left) nullspace of a rational matrix $R(\lambda)$ and the right (resp. left) nullspace of a polynomial system matrix of $R(\lambda)$.

Lemma 2.4.5. Let $P(\lambda)$ be a polynomial system matrix as in (2.7), and let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be its transfer function matrix. Then, the following statements hold:

(a) The linear map

$$T_r : \mathcal{N}_r(R) \longrightarrow \mathcal{N}_r(P)$$
$$x(\lambda) \longmapsto \begin{bmatrix} -A(\lambda)^{-1}B(\lambda) \\ I_m \end{bmatrix} x(\lambda)$$

is a bijection between the right nullspaces of $R(\lambda)$ and $P(\lambda)$.

(b) The linear map

$$T_{\ell} : \mathcal{N}_{\ell}(R) \longrightarrow \mathcal{N}_{\ell}(P)$$
$$y(\lambda)^{T} \longmapsto y(\lambda)^{T} \left[C(\lambda)A(\lambda)^{-1} \quad I_{p} \right]$$

is a bijection between the left nullspaces of $R(\lambda)$ and $P(\lambda)$.

Proof. We only prove part (a) since part (b) can be proved in a similar way. First, we observe that the map T_r is linear. Second, we notice that for any vector $x(\lambda)$, we have

$$\begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} -A(\lambda)^{-1}B(\lambda)x(\lambda) \\ x(\lambda) \end{bmatrix} = \begin{bmatrix} 0 \\ R(\lambda)x(\lambda) \end{bmatrix},$$

which shows that T_r maps vectors in the right nullspace of $R(\lambda)$ to vectors in the right nullspace of $P(\lambda)$. Finally, by (2.9) and the rank-nullity theorem, we have

$$\dim \mathcal{N}_r(P) = \dim \mathcal{N}_r(R). \tag{2.10}$$

Since the right nullspaces of $P(\lambda)$ and $R(\lambda)$ have the same dimension and the linear map T_r is clearly injective, we conclude that the map T_r is bijective.

Remark 2.4.6. Since the maps in Lemma 2.4.5 are bijections, they preserve linear independence. Hence, one can recover a basis of the right (resp. left) nullspace of $R(\lambda)$ from a basis of the right (resp. left) nullspace of $P(\lambda)$, and conversely. For instance, from part (a) in Lemma 2.4.5, we obtain that if $\{x_i(\lambda)\}_{i=1}^t$ is a basis of $\mathcal{N}_r(R)$, then $\left\{ \begin{bmatrix} -A(\lambda)^{-1}B(\lambda)x_i(\lambda) \\ x_i(\lambda) \end{bmatrix} \right\}_{i=1}^t$ is a basis of $\mathcal{N}_r(P)$. Conversely, if $\left\{ \begin{bmatrix} y_i(\lambda) \\ x_i(\lambda) \end{bmatrix} \right\}_{i=1}^t$ is a basis of $\mathcal{N}_r(R)$.

To recover minimal bases and minimal indices of rational matrices from polynomial system matrices $P(\lambda)$ one has to assume extra conditions on $P(\lambda)$. For results about recovery of minimal bases and minimal indices of rational matrices from polynomial system matrices we refer to [8, 91, 92].

2.4.1 Eigenvectors of polynomial system and transfer function matrices

We know from [7, Proposition 3.1] how to recover right eigenvectors of a polynomial system matrix $P(\lambda)$ from those of its transfer function $R(\lambda)$, and conversely. In Proposition 2.4.7 we state a extended version of [7, Proposition 3.1], that includes a result about bases of the right null-spaces of $P(\lambda)$ and $R(\lambda)$ evaluated at the eigenvalue of interest, that was introduced in [27, Proposition 5.1]. We state without proof the analogous result for left eigenvectors and null-spaces in Proposition 2.4.8, introduced in [27, Proposition 5.2].

For the sake of brevity, the following nomenclature is adopted: " (λ_0, x_0) is a solution of the REP $R(\lambda)x = 0$ " means that λ_0 is a finite eigenvalue of $R(\lambda)$ and x_0 is a right eigenvector corresponding to λ_0 , and " (λ_0, x_0) is a solution of the REP $x^T R(\lambda) = 0$ " means that λ_0 is a finite eigenvalue of $R(\lambda)$ and x_0 is a left eigenvector corresponding to λ_0 . Although it is not common in the literature, if $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ is singular, we call right and left eigenvectors of $R(\lambda)$ associated with an eigenvalue λ_0 to any nonzero vectors x and y satisfying $R(\lambda_0)x = 0$ and $y^T R(\lambda_0) = 0$, respectively.

In what follows, we assume that eigenvectors of the form $\begin{bmatrix} y \\ x \end{bmatrix}$ are partitioned conformable to the corresponding polynomial system matrix.

Proposition 2.4.7. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix and

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+p) \times (n+m)}$$

be any polynomial system matrix with $R(\lambda)$ as transfer function matrix.

- a) If $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix} \end{pmatrix}$ is a solution of the PEP $P(\lambda)z = 0$ such that det $A(\lambda_0) \neq 0$, then (λ_0, x_0) is a solution of the REP $R(\lambda)x = 0$.
- b) Moreover, if $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \dots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(P(\lambda_0))$, with det $A(\lambda_0) \neq 0$, then $\{x_1, \dots, x_t\}$ is a basis of $\mathcal{N}_r(R(\lambda_0))$.
- c) Conversely, if (λ_0, x_0) is a solution of the REP $R(\lambda)x = 0$ such that det $A(\lambda_0) \neq 0$ and y_0 is defined as the unique solution of $A(\lambda_0)y_0 + B(\lambda_0)x_0 = 0$, then $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix} \end{pmatrix}$ is a solution of the PEP $P(\lambda)z = 0$.
- d) Moreover, if $\{x_1, \ldots, x_t\}$ is a basis of $\mathcal{N}_r(R(\lambda_0))$, with det $A(\lambda_0) \neq 0$, and, for $i = 1, \ldots, t, y_i$ is defined as the unique solution of $A(\lambda_0)y_i + B(\lambda_0)x_i = 0$, then $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(P(\lambda_0))$.

Proof. The statements a) and c) are the results in [7, Proposition 3.1] stated here for a rectangular matrix $R(\lambda)$. The proofs are exactly the same as in [7] and, therefore, are omitted. To prove b) and d) we write

$$\begin{bmatrix} A(\lambda_0) & B(\lambda_0) \\ -C(\lambda_0) & D(\lambda_0) \end{bmatrix} = \begin{bmatrix} I_n & 0 \\ -C(\lambda_0)A(\lambda_0)^{-1} & I_p \end{bmatrix} \begin{bmatrix} A(\lambda_0) & 0 \\ 0 & R(\lambda_0) \end{bmatrix} \begin{bmatrix} I_n & A(\lambda_0)^{-1}B(\lambda_0) \\ 0 & I_m \end{bmatrix}.$$

Since det $A(\lambda_0) \neq 0$, rank $P(\lambda_0) = n + \operatorname{rank} R(\lambda_0)$. Therefore

$$\dim \mathcal{N}_r(P(\lambda_0)) = \dim \mathcal{N}_r(R(\lambda_0)).$$
(2.11)

Then b) and d) are obtained by using a) and c), respectively, taking (2.11) and the linear independence of the considered sets into account, and observing that $P(\lambda_0) \begin{bmatrix} y_0 \\ x_0 \end{bmatrix} = 0$ if and only if $y_0 = -A(\lambda_0)^{-1}B(\lambda_0)x_0$ and $R(\lambda_0)x_0 = 0$.

Proposition 2.4.8 is an analogous result to Proposition 2.4.7 for left eigenvectors and left null-spaces, and it can be proved in a similar way. It was introduced in [27, Proposition 5.2].

Proposition 2.4.8. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix and

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+p) \times (n+m)}$$

be any polynomial system matrix with $R(\lambda)$ as transfer function matrix.

- a) If $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}$ is a solution of the PEP $z^T P(\lambda) = 0$ such that det $A(\lambda_0) \neq 0$, then (λ_0, x_0) is a solution of the REP $x^T R(\lambda) = 0$.
- b) Moreover, if $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \dots, \begin{bmatrix} y_q \\ x_q \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_{\ell}(P(\lambda_0))$, with det $A(\lambda_0) \neq 0$, then $\{x_1, \dots, x_q\}$ is a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$.
- c) Conversely, if (λ_0, x_0) is a solution of the REP $x^T R(\lambda) = 0$ such that det $A(\lambda_0) \neq 0$, and y_0 is defined as the unique solution of $y_0^T A(\lambda_0) x_0^T C(\lambda_0) = 0$, then $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix} \end{pmatrix}$ is a solution of the PEP $z^T P(\lambda) = 0$.
- d) Moreover, if $\{x_1, \ldots, x_q\}$ is a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$, with det $A(\lambda_0) \neq 0$, and, for $i = 1, \ldots, q, y_i$ is defined as the unique solution of $y_i^T A(\lambda_0) x_i^T C(\lambda_0) = 0$, then $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_q \\ x_q \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_{\ell}(P(\lambda_0))$.
Remark 2.4.9. If $R(\lambda)$ is singular, then for any $\lambda_0 \in \overline{\mathbb{F}}$ that is not a pole of $R(\lambda)$, including those λ_0 that are not eigenvalues of $R(\lambda)$, $\mathcal{N}_r(R(\lambda_0)) \neq \{0\}$ or $\mathcal{N}_\ell(R(\lambda_0)) \neq \{0\}$. The reader can check easily that Propositions 2.4.7 and 2.4.8 remain valid for any $\lambda_0 \in \overline{\mathbb{F}}$ that is not a pole of $R(\lambda)$ in the case $R(\lambda)$ is singular.

As corollary, we obtain the following Lemma 2.4.10 that establishes bijections relating the right and left null spaces of a polynomial system matrix and those of its transfer function matrix, evaluated at a finite λ_0 . This result is analogous to that in Lemma 2.4.5 and was introduced in [29].

Lemma 2.4.10. Let $P(\lambda)$ be a polynomial system matrix as in (2.7), and let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be its transfer function matrix. Let $\lambda_0 \in \overline{\mathbb{F}}$ such that det $A(\lambda_0) \neq 0$. Then, the following statements hold:

(a) The linear map

$$F_r : \mathcal{N}_r(R(\lambda_0)) \longrightarrow \mathcal{N}_r(P(\lambda_0))$$
$$x \longmapsto \begin{bmatrix} -A(\lambda_0)^{-1}B(\lambda_0) \\ I_m \end{bmatrix} x$$

is a bijection between the right nullspaces over $\overline{\mathbb{F}}$ of $R(\lambda_0)$ and $P(\lambda_0)$.

(b) The linear map

$$F_{\ell} : \mathcal{N}_{\ell}(R(\lambda_0)) \longrightarrow \mathcal{N}_{\ell}(P(\lambda_0))$$
$$y^T \longmapsto y^T \left[C(\lambda_0) A(\lambda_0)^{-1} \quad I_p \right]$$

is a bijection between the left nullspaces over $\overline{\mathbb{F}}$ of $R(\lambda_0)$ and $P(\lambda_0)$.

2.5 Definitions of strong linearizations of polynomial and rational matrices in the literature

The standard method of dealing with the rational (and polynomial) eigenvalue problem consists of linearizing. That is, reformulating the corresponding rational matrix into a linear polynomial matrix in such a way that the eigenstructure can be exactly recovered, i.e., the zero structure, the polar structure, and the left and right null space structure.

2.5.1 Strong linearizations of polynomial matrices

In this subsection, we recall the classical definitions of linearization and strong linearization of polynomial matrices [42, 44].

Let $P(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ be a polynomial matrix. A linear polynomial matrix $L(\lambda) = B\lambda + A \in \mathbb{F}[\lambda]^{(p+s) \times (m+s)}$ is a *linearization* of $P(\lambda)$ in the sense of Gohberg, Lancaster and Rodman [44], or in the GLR-sense for short, if there exist unimodular matrices $U_1(\lambda) \in \mathbb{F}[\lambda]^{(p+s) \times (p+s)}$ and $U_2(\lambda) \in \mathbb{F}[\lambda]^{(m+s) \times (m+s)}$ such that

$$U_1(\lambda)L(\lambda)U_2(\lambda) = \begin{bmatrix} P(\lambda) & 0\\ 0 & I_s \end{bmatrix},$$

where I_s denotes the identity matrix of size any integer $s \ge 0$. The key property of a GLR-linearization is that it has the same finite eigenvalues with the same partial multiplicities as $P(\lambda)$. Furthermore, a linearization $L(\lambda)$ is said to be a strong linearization in the GLR-sense if rev $L(\lambda) = A\lambda + B$ is a linearization of the polynomial matrix rev_g $P(\lambda) = \lambda^g P(1/\lambda)$, where g is an integer greater than or equal to the degree of $P(\lambda)$, i.e., $g \ge \deg(P(\lambda))$, and is called grade of $P(\lambda)$ (see, for instance, [22]). Then, not only finite but infinite eigenvalues can be recovered from strong linearizations in the GLR-sense, together with their partial multiplicities. However, the minimal indices of a (strong) linearization $L(\lambda)$ in the GLR-sense may be completely unrelated to those of $P(\lambda)$ [22, Section 4], except for the fact that the number of left (resp. right) minimal indices of $L(\lambda)$ and $P(\lambda)$ are equal. Nevertheless, the strong linearizations in the GLR-sense that are used in practice have minimal indices that are simply related to those of the polynomial matrix through addition of a constant shift (see [26] and the references therein).

In [26] a wide family of (strong) linearizations for polynomial matrices is constructed, which are called (strong) block minimal bases pencils [26, Definition 3.1] and that uses the notion of minimal basis (recall Section 2.3). We introduce the family of (strong) block minimal bases pencils in the following definition, as it will be useful throughout this dissertation.

Definition 2.5.1. ((Strong) block minimal bases pencil) A block minimal bases pencil is a linear polynomial matrix over \mathbb{F} with the following structure

$$L(\lambda) = \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix},$$
(2.12)

where $K_1(\lambda)$ and $K_2(\lambda)$ are both minimal bases. Moreover, given a polynomial matrix $P(\lambda)$, it is said that $L(\lambda)$ is associated with $P(\lambda)$ if

$$N_2(\lambda)M(\lambda)N_1(\lambda)^T = P(\lambda),$$

where $N_1(\lambda)$ and $N_2(\lambda)$ are minimal bases dual to $K_1(\lambda)$ and $K_2(\lambda)$, respectively. In addition, if $K_1(\lambda)$ (resp. $K_2(\lambda)$) is a minimal basis with all its row degrees equal to 1 and with the row degrees of a minimal basis $N_1(\lambda)$ (resp. $N_2(\lambda)$) dual to $K_1(\lambda)$ (resp. $K_2(\lambda)$) all equal, then $L(\lambda)$ is called a strong block minimal bases pencil. **Theorem 2.5.2.** [26] A block minimal bases pencil $L(\lambda)$ as in (2.12) associated with a polynomial matrix $P(\lambda)$ is a linearization of $P(\lambda)$. Moreover, if the block minimal bases pencil $L(\lambda)$ is strong, then $L(\lambda)$ is a strong linearization of $P(\lambda)$ considered as a polynomial matrix of grade deg $N_1(\lambda) + \text{deg } N_2(\lambda) + 1$.

2.5.2 Strong linearizations of rational matrices

Next we present the definition of strong linearization for a rational matrix given in [6]. This definition contains the notion of first invariant order at infinity q_1 of a rational matrix $R(\lambda)$. For any non strictly proper rational matrix this number is $-\deg(D(\lambda))$ where $D(\lambda)$ is the polynomial part of $R(\lambda)$ in the expression (2.1); otherwise, $q_1 > 0$. More information can be found in [10, 6, 90].

Definition 2.5.3. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$. Let q_1 be its first invariant order at infinity and $g = \min(0, q_1)$. Let $n = \nu(R(\lambda))$. A strong linearization of $R(\lambda)$ is a linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1\lambda + A_0 & B_1\lambda + B_0 \\ -(C_1\lambda + C_0) & D_1\lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q)\times(n+r)}$$
(2.13)

such that the following conditions hold:

- (a) if n > 0 then $det(A_1\lambda + A_0) \neq 0$, and
- (b) if $\widehat{R}(\lambda) = (D_1\lambda + D_0) + (C_1\lambda + C_0)(A_1\lambda + A_0)^{-1}(B_1\lambda + B_0)$, \widehat{q}_1 is its first invariant order at infinity and $\widehat{g} = \min(0, \widehat{q}_1)$ then:
 - (i) there exist nonnegative integers s_1, s_2 , with $s_1 s_2 = q p = r m$, and unimodular matrices $U_1(\lambda) \in \mathbb{F}[\lambda]^{(p+s_1) \times (p+s_1)}$ and $U_2(\lambda) \in \mathbb{F}[\lambda]^{(m+s_1) \times (m+s_1)}$ such that

$$U_1(\lambda) \operatorname{diag}(R(\lambda), I_{s_1})U_2(\lambda) = \operatorname{diag}(\widehat{R}(\lambda), I_{s_2}), and$$

(ii) there exist biproper matrices $B_1(\lambda) \in \mathbb{F}_{pr}(\lambda)^{(p+s_1)\times(p+s_1)}$ and $B_2(\lambda) \in \mathbb{F}_{pr}(\lambda)^{(m+s_1)\times(m+s_1)}$ such that

$$B_1(\lambda)\operatorname{diag}(\lambda^g R(\lambda), I_{s_1})B_2(\lambda) = \operatorname{diag}(\lambda^{\widehat{g}}\widehat{R}(\lambda), I_{s_2}).$$

If, instead of $n = \nu(R(\lambda))$, in Definition 2.5.3 $L(\lambda)$ is considered to be a minimal polynomial system matrix and only condition (i) in part (b) holds, then $\mathcal{L}(\lambda)$ is just called *linearization* of $R(\lambda)$ in [6]. In addition, it is known [6] that if condition (i) in Definition 2.5.3 holds, then condition (ii) is equivalent to the existence of unimodular matrices $W_1(\lambda)$ and $W_2(\lambda)$ such that

$$W_1(\lambda) \operatorname{diag}\left(\frac{1}{\lambda^g} R\left(\frac{1}{\lambda}\right), I_{s_1}\right) W_2(\lambda) = \operatorname{diag}\left(\frac{1}{\lambda^{\widehat{g}}} \widehat{R}\left(\frac{1}{\lambda}\right), I_{s_2}\right).$$
 (2.14)

In Definition 2.5.3 it can always be taken $s_1 = 0$ or $s_2 = 0$, according to $p \ge q$ and $m \ge r$ or $q \ge p$ and $r \ge m$. We now consider $s_1 \ge 0$ and $s_2 = 0$. Notice that with this choice and with the notion of reversal given in Definition 2.2.1, (2.14) is equivalent to

$$W_1(\lambda) \operatorname{diag}\left(\operatorname{rev} R(\lambda), I_{s_1}\right) W_2(\lambda) = \operatorname{rev} \widehat{R}(\lambda).$$
(2.15)

Remark 2.5.4. Notice that Definition 2.5.3 extends the notion of strong linearization of polynomial matrices in the usual GLR-sense [63, Definition 2.5]. In particular, if $R(\lambda)$ is a polynomial matrix, then $n = \nu(R(\lambda)) = 0$. Therefore, a strong linearization $\mathcal{L}(\lambda)$ of $R(\lambda)$ is of the form $\mathcal{L}(\lambda) = D_1\lambda + D_0$, with $\widehat{R}(\lambda) = \mathcal{L}(\lambda)$, $g = q_1 = -\deg(R(\lambda))$ and $\widehat{g} = \widehat{q}_1 = -\deg(\mathcal{L}(\lambda))$.

By considering rational matrices expressed as in (2.1) and strong block minimal bases pencils associated to their polynomial parts, in [6, Theorem 5.11] is given a method to construct strong linearizations for rational matrices in the sense of Definition 2.5.3, which are called strong block minimal bases linearizations.

Theorem 2.5.5. [6] Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix, let $R(\lambda) = D(\lambda) + R_{sp}(\lambda)$ be its unique decomposition into its polynomial part $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ and its strictly proper part $R_{sp}(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$, and let $R_{sp}(\lambda) = C(\lambda I_n - A)^{-1}B$ be a minimal state-space realization of $R_{sp}(\lambda)$, where $n = \nu(R(\lambda)) = \nu(R_{sp}(\lambda))$. Assume that deg $(D(\lambda)) \geq 2$ and let

$$L(\lambda) = \left[\begin{array}{cc} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{array} \right]$$

be a strong block minimal bases pencil associated to $D(\lambda)$. Let $N_1(\lambda)$ and $N_2(\lambda)$ be minimal bases dual to $K_1(\lambda)$ and $K_2(\lambda)$, respectively, such that $N_2(\lambda)M(\lambda)N_1(\lambda)^T = D(\lambda)$ and $\deg(N_1(\lambda)) + \deg(N_2(\lambda)) + 1 = \deg(D(\lambda))$. Let \widehat{K}_1 and \widehat{K}_2 be constant matrices such that the matrices

$$U_i(\lambda) = \begin{bmatrix} K_i(\lambda) \\ \widehat{K}_i \end{bmatrix} \quad and \quad U_i(\lambda)^{-1} = \begin{bmatrix} \widehat{N}_i(\lambda)^T & N_i(\lambda)^T \end{bmatrix},$$

are unimodular for i = 1, 2. Then, for any nonsingular constant matrices $X, Y \in \mathbb{F}^{n \times n}$, the linear polynomial matrix

$$\mathcal{L}(\lambda) := \begin{bmatrix} X(\lambda I_n - A)Y & XB\widehat{K}_1 & 0\\ \hline -\widehat{K}_2^T CY & M(\lambda) & K_2(\lambda)^T\\ 0 & K_1(\lambda) & 0 \end{bmatrix},$$

is a strong linearization of $R(\lambda)$.

We will introduce a definition of local linearization of rational matrices in Chapter 4 that includes the above definitions of linearizations and strong linearizations of polynomial and rational matrices. In particular, we introduce in Chapter 4 linearizations satisfying minimality conditions that preserve pole and zero structures in a particular target set and/or at infinity. Then, in Chapter 4 a linearization will be said to be strong if the minimality conditions are satisfied in the whole underlying field and also at infinity. However, we will make a different and more flexible treatment of the conditions for a pencil to be a linearization in the strong sense, which allows us to construct linearizations for polynomial and rational matrices that can not be constructed with the existing definitions of strong linearizations in the literature. Examples of such linearizations are the linearizations introduced in Chapter 6, for arbitrary rational matrices from general representations, and the linearizations for polynomial and rational matrices preserving structures in Chapter 8.

Chapter 3

Strong linearizations of rational matrices with polynomial part expressed in an orthogonal basis

All the results in this chapter were introduced in [27], where a new family of strong linearizations of rational matrices is constructed according to Definition 2.5.3. For that, the corresponding rational matrix is considered as the sum of its polynomial part and its strictly proper part, and the polynomial part is expressed in a basis that satisfies a three term recurrence relation. Then, we combine the theory developed in [7] and the linearizations of polynomial matrices introduced in [36]. We also show how to recover eigenvectors of a rational matrix from those of its linearizations in this family. In addition, we present strong linearizations that preserve the structure of symmetric or Hermitian rational matrices. We complete the chapter by discussing how to extend the results in this chapter when the polynomial part is expressed in other bases. After the publication of [27], the recovery of minimal bases and minimal indices of the linearizations developed here were studied by other authors in [8].

A conclusion of the results presented in this chapter is that the combination of them with those in [7], allows us to use essentially all the strong linearizations of polynomial matrices developed in the literature to construct strong linearizations of any rational matrix in the sense of Definition 2.5.3, by expressing such a matrix in terms of its polynomial and strictly proper parts.

3.1 Some preliminaries

In the definition of strong linearization in [7] (recall Definition 2.5.3), it may seem that the least order $\nu(R(\lambda))$ has to be previously known in order to verify that a linear polynomial matrix as in (2.13) is a strong linearization of the rational matrix $R(\lambda)$. However, there are conditions to ensure that the size of $A_1\lambda + A_0$ is $n = \nu(R(\lambda))$. We state them in Proposition 3.1.1.

Proposition 3.1.1. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and let

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1 \lambda + A_0 & B_1 \lambda + B_0 \\ -(C_1 \lambda + C_0) & D_1 \lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q) \times (n+r)}$$

be a linear polynomial matrix with n > 0 and $\det(A_1\lambda + A_0) \neq 0$. Assume that there exist nonnegative integers s_1, s_2 , with $s_1 - s_2 = q - p = r - m$, and unimodular matrices $U_1(\lambda) \in \mathbb{F}[\lambda]^{(p+s_1)\times(p+s_1)}$ and $U_2(\lambda) \in \mathbb{F}[\lambda]^{(m+s_1)\times(m+s_1)}$ such that

 $U_1(\lambda)\operatorname{diag}(R(\lambda), I_{s_1})U_2(\lambda) = \operatorname{diag}(\widehat{R}(\lambda), I_{s_2}), \qquad (3.1)$

where $\widehat{R}(\lambda) = (D_1\lambda + D_0) + (C_1\lambda + C_0)(A_1\lambda + A_0)^{-1}(B_1\lambda + B_0)$. Then $n = \nu(R(\lambda))$ if and only if the following conditions hold:

a) A_1 is invertible, and

b) rank
$$\begin{bmatrix} A_1\mu + A_0\\ C_1\mu + C_0 \end{bmatrix}$$
 = rank $\begin{bmatrix} A_1\mu + A_0 & B_1\mu + B_0 \end{bmatrix}$ = n for all $\mu \in \overline{\mathbb{F}}$.

Proof. Condition b) is equivalent to $\mathcal{L}(\lambda)$ being a minimal polynomial system matrix, since det $(A_1\lambda + A_0) \neq 0$, see [78, Chapters 2 and 3]. By condition (3.1) and [6, Lemma 2.1], we have that $\nu(R(\lambda)) = \nu(\widehat{R}(\lambda))$. Assume that $n = \nu(R(\lambda))$. Thus, $\nu(\widehat{R}(\lambda)) = n$, and deg $(\det(A_1\lambda + A_0)) \geq \nu(\widehat{R}(\lambda)) = n$. However, deg $(\det(A_1\lambda + A_0)) \leq n$. Therefore, deg $(\det(A_1\lambda + A_0)) = n$ and deg $(\det(A_1\lambda + A_0)) = \nu(\widehat{R}(\lambda))$, which imply conditions a) and b), respectively. We assume now that conditions a) and b) hold. On the one hand, A_1 being invertible implies that deg $(\det(A_1\lambda + A_0)) =$ n. On the other hand, $\mathcal{L}(\lambda)$ being a minimal polynomial system matrix means that deg $(\det(A_1\lambda + A_0)) = \nu(\widehat{R}(\lambda))$. Therefore, $n = \nu(\widehat{R}(\lambda)) = \nu(R(\lambda))$.

Key Lemma Lemma 3.1.2 follows from Definition 2.5.3. It shows an easy way to obtain strong linearizations for a rational matrix $R(\lambda)$ from a particular strong linearization $\mathcal{L}(\lambda)$ by multiplying $\mathcal{L}(\lambda)$ by some appropriate matrices. This simple result is fundamental in this work, and we conjecture that it will be fundamental for constructing other families of strong linearizations of rational matrices.

Lemma 3.1.2. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix, and let

$$\mathcal{L}_1(\lambda) = \begin{bmatrix} A_1\lambda + A_0 & B_1\lambda + B_0 \\ -(C_1\lambda + C_0) & D_1\lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+(p+s))\times(n+(m+s))}$$

be a strong linearization of $R(\lambda)$. Consider $Q_1, Q_3 \in \mathbb{F}^{n \times n}$, $Q_2 \in \mathbb{F}^{(p+s) \times (p+s)}$, $Q_4 \in \mathbb{F}^{(m+s) \times (m+s)}$ nonsingular matrices, $W \in \mathbb{F}^{(p+s) \times n}$, and $Z \in \mathbb{F}^{n \times (m+s)}$. Then the linear polynomial matrix

$$\mathcal{L}_2(\lambda) = \left[\begin{array}{cc} Q_1 & 0 \\ W & Q_2 \end{array} \right] \mathcal{L}_1(\lambda) \left[\begin{array}{cc} Q_3 & Z \\ 0 & Q_4 \end{array} \right]$$

is a strong linearization of $R(\lambda)$.

Proof. Let us write

$$\mathcal{L}_2(\lambda) = \begin{bmatrix} A_2\lambda + \tilde{A}_0 & B_2\lambda + \tilde{B}_0 \\ -(C_2\lambda + \tilde{C}_0) & D_2\lambda + \tilde{D}_0 \end{bmatrix}.$$

We have $\det(A_2\lambda + \tilde{A}_0) \neq 0$ if n > 0, since $A_2\lambda + \tilde{A}_0 = Q_1(A_1\lambda + A_0)Q_3$. Let us consider the transfer functions $\hat{R}_1(\lambda)$, $\hat{R}_2(\lambda)$ of $\mathcal{L}_1(\lambda)$, $\mathcal{L}_2(\lambda)$, respectively. They satisfy $\hat{R}_2(\lambda) = Q_2\hat{R}_1(\lambda)Q_4$. Let q_1 be the first invariant order at infinity of $R(\lambda)$ and $g = \min(0, q_1)$. For i = 1, 2, let $\hat{g}_i = \min(0, \hat{q}_i)$, where \hat{q}_i is the first invariant order at infinity of $\hat{R}_i(\lambda)$. Since $\mathcal{L}_1(\lambda)$ is a strong linearization of $R(\lambda)$, there exist unimodular matrices $U_1(\lambda)$ and $U_2(\lambda)$ such that $U_1(\lambda) \operatorname{diag}(R(\lambda), I_s)U_2(\lambda) = \hat{R}_1(\lambda)$, and biproper matrices $B_1(\lambda)$ and $B_2(\lambda)$ such that $B_1(\lambda) \operatorname{diag}(\lambda^g R(\lambda), I_s)B_2(\lambda) = \lambda^{\hat{g}_1}\hat{R}_1(\lambda)$. By using the equality $\hat{R}_2(\lambda) = Q_2\hat{R}_1(\lambda)Q_4$, we have that $\hat{g}_1 = \hat{g}_2$, and by the same equality, we get

$$Q_2 U_1(\lambda) \operatorname{diag}(R(\lambda), I_s) U_2(\lambda) Q_4 = R_2(\lambda),$$

and

$$Q_2 B_1(\lambda) \operatorname{diag}(\lambda^g R(\lambda), I_s) B_2(\lambda) Q_4 = \lambda^{\widehat{g}_2} \widehat{R}_2(\lambda).$$

Then we obtain that conditions (a) and (b) in Definition 2.5.3 hold for $\mathcal{L}_2(\lambda)$.

By using Lemma 3.1.2, strong linearizations of a rational matrix $R(\lambda)$ expressed in the form (2.1) can be constructed from combining minimal state-space realizations of the strictly proper part $R_{sp}(\lambda)$ and strong linearizations of its polynomial part $D(\lambda)$. In particular, strong block minimal bases pencils associated to $D(\lambda)$ [26] can be used to construct a strong linearization for $R(\lambda)$, by using Theorem 2.5.5. Then one can use Lemma 3.1.2 to construct other strong linearizations for $R(\lambda)$ taking into account that strong block minimal bases linearizations of polynomial matrices unify many of the strong linearizations for polynomial matrices existing in the literature, in the sense that most of them are strictly equivalent to strong block minimal bases linearizations [26].

Remark 3.1.3. A first application of the key Lemma 3.1.2 is to construct strong linearizations of a rational matrix $R(\lambda)$ from any Fiedler-like strong linearization $L_F(\lambda)$ of its polynomial part $D(\lambda)$. For this purpose, note that [15, Theorems 3.8, 3.15, 3.16] guarantee that there exist permutation matrices Π_1 and Π_2 and a strong block minimal bases pencil $L(\lambda)$ associated to $D(\lambda)$ such that $L_F(\lambda) = \Pi_1 L(\lambda) \Pi_2$. In addition, Theorem 2.5.5 explains how to construct a strong linearization $\mathcal{L}(\lambda)$ of $R(\lambda)$ from $L(\lambda)$. Thus, according to Lemma 3.1.2, diag $(I_n, \Pi_1)\mathcal{L}(\lambda)$ diag (I_n, Π_2) is a strong linearization of $R(\lambda)$ based on $L_F(\lambda)$. This idea is used in [8], where it is proved that the families of Fiedler-like linearizations of rational matrices introduced in [2, 4, 17, 19] are, modulo permutations, particular instances of strong block minimal bases linearizations.

3.2 M_1 -strong linearizations

In this section and in Section 3.3, we present strong linearizations of square rational matrices $R(\lambda)$ with polynomial part $D(\lambda)$ expressed in an orthogonal basis. More precisely, we consider strong linearizations of $D(\lambda)$ that belong to the ansatz spaces $\mathbb{M}_1(D)$ or $\mathbb{M}_2(D)$, developed by H. Faßbender and P. Saltenberger in [36], and based on them, we construct strong linearizations of $R(\lambda)$ by using Lemma 3.1.2 and the strong linearizations presented in [6, Section 5.2].

We consider an arbitrary field \mathbb{F} throughout this chapter, although the results in [36] are stated only for the real field \mathbb{R} . Nevertheless, the results of [36] that are used in this chapter are also valid for any field \mathbb{F} . We consider a polynomial basis $\{\phi_j(\lambda)\}_{j=0}^{\infty}$ of $\mathbb{F}[\lambda]$, viewed as an \mathbb{F} -vector space, with $\phi_j(\lambda)$ a polynomial of degree j, that satisfies the following three-term recurrence relation:

$$\alpha_j \phi_{j+1}(\lambda) = (\lambda - \beta_j) \phi_j(\lambda) - \gamma_j \phi_{j-1}(\lambda) \quad j \ge 0$$
(3.2)

where $\alpha_j, \beta_j, \gamma_j \in \mathbb{F}, \alpha_j \neq 0, \phi_{-1}(\lambda) = 0$, and $\phi_0(\lambda) = 1$. Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix of degree k written in terms of this basis as follows

$$P(\lambda) = P_k \phi_k(\lambda) + P_{k-1} \phi_{k-1}(\lambda) + \dots + P_1 \phi_1(\lambda) + P_0 \phi_0(\lambda).$$
(3.3)

We define $\Phi_k(\lambda) = [\phi_{k-1}(\lambda) \cdots \phi_1(\lambda) \phi_0(\lambda)]^T$ and $V_P = \{v \otimes P(\lambda) : v \in \mathbb{F}^k\}$, and we consider the set of pencils

$$\mathbb{M}_1(P) = \{ L(\lambda) = \lambda X + Y : X, Y \in \mathbb{F}^{km \times km}, \ L(\lambda)(\Phi_k(\lambda) \otimes I_m) \in V_P \}.$$

A pencil $L(\lambda) \in \mathbb{M}_1(P)$, which verifies $L(\lambda)(\Phi_k(\lambda) \otimes I_m) = v \otimes P(\lambda)$ for some vector $v \in \mathbb{F}^k$, is said to have *right ansatz vector* v. A particular pencil in $\mathbb{M}_1(P)$ introduced in [36, page 63] is

$$F_{\Phi}^{P}(\lambda) = \begin{bmatrix} m_{\Phi}^{P}(\lambda) \\ M_{\Phi}(\lambda) \otimes I_{m} \end{bmatrix} \in \mathbb{F}[\lambda]^{km \times km}, \qquad (3.4)$$

where

$$m_{\Phi}^{P}(\lambda) = \begin{bmatrix} \frac{(\lambda - \beta_{k-1})}{\alpha_{k-1}} P_{k} + P_{k-1} & P_{k-2} - \frac{\gamma_{k-1}}{\alpha_{k-1}} P_{k} & P_{k-3} & \cdots & P_{1} & P_{0} \end{bmatrix},$$

and

$$M_{\Phi}(\lambda) = \begin{bmatrix} -\alpha_{k-2} & (\lambda - \beta_{k-2}) & -\gamma_{k-2} \\ & -\alpha_{k-3} & (\lambda - \beta_{k-3}) & -\gamma_{k-3} \\ & \ddots & \ddots & \ddots \\ & & -\alpha_1 & (\lambda - \beta_1) & -\gamma_1 \\ & & & -\alpha_0 & (\lambda - \beta_0) \end{bmatrix}.$$

Since $m_{\Phi}^{P}(\lambda)(\Phi_{k}(\lambda) \otimes I_{m}) = P(\lambda)$ and $(M_{\Phi}(\lambda) \otimes I_{m})(\Phi_{k}(\lambda) \otimes I_{m}) = 0$, we get that $F_{\Phi}^{P}(\lambda)(\Phi_{k}(\lambda) \otimes I_{m}) = e_{1} \otimes P(\lambda)$, where e_{1} is the first canonical vector of \mathbb{F}^{k} . Therefore, $F_{\Phi}^{P}(\lambda) \in \mathbb{M}_{1}(P)$ with right ansatz vector $e_{1} \in \mathbb{F}^{k}$. This particular example is very important because, by using it, we can obtain all the elements in $\mathbb{M}_{1}(P)$. This follows from the next theorem.

Theorem 3.2.1. [36, Theorem 1] Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix with degree $k \geq 2$. Then $L(\lambda) \in \mathbb{M}_1(P)$ with right ansatz vector $v \in \mathbb{F}^k$ if and only if

$$L(\lambda) = \begin{bmatrix} v \otimes I_m & H \end{bmatrix} F_{\Phi}^P(\lambda)$$

for some matrix $H \in \mathbb{F}^{km \times (k-1)m}$.

Remark 3.2.2. For the monomial basis $\{\phi_j(\lambda) = \lambda^j\}_{j=0}^{\infty}$ the space $\mathbb{M}_1(P)$ is denoted $\mathbb{L}_1(P)$ (see [63]). In this case $\alpha_j = 1$ and $\beta_j = \gamma_j = 0$ for all $j \ge 0$ in (3.2) and the matrix $F_{\Phi}^P(\lambda)$ is the first companion form of $P(\lambda)$.

It is known that $F_{\Phi}^{P}(\lambda)$ is a strong linearization of $P(\lambda)$ (see [5, Theorem 2] for regular polynomial matrices $P(\lambda)$, and [21, Section 7] for singular), but we can obtain this property as an immediate corollary of the next result.

Lemma 3.2.3. $F_{\Phi}^{P}(\lambda)$ is a strong block minimal bases pencil with only one block column associated to $P(\lambda)$ with sharp degree. Moreover, $\Phi_k(\lambda)^T \otimes I_m$ is a minimal basis dual to the minimal basis $M_{\Phi}(\lambda) \otimes I_m$.

Proof. Let us denote $M(\lambda) = m_{\Phi}^{P}(\lambda)$ and $K(\lambda) = M_{\Phi}(\lambda) \otimes I_{m}$. We consider

$$F_{\Phi}^{P}(\lambda) = \begin{bmatrix} M(\lambda) \\ K(\lambda) \end{bmatrix}$$

Note that $M_{\Phi}(\lambda_0)$ has full row rank for all $\lambda_0 \in \overline{\mathbb{F}}$ because $\alpha_i \neq 0$ for all $i \geq 0$. Also note that $M_{\Phi}(\lambda)$ is row reduced because its highest row degree coefficient matrix

$$[M_{\Phi}]_{hr} = \begin{bmatrix} 0 & 1 & 0 & & & \\ 0 & 1 & 0 & & \\ & \ddots & \ddots & \ddots & \\ & & 0 & 1 & 0 \\ & & & & 0 & 1 \end{bmatrix}$$

has full row rank. We conclude that $M_{\Phi}(\lambda)$ is a minimal basis, and therefore, $K(\lambda) = M_{\Phi}(\lambda) \otimes I_m$ is also a minimal basis [26, Corollary 2.4]. Let us denote $N(\lambda) = \Phi_k(\lambda)^T \otimes I_m$. Note that $\Phi_k(\lambda)^T$ is a minimal basis because $\phi_0(\lambda) = 1$, so $\Phi_k(\lambda_0)$ has rank 1 for all $\lambda_0 \in \overline{\mathbb{F}}$, and

$$[\Phi_k^T]_{hr} = \begin{bmatrix} \frac{1}{\alpha_0 \alpha_1 \cdots \alpha_{k-2}} & 0 & \cdots & 0 \end{bmatrix}$$

has also rank 1. Therefore, $N(\lambda) = \Phi_k(\lambda)^T \otimes I_m$ is also a minimal basis. Since $K(\lambda)N(\lambda)^T = (M_{\Phi}(\lambda) \otimes I_m)(\Phi_k(\lambda) \otimes I_m) = 0$ and $\begin{bmatrix} K(\lambda) \\ N(\lambda) \end{bmatrix}$ is a square matrix, we have that $K(\lambda)$ and $N(\lambda)$ are dual minimal bases. In addition, it is obvious that all the row degrees of $K(\lambda)$ are equal to 1 and all the row degrees of $\Phi_k(\lambda)^T \otimes I_m$ are equal to k - 1. Hence, $F_{\Phi}^P(\lambda)$ is a strong block minimal bases pencil associated to the polynomial matrix $M(\lambda)N(\lambda)^T = m_{\Phi}^P(\lambda)(\Phi_k(\lambda) \otimes I_m) = P(\lambda)$ and $\deg(P(\lambda)) = 1 + \deg(N(\lambda))$, which means that $F_{\Phi}^P(\lambda)$ has sharp degree.

Since every strong block minimal bases pencil is a strong linearization (see [26, Theorem 3.3]), the following corollary is straightforward.

Corollary 3.2.4. $F_{\Phi}^{P}(\lambda)$ is a strong linearization for $P(\lambda)$.

The proof of the next result is trivial because if $L(\lambda) = \begin{bmatrix} v \otimes I_m & H \end{bmatrix} F_{\Phi}^P(\lambda)$ with $\begin{bmatrix} v \otimes I_m & H \end{bmatrix}$ nonsingular then $L(\lambda)$ is strictly equivalent to $F_{\Phi}^P(\lambda)$.

Corollary 3.2.5. [36, Corollary 2.1] Let $L(\lambda) = [v \otimes I_m \quad H]F_{\Phi}^P(\lambda) \in \mathbb{M}_1(P)$. If $[v \otimes I_m \quad H]$ is nonsingular then $L(\lambda)$ is a strong linearization for $P(\lambda)$.

Remark 3.2.6. Although $F_{\Phi}^{P}(\lambda)$ is a strong block minimal bases pencil associated to $P(\lambda)$ this structure is not preserved in general when we multiply on the left by a nonsingular matrix $[v \otimes I_m \ H]$. For example, consider the polynomial matrix $P(\lambda) = I\lambda^3 + 2I\lambda^2 + I\lambda + S \in \mathbb{R}[\lambda]^{2\times 2}$ expressed in the monomial basis, where $S = \begin{bmatrix} 1 & 0 \\ 0 & 0 \end{bmatrix}$ and I stands for I_2 . In this case, the matrix $F_{\Phi}^{P}(\lambda)$ is $F_{\Phi}^{P}(\lambda) = \begin{bmatrix} \lambda I + 2I \ I \ S \\ -I \ \lambda I \ 0 \\ 0 \ -I \ \lambda I \end{bmatrix}$. Let $v = \begin{bmatrix} 1 & 1 \ 0 \end{bmatrix}^T$ and $H = \begin{bmatrix} 0 & 0 \\ I & 0 \\ 0 \ I \end{bmatrix}$ and let $L(\lambda) = \begin{bmatrix} \lambda I + 2I \ I \ S \\ \lambda I + I \ \lambda I + I \ S \\ 0 \ -I \ \lambda I \end{bmatrix}$. Notice that if $L(\lambda)$ were a strong

block minimal bases pencil associated to $P(\lambda)$, one of these two different situations would happen in (2.12):

1.
$$M(\lambda) = \begin{bmatrix} \lambda I + 2I & I & S \end{bmatrix}, K_1(\lambda) = \begin{bmatrix} \lambda I + I & \lambda I + I & S \\ 0 & -I & \lambda I \end{bmatrix}$$
 and $K_2(\lambda)$ empty.
2. $M(\lambda) = \begin{bmatrix} \lambda I + 2I \\ \lambda I + I \\ 0 \end{bmatrix}, K_2(\lambda)^T = \begin{bmatrix} I & S \\ \lambda I + I & S \\ -I & \lambda I \end{bmatrix}$ and $K_1(\lambda)$ empty.

In the first case, the matrix $K_1(\lambda)$ does not have full row rank for $\lambda = -1$. In the second case, the matrix $K_2(\lambda)$ does not have full row rank for $\lambda = 0$. Therefore, $L(\lambda)$ is not a strong block minimal bases pencil associated to $P(\lambda)$.

From the fact that $F_{\Phi}^{P}(\lambda)$ is a strong block minimal bases pencil, we can obtain strong linearizations for rational matrices by applying Theorem 5.11 in [6] (see Theorem 2.5.5). For this purpose, we prove first the following lemma.

Lemma 3.2.7. The matrix

$$U(\lambda) = \begin{bmatrix} M_{\Phi}(\lambda) \otimes I_m \\ e_k^T \otimes I_m \end{bmatrix} = \begin{bmatrix} M_{\Phi}(\lambda) \\ e_k^T \end{bmatrix} \otimes I_m$$

is unimodular, and its inverse has the form $U(\lambda)^{-1} = [\widehat{\Phi}_k(\lambda) \quad \Phi_k(\lambda) \otimes I_m]$ with $\widehat{\Phi}_k(\lambda) \in \mathbb{F}[\lambda]^{km \times (k-1)m}$.

Proof. Let us consider the matrix

$$\tilde{U}(\lambda) = \begin{bmatrix} M_{\Phi}(\lambda) \\ e_k^T \end{bmatrix} = \begin{bmatrix} -\alpha_{k-2} & (\lambda - \beta_{k-2}) & -\gamma_{k-2} \\ & -\alpha_{k-3} & (\lambda - \beta_{k-3}) & -\gamma_{k-3} \\ & \ddots & \ddots & \ddots \\ & & -\alpha_1 & (\lambda - \beta_1) & -\gamma_1 \\ & & & -\alpha_0 & (\lambda - \beta_0) \\ 0 & & \cdots & 0 & 1 \end{bmatrix}$$

Since $\tilde{U}(\lambda)$ is upper triangular, its determinant is $(-\alpha_{k-2})\cdots(-\alpha_0)$, i.e., a constant different from zero. Therefore, $\tilde{U}(\lambda)$ is unimodular. Finally, note that $\tilde{U}(\lambda)\Phi_k(\lambda) = e_k \in \mathbb{F}^k$. Thus $\Phi_k(\lambda)$ is the last column of $\tilde{U}(\lambda)^{-1}$.

Theorem 3.2.8. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix, let $R(\lambda) = D(\lambda) + R_{sp}(\lambda)$ be its unique decomposition into its polynomial part $D(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ and its strictly proper part $R_{sp}(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$, and let $R_{sp}(\lambda) = C(\lambda I_n - A)^{-1}B$ be a minimal state-space realization of $R_{sp}(\lambda)$. Assume that $\deg(D(\lambda)) \geq 2$. Write $D(\lambda)$ in terms of the polynomial basis $\{\phi_j(\lambda)\}_{j=0}^{\infty}$ satisfying the three-term recurrence relation (3.2), as

$$D(\lambda) = D_k \phi_k(\lambda) + D_{k-1} \phi_{k-1}(\lambda) + \dots + D_1 \phi_1(\lambda) + D_0 \phi_0(\lambda)$$
(3.5)

with $D_k \neq 0$, and let $F_{\Phi}^D(\lambda)$ be the matrix pencil defined as in (3.4). Then, for any nonsingular matrices $X, Y \in \mathbb{F}^{n \times n}$ the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -CY & F_{\Phi}^D(\lambda) \\ 0_{(k-1)m \times n} & F_{\Phi}^D(\lambda) \end{bmatrix}$$

is a strong linearization of $R(\lambda)$.

Proof. Lemmas 3.2.3 and 3.2.7 allow us to apply [6, Theorem 5.11], with $K_1(\lambda) = M_{\Phi}(\lambda) \otimes I_m$, $\widehat{K}_1 = e_k^T \otimes I_m$, $K_2(\lambda)^T$ empty and $\widehat{K}_2^T = I_m$.

Then, from combining Lemma 3.1.2 and Theorem 3.2.8 we obtain strong linearizations of a rational matrix from strong linearizations in $\mathbb{M}_1(D)$ of its polynomial part.

Theorem 3.2.9. Under the same assumptions as in Theorem 3.2.8, let $v \in \mathbb{F}^k$, $H \in \mathbb{F}^{km \times (k-1)m}$ with $[v \otimes I_m \quad H]$ nonsingular and let $L(\lambda) = [v \otimes I_m \quad H]F_{\Phi}^D(\lambda) \in \mathbb{M}_1(D)$. Then, the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

is a strong linearization of $R(\lambda)$.

Proof. Set $K = \begin{bmatrix} v \otimes I_m & H \end{bmatrix}$. If K is nonsingular then, by Lemma 3.1.2 and Theorem 3.2.8,

$$\mathcal{L}(\lambda) = \begin{bmatrix} I_n & 0\\ 0 & K \end{bmatrix} \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -CY & F_{\Phi}^D(\lambda) \end{bmatrix}$$
$$= \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}.$$

is a strong linearization of $R(\lambda)$.

The strong linearizations of square rational matrices constructed in Theorem 3.2.9 will be called \mathbb{M}_1 -strong linearizations.

3.3 M_2 -strong linearizations

In this section we obtain strong linearizations of a square rational matrix from the transposed version of $\mathbb{M}_1(P)$, where $P(\lambda)$ is the polynomial matrix in (3.3). Since the proofs of the results are similar to those in Section 3.2, they are omitted for brevity. We define $W_P = \{w^T \otimes P(\lambda) : w \in \mathbb{F}^k\}$, and we consider the set of pencils

$$\mathbb{M}_2(P) = \{ L(\lambda) = \lambda X + Y : X, Y \in \mathbb{F}^{km \times km}, \ (\Phi_k(\lambda)^T \otimes I_m) L(\lambda) \in W_P \}$$

A pencil $L(\lambda) \in \mathbb{M}_2(P)$, which verifies $(\Phi_k(\lambda)^T \otimes I_m)L(\lambda) = w^T \otimes P(\lambda)$ for some vector $w \in \mathbb{F}^k$, is said to have *left ansatz vector* w. Pencils in $\mathbb{M}_2(P)$ are characterized in [36, Theorem 2]. We need the definition of the block-transpose of a $km \times lm$ pencil $L(\lambda)$. If we express $L(\lambda)$ as $L(\lambda) = \sum_{i=1}^k \sum_{j=1}^l e_i e_j^T \otimes L_{ij}(\lambda)$ for certain $m \times m$ pencils $L_{ij}(\lambda)$, where e_i denotes the *i*th canonical vector in \mathbb{F}^k , and e_j the *j*th canonical vector in \mathbb{F}^l , $L(\lambda)^{\mathcal{B}} = \sum_{i=1}^k \sum_{j=1}^l e_j e_i^T \otimes L_{ij}(\lambda)$ is called the *block-transpose* of $L(\lambda)$ (see [53]). Notice that $F_{\Phi}^P(\lambda)^{\mathcal{B}} = [m_{\Phi}^P(\lambda)^{\mathcal{B}} \quad M_{\Phi}(\lambda)^T \otimes I_m]$.

Theorem 3.3.1. [36, Theorem 2] Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix with degree $k \geq 2$. Then $L(\lambda) \in \mathbb{M}_2(P)$ with left ansatz vector $w \in \mathbb{F}^k$ if and only if

$$L(\lambda) = F_{\Phi}^{P}(\lambda)^{\mathcal{B}} \left[\begin{array}{c} w^{T} \otimes I_{m} \\ H^{\mathcal{B}} \end{array} \right]$$

for some matrix $H \in \mathbb{F}^{km \times (k-1)m}$ partitioned into $k \times (k-1)$ blocks each of size $m \times m$.

The vector space $\mathbb{M}_2(P)$ reduces to the well-known space $\mathbb{L}_2(P)$ when $\{\phi_k(\lambda)\}_{k=0}^{\infty}$ is the monomial basis, see [63]. Lemma 3.3.2 is for $\mathbb{M}_2(P)$ the counterpart of Lemma 3.2.3 for $\mathbb{M}_1(P)$ and can be used to proceed with $\mathbb{M}_2(P)$ analogously as we did with $\mathbb{M}_1(P)$.

Lemma 3.3.2. $F_{\Phi}^{P}(\lambda)^{\mathcal{B}}$ is a strong block minimal bases pencil with only one block row associated to $P(\lambda)$ with sharp degree.

In particular, Lemma 3.3.2 allows us to apply [6, Theorem 5.11] to the strong linearization $F_{\Phi}^D(\lambda)^{\mathcal{B}}$ of the polynomial part of a square rational matrix, with $K_2(\lambda) = M_{\Phi}(\lambda) \otimes I_m$, $\hat{K}_2 = e_k^T \otimes I_m$, $K_1(\lambda)$ empty and $\hat{K}_1 = I_m$. Thus, we get the following results to obtain strong linearizations of a square rational matrix $R(\lambda) = D(\lambda) + R_{sp}(\lambda)$ expressed as in (2.1) from strong linearizations in $\mathbb{M}_2(D)$.

Theorem 3.3.3. Under the same assumptions as in Theorem 3.2.8, the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB & 0_{n \times (k-1)m} \\ 0_{(k-1)m \times n} & & F_{\Phi}^D(\lambda)^{\mathcal{B}} \\ -CY & & & F_{\Phi}^D(\lambda)^{\mathcal{B}} \end{bmatrix}$$

is a strong linearization of $R(\lambda)$.

Theorem 3.3.4. Under the same assumptions as in Theorem 3.2.8, let $w \in \mathbb{F}^k$, $H \in \mathbb{F}^{k_m \times (k-1)m}$ with $\begin{bmatrix} w^T \otimes I_m \\ H^{\mathcal{B}} \end{bmatrix}$ nonsingular and let $L(\lambda) = F_{\Phi}^D(\lambda)^{\mathcal{B}} \begin{bmatrix} w^T \otimes I_m \\ H^{\mathcal{B}} \end{bmatrix} \in \mathbb{M}_2(D)$. Then the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB(w^T \otimes I_m) \\ 0_{(k-1)m \times n} & L(\lambda) \\ -CY & L(\lambda) \end{bmatrix}$$

is a strong linearization of $R(\lambda)$.

Proof. We apply Lemma 3.1.2 by multiplying on the right the matrix $\mathcal{L}(\lambda)$ in Theorem 3.3.3 by the matrix $\begin{bmatrix} I_n & 0\\ 0 & K \end{bmatrix}$ with $K = \begin{bmatrix} w^T \otimes I_m \\ H^{\mathcal{B}} \end{bmatrix}$ nonsingular.

The strong linearizations of rational matrices constructed in Theorem 3.3.4 will be called M_2 -strong linearizations.

3.4 Recovering eigenvectors from M_1 - and M_2 -strong linearizations of rational matrices

In this section we will recover right and left eigenvectors of a rational matrix. These eigenvectors will be obtained without essentially computational cost from the right and left eigenvectors of the strong linearizations that we have constructed in Theorems 3.2.9 and 3.3.4.

3.4.1 Eigenvectors from M_1 -strong linearizations

We consider in this subsection the linearizations that we have constructed in Theorem 3.2.9, which we called \mathbb{M}_1 -strong linearizations. We will recover the eigenvectors of a rational matrix $R(\lambda)$ from those of its \mathbb{M}_1 -strong linearizations, and conversely. Lemma 3.4.1 will be used for this purpose.

Lemma 3.4.1. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_1 -strong linearization of $R(\lambda)$, and let $\widehat{R}(\lambda)$ be the transfer function of $\mathcal{L}(\lambda)$. Then

$$\widehat{R}(\lambda)(\Phi_k(\lambda) \otimes I_m) = v \otimes R(\lambda).$$
(3.6)

Proof. We consider the transfer function of the matrix $\mathcal{L}(\lambda)$,

$$\widehat{R}(\lambda) = L(\lambda) + \begin{bmatrix} 0_{km \times (k-1)m} & (v \otimes I_m)C(\lambda I_n - A)^{-1}B \end{bmatrix}.$$

Let $D(\lambda)$ be the polynomial part of $R(\lambda)$. Since $L(\lambda)$ belongs to $\mathbb{M}_1(D)$, $L(\lambda)(\Phi_k(\lambda) \otimes I_m) = v \otimes D(\lambda) = (v \otimes I_m)D(\lambda)$. Therefore, we obtain

$$\widehat{R}(\lambda)(\Phi_k(\lambda) \otimes I_m) = (L(\lambda) + \begin{bmatrix} 0_{km \times (k-1)m} & (v \otimes I_m)C(\lambda I_n - A)^{-1}B \end{bmatrix})(\Phi_k(\lambda) \otimes I_m)$$
$$= (v \otimes I_m)D(\lambda) + (v \otimes I_m)C(\lambda I_n - A)^{-1}B$$
$$= (v \otimes I_m)R(\lambda).$$

Remark 3.4.2. Since $\mathcal{L}(\lambda)$ is a strong linearization of the rational matrix $R(\lambda)$ we have, by Definition 2.5.3, that there are unimodular matrices $U(\lambda), V(\lambda) \in \mathbb{F}[\lambda]^{km \times km}$ such that

$$U(\lambda)\widehat{R}(\lambda)V(\lambda) = \operatorname{diag}(R(\lambda), I_{(k-1)m}).$$
(3.7)

Thus, if we consider a finite eigenvalue λ_0 of $R(\lambda)$ then it is also an eigenvalue of the transfer function $\widehat{R}(\lambda)$ and

$$\dim \mathcal{N}_r(R(\lambda_0)) = \dim \mathcal{N}_r(\widehat{R}(\lambda_0)).$$
(3.8)

By [6, Theorem 3.10], $det(\lambda_0 I_n - A) \neq 0$. Thus, by Proposition 2.4.7,

$$\dim \mathcal{N}_r(\widehat{R}(\lambda_0)) = \dim \mathcal{N}_r(\mathcal{L}(\lambda_0)).$$
(3.9)

By (3.7) and Proposition 2.4.8, we have the same equalities for the dimensions of the left null-spaces, i.e.,

dim
$$\mathcal{N}_{\ell}(R(\lambda_0)) = \dim \mathcal{N}_{\ell}(\widehat{R}(\lambda_0))$$
 and dim $\mathcal{N}_{\ell}(\widehat{R}(\lambda_0)) = \dim \mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$. (3.10)

Moreover, notice that since $R(\lambda)$ is square, dim $\mathcal{N}_r(R(\lambda_0)) = \dim \mathcal{N}_\ell(R(\lambda_0))$.

A consequence of Lemma 3.4.1 is that we can recover very easily right eigenvectors of a rational matrix $R(\lambda)$ from the eigenvectors of the transfer function $\widehat{R}(\lambda)$ of any \mathbb{M}_1 -strong linearization of $R(\lambda)$. We state that in Theorem 3.4.3, and we emphasize that this result is in the spirit of the one presented in [36, Proposition 3.1] for polynomial matrices $P(\lambda)$ and their strong linearizations in $\mathbb{M}_1(P)$.

Theorem 3.4.3. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, and let $\widehat{R}(\lambda)$ be the transfer function of an \mathbb{M}_1 -strong linearization

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

of $R(\lambda)$. Let λ_0 be a finite eigenvalue of $R(\lambda)$. Then, $u \in \mathcal{N}_r(R(\lambda_0))$ if and only if $\Phi_k(\lambda_0) \otimes u \in \mathcal{N}_r(\widehat{R}(\lambda_0))$. Moreover, $\{u_1, \ldots, u_t\}$ is a basis of $\mathcal{N}_r(R(\lambda_0))$ if and only if $\{\Phi_k(\lambda_0) \otimes u_1, \ldots, \Phi_k(\lambda_0) \otimes u_t\}$ is a basis of $\mathcal{N}_r(\widehat{R}(\lambda_0))$.

Proof. By Lemma 3.4.1, $\widehat{R}(\lambda_0)(\Phi_k(\lambda_0) \otimes I_m) = v \otimes R(\lambda_0)$. Thus, it is easy to see that $u \in \mathcal{N}_r(R(\lambda_0))$ if and only if $\Phi_k(\lambda_0) \otimes u \in \mathcal{N}_r(\widehat{R}(\lambda_0))$. Consider $\{u_1, \ldots, u_t\}$ a basis of $\mathcal{N}_r(R(\lambda_0))$. Therefore, as dim $\mathcal{N}_r(R(\lambda_0)) = \dim \mathcal{N}_r(\widehat{R}(\lambda_0))$, an immediate linear independence argument proves that $\{\Phi_k(\lambda_0) \otimes u_1, \ldots, \Phi_k(\lambda_0) \otimes u_t\}$ is a basis of $\mathcal{N}_r(\widehat{R}(\lambda_0))$, and conversely.

In addition, by using Proposition 2.4.7, we can recover the right eigenvectors of the transfer function $\widehat{R}(\lambda)$ from the right eigenvectors of the linearization $\mathcal{L}(\lambda)$, and conversely. In particular, if $\left(\lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}\right)$ is a solution of the polynomial eigenvalue problem $\mathcal{L}(\lambda)z = 0$ such that $\det(\lambda_0 I_n - A) \neq 0$, then (λ_0, x_0) is a solution of the rational eigenvalue problem $\widehat{R}(\lambda)x = 0$.

In what follows, if we have a vector $\begin{bmatrix} y \\ x \end{bmatrix}$, with $y \in \overline{\mathbb{F}}^{n \times 1}$ and $x \in \overline{\mathbb{F}}^{km \times 1}$, we will consider the vector x partitioned as $x = \begin{bmatrix} x^{(1)} & x^{(2)} & \cdots & x^{(k)} \end{bmatrix}^T$ with $x^{(j)} \in \overline{\mathbb{F}}^{m \times 1}$ for $j = 1, \ldots, k$. Recall also in Theorem 3.4.4 that, as we have explained in Remark 3.4.2, if $\lambda_0 \in \overline{\mathbb{F}}$ is a finite eigenvalue of $R(\lambda)$ then $\det(\lambda_0 I_n - A) \neq 0$. However, if λ_0 is an eigenvalue of $\mathcal{L}(\lambda)$, then, according to [6, Theorem 3.10], λ_0 might be a zero of $R(\lambda)$ that is simultaneously a pole and, therefore, $\det(\lambda_0 I_n - A) = 0$, and λ_0 is not an eigenvalue of $R(\lambda)$. This is the reason why the condition $\det(\lambda_0 I_n - A) \neq 0$ is assumed in parts a) and b) of Theorem 3.4.4.

Theorem 3.4.4. (Recovery of right eigenvectors from \mathbb{M}_1 -strong linearizations) Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, and let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_1 -strong linearization of $R(\lambda)$.

- a) If $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}$ is a solution of the LEP $\mathcal{L}(\lambda)z = 0$ such that $\det(\lambda_0 I_n A) \neq 0$, then $(\lambda_0, x_0^{(k)})$ is a solution of the REP $R(\lambda)x = 0$.
- b) Moreover, if $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \dots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(\mathcal{L}(\lambda_0))$, with $\det(\lambda_0 I_n A) \neq 0$, then $\{x_1^{(k)}, \dots, x_t^{(k)}\}$ is a basis of $\mathcal{N}_r(R(\lambda_0))$.
- c) Conversely, if (λ_0, u_0) is a solution of the REP $R(\lambda)x = 0$, $x_0 = \Phi_k(\lambda_0) \otimes u_0$ and y_0 is defined as the unique solution of $(\lambda_0 I_n - A)Yy_0 + Bu_0 = 0$, then $\left(\lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}\right)$ is a solution of the LEP $\mathcal{L}(\lambda)z = 0$.
- d) Moreover, if $\{u_1, \ldots, u_t\}$ is a basis of $\mathcal{N}_r(R(\lambda_0))$ and, for $i = 1, \ldots, t, x_i = \Phi_k(\lambda_0) \otimes u_i$ and y_i is defined as the unique solution of $(\lambda_0 I_n A)Yy_i + Bu_i = 0$, then $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(\mathcal{L}(\lambda_0))$.

Proof. By Proposition 2.4.7, if $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix} \end{pmatrix}$ is a solution of the LEP $\mathcal{L}(\lambda)z = 0$ such that $\det(\lambda_0 I_n - A) \neq 0$, then (λ_0, x_0) is a solution of the REP $\widehat{R}(\lambda)x = 0$, where

 $\widehat{R}(\lambda)$ is the transfer function matrix of $\mathcal{L}(\lambda)$. By Theorem 3.4.3, x_0 has the form $x_0 = \Phi_k(\lambda_0) \otimes u$ for some $u \in \mathcal{N}_r(R(\lambda_0))$. Since $\phi_0(\lambda) = 1$ we have that $u = x_0^{(k)}$, which proves a). The converse c) is proved analogously. The implications b) and d) are consequences of (3.8), (3.9), basic arguments of linear independence, and the fact that $\mathcal{L}(\lambda_0) \begin{bmatrix} y_0 \\ x_0 \end{bmatrix} = 0$ if and only if $(\lambda_0 I_n - A)Yy_0 + XBx_0^{(k)} = 0$ and $\widehat{R}(\lambda_0)x_0 = 0$.

Next, we pay attention to the recovery of left eigenvectors.

Theorem 3.4.5. (Recovery of left eigenvectors from M_1 -strong linearizations)

Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_1 -strong linearization of $R(\lambda)$, and let $\widehat{R}(\lambda)$ be the transfer function of $\mathcal{L}(\lambda)$.

- a) If $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix} \end{pmatrix}$ is a solution of the LEP $z^T \mathcal{L}(\lambda) = 0$ such that $\det(\lambda_0 I_n A) \neq 0$, then $(\lambda_0, (v^T \otimes I_m)x_0)$ is a solution of the REP $x^T R(\lambda) = 0$.
- b) Moreover, if $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \dots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$, with $\det(\lambda_0 I_n A) \neq 0$, then $\{ (v^T \otimes I_m) x_1, \dots, (v^T \otimes I_m) x_t \}$ is a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$.
- c) Conversely, if (λ_0, u_0) is a solution of the REP $x^T R(\lambda) = 0$, then there exists $x_0 \in \mathcal{N}_{\ell}(\widehat{R}(\lambda_0))$ such that $u_0 = (v^T \otimes I_m)x_0$ and if y_0 is defined as the unique solution of $y_0^T X(\lambda_0 I_n A) u_0^T C = 0$, then $\left(\lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}\right)$ is a solution of the LEP $z^T \mathcal{L}(\lambda) = 0$.
- d) Moreover, if $\{u_1, \ldots, u_t\}$ is a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$ then, for $i = 1, \ldots, t$, there exists $x_i \in \mathcal{N}_{\ell}(\widehat{R}(\lambda_0))$ such that $u_i = (v^T \otimes I_m)x_i$ and if y_i is defined as the unique solution of $y_i^T X(\lambda_0 I_n A) u_i^T C = 0$, then $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$.

Proof. We consider the transfer function of $\mathcal{L}(\lambda)$, $\widehat{R}(\lambda) = L(\lambda) + [0_{km \times (k-1)m} \quad (v \otimes I_m)C(\lambda I_n - A)^{-1}B]$. If $\left(\lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}\right)$ is a solution of the LEP $z^T \mathcal{L}(\lambda) = 0$ such that $\det(\lambda_0 I_n - A) \neq 0$, by using Proposition 2.4.8 *a*) applied to $\mathcal{L}(\lambda)$, we get

$$x_0^T \widehat{R}(\lambda_0) = x_0^T L(\lambda_0) + \begin{bmatrix} 0_{1 \times (k-1)m} & x_0^T (v \otimes I_m) C(\lambda_0 I_n - A)^{-1}B \end{bmatrix} = 0, \quad (3.11)$$

where $x_0 \neq 0$ since (λ_0, x_0) is a solution of the REP $x^T \widehat{R}(\lambda) = 0^{-1}$. In addition, by Lemma 3.4.1, $x_0^T \widehat{R}(\lambda_0)(\Phi_k(\lambda_0) \otimes I_m) = x_0^T(v \otimes I_m)R(\lambda_0)$. Therefore $x_0^T(v \otimes I_m)R(\lambda_0) = 0$. To see that $(v^T \otimes I_m)x_0$ is a left eigenvector of $R(\lambda_0)$, we only need to prove that $x_0^T(v \otimes I_m) \neq 0$. Let us suppose that $x_0^T(v \otimes I_m) = 0$, and let us get a contradiction. In this case $x_0^T(v \otimes I_m)C(\lambda_0I_n - A)^{-1}B = 0$ and, therefore, $x_0^T L(\lambda_0) = x_0^T[v \otimes I_m - H]F_{\Phi}^D(\lambda_0) = 0$ by (3.11). We call $w^T = x_0^T[v \otimes I_m - H]$ and we consider w partitioned as $w = (w_i)_{i=1}^k$ with $w_i \in \overline{\mathbb{F}}^{m \times 1}$. We have that $w_1^T = x_0^T(v \otimes I_m) = 0$. Therefore $[0 \quad w_2^T \quad \cdots \quad w_k^T]F_{\Phi}^D(\lambda_0) = 0$. This implies $-\alpha_{k-2}w_2^T = 0$ and thus $w_2 = 0$, since $\alpha_{k-2} \neq 0$. Therefore $[0 \quad w_3^T \quad \cdots \quad w_k^T]F_{\Phi}^D(\lambda_0) = 0$ and $w_3 = 0$. Proceeding in this way it is easy to prove that $w_i = 0$ for $i = 2, \cdots, k$. Thus $x_0^T[v \otimes I_m - H] = 0$ which is a contradiction because $[v \otimes I_m - H]$ is assumed to be regular and $x_0 \neq 0$. This proves a).

The implication b) is proved as follows. From a), the vectors $(v^T \otimes I_m)x_1, \ldots, (v^T \otimes I_m)x_t$ belong to $\mathcal{N}_{\ell}(R(\lambda_0))$. Therefore, as a consequence of (3.10), if we prove that $\{(v^T \otimes I_m)x_1, \ldots, (v^T \otimes I_m)x_t\}$ is linearly independent, then b) is proved. For this purpose, let $\alpha_1, \ldots, \alpha_t \in \overline{\mathbb{F}}$ be arbitrary scalars such that at least one is different from zero. Thus $0 \neq \begin{bmatrix} \alpha_1 y_1 + \cdots + \alpha_t y_t \\ \alpha_1 x_1 + \cdots + \alpha_t x_t \end{bmatrix} \in \mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$, and, from part a), $x = (v^T \otimes I_m)(\alpha_1 x_1 + \cdots + \alpha_t x_t) \neq 0$ and $x \in \mathcal{N}_{\ell}(R(\lambda_0))$.

For proving c), we prove first that there exists a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$ of the form $\{(v^T \otimes I_m)x_1, \ldots, (v^T \otimes I_m)x_t\}$, where $\{x_1, \ldots, x_t\}$ is a basis of $\mathcal{N}_{\ell}(\widehat{R}(\lambda_0))$. To this purpose, let $\left\{ \begin{bmatrix} y_1\\x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t\\x_t \end{bmatrix} \right\}$ be a basis of $\mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$. Then, Proposition 2.4.8 b) applied to $\mathcal{L}(\lambda)$ implies that $\{x_1, \ldots, x_t\}$ is a basis of $\mathcal{N}_{\ell}(\widehat{R}(\lambda_0))$ and Theorem 3.4.5 b) that $\{(v^T \otimes I_m)x_1, \ldots, (v^T \otimes I_m)x_t\}$ is a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$. Then, if (λ_0, u_0) is a solution of the REP $x^T R(\lambda) = 0$, u_0 can be written as $u_0 = (v^T \otimes I_m) \sum_{i=1}^t a_i x_i$ with $a_i \in \overline{\mathbb{F}}$, and we define $x_0 = \sum_{i=1}^t a_i x_i \in \mathcal{N}_{\ell}(\widehat{R}(\lambda_0))$. Finally, Proposition 2.4.8 c) applied to the solution (λ_0, x_0) of the REP $x^T \widehat{R}(\lambda) = 0$ and to $\mathcal{L}(\lambda)$, and the fact that det $(\lambda_0 I_n - A) \neq 0$ imply that if y_0 is the unique solution of $y_0^T X(\lambda_0 I_n - A) - x_0^T(v \otimes I_m)C = 0$, which is equivalent to $y_0^T X(\lambda_0 I_n - A) - u_0^T C = 0$, then $\left(\lambda_0, \begin{bmatrix} y_0\\x_0 \end{bmatrix}\right)$ is a solution of the LEP $z^T \mathcal{L}(\lambda) = 0$.

Finally, the proof of d) proceeds as follows. From part c), we obtain that the vectors x_1, \ldots, x_t satisfying $u_i = (v^T \otimes I_m) x_i$ exist, and that the vectors $\begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix}$

¹With the notation of Proposition 2.4.8, it is easy to see that $[y_0^T x_0^T]P(\lambda_0) = 0$ if and only if $y_0^T A(\lambda_0) - x_0^T C(\lambda_0) = 0$ and $x_0^T R(\lambda_0) = 0$. Thus, $x_0 = 0$ and det $A(\lambda_0) \neq 0$ imply $y_0 = 0$. Therefore, any left eigenvector of $P(\lambda)$ corresponding to the finite eigenvalue λ_0 must have $x_0 \neq 0$.

belong to $\mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$. Therefore, taking (3.10) into account, it only remains to prove that $\begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix}$ are linearly independent. This is easily proved by contradiction: If $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is linearly dependent, then $\{x_1, \ldots, x_t\}$ is linearly dependent, and $\{u_1, \ldots, u_t\}$ is linearly dependent, which is a contradiction since $\{u_1, \ldots, u_t\}$ is a basis.

Remark 3.4.6. Analogously to Remark 2.4.9, if $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ is singular, then the results on null-spaces proved so far in Section 3.4.1 are valid for any $\lambda_0 \in \overline{\mathbb{F}}$ that satisfies $\det(\lambda_0 I_n - A) \neq 0$.

Finally, we study the recovery of the eigenvectors corresponding to the infinite eigenvalue from \mathbb{M}_1 -strong linearizations.

Theorem 3.4.7. (Recovery of eigenvectors associated to infinity from \mathbb{M}_1 strong linearizations) Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_1 -strong linearization of $R(\lambda)$, and let D_k be the leading matrix coefficient of the polynomial part of $R(\lambda)$ as in (3.5). Then the following statements hold:

a) $\mathcal{N}_r(\operatorname{rev} R(0)) = \mathcal{N}_r(D_k) \text{ and } x_0 \in \mathcal{N}_r(D_k) \text{ if and only if } \begin{bmatrix} 0\\e_1 \otimes x_0 \end{bmatrix} \in \mathcal{N}_r(\operatorname{rev} \mathcal{L}(0)).$ Moreover, $\{x_1, \dots, x_q\}$ is a basis of $\mathcal{N}_r(\operatorname{rev} R(0))$ if and only if $\left\{ \begin{bmatrix} 0\\e_1 \otimes x_1 \end{bmatrix}, \dots, \begin{bmatrix} 0\\e_1 \otimes x_q \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(\operatorname{rev} \mathcal{L}(0)).$ b) $\mathcal{N}_\ell(\operatorname{rev} R(0)) = \mathcal{N}_\ell(D_k) \text{ and } \begin{bmatrix} 0\\x_0 \end{bmatrix} \in \mathcal{N}_\ell(\operatorname{rev} \mathcal{L}(0)) \text{ if and only if } (v^T \otimes I_m) x_0 \in \mathbb{N}_\ell(\operatorname{rev} \mathcal{L}(0))$

$$\mathcal{N}_{\ell}(D_k). \text{ Moreover, } \left\{ \begin{bmatrix} 0 \\ x_1 \end{bmatrix}, \dots, \begin{bmatrix} 0 \\ x_q \end{bmatrix} \right\} \text{ is a basis of } \mathcal{N}_{\ell}(\operatorname{rev} \mathcal{L}(0)) \text{ if and only} \\ \text{ if } \left\{ (v^T \otimes I_m) x_1, \dots, (v^T \otimes I_m) x_q \right\} \text{ is a basis of } \mathcal{N}_{\ell}(\operatorname{rev} R(0)).$$

Proof. Notice that from (3.4),

$$F_{\Phi}^{D}(\lambda) = \lambda \begin{bmatrix} \alpha_{k-1}^{-1}D_{k} & 0\\ 0 & I_{(k-1)m} \end{bmatrix} + F_{\Phi}^{D}(0).$$

We consider

$$L(\lambda) = [v \otimes I_m \quad H] F_{\Phi}^D(\lambda) = [\alpha_{k-1}^{-1}(v \otimes D_k) \quad H] \lambda + L(0) =: L_1 \lambda + L_0$$

and let $\widehat{R}(\lambda)$ be the transfer function matrix of $\mathcal{L}(\lambda)$. We have that rev $\mathcal{L}(0) = \begin{bmatrix} \frac{XY \mid 0}{0 \mid L_1} \end{bmatrix}$ and rev $\widehat{R}(0) = \text{rev } L(0) = L_1$. Moreover, rev $R(0) = \alpha_0^{-1} \alpha_1^{-1} \cdots \alpha_{k-1}^{-1} D_k$, that is, the coefficient of λ^k in $D(\lambda)$. Therefore, $\mathcal{N}_r(\text{rev } R(0)) = \mathcal{N}_r(D_k), \mathcal{N}_\ell(\text{rev } R(0)) = \mathcal{N}_\ell(D_k)$ and ∞ is an eigenvalue of $R(\lambda)$ if and only if D_k is singular. In addition, every right (respectively left) eigenvector w of rev $\mathcal{L}(0)$ has the form $w = \begin{bmatrix} 0 \\ x_0 \end{bmatrix}$ for some $x_0 \in \mathcal{N}_r(L_1)$ (respectively $x_0 \in \mathcal{N}_\ell(L_1)$). By Lemma 3.4.1, we have

$$\lambda \widehat{R}\left(\frac{1}{\lambda}\right) \left(\lambda^{k-1} \Phi_k\left(\frac{1}{\lambda}\right) \otimes I_m\right) = v \otimes \lambda^k R\left(\frac{1}{\lambda}\right).$$

Therefore,

 $\operatorname{rev} \widehat{R}(0)(\operatorname{rev} \Phi_k(0) \otimes I_m) = (v \otimes I_m) \operatorname{rev} R(0).$

Since rev $\Phi_k(0) = \alpha_0^{-1} \alpha_1^{-1} \cdots \alpha_{k-2}^{-1} e_1$, we obtain

$$\alpha_0^{-1}\alpha_1^{-1}\cdots\alpha_{k-2}^{-1}\operatorname{rev}\widehat{R}(0)(e_1\otimes I_m)=(v\otimes I_m)\operatorname{rev}R(0).$$

In addition, by (2.15), there exist unimodular matrices $W_1(\lambda)$ and $W_2(\lambda)$ such that

$$W_1(0) \operatorname{diag} \left(\operatorname{rev} R(0), I_{(k-1)m} \right) W_2(0) = \operatorname{rev} R(0),$$

which implies that $\dim \mathcal{N}_r(\operatorname{rev} R(0)) = \dim \mathcal{N}_r(\operatorname{rev} \widehat{R}(0))$ and $\dim \mathcal{N}_\ell(\operatorname{rev} R(0)) = \dim \mathcal{N}_\ell(\operatorname{rev} \widehat{R}(0))$. Finally *a*) and *b*) follow from the results above by using similar arguments to the ones we used in the recovery of eigenvectors associated to finite eigenvalues.

3.4.2 Eigenvectors from M_2 -strong linearizations

If we proceed analogously as we did with M_1 -strong linearizations, and we use Lemma 3.4.8, then we get Theorems 3.4.9, 3.4.10 and 3.4.12 to recover right and left eigenvectors of a rational matrix from those of its M_2 -strong linearizations. The proofs are essentially the same as those in Section 3.4.1 by interchanging the roles of left and right eigenvectors, and they are omitted for brevity.

Lemma 3.4.8. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB(w^T \otimes I_m) \\ 0_{(k-1)m \times n} & L(\lambda) \\ -CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_2 -strong linearization of $R(\lambda)$, and let $\widehat{R}(\lambda)$ be the transfer function of $\mathcal{L}(\lambda)$. Then

$$(\Phi_k(\lambda)^T \otimes I_m)\widehat{R}(\lambda) = w^T \otimes R(\lambda).$$
(3.12)

Theorem 3.4.9. (Recovery of right eigenvectors from \mathbb{M}_2 -strong linearizations) Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB(w^T \otimes I_m) \\ 0_{(k-1)m \times n} & L(\lambda) \\ -CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_2 -strong linearization of $R(\lambda)$, and let $\widehat{R}(\lambda)$ be the transfer function of $\mathcal{L}(\lambda)$.

- a) If $\left(\lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}\right)$ is a solution of the LEP $\mathcal{L}(\lambda)z = 0$ such that $\det(\lambda_0 I_n A) \neq 0$ then, $(\lambda_0, (w^T \otimes I_m)x_0)$ is a solution of the REP $R(\lambda)x = 0$.
- b) Moreover, if $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \dots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(\mathcal{L}(\lambda_0))$, with $\det(\lambda_0 I_n A) \neq 0$, then $\{ (w^T \otimes I_m) x_1, \dots, (w^T \otimes I_m) x_t \}$ is a basis of $\mathcal{N}_r(R(\lambda_0))$.
- c) Conversely, if (λ_0, u_0) is a solution of the REP $R(\lambda)x = 0$, then there exists $x_0 \in \mathcal{N}_r(\widehat{R}(\lambda_0))$ such that $u_0 = (w^T \otimes I_m)x_0$ and if y_0 is defined as the unique solution of $(\lambda_0 I_n A)Yy_0 + Bu_0 = 0$, then $\left(\lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}\right)$ is a solution of the LEP $\mathcal{L}(\lambda)z = 0$.
- d) Moreover, if $\{u_1, \ldots, u_t\}$ is a basis of $\mathcal{N}_r(R(\lambda_0))$ then, for $i = 1, \ldots, t$, there exists $x_i \in \mathcal{N}_r(\widehat{R}(\lambda_0))$ such that $u_i = (w^T \otimes I_m)x_i$ and if y_i is defined as the unique solution of $(\lambda_0 I_n A)Yy_i + Bu_i = 0$, then $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(\mathcal{L}(\lambda_0))$.

Theorem 3.4.10. (Recovery of left eigenvectors from \mathbb{M}_2 -strong linearizations) Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, and let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB(w^T \otimes I_m) \\ 0_{(k-1)m \times n} & L(\lambda) \\ -CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_2 -strong linearization of $R(\lambda)$.

- a) If $\begin{pmatrix} \lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}$ is a solution of the LEP $z^T \mathcal{L}(\lambda) = 0$ such that $\det(\lambda_0 I_n A) \neq 0$, then $(\lambda_0, x_0^{(k)})$ is a solution of the REP $x^T R(\lambda) = 0$.
- b) Moreover, if $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \dots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$, with $\det(\lambda_0 I_n A) \neq 0$, then $\{x_1^{(k)}, \dots, x_t^{(k)}\}$ is a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$.

- c) Conversely, if (λ_0, u_0) is a solution of the REP $x^T R(\lambda) = 0$, $x_0 = \Phi_k(\lambda_0) \otimes u_0$ and y_0 is defined as the unique solution of $y_0^T X(\lambda_0 I_n - A) - u_0^T C = 0$, then $\left(\lambda_0, \begin{bmatrix} y_0 \\ x_0 \end{bmatrix}\right)$ is a solution of the LEP $z^T \mathcal{L}(\lambda) = 0$.
- d) Moreover, if $\{u_1, \ldots, u_t\}$ is a basis of $\mathcal{N}_{\ell}(R(\lambda_0))$ and, for $i = 1, \ldots, t, x_i = \Phi_k(\lambda_0) \otimes u_i$ and y_i is defined as the unique solution of $y_i^T X(\lambda_0 I_n A) u_i^T C = 0$, then $\left\{ \begin{bmatrix} y_1 \\ x_1 \end{bmatrix}, \ldots, \begin{bmatrix} y_t \\ x_t \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$.

Remark 3.4.11. Analogously to Remarks 2.4.9 and 3.4.6, if $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ is singular, then the results on null-spaces in Theorems 3.4.9 and 3.4.10 hold for any $\lambda_0 \in \overline{\mathbb{F}}$ such that $\det(\lambda_0 I_n - A) \neq 0$.

Theorem 3.4.12. (Recovery of eigenvectors associated to infinity from \mathbb{M}_2 -strong linearizations) Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix with polynomial part of degree $k \geq 2$, let

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB(w^T \otimes I_m) \\ 0_{(k-1)m \times n} & L(\lambda) \\ -CY & L(\lambda) \end{bmatrix}$$

be an \mathbb{M}_2 -strong linearization of $R(\lambda)$, and let D_k be the leading matrix coefficient of the polynomial part of $R(\lambda)$ as in (3.5). Then the following statements hold:

- a) $\mathcal{N}_r(\operatorname{rev} R(0)) = \mathcal{N}_r(D_k) \text{ and } \begin{bmatrix} 0\\x_0 \end{bmatrix} \in \mathcal{N}_r(\operatorname{rev} \mathcal{L}(0)) \text{ if and only if } (w^T \otimes I_m) x_0 \in \mathcal{N}_r(D_k).$ Moreover, $\left\{ \begin{bmatrix} 0\\x_1 \end{bmatrix}, \dots, \begin{bmatrix} 0\\x_q \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_r(\operatorname{rev} \mathcal{L}(0))$ if and only if $\left\{ (w^T \otimes I_m) x_1, \dots, (w^T \otimes I_m) x_q \right\}$ is a basis of $\mathcal{N}_r(\operatorname{rev} R(0)).$
- b) $\mathcal{N}_{\ell}(\operatorname{rev} R(0)) = \mathcal{N}_{\ell}(D_k) \text{ and } x_0 \in \mathcal{N}_{\ell}(D_k) \text{ if and only if } \begin{bmatrix} 0\\e_1 \otimes x_0 \end{bmatrix} \in \mathcal{N}_{\ell}(\operatorname{rev} \mathcal{L}(0)).$ Moreover, $\{x_1, \ldots, x_q\}$ is a basis of $\mathcal{N}_{\ell}(\operatorname{rev} R(0))$ if and only if $\left\{ \begin{bmatrix} 0\\e_1 \otimes x_1 \end{bmatrix}, \ldots, \begin{bmatrix} 0\\e_1 \otimes x_q \end{bmatrix} \right\}$ is a basis of $\mathcal{N}_{\ell}(\operatorname{rev} \mathcal{L}(0)).$

3.5 Symmetric realizations of symmetric rational matrices

In this section and in the next one our aim is to obtain a strong linearization of a symmetric rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$, i.e., $R(\lambda)^T = R(\lambda)$, that preserves its

symmetric structure. We write $R(\lambda)$ as

$$R(\lambda) = D(\lambda) + R_{sp}(\lambda) \tag{3.13}$$

with $D(\lambda)$ its polynomial part and $R_{sp}(\lambda)$ its strictly proper part. Since (3.13) is a unique decomposition we obtain the following result just by taking transposes.

Proposition 3.5.1. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a symmetric rational matrix. Then the matrices $D(\lambda)$ and $R_{sp}(\lambda)$ in (3.13) are also symmetric.

Proposition 3.5.5 is the main result in this section and shows that any symmetric strictly proper rational matrix admits a state-space realization that reveals transparently the symmetry. In order to state concisely Proposition 3.5.5, we will use the following definition.

Definition 3.5.2. Let $R_{sp}(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a symmetric strictly proper rational matrix and let $n = \nu(R_{sp}(\lambda))$ be the least order of $R_{sp}(\lambda)$. A symmetric minimal state-space realization of $R_{sp}(\lambda)$ is an expression of the form

$$R_{sp}(\lambda) = W(S_1\lambda - S_2)^{-1}W^T$$

where $S_1, S_2 \in \mathbb{F}^{n \times n}$ are symmetric matrices with S_1 nonsingular and $W \in \mathbb{F}^{m \times n}$.

We remark that the realization described in Definition 3.5.2 is equivalent to [11, Definition 4.44] for a minimal state-space realization. However, in Definition 3.5.2 we express strictly proper matrices in a form more convenient for the goals of this chapter. In particular, we will see in Section 3.6 that by combining a symmetric minimal state-space realization of the matrix $R_{sp}(\lambda)$ in (3.13) and a symmetric strong block minimal bases pencil associated to $D(\lambda)$, we can construct symmetric strong linearizations of $R(\lambda)$. The next technical lemma is used in the proof of Proposition 3.5.5.

Lemma 3.5.3. Let $R_{sp}(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a symmetric strictly proper rational matrix and let $R_{sp}(\lambda) = C(\lambda I_n - A)^{-1}B$ be a minimal state-space realization of $R_{sp}(\lambda)$. Then there exists a unique nonsingular and symmetric matrix $S \in \mathbb{F}^{n \times n}$ such that $A^T = S^{-1}AS$ and $C^T = S^{-1}B$.

Proof. As $R_{sp}(\lambda)$ is symmetric, $R_{sp}(\lambda) = B^T (\lambda I_n - A^T)^{-1} C^T$ is also a minimal statespace realization of $R_{sp}(\lambda)$ since both have the same minimal order n. Therefore, by [48, Proposition 3.3.2], the realizations (A, B, C) and (A^T, C^T, B^T) are similar and there exists a unique nonsingular matrix $S \in \mathbb{F}^{n \times n}$ such that

$$A^{T} = S^{-1}AS, \quad C^{T} = S^{-1}B, \quad B^{T} = CS.$$
 (3.14)

The fact that (A, B, C) is a minimal realization of $R_{sp}(\lambda)$ is equivalent to that (A, B) and (A, C) are controllable and observable, respectively (see [78, Chapter

3]). That means that the controllability matrix of (A, B) and the observability matrix of (A, C), i.e.,

$$\mathcal{C}(A,B) = \begin{bmatrix} B & AB & A^2B & \cdots & A^{n-1}B \end{bmatrix} \text{ and } \mathcal{O}(A,C) = \begin{bmatrix} C \\ CA \\ CA^2 \\ \vdots \\ CA^{n-1} \end{bmatrix},$$

have both rank *n*. From the equalities in (3.14) it is easy to see that $S^{-1}\mathcal{C}(A, B) = \mathcal{O}(A, C)^T$, and $S^{-T}\mathcal{C}(A, B) = \mathcal{O}(A, C)^T$. As $\mathcal{C}(A, B)$ has full row rank, we deduce that $S = S^T$.

Remark 3.5.4. Notice that the system similarity matrix S between the realizations in Lemma 3.5.3 is given by $S = \mathcal{O}(A, C)^+ \mathcal{C}(A, B)^T = \mathcal{C}(A, B)(\mathcal{O}(A, C)^T)^\dagger$ where + denotes any left inverse and \dagger denotes any right inverse. Notice also that these left and right inverses exist because (A, B, C) is a minimal realization of $R_{sp}(\lambda)$ and that they can be taken to be the Moore–Penrose inverses. Thus S can be efficiently computed when $\mathbb{F} = \mathbb{R}, \mathbb{C}$.

Proposition 3.5.5. Any symmetric strictly proper rational matrix has a symmetric minimal state-space realization.

Proof. As said in Section 2.4, any strictly proper rational matrix $R_{sp}(\lambda)$ admits a minimal state-space realization, that is, $R_{sp}(\lambda) = C(\lambda I_n - A)^{-1}B$ [78]. By Lemma 3.5.3, there exists a unique nonsingular and symmetric matrix S such that $A^T = S^{-1}AS$ and $C^T = S^{-1}B$. Thus, $R_{sp}(\lambda) = C(\lambda I_n - A)^{-1}SC^T = C(\lambda S^{-1} - S^{-1}A)^{-1}C^T$, and $S^{-1}A$ is symmetric, as $(S^{-1}A)^T = A^TS^{-1} = S^{-1}A$.

Remark 3.5.6. We can construct a symmetric minimal state-space realization of a symmetric strictly proper rational matrix $R_{sp}(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ without previously considering a non-symmetric minimal state-space realization of $R_{sp}(\lambda)$, in contrast to what we have done in the proof of Proposition 3.5.5. For this purpose we require \mathbb{F} not to be a field of characteristic 2. Let $R_{sp}(\lambda) = G_1 \lambda^{-1} + G_2 \lambda^{-2} + \cdots$ be the Laurent series of $R_{sp}(\lambda)$, which converges for $|\lambda|$ large enough. Let $n = \nu(R_{sp}(\lambda))$ be the least order of $R_{sp}(\lambda)$. We consider the block Hankel matrix

$$H_{n} = \begin{bmatrix} G_{1} & G_{2} & \cdots & G_{n} \\ G_{2} & G_{3} & \cdots & G_{n+1} \\ \vdots & \vdots & \ddots & \vdots \\ G_{n} & G_{n+1} & \cdots & G_{2n-1} \end{bmatrix}$$
(3.15)

and follow in a symmetric way the three steps of the algorithm in [48, Section 3.4] to get a symmetric minimal state-space realization from the Hankel matrix. Notice that

the Hankel matrix is symmetric since $R_{sp}(\lambda)$ is symmetric, which implies $G_i = G_i^T$ for all $i \ge 1$, and rank $H_n = n$ by [48, Proposition 3.3.2]. Therefore we can write

$$H_n = X \begin{bmatrix} K & 0 \\ 0 & 0 \end{bmatrix} X^T = X \begin{bmatrix} K \\ 0 \end{bmatrix} \begin{bmatrix} I_n & 0 \end{bmatrix} X^T$$

with X nonsingular and $K \in \mathbb{F}^{n \times n}$ diagonal (see [62, Theorem 34.1]). Let us denote

$$\Gamma = X \begin{bmatrix} K \\ 0 \end{bmatrix}$$
 and $\Lambda = \begin{bmatrix} I_n & 0 \end{bmatrix} X^T$.

We have that $H_n = \Gamma \Lambda$. We write $X = \begin{bmatrix} X_1 & X_2 \end{bmatrix}$, where $X_1 = \begin{bmatrix} X_{i1} \end{bmatrix}_{i=1}^n$ with $X_{i1} \in \mathbb{F}^{m \times n}$ for $i = 1, \ldots, n$. Thus

$$\Gamma = \begin{bmatrix} X_{11}K \\ \vdots \\ X_{n1}K \end{bmatrix} \text{ and } \Lambda = \begin{bmatrix} X_{11}^T & \cdots & X_{n1}^T \end{bmatrix}.$$

We define

$$R = \begin{bmatrix} G_2 & G_3 & \cdots & G_{n+1} \\ G_3 & G_4 & \cdots & G_{n+2} \\ \vdots & \vdots & \ddots & \vdots \\ G_{n+1} & G_{n+2} & \cdots & G_{2n} \end{bmatrix}$$

and we set $C = X_{11}K$, $B = X_{11}^T$ and $A = \Gamma^+ R \Lambda^+$, with $\Gamma^+ = [K^{-1} \quad 0] X^{-1}$ and $\Lambda^+ = X^{-T} \begin{bmatrix} I_n \\ 0 \end{bmatrix}$. Thus $A = [K^{-1} \quad 0] X^{-1} R X^{-T} \begin{bmatrix} I_n \\ 0 \end{bmatrix}$ and, by [48, Theorem 3.4.1], (A, B, C) is a minimal realization for $R_{sp}(\lambda)$. Therefore

$$R_{sp}(\lambda) = X_{11}K \left(\lambda I_n - \begin{bmatrix} K^{-1} & 0 \end{bmatrix} X^{-1}RX^{-T} \begin{bmatrix} I_n \\ 0 \end{bmatrix} \right)^{-1} X_{11}^T$$
$$= X_{11} \left(\lambda K^{-1} - \begin{bmatrix} K^{-1} & 0 \end{bmatrix} X^{-1}RX^{-T} \begin{bmatrix} K^{-1} \\ 0 \end{bmatrix} \right)^{-1} X_{11}^T$$

Finally we set $W = X_{11}$, $S_1 = K^{-1}$ and $S_2 = \begin{bmatrix} K^{-1} & 0 \end{bmatrix} X^{-1} R X^{-T} \begin{bmatrix} K^{-1} \\ 0 \end{bmatrix}$, and we obtain a symmetric minimal state-space realization of $R_{sp}(\lambda)$.

In the particular case $R_{sp}(\lambda) \in \mathbb{R}(\lambda)^{m \times m}$, which is of significant importance in applications, the Hankel matrix H_n is real and symmetric. Therefore, we can write

$$H_n = P \left[\begin{array}{cc} K & 0 \\ 0 & 0 \end{array} \right] P^T$$

with *P* orthogonal, i.e., $P^{-1} = P^T$, and *K* a diagonal matrix that has the eigenvalues different from zero of H_n at the diagonal elements. In this case, let $P = \begin{bmatrix} P_1 & P_2 \end{bmatrix}$,

where $P_1 = [P_{i1}]_{i=1}^n$ with $P_{i1} \in \mathbb{F}^{m \times n}$ for $i = 1, \ldots, n$. Then we obtain $R_{sp}(\lambda) = P_{11}(\lambda K^{-1} - K^{-1}P_1^T R P_1 K^{-1})^{-1} P_{11}^T$. That is, $R_{sp}(\lambda)$ has a symmetric minimal statespace realization $R_{sp}(\lambda) = W(\lambda S_1 - S_2)^{-1} W^T$ where $W = P_{11}, S_1 = K^{-1}$ and $S_2 = K^{-1} P_1^T R P_1 K^{-1}$.

From Proposition 3.5.5 and Remark 3.5.6 we know how to write the strictly proper part $R_{sp}(\lambda)$ of a symmetric rational matrix $R(\lambda)$ as a symmetric minimal state-space realization with or without having in advance a particular non-symmetric minimal state-space realization of $R_{sp}(\lambda)$. Moreover, it is worth to emphasize that in many applications of symmetric REPs, this can be done very easily from the data of the model without any computational cost (see [6, Section 5.3] or [79, Section 4]).

3.6 Symmetric strong linearizations for symmetric rational matrices

In this section symmetric strong linearizations for symmetric rational matrices will be constructed. We start with Example 3.6.1 in which we construct a symmetric strong linearization of a symmetric rational matrix when the polynomial part has odd degree. We will use Proposition 3.5.5 and a particular symmetric strong block minimal bases pencil associated to its polynomial part with sharp degree. After that, we present symmetric strong linearizations for symmetric rational matrices in which the polynomial part may have even or odd degree but the leading coefficient must be nonsingular. In order to get these results, we need to study symmetric strong linearizations in the polynomial case.

Example 3.6.1. Let $R(\lambda) = D(\lambda) + R_{sp}(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a symmetric rational matrix. Consider the polynomial part $D(\lambda)$ written in terms of the monomial basis $D(\lambda) = D_k \lambda^k + D_{k-1} \lambda^{k-1} + \cdots + D_0 \in \mathbb{F}[\lambda]^{m \times m}$, with k > 1 and $D_k \neq 0$, and the matrices

$$L_p(\lambda) = \begin{bmatrix} -1 & \lambda & & \\ & -1 & \lambda & \\ & & \ddots & \ddots \\ & & & -1 & \lambda \end{bmatrix} \in \mathbb{F}[\lambda]^{p \times (p+1)},$$
(3.16)

and

$$\Lambda_p(\lambda)^T = \begin{bmatrix} \lambda^p & \cdots & \lambda & 1 \end{bmatrix} \in \mathbb{F}[\lambda]^{1 \times (p+1)}.$$
(3.17)

A block Kronecker linearization of $D(\lambda)$ is a pencil

$$L(\lambda) = \begin{bmatrix} M(\lambda) & L_{\eta}(\lambda)^T \otimes I_m \\ \hline L_{\varepsilon}(\lambda) \otimes I_m & 0 \end{bmatrix} \begin{cases} \eta + 1 \end{pmatrix}_{\varepsilon m} \\ \vdots \\ \eta m \end{cases}$$
(3.18)

such that $D(\lambda) = (\Lambda_{\eta}(\lambda)^T \otimes I_m) M(\lambda) (\Lambda_{\varepsilon}(\lambda) \otimes I_m)$ (see [26, Definition 4.1]). Recall that block Kronecker linearizations are particular cases of strong block minimal bases pencils [26]. If the polynomial part $D(\lambda)$ has odd degree k = 2q + 1 we can consider the symmetric block Kronecker linearization in which

$$M(\lambda) = \begin{bmatrix} D_{2q+1}\lambda + D_{2q} & & \\ & D_{2q-1}\lambda + D_{2q-2} & \\ & & \ddots & \\ & & & D_{1}\lambda + D_{0} \end{bmatrix}$$

and $\varepsilon = \eta = q$. Proposition 3.5.5 allows us to write $R_{sp}(\lambda) = W(\lambda S_1 - S_2)^{-1}W^T$ with S_1 and S_2 symmetric and S_1 nonsingular. Applying [6, Theorem 5.11] with $Y = -S_1X^T$ for any nonsingular matrix $X \in \mathbb{F}^{n \times n}$, $C = WS_1^{-1}$, $A = S_2S_1^{-1}$, $B = W^T$, and $\widehat{K}_1 = \widehat{K}_2 = e_{q+1}^T \otimes I_m$, we obtain that the linear polynomial matrix

	$X(S_2 - \lambda S_1)X^T$	$0 XW^T$	0
$\mathcal{L}(\lambda) =$	$\begin{array}{c} 0 \\ WX^T \end{array}$	$M(\lambda)$	$L_q(\lambda)^T\otimes I_m$
	0	$L_q(\lambda)\otimes I_m$	0

is a symmetric strong linearization of $R(\lambda)$.

Remark 3.6.2. The approach in Example 3.6.1 can be extended to other symmetric strong block minimal bases pencils of the symmetric polynomial part $D(\lambda)$ of $R(\lambda) = R(\lambda)^T$ to construct other symmetric strong linearizations of $R(\lambda)$, as long as $D(\lambda)$ has odd-degree. See, for instance, the pencils considered in [30]. However, the linearization in Example 3.6.1 is particularly simple and, in view of the results in [14], we expect that it will have favourable numerical properties.

Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix of degree k. A $km \times km$ pencil $L(\lambda)$ is called *block-symmetric* if $L(\lambda) = L(\lambda)^{\mathcal{B}}$, where $L(\lambda)$ is viewed as a block partitioned pencil with $k \times k$ blocks each of them of size $m \times m$. Notice that a pencil $L(\lambda)$ satisfies $L(\lambda)(\Phi_k(\lambda) \otimes I_m) = v \otimes P(\lambda)$ for some vector $v \in \mathbb{F}^k$ if and only if $L(\lambda)^{\mathcal{B}}$ satisfies $(\Phi_k(\lambda)^T \otimes I_m)L(\lambda)^{\mathcal{B}} = v^T \otimes P(\lambda)$. Thus, if $L(\lambda) \in \mathbb{M}_1(P)$ is block-symmetric, then $L(\lambda) \in \mathbb{M}_1(P) \cap \mathbb{M}_2(P)$. This intersection space was introduced in [36], it is called *double generalized ansatz space*, and it is denoted by

$$\mathbb{DM}(P) = \mathbb{M}_1(P) \cap \mathbb{M}_2(P).$$

If $\{\phi_j(\lambda)\}_{j=0}^{\infty}$ is the monomial basis, the space $\mathbb{DM}(P)$ is denoted $\mathbb{DL}(P)$ and was introduced originally in [63]. In [36, Corollary 6] it is shown that if a pencil $L(\lambda)$ belongs to $\mathbb{DM}(P)$ then its right and left ansatz vectors are the same, which is called simply *ansatz vector*, and that

$$\mathbb{DM}(P) = \{ L(\lambda) \in \mathbb{M}_1(P) : L(\lambda) = L(\lambda)^{\mathcal{B}} \}.$$

In fact, if $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ is a symmetric polynomial matrix we obtain that any pencil in $\mathbb{DM}(P)$ must be symmetric. This result is not in [36], and we state it in Theorem 3.6.5. For its proof, we use Lemmas 3.6.3 and 3.6.4.

Lemma 3.6.3. Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix of degree $k \geq 2$ and let $L(\lambda) \in \mathbb{DM}(P)$ with ansatz vector $0 \in \mathbb{F}^k$. Then $L(\lambda) = 0$.

Proof. Notice that from (3.4),

$$F_{\Phi}^{P}(\lambda) = \lambda \begin{bmatrix} \alpha_{k-1}^{-1} P_{k} & 0\\ 0 & I_{(k-1)m} \end{bmatrix} + F_{\Phi}^{P}(0), \qquad (3.19)$$

where $P(\lambda)$ is expressed as in (3.3). From [36, Corollary 6], $L(\lambda)$ must have the form

$$L(\lambda) = \begin{bmatrix} 0_{km \times m} & H \end{bmatrix} F_{\Phi}^{P}(\lambda) = \begin{bmatrix} 0 & H \end{bmatrix} \lambda + \begin{bmatrix} 0 & H \end{bmatrix} F_{\Phi}^{P}(0) = \begin{bmatrix} 0 \\ H^{\mathcal{B}} \end{bmatrix} \lambda + F_{\Phi}^{P}(0)^{\mathcal{B}} \begin{bmatrix} 0 \\ H^{\mathcal{B}} \end{bmatrix}.$$

As $L(\lambda)$ is block symmetric, H must have the form $H = \begin{bmatrix} 0 \\ W \end{bmatrix}$ where W is a $(k-1)m \times (k-1)m$ block symmetric matrix. Let $W = [W_{ij}]_{i,j=1}^{k-1}$ with $W_{ij} \in \mathbb{F}^{m \times m}$. Then

$$[0_{km \times m} \quad H]F_{\Phi}^{P}(0) = \begin{bmatrix} 0 & 0 & \cdots & 0 \\ -\alpha_{k-2}W_{11} & * & \cdots & * \\ -\alpha_{k-2}W_{21} & * & \cdots & * \\ \vdots & & & \\ -\alpha_{k-2}W_{(k-1)1} & * & \cdots & * \end{bmatrix}.$$

Notice that $\begin{bmatrix} 0_{km\times m} & H \end{bmatrix} F_{\Phi}^{P}(0)$ is also block symmetric because of the block symmetry of $L(\lambda)$. Then, we obtain that $W_{i1} = W_{1i} = 0$ with $i = 1, \ldots, k - 1$. Next, we proceed by induction. Let $j \in \{2, \ldots, k - 1\}$ and suppose that $W_{it} = W_{ti} = 0$ for all $i = 1, \ldots, k - 1$ and $t = 1, \ldots, j - 1$. Then,

$$[0_{km \times m} \quad H]F_{\Phi}^{P}(0) = \begin{pmatrix} (j) \\ 0 & \cdots & 0 & 0 & 0 & \cdots & 0 \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ 0 & \cdots & 0 & 0 & 0 & \cdots & 0 \\ 0 & \cdots & 0 & -\alpha_{k-(j+1)}W_{jj} & * & \cdots & * \\ 0 & \cdots & 0 & -\alpha_{k-(j+1)}W_{(j+1)j} & * & \cdots & * \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ 0 & \cdots & 0 & -\alpha_{k-(j+1)}W_{(k-1)j} & * & \cdots & * \end{bmatrix}.$$

Therefore, $W_{ij} = W_{ji} = 0$ with i = 1, ..., k - 1. By induction, H = 0 and $L(\lambda) = 0$.

Theorem 3.4 in [53] states that for each $v \in \mathbb{F}^k$ there is a uniquely determined pencil in $\mathbb{DL}(P)$ with ansatz vector v. We show this result extended to the space $\mathbb{DM}(P)$ in the following lemma.

Lemma 3.6.4. Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix of degree $k \geq 2$. For each $v \in \mathbb{F}^k$ there is only one pencil in $\mathbb{DM}(P)$ with ansatz vector v.

Proof. We consider the linear map $\mathbb{DM}(P) \longrightarrow \mathbb{F}^k$ that associates to any pencil $L(\lambda)$ in $\mathbb{DM}(P)$ its ansatz vector $v \in \mathbb{F}^k$. By Lemma 3.6.3 this map is injective and by [36, Corollary 7] dim $(\mathbb{DM}(P)) = k$. Therefore, the map is bijective.

Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a symmetric polynomial matrix, and let us define the set

$$\mathbb{S}(P) = \{ L(\lambda) \in \mathbb{M}_1(P) : L(\lambda) = L(\lambda)^T \}.$$

The elements in $\mathbb{S}(P)$ are in $\mathbb{DM}(P)$ because if $L(\lambda) = [v \otimes I_m \quad H]F_{\Phi}^P(\lambda) \in \mathbb{S}(P)$ then $L(\lambda)^T = F_{\Phi}^P(\lambda)^{\mathcal{B}} \begin{bmatrix} v^T \otimes I_m \\ H^T \end{bmatrix} \in \mathbb{M}_2(P)$, since in the case $P(\lambda)$ is symmetric $F_{\Phi}^P(\lambda)^T = F_{\Phi}^P(\lambda)^{\mathcal{B}}$, and $L(\lambda) = L(\lambda)^T$. Moreover, Theorem 3.6.5 shows that $\mathbb{S}(P)$ and $\mathbb{DM}(P)$ are equal.

Theorem 3.6.5. Let $P(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a symmetric polynomial matrix of degree $k \geq 2$. Then

$$\mathbb{DM}(P) = \mathbb{S}(P).$$

Proof. We have already seen that $\mathbb{S}(P) \subseteq \mathbb{DM}(P)$. To see the other inclusion we only have to use Lemma 3.6.4 and [36, Corollary 6], and notice that if $L(\lambda) \in \mathbb{DM}(P)$ with $P(\lambda)$ symmetric then

$$L(\lambda) = \begin{bmatrix} v \otimes I_m & H \end{bmatrix} F_{\Phi}^P(\lambda) = F_{\Phi}^P(\lambda)^{\mathcal{B}} \begin{bmatrix} v^T \otimes I_m \\ H^{\mathcal{B}} \end{bmatrix},$$

and

$$L(\lambda)^T = \begin{bmatrix} v \otimes I_m & (H^{\mathcal{B}})^T \end{bmatrix} F_{\Phi}^P(\lambda) = F_{\Phi}^P(\lambda)^{\mathcal{B}} \begin{bmatrix} v^T \otimes I_m \\ H^T \end{bmatrix}$$

which implies that $L(\lambda)^T \in \mathbb{DM}(P)$ and that $L(\lambda)$ and $L(\lambda)^T$ have the same ansatz vector. So, by Lemma 3.6.4, $L(\lambda) = L(\lambda)^T$ and $L(\lambda) \in \mathbb{S}(P)$.

Therefore, if $P(\lambda)$ is a symmetric polynomial matrix all the pencils in $\mathbb{DM}(P)$ are also symmetric. In order to find linearizations in $\mathbb{DM}(P)$ we have to consider only regular polynomials $P(\lambda)$ because by [36, Theorem 7] if $P(\lambda)$ is a singular polynomial matrix then none of the pencils in $\mathbb{DM}(P)$ is a linearization for $P(\lambda)$.

In Theorem 3.6.9, we construct symmetric strong linearizations for a symmetric rational matrix from a particular symmetric strong linearization of its polynomial part $D(\lambda)$ when the leading coefficient D_k of $D(\lambda)$ is nonsingular. This particular strong linearization is the pencil in $\mathbb{DM}(D)$ with ansatz vector e_k , i.e., the last vector in the canonical basis of \mathbb{F}^k . Some properties of this pencil are studied in Lemma 3.6.6 and Corollary 3.6.7.

Lemma 3.6.6. Let $D(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix with degree $k \geq 2$ and let $L(\lambda) = [e_k \otimes I_m \quad H] F_{\Phi}^D(\lambda) \in \mathbb{DM}(D)$. Then $[e_k \otimes I_m \quad H]$ is nonsingular if and only if the leading matrix coefficient D_k of $D(\lambda)$ is nonsingular.

Proof. Let $L(\lambda) = [e_k \otimes I_m \quad H] F_{\Phi}^D(\lambda) \in \mathbb{DM}(D)$. We write, by using (3.19) and [36, Corollary 6],

$$L(\lambda) = \begin{bmatrix} e_k \otimes I_m & H \end{bmatrix} F_{\Phi}^D(\lambda) = \begin{bmatrix} e_k \otimes \alpha_{k-1}^{-1} D_k & H \end{bmatrix} \lambda + \begin{bmatrix} e_k \otimes I_m & H \end{bmatrix} F_{\Phi}^D(0)$$
$$= \begin{bmatrix} e_k^T \otimes \alpha_{k-1}^{-1} D_k \\ H^{\mathcal{B}} \end{bmatrix} \lambda + F_{\Phi}^D(0)^{\mathcal{B}} \begin{bmatrix} e_k^T \otimes I_m \\ H^{\mathcal{B}} \end{bmatrix}$$

Then $H = \begin{bmatrix} 0_{m \times (k-2)m} & \alpha_{k-1}^{-1}D_k \\ H' \end{bmatrix}$ for some $(k-1)m \times (k-1)m$ block symmetric matrix H'. Let $H' = [H'_{ij}]_{i,j=1}^{k-1}$ with $H'_{ij} \in \mathbb{F}^{m \times m}$. If we calculate the first block row and block column of the product $[e_k \otimes I_m \quad H]F^D_{\Phi}(0)$ we obtain

$$\begin{bmatrix} 0 & 0 & \cdots & 0 & -\frac{\alpha_0}{\alpha_{k-1}}D_k & -\frac{\beta_0}{\alpha_{k-1}}D_k \\ -\alpha_{k-2}H'_{11} & * & \cdots & * & * & * \\ -\alpha_{k-2}H'_{21} & * & \cdots & * & * & * \\ \vdots & & & & & \\ -\alpha_{k-2}H'_{(k-2)1} & * & \cdots & * & * & * \\ -\frac{\beta_{k-1}}{\alpha_{k-1}}D_k + D_{k-1} - \alpha_{k-2}H'_{(k-1)1} & * & \cdots & * & * & * \end{bmatrix}$$

Since $[e_k \otimes I_m \quad H]F^D_{\Phi}(0)$ is block symmetric we obtain

$$H'_{1i} = H'_{i1} = 0$$
 for $i = 1, \dots, k - 3$ (3.20)

and

$$-\alpha_{k-2}H'_{(k-2)1} = -\frac{\alpha_0}{\alpha_{k-1}}D_k.$$

Thus,

$$H'_{(k-2)1} = H'_{1(k-2)} = \frac{\alpha_0}{\alpha_{k-1}\alpha_{k-2}} D_k.$$
(3.21)

Using (3.20) and (3.21) and calculating the second block row and block column of the product $[e_k \otimes I_m \quad H]F^D_{\Phi}(0)$ as before, we obtain

$$H'_{2i} = H'_{i2} = 0$$
 for $i = 1, \dots, k - 4$

and

$$-\alpha_{k-3}H'_{(k-3)2} = -\alpha_1H'_{1(k-2)}$$

Thus,

$$H'_{(k-3)2} = H'_{2(k-3)} = \frac{\alpha_0 \alpha_1}{\alpha_{k-1} \alpha_{k-2} \alpha_{k-3}} D_k.$$

In general, an induction argument proves that

$$H'_{(k-j)i} = H'_{i(k-j)} = \frac{\alpha_0 \alpha_1 \cdots \alpha_{i-1}}{\alpha_{k-1} \alpha_{k-2} \cdots \alpha_{k-j}} D_k \text{ for } j-i=1,$$

and the matrix $\begin{bmatrix} e_k \otimes I_m & H \end{bmatrix}$ has the following block anti-triangular form

$$\begin{bmatrix} 0 & 0 & \cdots & 0 & 0 & 0 & \alpha_{k-1}^{-1}D_k \\ 0 & 0 & \cdots & 0 & 0 & \frac{\alpha_0}{\alpha_{k-1}\alpha_{k-2}}D_k & * \\ 0 & 0 & \cdots & 0 & \frac{\alpha_0\alpha_1}{\alpha_{k-1}\alpha_{k-2}\alpha_{k-3}}D_k & * & * \\ \vdots & & & \ddots & & \\ 0 & 0 & \frac{\alpha_0\alpha_1}{\alpha_{k-1}\alpha_{k-2}\alpha_{k-3}}D_k & * & * & * & * \\ 0 & \frac{\alpha_0}{\alpha_{k-1}\alpha_{k-2}}D_k & * & * & * & * & * \\ I_m & * & * & * & * & * & * & * \\ \end{bmatrix}$$

Therefore $\begin{bmatrix} e_k \otimes I_m & H \end{bmatrix}$ is nonsingular if and only if D_k is nonsingular.

Corollary 3.6.7. Let $D(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ be a polynomial matrix with degree $k \geq 2$ and leading matrix coefficient D_k , and let $L(\lambda) = [e_k \otimes I_m \quad H]F_{\Phi}^D(\lambda) \in \mathbb{DM}(D)$. Then the following statements hold:

- 1. $L(\lambda)$ is a strong linearization of $D(\lambda)$ if D_k is nonsingular.
- 2. If $D(\lambda)$ is regular, $L(\lambda)$ is a strong linearization of $D(\lambda)$ if and only if D_k is nonsingular.

Proof. Item 1. follows from Lemma 3.6.6 and Corollary 3.2.5. Item 2. follows from Lemma 3.6.6 and [36, Theorem 3].

Computing the pencil in $\mathbb{DM}(P)$ with ansatz vector e_k , or with any other ansatz vector v, may be difficult. In general, one can follow the procedure in [36, Section 7] or use the MATLAB code in [72, Subsection 7.1]. However, if the recurrence relation (3.2) is simple and k is low, then the computation can be performed easily by hand, as we illustrate in Example 3.6.8.

Example 3.6.8. For a second degree polynomial matrix $D(\lambda) = D_2\phi_2(\lambda) + D_1\phi_1(\lambda) + D_0\phi_0(\lambda)$ expressed in terms of a polynomial basis satisfying (3.2), the pencil $L(\lambda) \in \mathbb{DM}(D)$ with ansatz vector e_2 is

$$L(\lambda) = \begin{bmatrix} -\frac{\alpha_0}{\alpha_1} D_2 & \frac{\lambda - \beta_0}{\alpha_1} D_2 \\ \frac{\lambda - \beta_0}{\alpha_1} D_2 & \left(\frac{\beta_0 - \beta_1}{\alpha_0 \alpha_1} (\lambda - \beta_0) - \frac{\gamma_1}{\alpha_1} \right) D_2 + \frac{\lambda - \beta_0}{\alpha_0} D_1 + D_0 \end{bmatrix}.$$

This can be obtained, for instance, by computing the matrix H' as in the proof of Lemma 3.6.6. For example, Chebyshev polynomials of the first kind $\{\phi_j(\lambda)\}_{j=0}^{\infty}$ satisfy the following three-term recurrence relation:

$$\frac{1}{2}\phi_{j+1}(\lambda) = \lambda\phi_j(\lambda) - \frac{1}{2}\phi_{j-1}(\lambda) \quad j \ge 1$$
(3.22)

where $\phi_{-1}(\lambda) = 0$, $\phi_0(\lambda) = 1$ and $\phi_1(\lambda) = \lambda$. Therefore, $\alpha_0 = 1$, $\alpha_j = \gamma_j = \frac{1}{2}$ for $j \ge 1$, $\beta_j = 0$ $j \ge 0$ and

$$L(\lambda) = \begin{bmatrix} -2D_2 & 2\lambda D_2 \\ 2\lambda D_2 & \lambda D_1 + D_0 - D_2 \end{bmatrix}.$$

Chebyshev polynomials of the second kind satisfy the same recurrence relation with $\phi_1(\lambda) = 2\lambda$. Thus, $\alpha_j = \gamma_j = \frac{1}{2}$, $\beta_j = 0$ for $j \ge 0$ and

$$L(\lambda) = \begin{bmatrix} -D_2 & 2\lambda D_2 \\ 2\lambda D_2 & 2\lambda D_1 + D_0 - D_2 \end{bmatrix}.$$

For a cubic polynomial matrix $D(\lambda) = D_3\phi_3(\lambda) + D_2\phi_2(\lambda) + D_1\phi_1(\lambda) + D_0\phi_0(\lambda)$ expressed in terms of Chebyshev polynomials of the first kind, the pencil $L(\lambda) \in \mathbb{DM}(D)$ with ansatz vector e_3 is

$$L(\lambda) = \begin{bmatrix} 0 & -2D_3 & 2\lambda D_3 \\ -2D_3 & 4\lambda D_3 - 2D_2 & 2\lambda D_2 - 2D_3 \\ 2\lambda D_3 & 2\lambda D_2 - 2D_3 & \lambda(D_1 + D_3) + D_0 - D_2 \end{bmatrix}.$$

If $D(\lambda)$ is expressed in terms of Chebyshev polynomials of the second kind we obtain

$$L(\lambda) = \begin{bmatrix} 0 & -D_3 & 2\lambda D_3 \\ -D_3 & 2\lambda D_3 - D_2 & 2\lambda D_2 - D_3 \\ 2\lambda D_3 & 2\lambda D_2 - D_3 & 2\lambda D_1 + D_0 - D_2 \end{bmatrix}.$$

By using Theorems 3.2.9 or 3.3.4, Theorem 3.6.5, Lemma 3.6.6 and Proposition 3.5.5, we obtain in Theorem 3.6.9 symmetric strong linearizations of a symmetric rational matrix when the leading coefficient of its polynomial part is nonsingular as we announced.

Theorem 3.6.9. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a symmetric rational matrix and let $R(\lambda) = D(\lambda) + R_{sp}(\lambda)$ be its unique decomposition into its polynomial part $D(\lambda) \in \mathbb{F}[\lambda]^{m \times m}$ and its strictly proper part $R_{sp}(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$. Assume that $\deg(D(\lambda)) = k \geq 2$ and let $n = \nu(R(\lambda))$. Consider a symmetric minimal state-space realization of $R_{sp}(\lambda)$, i.e., $R_{sp}(\lambda) = W(\lambda S_1 - S_2)^{-1}W^T$ as in Definition 3.5.2, and $L(\lambda) \in \mathbb{DM}(D)$ with ansatz vector e_k . If the leading matrix coefficient D_k of $D(\lambda)$ is nonsingular then, for any nonsigular matrix $Z \in \mathbb{F}^{n \times n}$, the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} \frac{Z(S_2 - \lambda S_1)Z^T \mid 0_{n \times (k-1)m} \quad ZW^T}{0_{(k-1)m \times n}} \\ WZ^T \quad L(\lambda) \end{bmatrix}$$
(3.23)

is a symmetric strong linearization of $R(\lambda)$.

Proof. Let $L(\lambda) = [e_k \otimes I_m \quad H] F_{\Phi}^D(\lambda)$ be the pencil in $\mathbb{DM}(D)$ with ansatz vector e_k . Since D_k is nonsingular, the matrix $[e_k \otimes I_m \quad H]$ is also nonsingular by using Lemma 3.6.6. Notice that if $R_{sp}(\lambda) = W(\lambda S_1 - S_2)^{-1}W^T$ is a symmetric minimal state-space realization of $R_{sp}(\lambda)$ then $R_{sp}(\lambda) = W(\lambda I_n - S_1^{-1}S_2)^{-1}S_1^{-1}W^T$ is a minimal state-space realization. It only remains to consider Theorem 3.2.9 with $X = ZS_1$ and $Y = -Z^T$. Equivalently, we can consider Theorem 3.3.4 with $X = ZS_1$ and $Y = -Z^T$.

Example 3.6.10. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a symmetric rational matrix and write $R(\lambda) = D(\lambda) + R_{sp}(\lambda)$ as the sum of its polynomial part and its strictly proper part. Suppose that

$$D(\lambda) = D_k \lambda^k + D_{k-1} \lambda^{k-1} + \dots + D_1 \lambda + D_0,$$

with $k \geq 2$ and D_k nonsingular, and write $R_{sp}(\lambda) = W(\lambda S_1 - S_2)^{-1}W^T$ as a symmetric minimal state-space realization. For the monomial basis we obtain by [53, Theorem 3.5] that the pencil $L(\lambda) \in \mathbb{DL}(D)$ with ansatz vector e_k is

$$L(\lambda) = \lambda \begin{bmatrix} & & D_k \\ & \ddots & D_{k-1} \\ & \ddots & \ddots & \vdots \\ & \ddots & \ddots & D_2 \\ D_k & D_{k-1} & \cdots & D_2 & D_1 \end{bmatrix} - \begin{bmatrix} & & D_k \\ & \ddots & D_{k-1} \\ & \ddots & \ddots & \vdots \\ D_k & D_{k-1} & \cdots & D_2 \\ & & & -D_0 \end{bmatrix}.$$

Then, by Theorem 3.6.9 with $Z = I_n$, the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} S_2 - \lambda S_1 & 0_{n \times (k-1)m} & W^T \\ 0_{(k-1)m \times n} & L(\lambda) \\ W & L(\lambda) \end{bmatrix}$$

is a symmetric strong linearization of $R(\lambda)$.

We can obtain infinitely many symmetric strong linearizations by using Theorem 3.6.9 and Lemma 3.1.2.

Corollary 3.6.11. Under the same assumptions as in Theorem 3.6.9, consider the symmetric strong linearization $\mathcal{L}(\lambda)$ in (3.23). Let $Q \in \mathbb{F}^{n \times n}$, $P \in \mathbb{F}^{km \times km}$ be nonsingular matrices and $R \in \mathbb{F}^{km \times n}$. Then

$$\widehat{\mathcal{L}}(\lambda) = \begin{bmatrix} Q & 0 \\ R & P \end{bmatrix} \mathcal{L}(\lambda) \begin{bmatrix} Q^T & R^T \\ 0 & P^T \end{bmatrix}$$

is a symmetric strong linearization of $R(\lambda)$.

3.7 Hermitian strong linearizations for Hermitian rational matrices

In this section we extend the results in Sections 3.5 and 3.6 from symmetric to Hermitian rational matrices. Since most of the arguments are similar to those in the symmetric case, we limit ourselves to state the main results, and most of the proofs are omitted. We consider the ring of polynomials $\mathbb{C}[\lambda]$ and a polynomial basis $\{\phi_j(\lambda)\}_{i=0}^{\infty}$ that satisfies the three-term recurrence relation:

$$\alpha_j \phi_{j+1}(\lambda) = (\lambda - \beta_j) \phi_j(\lambda) - \gamma_j \phi_{j-1}(\lambda) \quad j \ge 0$$

as in (3.2), with $\alpha_j, \beta_j, \gamma_j \in \mathbb{R}, \alpha_j \neq 0, \phi_{-1}(\lambda) = 0$, and $\phi_0(\lambda) = 1$. Let $P(\lambda) \in \mathbb{C}[\lambda]^{m \times m}$ be a polynomial matrix of degree k written in terms of this basis, i.e., $P(\lambda) = \sum_{i=0}^{k} P_i \phi_i(\lambda)$ with $P_i \in \mathbb{C}^{m \times m}$. Suppose that $P(\lambda)$ is Hermitian, i.e., $P(\lambda)^* = k$

 $P(\overline{\lambda})$ or, equivalently, $P^*(\lambda) = P(\lambda)$, where $P^*(\lambda)$ is defined as $P^*(\lambda) = \sum_{i=0}^{k} P_i^* \phi_i(\lambda)$

with P_i^* the conjugate transpose of $P_i \in \mathbb{C}^{m \times m}$. We also consider the set of pencils

$$\mathbb{H}(P) = \{ \lambda X + Y \in \mathbb{M}_1(P) : X^* = X, Y^* = Y \}.$$

That is, $\mathbb{H}(P)$ is the set of pencils in $\mathbb{M}_1(P)$ that are Hermitian. Theorem 3.7.1 shows that the elements of $\mathbb{H}(P)$ are in $\mathbb{D}\mathbb{M}(P)$, and that, in fact, they are the pencils in $\mathbb{D}\mathbb{M}(P)$ with real ansatz vector. The proof of Theorem 3.7.1 is ommitted for brevity since it is similar to the proof of [53, Theorem 6.1], which is Theorem 3.7.1 in the particular case $\phi_i(\lambda) = \lambda^j$ for $j \geq 0$.

Theorem 3.7.1. Let $P(\lambda) \in \mathbb{C}[\lambda]^{m \times m}$ be a Hermitian polynomial matrix. Then $\mathbb{H}(P)$ is the subset of all pencils in $\mathbb{DM}(P)$ with real ansatz vector.
Let $R(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$ be a Hermitian rational matrix, i.e., a rational matrix satisfying $R(\lambda)^* = R(\overline{\lambda})$. Consider $R(\lambda) = D(\lambda) + R_{sp}(\lambda)$ as in (2.1). Then $D(\lambda)$ and $R_{sp}(\lambda)$ are also Hermitian. For Hermitian strictly proper rational matrices we introduce the notion of Hermitian minimal state-space realizations, in the spirit of Definition 3.5.2.

Definition 3.7.2. Let $R_{sp}(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$ be a Hermitian strictly proper rational matrix and let $n = \nu(R_{sp}(\lambda))$. A Hermitian minimal state-space realization of $R_{sp}(\lambda)$ is an expression of the form

$$R_{sp}(\lambda) = W(\lambda H_1 - H_2)^{-1} W^*$$

where $H_1, H_2 \in \mathbb{C}^{n \times n}$ are Hermitian matrices, with H_1 nonsingular, and $W \in \mathbb{C}^{m \times n}$.

Following arguments similar to those in Lemma 3.5.3 and Proposition 3.5.5, it is easy to see that the strictly proper part of a Hermitian rational matrix has a Hermitian minimal state-space realization.

Proposition 3.7.3. Any Hermitian strictly proper rational matrix has a Hermitian minimal state-space realization.

Proof. In order to obtain a Hermitian minimal state-space realization of $R_{sp}(\lambda)$, we can consider a minimal state-space realization $R_{sp}(\lambda) = C(\lambda I_n - A)^{-1}B$. We prove analogously to Lemma 3.5.3 that there exists a unique nonsingular and Hermitian matrix $H \in \mathbb{C}^{n \times n}$ such that $A^* = H^{-1}AH$ and $C^* = H^{-1}B$. Therefore, $R_{sp}(\lambda) = C(\lambda H^{-1} - H^{-1}A)^{-1}C^*$ is a Hermitian minimal state-space realization of $R_{sp}(\lambda)$.

Remark 3.7.4. Another constructive way to prove Proposition 3.7.3 is to consider the Hankel matrix H_n of $R_{sp}(\lambda)$ defined in (3.15), that is also Hermitian, and write

$$H_n = U \left[\begin{array}{cc} K & 0 \\ 0 & 0 \end{array} \right] U^*$$

with U unitary, i.e., $U^{-1} = U^*$, and K a diagonal matrix that has the eigenvalues different from zero of H_n at the diagonal elements. Then proceed as in the last paragraph of Remark 3.5.6 to get a Hermitian minimal state-space realization. Notice that K is Hermitian because the eigenvalues of H_n are real.

By using Proposition 3.7.3 and Theorem 3.7.1, we obtain in Theorem 3.7.5 Hermitian strong linearizations of a Hermitian rational matrix when the leading coefficient of its polynomial part is nonsingular, analogously as we did in Theorem 3.6.9 for the symmetric case.

Theorem 3.7.5. Let $R(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$ be a Hermitian rational matrix and let $R(\lambda) = D(\lambda) + R_{sp}(\lambda)$ be its unique decomposition into its polynomial part $D(\lambda) \in \mathbb{C}[\lambda]^{m \times m}$ and its strictly proper part $R_{sp}(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$. Assume that $\deg(D(\lambda)) =$

 $k \geq 2$ and let $n = \nu(R(\lambda))$. Consider a Hermitian minimal state-space realization of $R_{sp}(\lambda)$, i.e., $R_{sp}(\lambda) = W(\lambda H_1 - H_2)^{-1}W^*$ as in Definition 3.7.2, and $L(\lambda) \in \mathbb{DM}(D)$ with ansatz vector e_k . If the leading matrix coefficient D_k of $D(\lambda)$ is nonsingular then, for any nonsigular matrix $Z \in \mathbb{C}^{n \times n}$, the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} Z(H_2 - \lambda H_1)Z^* & 0_{n \times (k-1)m} & ZW^* \\ 0_{(k-1)m \times n} & L(\lambda) \\ WZ^* & L(\lambda) \end{bmatrix}$$
(3.24)

is a Hermitian strong linearization of $R(\lambda)$.

As in Corollary 3.6.11, we can obtain infinitely many Hermitian strong linearizations by using Theorem 3.7.5 and Lemma 3.1.2.

Corollary 3.7.6. Under the same assumptions as in Theorem 3.7.5, consider the Hermitian strong linearization $\mathcal{L}(\lambda)$ in (3.24). Let $Q \in \mathbb{C}^{n \times n}$, $P \in \mathbb{C}^{km \times km}$ be nonsingular matrices and $R \in \mathbb{C}^{km \times n}$. Then

$$\widehat{\mathcal{L}}(\lambda) = \begin{bmatrix} Q & 0 \\ R & P \end{bmatrix} \mathcal{L}(\lambda) \begin{bmatrix} Q^* & R^* \\ 0 & P^* \end{bmatrix}$$

is a Hermitian strong linearization of $R(\lambda)$.

3.8 Strong linearizations of rational matrices with polynomial part expressed in other polynomial bases

Polynomial bases $\{\phi_j(\lambda)\}_{j=0}^{\infty}$ satisfying a three-term recurrence relation as in (3.2) are by far the most useful in applications. However, from a theoretical point of view, a natural question is whether or not the results in this chapter can be extended to other polynomial bases. The goal of this section is to show that this can be done by using exactly the same tools that we have used in previous sections, that is, [6, Theorem 5.11], our key Lemma 3.1.2, and the results in [36]. Since the arguments in this section are very similar to the ones previously used, we will simply sketch the main ideas.

Let $D(\lambda)$ be the polynomial part of a rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$, with $\deg(D(\lambda)) = k \geq 2$. Let us consider, motivated by (3.4) and its properties, a polynomial basis $\{\psi_j(\lambda)\}_{j=0}^{\infty}$ of $\mathbb{F}[\lambda]$, with $\psi_j(\lambda)$ a polynomial of degree j, that satisfies a linear relation:

$$M_{\Psi}(\lambda)\Psi_k(\lambda) = 0, \qquad (3.25)$$

where $M_{\Psi}(\lambda) \in \mathbb{F}[\lambda]^{(k-1)\times k}$ is a minimal basis with all its row degrees equal to 1, and $\Psi_k(\lambda) = [\psi_{k-1}(\lambda) \cdots \psi_1(\lambda) \psi_0(\lambda)]^T$ with $\Psi_k(\lambda_0) \neq 0$ for all $\lambda_0 \in \overline{\mathbb{F}}$. Then there exists a vector $w \in \mathbb{F}^k$ such that

$$U(\lambda) = \begin{bmatrix} M_{\Psi}(\lambda) \otimes I_m \\ w^T \otimes I_m \end{bmatrix}$$
(3.26)

is unimodular, and its inverse has the form $U(\lambda)^{-1} = [\widehat{\Psi}_k(\lambda) \quad \Psi_k(\lambda) \otimes I_m]$ with $\widehat{\Psi}_k(\lambda) \in \mathbb{F}[\lambda]^{km \times (k-1)m}$ (see [6, Lemma 5.5]). Let

$$F_{\Psi}^{D}(\lambda) = \begin{bmatrix} m_{\Psi}^{D}(\lambda) \\ M_{\Psi}(\lambda) \otimes I_{m} \end{bmatrix} \in \mathbb{F}[\lambda]^{km \times km}$$
(3.27)

be a pencil such that $m_{\Psi}^{D}(\lambda)(\Psi_{k}(\lambda)\otimes I_{m}) = D(\lambda)$. Then, $F_{\Psi}^{D}(\lambda)$ is a strong block minimal bases pencil associated to $D(\lambda)$ with sharp degree which verifies $F_{\Psi}^{D}(\lambda)(\Psi_{k}(\lambda)\otimes I_{m}) = e_{1} \otimes D(\lambda)$. Thus, we can apply [6, Theorem 5.11] and Lemma 3.1.2 in order to construct strong linearizations of $R(\lambda)$ from pencils of the form $L(\lambda) = [v \otimes I_{m} \quad H]F_{\Psi}^{D}(\lambda)$ with $v \in \mathbb{F}^{k}$ and $[v \otimes I_{m} \quad H]$ nonsingular. Notice that pencils $L(\lambda)$ of this form verify the ansatz relation $L(\lambda)(\Psi_{k}(\lambda) \otimes I_{m}) = v \otimes D(\lambda)$. In summary, with those arguments, we obtain the following result that is the generalization of Theorem 3.2.9 for polynomial bases as in (5.19).

Theorem 3.8.1. Let $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$ be a rational matrix written as in (2.1), and let $R_{sp}(\lambda) = C(\lambda I_n - A)^{-1}B$ be a minimal state-space realization of $R_{sp}(\lambda)$. Assume that deg $(D(\lambda)) \ge 2$ and write $D(\lambda)$ in terms of a polynomial basis $\{\psi_j(\lambda)\}_{j=0}^{\infty}$ satisfying (5.19), as

$$D(\lambda) = D_k \psi_k(\lambda) + D_{k-1} \psi_{k-1}(\lambda) + \dots + D_1 \psi_1(\lambda) + D_0 \psi_0(\lambda)$$
(3.28)

with $D_k \neq 0$. Let $L(\lambda) = [v \otimes I_m \quad H] F_{\Psi}^D(\lambda)$ with $[v \otimes I_m \quad H]$ nonsingular and $F_{\Psi}^D(\lambda)$ as in (3.27). Let $w \in \mathbb{F}^k$ be the vector in (3.26). Then, for any nonsingular matrices $X, Y \in \mathbb{F}^{n \times n}$ the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB(w^T \otimes I_m) \\ -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

is a strong linearization of $R(\lambda)$.

In a similar manner, the results in Sections 3.3 and 3.4 can be extended to square rational matrices with polynomial parts expressed in terms of polynomial bases as in (5.19).

In Example 3.8.2 we consider degree-graded polynomial bases presented in [36, Section 9], and we construct strong linearizations of square rational matrices by expressing the polynomial parts in terms of these bases and using Theorem 3.8.1.

Example 3.8.2. Let $\{\psi_j(\lambda)\}_{j=0}^{\infty}$ be a degree-graded polynomial basis of $\mathbb{F}[\lambda]$ that satisfies the following recurrence relation:

$$\psi_j(\lambda) = (\lambda - \alpha_j)\psi_{j-1}(\lambda) + \sum_{i=0}^{j-2} \beta_j^i \psi_i(\lambda) \quad j \ge 1$$

where $\alpha_j \in \mathbb{F}$ for $j \geq 1$, $\beta_j^i \in \mathbb{F}$ for $j \geq 2$, $0 \leq i \leq j-2$ and $\psi_0(\lambda) = 1$. Let $R(\lambda) = D(\lambda) + C(\lambda I_n - A)^{-1}B$ be an $m \times m$ rational matrix written as in Theorem 3.8.1. We express the polynomial part $D(\lambda)$ in terms of the polynomial basis $\{\psi_j(\lambda)\}_{j=0}^{\infty}$, as in (3.28). Let us denote $\Psi_k(\lambda) = [\psi_{k-1}(\lambda) \cdots \psi_1(\lambda) \psi_0(\lambda)]^T$ and consider the following pencil $G_{\Psi}^D(\lambda)$ introduced in [36, Section 9]:

$$G_{\Psi}^{D}(\lambda) = \begin{bmatrix} m_{\Psi}^{D}(\lambda) \\ M_{\Psi}(\lambda) \otimes I_{m} \end{bmatrix} \in \mathbb{F}[\lambda]^{km \times km},$$

where

$$m_{\Psi}^{D}(\lambda) = \begin{bmatrix} (\lambda - \alpha_{k})D_{k} + D_{k-1} & \beta_{k}^{k-2}D_{k} + D_{k-2} & \cdots & \beta_{k}^{1}D_{k} + D_{1} & \beta_{k}^{0}D_{k} + D_{0} \end{bmatrix},$$

and

$$M_{\Psi}(\lambda) = \begin{bmatrix} -1 & (\lambda - \alpha_{k-1}) & \beta_{k-1}^{k-3} & \beta_{k-1}^{k-4} & \cdots & \beta_{k-1}^{2} & \beta_{k-1}^{1} & \beta_{k-1}^{0} \\ & -1 & (\lambda - \alpha_{k-2}) & \beta_{k-2}^{k-4} & \cdots & \beta_{k-2}^{2} & \beta_{k-2}^{1} & \beta_{k-2}^{0} \\ & & \ddots & \ddots & \ddots & \vdots & \vdots & \vdots \\ & & & & -1 & (\lambda - \alpha_{2}) & \beta_{2}^{0} \\ & & & & & -1 & (\lambda - \alpha_{1}) \end{bmatrix}.$$

The matrix $G_{\Psi}^{D}(\lambda)$ verifies that $G_{\Psi}^{D}(\lambda)(\Psi_{k}(\lambda) \otimes I_{m}) = e_{1} \otimes D(\lambda)$. Moreover, $G_{\Psi}^{D}(\lambda)$ is a strong block minimal bases pencil associated to $D(\lambda)$ with sharp degree. It can be proved, as in [36, Theorem 1], that any pencil $L(\lambda)$ that verifies $L(\lambda)(\Psi_{k}(\lambda) \otimes I_{m}) =$ $v \otimes D(\lambda)$ for some vector $v \in \mathbb{F}^{k}$ can be written as $L(\lambda) = [v \otimes I_{m} \quad H]G_{\Psi}^{D}(\lambda)$ for some matrix $H \in \mathbb{F}^{km \times (k-1)m}$. If we consider a pencil $L(\lambda)$ of this form with $[v \otimes I_{m} \quad H]$ nonsingular we can obtain strong linearizations for $R(\lambda)$. In particular, we have that conditions in Theorem 3.8.1 hold, and we can apply it with $w = e_{k}$. Then, we have that for any nonsingular matrices $X, Y \in \mathbb{F}^{n \times n}$ the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & 0_{n \times (k-1)m} & XB \\ \hline -(v \otimes I_m)CY & L(\lambda) \end{bmatrix}$$

is a strong linearization of $R(\lambda)$.

Chapter 4

Local linearizations of rational matrices

In this chapter, we present a definition of local linearizations of rational matrices and study their properties. This new theory of local linearizations captures and explains rigorously the properties of all the different pencils that have been constructed from the 1970's in the literature for computing zeros, poles and eigenvalues of rational matrices. The results in this chapter appear in [28] and [29].

Local linearizations are pencils associated to a rational matrix that preserve its structure of zeros and poles in subsets of any algebraically closed field \mathbb{F} , in the whole \mathbb{F} and also at infinity. In practice, one is often interested in studying the pole and zero structure of rational matrices not in the whole space $\mathbb{F} \cup \{\infty\}$ but in a particular region (see [46, 47, 60, 35]). For instance, this happens when a REP arises from approximating a NLEP, since the approximation is usually reliable only in a target region. As a consequence, the eigenvalues (those zeros that are not poles) of the approximating REP need to be computed only in that region. In this scenario, one can use local linearizations of the corresponding rational matrix which contain the information about the poles and zeros in the target region, but possibly not in the whole space $\mathbb{F} \cup \{\infty\}$. In general, the pencils in [46, 47, 60, 35] do not satisfy the definitions of linearizations and strong linearizations of rational matrices introduced in [6]. Thereby local linearizations provide extra flexibility in solving NLEPs. We assume throughout this chapter that \mathbb{F} is an algebraically closed field.

4.1 Polynomial system matrices minimal in a set and at infinity

In this section, we extend the concept of minimal polynomial system matrices from the classical scenario to a local notion. Some of the definitions in this section can also be found in [16] expressed in an abstract algebraic language.

4.1.1 Polynomial system matrices minimal in a set

Consider a polynomial system matrix

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix}$$
(4.1)

with transfer function matrix $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$. The next Definition 4.1.1 extends to subsets of \mathbb{F} the classical definition of minimal, or with least order, polynomial system matrices introduced by Rosenbrock in [78].

Definition 4.1.1 (Polynomial system matrix minimal in a subset of \mathbb{F}). Let $\Sigma \subseteq \mathbb{F}$ be nonempty. The polynomial system matrix $P(\lambda)$ in (4.1), with n > 0, is said to be minimal in Σ if, for each $\lambda_0 \in \Sigma$, the following condition holds:

$$\operatorname{rank} \begin{bmatrix} A(\lambda_0) \\ C(\lambda_0) \end{bmatrix} = \operatorname{rank} \begin{bmatrix} A(\lambda_0) & B(\lambda_0) \end{bmatrix} = n.$$
(4.2)

Rosenbrock's definition coincides with Definition 4.1.1 when $\Sigma = \mathbb{F}$.

Remark 4.1.2. Notice that nrank $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix} = \operatorname{nrank} \begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix} = n$ since $A(\lambda)$ is regular. Thus, the rank condition (4.2) holds if and only if λ_0 is neither an eigenvalue of $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix}$ nor of $\begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix}$.

Remark 4.1.3. For convenience, if n = 0 in (4.1), we adopt the agreement that $P(\lambda)$ is minimal at every point $\lambda_0 \in \mathbb{F}$.

In the next example, we illustrate Definition 4.1.1 with a rational matrix and a polynomial system matrix taken from the reference [35] dealing with numerical algorithms for solving NLEPs via rational approximation. We advance that we will use Example 4.1.4 several times for illustrating different concepts. In this respect, we emphasize that [35] does not mention polynomial system matrices at all.

Example 4.1.4. Let $R(\lambda)$ be a rational matrix of the form

$$R(\lambda) = -B_0 + \lambda A_0 + \frac{B_1}{\lambda - \sigma_1} + \dots + \frac{B_s}{\lambda - \sigma_s} \in \mathbb{C}(\lambda)^{p \times p}, \qquad (4.3)$$

with $A_0, B_0, \ldots, B_s \in \mathbb{C}^{p \times p}, \sigma_1, \ldots, \sigma_s \in \mathbb{C}$, and $\sigma_i \neq \sigma_j$ if $i \neq j$. Let us consider the linear polynomial matrix

$$P(\lambda) = \begin{bmatrix} (\lambda - \sigma_1)I & & I \\ & \ddots & & \vdots \\ & & (\lambda - \sigma_s)I & I \\ \hline & -B_1 & \cdots & -B_s & \lambda A_0 - B_0 \end{bmatrix}.$$

These matrices are introduced in [35] to tackle a NLEP $T(\lambda)v = 0$, in a certain region $\Omega \subseteq \mathbb{C}$, where the matrix $T(\lambda)$ is of the form $T(\lambda) = -B_0 + \lambda A_0 + f_1(\lambda)A_1 + \cdots + f_q(\lambda)A_q$, with $A_0, A_1, \ldots, A_q \in \mathbb{C}^{p \times p}$ and $f_i : \Omega \subseteq \mathbb{C} \longrightarrow \mathbb{C}$, $i = 1, \ldots, q$, being scalar functions nonlinear in the variable λ and holomorphic in Ω . For solving a NLEP of this form, the nonlinear matrix $T(\lambda)$ is approximated in Ω by a rational matrix $R(\lambda)$ as in (4.3), and $P(\lambda)$ is considered to linearize $R(\lambda)$. It is easy to see that $P(\lambda)$ is, in fact, a linear polynomial system matrix of $R(\lambda)$, by setting the matrix diag $((\lambda - \sigma_1)I, \ldots, (\lambda - \sigma_s)I)$ as state matrix $A(\lambda)$ in (4.1). Moreover, without any assumption, $P(\lambda)$ is minimal in $\Sigma := \mathbb{C} \setminus \{\sigma_1, \ldots, \sigma_s\}$. In particular, and according to [35], Ω is a subset of Σ . Therefore, $P(\lambda)$ is minimal in the target set Ω . For completeness, notice that a polynomial system matrix as $P(\lambda)$ is minimal in \mathbb{C} if and only if all the matrices B_1, \ldots, B_s are nonsingular.

The next result provides the pole and zero elementary divisors of a rational matrix $R(\lambda)$ in a subset Σ from any polynomial system matrix of $R(\lambda)$ minimal in Σ . This result is the counterpart of [78, Chapter 3, Theorem 4.1] for polynomial system matrices minimal in a particular subset instead of polynomial system matrices of least order.

Theorem 4.1.5. Let $\Sigma \subseteq \mathbb{F}$ be nonempty. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and let

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+p) \times (n+m)}$$

be a polynomial system matrix minimal in Σ whose transfer function matrix is $R(\lambda)$. Then the elementary divisors of $A(\lambda)$ in Σ are the pole elementary divisors of $R(\lambda)$ in Σ , and the elementary divisors of $P(\lambda)$ in Σ are the zero elementary divisors of $R(\lambda)$ in Σ .

Proof. We give the proof for a finite point $\lambda_0 \in \Sigma$. Then, the result can be extended to Σ in a natural way. Let us consider the Smith normal form of $\begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix}$. Namely, $U(\lambda) \begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix} V(\lambda) = \begin{bmatrix} S(\lambda) & 0 \end{bmatrix}$ with $U(\lambda)$ and $V(\lambda)$ unimodular matrices. Observe that $S(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ is regular since nrank $\begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix} = n$. We set $H_1(\lambda) := S(\lambda)^{-1}U(\lambda)$. Since $P(\lambda)$ is minimal at λ_0 , $S(\lambda)$ has no zeros at λ_0 . Therefore, $H_1(\lambda)$ is invertible at λ_0 . Moreover, $\begin{bmatrix} H_1(\lambda)A(\lambda) & H_1(\lambda)B(\lambda) \end{bmatrix}$ is a polynomial matrix, as it is equal to $\begin{bmatrix} I_n & 0 \end{bmatrix} V(\lambda)^{-1}$, has full row normal rank, and has no zeros in \mathbb{F} . Now, let us consider the Smith normal form of the polynomial matrix $\begin{bmatrix} H_1(\lambda)A(\lambda) \\ -C(\lambda) \end{bmatrix}$. Namely, $\widetilde{U}(\lambda) \begin{bmatrix} H_1(\lambda)A(\lambda) \\ -C(\lambda) \end{bmatrix} \widetilde{V}(\lambda) = \begin{bmatrix} \widetilde{S}(\lambda) \\ 0 \end{bmatrix}$ with $\widetilde{U}(\lambda)$ and $\widetilde{V}(\lambda)$ unimodular matrices. Observe that $\widetilde{S}(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ is regular since $H_1(\lambda)$ is regular and nrank $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix} = n$. We set $H_2(\lambda) := \widetilde{V}(\lambda)\widetilde{S}(\lambda)^{-1}$. Moreover, the matrix $\begin{bmatrix} H_1(\lambda)A(\lambda)H_2(\lambda) \\ -C(\lambda)H_2(\lambda) \end{bmatrix}$ is also polynomial, as it is equal to $\widetilde{U}(\lambda)^{-1} \begin{bmatrix} I_n \\ 0 \end{bmatrix}$, has full column

normal rank, and has no zeros in \mathbb{F} . Since $P(\lambda)$ is minimal at λ_0 and $H_1(\lambda)$ is invertible at λ_0 , $\tilde{S}(\lambda)$ has no zeros at λ_0 . Therefore, $H_2(\lambda)$ is invertible at λ_0 . Let us define now the polynomial system matrix

$$\widetilde{P}(\lambda) := \begin{bmatrix} H_1(\lambda) & 0\\ 0 & I_p \end{bmatrix} \begin{bmatrix} A(\lambda) & B(\lambda)\\ -C(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} H_2(\lambda) & 0\\ 0 & I_m \end{bmatrix} = \begin{bmatrix} H_1(\lambda)A(\lambda)H_2(\lambda) & H_1(\lambda)B(\lambda)\\ -C(\lambda)H_2(\lambda) & D(\lambda) \end{bmatrix}.$$

We claim that $\widetilde{P}(\lambda)$ is a minimal polynomial system matrix in \mathbb{F} or in the classical sense of [78]. For that, it remains to prove that the matrix

$$Z(\lambda) := \begin{bmatrix} H_1(\lambda)A(\lambda)H_2(\lambda) & H_1(\lambda)B(\lambda) \end{bmatrix}$$

has full row rank for all $\lambda \in \mathbb{F}$. Let us suppose that there exists $\lambda_1 \in \mathbb{F}$ such that rank $Z(\lambda_1) < n$. We know that rank $\begin{bmatrix} H_1(\lambda_1)A(\lambda_1)\widetilde{V}(\lambda_1) & H_1(\lambda_1)B(\lambda_1) \end{bmatrix} = n$ since the Smith normal form of $\begin{bmatrix} H_1(\lambda)A(\lambda) & H_1(\lambda)B(\lambda) \end{bmatrix}$ is equal to $\begin{bmatrix} I_n & 0 \end{bmatrix}$ and $\widetilde{V}(\lambda)$ is unimodular. On the other hand, we have that

$$\operatorname{rank} \begin{bmatrix} H_1(\lambda_1)A(\lambda_1)\widetilde{V}(\lambda_1) & H_1(\lambda_1)B(\lambda_1) \end{bmatrix} = \operatorname{rank} \begin{pmatrix} Z(\lambda_1) \begin{bmatrix} \widetilde{S}(\lambda_1) & 0\\ 0 & I_m \end{bmatrix} \end{pmatrix}$$

and rank $\begin{pmatrix} Z(\lambda_1) \begin{bmatrix} \tilde{S}(\lambda_1) & 0 \\ 0 & I_m \end{bmatrix} \end{pmatrix} \leq \operatorname{rank} Z(\lambda_1) < n$, which is a contradiction. Therefore, $\tilde{P}(\lambda)$ is a minimal polynomial system matrix. Its transfer function matrix is $R(\lambda)$. Then, by [78, Chapter 3, Theorem 4.1], we know that the zero elementary divisors of $R(\lambda)$ are the elementary divisors of $\tilde{P}(\lambda)$, and that the pole elementary divisors of $R(\lambda)$ are the elementary divisors of $H_1(\lambda)A(\lambda)H_2(\lambda)$. Finally, the result

follows by taking into account that $P(\lambda) \sim_{\lambda_0} \tilde{P}(\lambda)$ and $A(\lambda) \sim_{\lambda_0} H_1(\lambda)A(\lambda)H_2(\lambda)$, since $H_1(\lambda)$ and $H_2(\lambda)$ are both invertible at λ_0 .

Example 4.1.6. If Theorem 4.1.5 is applied in Example 4.1.4, we obtain that (without any hypothesis) the eigenvalues of $P(\lambda)$ in Σ coincide exactly with the zeros of $R(\lambda)$ in Σ , with exactly the same multiplicities (geometric, algebraic and partial). In addition, all the zeros of $R(\lambda)$ in Σ are, in fact, eigenvalues of $R(\lambda)$ because the only potential poles of $R(\lambda)$ are $\sigma_1, \ldots, \sigma_s$. This result is stronger than Lemma 3.1 and Corollary 3.2 in [35] from two perspectives: [35] deals with determinants and, so, only gives information on algebraic multiplicities, and the requirements in [35] impose the additional hypothesis that A_0 is nonsingular. Note that, under the assumption that all the matrices B_1, \ldots, B_s are nonsingular, $P(\lambda)$ and $A(\lambda)$ allow us to obtain the complete information on finite zeros and poles (including all the multiplicities) of $R(\lambda)$ in \mathbb{C} .

4.1.2 Polynomial system matrices minimal at infinity

Theorem 4.1.5 characterizes polynomial system matrices that contain the information of the invariant orders at finite points of their transfer functions. The extension of these results for including the information at infinity is an old problem that has been considered in classical papers as, for instance, in [91, 92]. However, a satisfactory solution has been found, so far, only for polynomial system matrices with state matrix $A(\lambda)$ being a linear polynomial matrix and the other blocks $B(\lambda)$, $C(\lambda)$, $D(\lambda)$ being constant matrices. In other cases, recovering the information at infinity requires to embed the polynomial system matrix into a larger matrix. In this section, we propose a new approach for obtaining a counterpart of Theorem 4.1.5 at infinity.

First, we introduce the notion of g-reversal of a rational matrix in Definition 4.1.7, where g is any integer.

Definition 4.1.7 (g-reversal of a rational matrix). Let $R(\lambda)$ be a rational matrix, and let g be an integer. We define the g-reversal of $R(\lambda)$ as the rational matrix

$$\operatorname{rev}_g R(\lambda) := \lambda^g R\left(\frac{1}{\lambda}\right).$$

Consider now $R(\lambda)$ expressed as the sum of its polynomial and strictly proper parts. Namely,

$$R(\lambda) = D(\lambda) + R_{sp}(\lambda)$$

where $D(\lambda)$ is a polynomial matrix and $R_{sp}(\lambda)$ is a strictly proper rational matrix. If $g = \deg(D(\lambda))$, whenever $R(\lambda)$ is not strictly proper, or g = 0, if $R(\lambda)$ is strictly proper, then the g-reversal is called the *reversal of* $R(\lambda)$ and is denoted by just rev $R(\lambda)$ (recall Definition 2.2.1).

Definition 4.1.7 extends the definition of g-reversal for polynomial matrices (see, for instance, [22, Definition 2.12]). However, in the definition of g-reversal of a polynomial matrix considered previously in the literature, g is always taken larger than or equal to the degree of the polynomial matrix, while in Definition 4.1.7 we only ask for g to be an integer.

Given a polynomial system matrix $P(\lambda)$ as in (4.1) of degree d, we have that

$$\operatorname{rev} P(\lambda) = \begin{bmatrix} \operatorname{rev}_d A(\lambda) & \operatorname{rev}_d B(\lambda) \\ -\operatorname{rev}_d C(\lambda) & \operatorname{rev}_d D(\lambda) \end{bmatrix}$$

is also a polynomial matrix. Moreover, $\operatorname{rev}_d A(\lambda)$ is regular since $A(\lambda)$ is regular. Therefore, $\operatorname{rev} P(\lambda)$ is also a polynomial system matrix. By using the notion of reversal, we introduce the concept of minimality at infinity of a polynomial system matrix.

Definition 4.1.8 (Polynomial system matrix minimal at infinity). The polynomial system matrix $P(\lambda)$ in (4.1) is minimal at ∞ if rev $P(\lambda)$ is minimal at 0.

Example 4.1.9. The polynomial system matrix in Example 4.1.4 is minimal at ∞ since rev $P(\lambda)$ is, obviously, minimal at 0.

Remark 4.1.10. A polynomial system matrix $P(\lambda)$ as in (4.1), with deg $(P(\lambda)) = d$ and n > 0, is minimal at ∞ if and only if

$$\operatorname{rank} \begin{bmatrix} \operatorname{rev}_d A(0) \\ \operatorname{rev}_d C(0) \end{bmatrix} = \operatorname{rank} \begin{bmatrix} \operatorname{rev}_d A(0) & \operatorname{rev}_d B(0) \end{bmatrix} = n.$$

More precisely, let A_d , B_d and C_d be the matrix coefficients of λ^d in $A(\lambda)$, $B(\lambda)$ and $C(\lambda)$, respectively. Then $P(\lambda)$ is minimal at ∞ if and only if

$$\operatorname{rank} \begin{bmatrix} A_d \\ C_d \end{bmatrix} = \operatorname{rank} \begin{bmatrix} A_d & B_d \end{bmatrix} = n.$$

Notice that if d = 0 then $P(\lambda)$ is a constant polynomial system matrix, and A_0 must be invertible. Therefore, in this case, the rank condition above is automatically satisfied, and $P(\lambda)$ is minimal at ∞ .

Theorem 4.1.11 is essentially the counterpart of Theorem 4.1.5 at infinity. We state it in terms of reversals and their elementary divisors at 0 as we only have defined elementary divisors for finite points. The implications of Theorem 4.1.11 on the structure at infinity are made explicit in Theorem 4.1.13.

Theorem 4.1.11. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and let

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+p) \times (n+m)}$$

be a polynomial system matrix of degree d minimal at ∞ whose transfer function matrix is $R(\lambda)$. Then the elementary divisors of $\operatorname{rev}_d A(\lambda)$ at 0 are the pole elementary divisors of $\operatorname{rev}_d R(\lambda)$ at 0, and the elementary divisors of $\operatorname{rev} P(\lambda)$ at 0 are the zero elementary divisors of $\operatorname{rev}_d R(\lambda)$ at 0.

Proof. It can be easily proved that the transfer function matrix of rev $P(\lambda)$ is rev_d $R(\lambda)$. The theorem then follows by applying Theorem 4.1.5, since rev $P(\lambda)$ is minimal at 0.

Once we have obtained the elementary divisors of the *d*-reversal of a rational matrix at 0, from one of its polynomial system matrices of degree *d* minimal at ∞ , we can then obtain its invariant orders at infinity as we state in Theorem 4.1.13. For proving that, we use Lemma 4.1.12.

Lemma 4.1.12. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ with nrank $R(\lambda) = r$, and let g be an integer. Let e_1, \ldots, e_r be the invariant orders of $\operatorname{rev}_g R(\lambda)$ at 0, and let q_1, \ldots, q_r be the invariant orders at infinity of $R(\lambda)$. Then

$$e_i = q_i + g \quad i = 1, \dots, r.$$
 (4.4)

Proof. From the local Smith–McMillan form at infinity of $R(\lambda)$, there exist biproper rational matrices $B_1(\lambda)$ and $B_2(\lambda)$ such that

$$R(\lambda) = B_1(\lambda) \operatorname{diag}\left((1/\lambda)^{q_1}, \dots, (1/\lambda)^{q_r}, 0_{(p-r)\times(m-r)}\right) B_2(\lambda).$$

Let us perform the transformation $\lambda \mapsto 1/\lambda$ on the variable of the equation above. Thus,

$$R(1/\lambda) = B_1(1/\lambda) \operatorname{diag} \left(\lambda^{q_1}, \dots, \lambda^{q_r}, 0_{(p-r)\times(m-r)}\right) B_2(1/\lambda).$$

By [10, Lemma 6.9], $B_1(1/\lambda)$ and $B_2(1/\lambda)$ are invertible at 0. We now multiply the previous equation by λ^g , and we get that $q_i + g$ for $i = 1, \ldots, r$ are the invariant orders of rev_g $R(\lambda)$ at 0.

Theorem 4.1.13. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ with nrank $R(\lambda) = r$ and let

$$P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+p) \times (n+m)}$$

be a polynomial system matrix of degree d minimal at ∞ whose transfer function matrix is $R(\lambda)$. Let $e_1 \leq \cdots \leq e_s$ be the partial multiplicities of $\operatorname{rev}_d A(\lambda)$ at 0 and let $\tilde{e}_1 \leq \cdots \leq \tilde{e}_u$ be the partial multiplicities of $\operatorname{rev} P(\lambda)$ at 0. Then the invariant orders at infinity $q_1 \leq \cdots \leq q_r$ of $R(\lambda)$ are

$$(q_1, q_2, \dots, q_r) = (-e_s, -e_{s-1}, \dots, -e_1, \underbrace{0, \dots, 0}_{r-s-u}, \widetilde{e}_1, \widetilde{e}_2, \dots, \widetilde{e}_u) - (d, d, \dots, d).$$

Proof. By Theorem 4.1.11, we know that e_i and \tilde{e}_j with $i = 1, \ldots, s$ and $j = 1, \ldots, u$ are the pole and zero partial multiplicities of $\operatorname{rev}_d R(\lambda)$ at 0, respectively. Thus, the invariant orders of $\operatorname{rev}_d R(\lambda)$ at 0 are

$$-e_s \leq -e_{s-1} \leq \cdots \leq -e_1 < \underbrace{0 = \cdots = 0}_{r-s-u} < \widetilde{e}_1 \leq \cdots \leq \widetilde{e}_u.$$

Then the use of Lemma 4.1.12 completes the proof.

Example 4.1.14. By combining Theorem 4.1.13 and Example 4.1.9, we see that $P(\lambda)$ contains the complete information about the invariant orders at ∞ of $R(\lambda)$ (without imposing any hypothesis). Note that, in this case, d = 1 and that the 1-reversal of the state matrix, i.e., $\operatorname{rev}_1 A(\lambda) = \operatorname{diag}((1 - \lambda \sigma_1)I, \dots, (1 - \lambda \sigma_s)I)$, has no partial multiplicities at 0. This result on the relationship between the infinite structure of $R(\lambda)$ and the reversal of $P(\lambda)$ is not mentioned in [35].

For polynomial system matrices that are minimal at infinity and, also, at every finite point, we state Definition 4.1.15 about strong minimality. This definition was introduced in [32, Definition 3.3]. However, in [32] the definition is given in terms of eigenvalues instead of minimality at every point, but both definitions are equivalent. We introduce the definition in [32] in Chapter 7.

Definition 4.1.15 (Strongly minimal polynomial system matrix). The polynomial system matrix $P(\lambda)$ in (4.1) is strongly minimal if it is minimal at each point of $\mathbb{F} \cup \{\infty\}$.

We emphasize that, as a consequence of Theorems 4.1.5 and 4.1.13, strongly minimal polynomial system matrices contain all the information about the invariant orders of their transfer function matrices, both at finite points and at infinity.

4.2 Local linearizations of rational matrices

In this section, we give separately the definitions of linearizations of rational matrices in subsets of \mathbb{F} and at infinity. These linearizations will be useful in order to study the pole and zero structure of rational matrices in different sets containing infinity or not. In particular, and as an application of these definitions, we will study in Chapter 5 the structure of the linearizations that appear in [47] and [60].

4.2.1 Linearizations in a set

In this subsection we introduce the definition of linearization of a rational matrix in a set not containing infinity and study some of its properties.

Definition 4.2.1 (Linearization in a subset of \mathbb{F}). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and let $\Sigma \subseteq \mathbb{F}$ be nonempty. Let

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1 \lambda + A_0 & B_1 \lambda + B_0 \\ -(C_1 \lambda + C_0) & D_1 \lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q) \times (n+r)}$$
(4.5)

be a linear polynomial system matrix with state matrix $A_1\lambda + A_0$ and let

$$\widehat{R}(\lambda) = (D_1\lambda + D_0) + (C_1\lambda + C_0)(A_1\lambda + A_0)^{-1}(B_1\lambda + B_0) \in \mathbb{F}(\lambda)^{q \times r}$$

be its transfer function matrix. $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Σ if the following conditions hold:

- (a) $\mathcal{L}(\lambda)$ is minimal in Σ , and
- (b) there exist nonnegative integers s_1, s_2 satisfying $s_1 s_2 = q p = r m$, such that

$$\operatorname{diag}(R(\lambda), I_{s_1}) \sim_{\Sigma} \operatorname{diag}(R(\lambda), I_{s_2}).$$

$$(4.6)$$

Linearizations of rational matrices are polynomial system matrices and their definition includes a specific partition. Thus, a fixed linear polynomial matrix may be partitioned in different ways giving rise to different linearizations of the same or of different rational matrices, or in different subsets. To deal with different partitions, we will use expressions as " $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Σ with state matrix $A_1\lambda + A_0$ " when it is necessary for avoiding any ambiguity. The expression " $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Σ with empty state matrix" will cover the case n = 0 in (4.5), which does not give us pole information since the pole information is contained in the state matrix (see Remark 4.2.7).

Remark 4.2.2. We remark the following extreme cases since they are important in applications and make Definition 4.2.1 very general:

- 1. $\widehat{R}(\lambda) = R(\lambda)$. That is, the transfer function matrix of $\mathcal{L}(\lambda)$ is the desired rational matrix $R(\lambda)$. Then we just have to check condition (a). It follows that any linear polynomial system matrix $\mathcal{L}(\lambda)$ is a linearization of its transfer function matrix in the sets where $\mathcal{L}(\lambda)$ is minimal.
- 2. n = 0. Then it is not necessary to take into account condition (a) (it is automatically satisfied by the agreement in Remark 4.1.3) and, therefore, we just have to check condition (b) with $\widehat{R}(\lambda) = D_1 \lambda + D_0 = \mathcal{L}(\lambda)$. That is,

diag $(R(\lambda), I_{s_1}) \sim_{\Sigma}$ diag $(\mathcal{L}(\lambda), I_{s_2})$.

Notice that if we want a linearization of $R(\lambda)$ in $\Sigma = \mathbb{F}$ we cannot consider the case n = 0 unless $R(\lambda)$ is polynomial.

In condition (4.6), one can always take $s_1 = 0$ or $s_2 = 0$, according to $p \ge q$ and $m \ge r$ or $q \ge p$ and $r \ge m$, respectively. This is a consequence of the local Smith–McMillan forms of diag $(R(\lambda), I_{s_1})$ and diag $(\hat{R}(\lambda), I_{s_2})$ being equivalent to each other in Σ . In the rest of the results of this subsection, we will consider $s := s_1 \ge 0$ and $s_2 = 0$, since it corresponds to the most interesting situation in applications.

Remark 4.2.3. If we have a linearization of $R(\lambda)$ in a set Σ then, for each point $\mu \in \Sigma$, there exist rational matrices $H_1^{\mu}(\lambda)$ and $H_2^{\mu}(\lambda)$ invertible at μ such that $H_1^{\mu}(\lambda) \operatorname{diag}(R(\lambda), I_s) H_2^{\mu}(\lambda) = \widehat{R}(\lambda)$. In principle, for different values of $\mu \in \Sigma$, the rational matrices $H_1^{\mu}(\lambda)$ (respectively, $H_2^{\mu}(\lambda)$) may be different from each other, that is, $H_1^{\mu}(\lambda)$ (resp., $H_2^{\mu}(\lambda)$) depends on μ . However, Proposition 2.1.5 implies that the existence of $H_1^{\mu}(\lambda)$ and $H_2^{\mu}(\lambda)$ for each $\mu \in \Sigma$ is equivalent to the existence of two rational matrices $H_1(\lambda)$ and $H_2(\lambda)$ both invertible in Σ (and independent of μ) such that $H_1(\lambda) \operatorname{diag}(R(\lambda), I_s) H_2(\lambda) = \widehat{R}(\lambda)$.

Remark 4.2.4. When $\Sigma = \mathbb{F}$, in Definition 4.2.1, condition (4.6) is satisfied with unimodular equivalence. Therefore, a linearization in \mathbb{F} , or at every point of \mathbb{F} , is a linearization in the sense of [7, Definition 3.2] and vice versa.

The next result gives the relation between the Smith–McMillan forms at a finite point of the rational matrices $R(\lambda)$ and $\operatorname{diag}(R(\lambda), I_s)$, with s > 0. It is motivated by (4.6). **Lemma 4.2.5.** Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and let diag $((\lambda - \lambda_0)^{\nu_1}, \dots, (\lambda - \lambda_0)^{\nu_k}, (\lambda - \lambda_0)^{\nu_{k+1}}, \dots, (\lambda - \lambda_0)^{\nu_r}, 0_{(p-r) \times (m-r)})$ be the Smith–McMillan form at $\lambda_0 \in \mathbb{F}$ of $R(\lambda)$, with $\nu_i \leq 0$ for $i = 1, \dots, k$ and $\nu_i > 0$ for $i = k + 1, \dots, r$. Then the Smith–McMillan form at λ_0 of diag $(R(\lambda), I_s)$ is diag $((\lambda - \lambda_0)^{\nu_1}, \dots, (\lambda - \lambda_0)^{\nu_k}, I_s, (\lambda - \lambda_0)^{\nu_{k+1}}, \dots, (\lambda - \lambda_0)^{\nu_r}, 0_{(p-r) \times (m-r)})$.

Proof. Let $M(\lambda) := \text{diag}\left((\lambda - \lambda_0)^{\nu_1}, \dots, (\lambda - \lambda_0)^{\nu_r}, 0_{(p-r)\times(m-r)}\right)$ be the local Smith-McMillan form of $R(\lambda)$ at λ_0 . Then, $R(\lambda) = R_1(\lambda)M(\lambda)R_2(\lambda)$ for some rational matrices $R_1(\lambda)$ and $R_2(\lambda)$ invertible at λ_0 . Moreover,

$$\operatorname{diag}(R(\lambda), I_s) = \operatorname{diag}(R_1(\lambda), I_s) \operatorname{diag}(M(\lambda), I_s) \operatorname{diag}(R_2(\lambda), I_s).$$

Therefore, since the matrices diag $(R_1(\lambda), I_s)$ and diag $(R_2(\lambda), I_s)$ are invertible at λ_0 , the local Smith–McMillan form of diag $(R(\lambda), I_s)$ at λ_0 is diag $(M(\lambda), I_s)$ up to a permutation.

Theorem 4.2.6 states the spectral information that one can obtain from local linearizations in the spirit of [7, Theorem 3.10].

Theorem 4.2.6 (Spectral characterization of linearizations in a subset of \mathbb{F}). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}, \Sigma \subseteq \mathbb{F}$ nonempty and let

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1 \lambda + A_0 & B_1 \lambda + B_0 \\ -(C_1 \lambda + C_0) & D_1 \lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+(p+s)) \times (n+(m+s))}$$

be a linear polynomial system matrix, with state matrix $A_1\lambda + A_0$, minimal in Σ . Then $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Σ if and only if the following conditions hold:

- (a) nrank $\mathcal{L}(\lambda) = \operatorname{nrank} R(\lambda) + n + s$,
- (b) the pole elementary divisors of R(λ) in Σ are the elementary divisors of A₁λ + A₀ in Σ, and the zero elementary divisors of R(λ) in Σ are the elementary divisors of L(λ) in Σ.

Proof. We give the proof for a point $\lambda_0 \in \Sigma$. Then, the result can be extended to Σ in a natural way. Let $\widehat{R}(\lambda)$ be the transfer function matrix of $\mathcal{L}(\lambda)$. First, assume that $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ at λ_0 . By (2.9), nrank $\widehat{R}(\lambda) = \operatorname{nrank} \mathcal{L}(\lambda) - n$. And, by Lemma 4.2.5, nrank $\widehat{R}(\lambda) = \operatorname{nrank} R(\lambda) + s$. Then, nrank $\mathcal{L}(\lambda) = \operatorname{nrank} R(\lambda) + n + s$. By Lemma 4.2.5, we also have that $R(\lambda)$ and $\widehat{R}(\lambda)$ have the same pole and zero elementary divisors at λ_0 . Then (b) follows from Theorem 4.1.5, since the pole elementary divisors of $\widehat{R}(\lambda)$ at λ_0 are the elementary divisors of $A_1\lambda + A_0$ at λ_0 , and the zero elementary divisors of $\widehat{R}(\lambda)$ at λ_0 are the elementary divisors of $\mathcal{L}(\lambda)$ at λ_0 . For the converse, suppose that diag $((\lambda - \lambda_0)^{\nu_1}, \ldots, (\lambda - \lambda_0)^{\nu_k}, (\lambda - \lambda_0)^{\nu_{k+1}}, \ldots, (\lambda - \lambda_0)^{\nu_r}, 0_{(p-r)\times(m-r)})$ is the Smith–McMillan form at λ_0 of $R(\lambda)$, with $\nu_i \leq 0$ for $i = 1, \ldots, k$ and $\nu_i > 0$ for $i = k + 1, \ldots, r$. From (b) and Theorem 4.1.5, the pole and zero elementary divisors of $R(\lambda)$ and $\widehat{R}(\lambda)$ are the same. Moreover, by (2.9) and (a), nrank $\widehat{R}(\lambda) = \operatorname{nrank} R(\lambda) + s$. Therefore, the Smith–McMillan form at λ_0 of $\widehat{R}(\lambda)$ must be diag $((\lambda - \lambda_0)^{\nu_1}, \ldots, (\lambda - \lambda_0)^{\nu_k}, I_s, (\lambda - \lambda_0)^{\nu_{k+1}}, \ldots, (\lambda - \lambda_0)^{\nu_r}, 0_{(p-r)\times(m-r)})$. This is also the Smith–McMillan form at λ_0 of diag $(R(\lambda), I_s)$, as stated in the previous lemma. Thus, diag $(R(\lambda), I_s) \sim_{\lambda_0} \widehat{R}(\lambda)$.

Remark 4.2.7. Notice that if n = 0 in Theorem 4.2.6 then we can not obtain pole information in Σ from the linearization $\mathcal{L}(\lambda)$ since the state matrix is empty.

Example 4.2.8. Consider Example 4.1.4. By combining the discussion in that example with Remark 4.2.2(case 1), we immediately obtain that $P(\lambda)$ is a linearization of $R(\lambda)$ in Σ . With a bit more effort, it is also easy to obtain the following stronger result: $P(\lambda)$ is a linearization of $R(\lambda)$ in $\mathbb{C}\setminus\Pi$ where $\Pi := \{\sigma_i : B_i \text{ is singular for } 1 \leq i \leq s\}$.

4.2.2 Linearizations at infinity and in sets containing infinity

Our definition of linearization of a rational matrix at infinity is based on the notion of g-reversal of a rational matrix introduced in Definition 4.1.7.

Definition 4.2.9 (Linearization at infinity of grade g). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$. Let

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1 \lambda + A_0 & B_1 \lambda + B_0 \\ -(C_1 \lambda + C_0) & D_1 \lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q) \times (n+r)}$$
(4.7)

be a linear polynomial system matrix with state matrix $A_1\lambda + A_0$ and let

$$\widehat{R}(\lambda) = (D_1\lambda + D_0) + (C_1\lambda + C_0)(A_1\lambda + A_0)^{-1}(B_1\lambda + B_0) \in \mathbb{F}(\lambda)^{q \times r}$$

be its transfer function matrix. Let g be an integer. $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ at ∞ of grade g if the following conditions hold:

- (a) rev $\mathcal{L}(\lambda)$ is minimal at 0, and
- (b) there exist nonnegative integers s_1, s_2 , with $s_1 s_2 = q p = r m$, such that

$$\operatorname{diag}(\operatorname{rev}_{g} R(\lambda), I_{s_{1}}) \sim_{0} \operatorname{diag}(\operatorname{rev}_{\ell} R(\lambda), I_{s_{2}}), \tag{4.8}$$

where $\ell = \deg(\mathcal{L}(\lambda))$.

Observe that Definition 4.2.9 allows, for completeness, the possibility of $\ell = \deg(\mathcal{L}(\lambda))$ being equal to 0. We admit that this case has a very limited interest in applications, since it corresponds to $\mathcal{L}(\lambda)$ and $\operatorname{rev}_{\ell} \widehat{R}(\lambda) = \widehat{R}(\lambda)$ being constant

matrices. However, it includes linearizations at ∞ of rational matrices $R(\lambda)$ such that, for some integer g, $\operatorname{rev}_g R(\lambda)$ has all its invariant orders at 0 equal to zero. Moreover, notice that, in any case, $\operatorname{rev} \mathcal{L}(\lambda)$ is also a linear polynomial system matrix since $\operatorname{rev}_\ell(A_1\lambda + A_0)$ is nonsingular. We then have the following characterization of linearizations at infinity, which follows from Definition 4.2.1 and the fact that $\operatorname{rev}_\ell \widehat{R}(\lambda)$ with $\ell = \operatorname{deg}(\mathcal{L}(\lambda))$ is the transfer function matrix of $\operatorname{rev} \mathcal{L}(\lambda)$.

Proposition 4.2.10. A linear polynomial system matrix $\mathcal{L}(\lambda)$ as in (4.7) is a linearization of a rational matrix $R(\lambda)$ at ∞ of grade g if and only if rev $\mathcal{L}(\lambda)$ is a linearization of rev_g $R(\lambda)$ at 0.

Conditions (a) and (b) in Definition 4.2.9 can be stated in a different way as we show in Remarks 4.2.11 and 4.2.13, respectively.

Remark 4.2.11. As a particular case of what is discussed in Remark 4.1.10, condition (a) in Definition 4.2.9 is equivalent to

$$\operatorname{rank} \begin{bmatrix} A_1 \\ C_1 \end{bmatrix} = \operatorname{rank} \begin{bmatrix} A_1 & B_1 \end{bmatrix} = n, \tag{4.9}$$

if $\mathcal{L}(\lambda)$ is nonconstant, i.e., if $\ell = 1$. If $\mathcal{L}(\lambda)$ is constant, i.e., $\ell = 0$, condition (a) is automatically satisfied since $\mathcal{L}(\lambda)$ is a polynomial system matrix and, therefore, A_0 is invertible. We emphasize that when a nonconstant linear polynomial system matrix $\mathcal{L}(\lambda)$ as in (4.7) satisfies condition (4.9) then $\mathcal{L}(\lambda)$ is a linearization of its transfer function matrix $\widehat{R}(\lambda)$ at ∞ of grade 1. If $\mathcal{L}(\lambda)$ is constant then $\mathcal{L}(\lambda)$ is a linearization of $\widehat{R}(\lambda)$ at ∞ of grade 0.

Example 4.2.12. Consider the matrices in Example 4.1.4. By Remark 4.2.11, the linear polynomial system matrix $P(\lambda)$ is a linearization of $R(\lambda)$ at ∞ of grade 1.

Remark 4.2.13. By performing the transformation $\lambda \mapsto 1/\lambda$, condition (b) in Definition 4.2.9 is equivalent to diag $((1/\lambda)^g R(\lambda), I_{s_1}) \sim_{\infty} \text{diag}((1/\lambda)^{\ell} \widehat{R}(\lambda), I_{s_2})$.

We state in Theorem 4.2.14 a characterization of linearizations at ∞ analogous to the one in Theorem 4.2.6 for linearizations at finite points. In this characterization, we consider the most usual situation $s_1 := s \ge 0$ and $s_2 = 0$, assuming $q \ge p$ and $r \ge m$. Its proof is omitted since it follows immediately from Theorem 4.2.6 and Proposition 4.2.10.

Theorem 4.2.14 (Spectral characterization of linearizations at infinity). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and let

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1\lambda + A_0 & B_1\lambda + B_0 \\ -(C_1\lambda + C_0) & D_1\lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+(p+s))\times(n+(m+s))}$$

be a linear polynomial system matrix, with state matrix $A_1\lambda + A_0$, such that rev $\mathcal{L}(\lambda)$ is minimal at 0. Let $\ell = \deg(\mathcal{L}(\lambda))$. Then $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ at ∞ of grade g if and only if the following conditions hold:

- (a) nrank $\mathcal{L}(\lambda) = \operatorname{nrank} R(\lambda) + n + s$,
- (b) the pole elementary divisors of rev_g R(λ) at 0 are the elementary divisors of rev_ℓ(A₁λ + A₀) at 0, and the zero elementary divisors of rev_g R(λ) at 0 are the elementary divisors of rev L(λ) at 0.

Next, we study in Proposition 4.2.15 how to recover the invariant orders at infinity of rational matrices from linearizations at infinity of grade g. Its proof is analogous to the one for Theorem 4.1.13. It follows from combining Theorem 4.2.14 and Lemma 4.1.12.

Proposition 4.2.15. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ with nrank $R(\lambda) = r$, and let

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1 \lambda + A_0 & B_1 \lambda + B_0 \\ -(C_1 \lambda + C_0) & D_1 \lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+(p+s)) \times (n+(m+s))}$$

be a linearization at infinity of grade g of $R(\lambda)$ with $\ell = \deg(\mathcal{L}(\lambda))$. Let $e_1 \leq \cdots \leq e_t$ be the partial multiplicities of $\operatorname{rev}_{\ell}(A_1\lambda + A_0)$ at 0, and let $\tilde{e}_1 \leq \cdots \leq \tilde{e}_u$ be the partial multiplicities of $\operatorname{rev} \mathcal{L}(\lambda)$ at 0. Then the invariant orders at infinity $q_1 \leq q_2 \leq \cdots \leq q_r$ of $R(\lambda)$ are

$$(q_1, q_2, \dots, q_r) = (-e_t, -e_{t-1}, \dots, -e_1, \underbrace{0, \dots, 0}_{r-t-u}, \widetilde{e}_1, \widetilde{e}_2, \dots, \widetilde{e}_u) - (g, g, \dots, g).$$

A linear polynomial system matrix that satisfies Definition 4.2.1 in \mathbb{F} and Definition 4.2.9, for a certain grade g, allows us to recover the complete information about the poles and zeros of the corresponding rational matrix, finite and at infinity. This is due to Theorem 4.2.6 and Proposition 4.2.15. This important case leads us to introduce the following definition.

Definition 4.2.16 (g-strong linearization). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and let g be an integer. A linear polynomial system matrix $\mathcal{L}(\lambda)$ is said to be a strong linearization of grade g, or a g-strong linearization, of $R(\lambda)$ if $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in \mathbb{F} and also at ∞ of grade g.

Example 4.2.17. Consider again the matrices in Example 4.1.4. Then the linear polynomial system matrix $P(\lambda)$ is a 1-strong linearization of $R(\lambda)$ if and only if all the matrices B_1, \ldots, B_s are nonsingular.

In Example 4.2.18 we consider a linear polynomial system matrix $\mathcal{L}(\lambda)$ that is a linearization of a rational matrix $R(\lambda)$ in $\mathbb{F} \cup \{\infty\}$. However, it is not a strong linearization in the sense of [7, Definition 3.4]. In particular, the grade of $\mathcal{L}(\lambda)$ as linearization at ∞ is not equal to the degree of the polynomial part of $R(\lambda)$. Actually, the grade is less than the degree of the polynomial part. Example 4.2.18. Let us consider the rational matrix

$$R(\lambda) = \left[\begin{array}{cc} \frac{\lambda^2 + \lambda - 1}{\lambda} & -\frac{1}{\lambda} \\ -1 & -\lambda^2 + \lambda - 2 \end{array}\right].$$

It can be easily proved that

$$\mathcal{L}(\lambda) = \begin{bmatrix} \lambda & 0 & 1 & 1 \\ 0 & 1 & 0 & \lambda \\ \hline 1 & 0 & \lambda + 1 & 0 \\ \lambda & \lambda & 0 & \lambda - 1 \end{bmatrix} := \begin{bmatrix} A_1\lambda + A_0 & B_1\lambda + B_0 \\ \hline -(C_1\lambda + C_0) & D_1\lambda + D_0 \end{bmatrix}$$

is a linear polynomial system matrix of $R(\lambda)$, with state matrix $A_1\lambda + A_0$. Moreover, note that $\mathcal{L}(\lambda)$ is minimal for all $\lambda_0 \in \mathbb{F}$. Therefore, by Remark 4.2.2(case 1), $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in \mathbb{F} . By Remark 4.2.11, $\mathcal{L}(\lambda)$ is also a linearization of $R(\lambda)$ at ∞ of grade 1 since

$$\operatorname{rank} \begin{bmatrix} A_1 \\ C_1 \end{bmatrix} = \operatorname{rank} \begin{bmatrix} A_1 & B_1 \end{bmatrix} = 2.$$

Thus, $\mathcal{L}(\lambda)$ is a 1-strong linearization of $R(\lambda)$, according to Definition 4.2.16. However, $\mathcal{L}(\lambda)$ is not a strong linearization according to [7, Definition 3.4] since A_1 is singular. Nevertheless, we can recover easily the invariant orders at ∞ from $\mathcal{L}(\lambda)$ by applying Proposition 4.2.15 with g = 1. For this purpose, note that rev $\mathcal{L}(\lambda)$ does not have elementary divisors at 0, since rev $\mathcal{L}(\lambda)$ is invertible at 0. Moreover, the only elementary divisor at 0 of $A_1 + A_0\lambda$ is λ . Therefore, the invariant orders at infinity of $R(\lambda)$ are -2 and -1 by Proposition 4.2.15. The invariant orders of $R(\lambda)$ at any finite point can be recovered from $\mathcal{L}(\lambda)$ by using Theorem 4.2.6. It is worthwhile to emphasize that the grade of $\mathcal{L}(\lambda)$ as linearization at ∞ of $R(\lambda)$ is different from the degree of the polynomial part of $R(\lambda)$.

4.3 Block full rank pencils

In this section, we introduce a wide family of pencils that are linearizations with empty state matrix of rational matrices. Thus, they give us information about zeros locally, i.e., in subsets of \mathbb{F} and/or at ∞ under some conditions. These pencils will be called block full rank pencils, since they generalize the block minimal bases pencils introduced in [26, Definition 3.1]. The definition of block full rank pencils is motivated by the fact that most of the linearizations for rational approximations of NLEPs that have been constructed so far are pencils of this type. The structure of block full rank pencils is extended in Section 4.4 to introduce a wide family of pencils that give us information not only about zeros but also about poles of rational matrices locally. They will be called block full rank linearizations.

The key results in this section are Theorems 4.3.5 and 4.3.7, which will be applied in the following section to establish rigorously and very easily properties of the linearizations used in [47] and [60]. Note that, according to Theorem 4.2.6, the results in this section are not useful for studying, or computing, the finite poles of rational matrices because the considered linearizations have empty state matrix. This may be a drawback in certain situations, but we emphasize again that it is not in the development of algorithms for solving large-scale NLEPs via rational approximations [46, 47, 60, 35]. This is due to the fact that, in those cases, the poles of the rational matrix are chosen for constructing the approximation and/or are located outside the corresponding target set.

Definition 4.3.1. (Block full rank pencil) A block full rank pencil is a linear polynomial matrix over \mathbb{F} with the following structure

$$L(\lambda) = \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix}$$
(4.10)

where $K_1(\lambda)$ and $K_2(\lambda)$ are pencils with full row normal rank.

Definition 4.3.1 includes the cases when $K_1(\lambda)$ or $K_2(\lambda)$ are empty matrices. That is, when $L(\lambda) = \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \end{bmatrix}$, $L(\lambda) = \begin{bmatrix} M(\lambda) \\ K_1(\lambda) \end{bmatrix}$ or $L(\lambda) = M(\lambda)$.

We introduce some auxiliary concepts and results before establishing the most important properties of block full rank pencils in Theorems 4.4.1 and 4.3.7. We will say that a rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ has full row rank in $\Sigma \subseteq \mathbb{F}$ if, for all $\lambda_0 \in \Sigma$, $R(\lambda_0) \in \mathbb{F}^{p \times m}$, i.e., $R(\lambda)$ is defined at λ_0 , and rank $R(\lambda_0) = p$. Observe that this implies that $R(\lambda)$ has no poles in Σ . The following lemma connects rational matrices with full row rank in a set Σ with minimal bases, and establishes other properties that will be used later.

Lemma 4.3.2. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix with full row normal rank and let $T(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$ be a minimal basis of the row space of $R(\lambda)$. Then the following statements hold:

- (a) There exists a unique regular rational matrix $S(\lambda) \in \mathbb{F}(\lambda)^{p \times p}$ such that $R(\lambda) = S(\lambda)T(\lambda)$.
- (b) $R(\lambda)$ has full row rank in $\Sigma \subseteq \mathbb{F}$ if and only if $S(\lambda)$ in (a) is invertible in Σ .
- (c) $R(\lambda)$ is a polynomial matrix if and only if $S(\lambda)$ in (a) is a polynomial matrix.
- (d) If $R(\lambda)$ is a matrix pencil, then $S(\lambda)$ in (a) and $T(\lambda)$ are both matrix pencils.

Proof. Part (a). Each row of $S(\lambda)$ is uniquely defined because its entries are the unique rational coefficients that allow us to express the corresponding row of $R(\lambda)$ as a unique linear combination of the rows of $T(\lambda)$. Moreover, $S(\lambda)$ must be regular since, otherwise, there would exist a nonzero vector $y(\lambda) \in \mathbb{F}(\lambda)^{p\times 1}$ such that $y(\lambda)^T S(\lambda) = 0$. So, $y(\lambda)^T R(\lambda) = 0$, which contradicts that rank $R(\lambda) = p$.

Part (b). It is obvious that if $S(\lambda)$ is invertible in Σ , then $R(\lambda)$ has full row rank in Σ , because $T(\lambda)$ is defined in Σ , as $T(\lambda)$ is a polynomial matrix, and $T(\lambda)$ has full row rank in Σ , since $T(\lambda)$ is a minimal basis. The proof of the converse implication starts by proving that if $R(\lambda)$ has full row rank in Σ , then $S(\lambda)$ is defined in Σ . To see this, note that the Smith form of $T(\lambda)$ is $[I_p \ 0]$, because $T(\lambda)$ is a minimal basis and, therefore, does not have finite zeros. Thus, there exist unimodular matrices $U(\lambda)$ and $V(\lambda)$ such that $T(\lambda) = U(\lambda) [I_p \ 0] V(\lambda)$, and $R(\lambda)V(\lambda)^{-1} = [S(\lambda)U(\lambda) \ 0]$. This shows that $C(\lambda) := S(\lambda)U(\lambda)$ is defined in Σ , because $R(\lambda)$ and $V(\lambda)^{-1}$ are both defined in $\Sigma \ (R(\lambda)$ by hypothesis and $V(\lambda)^{-1}$ because is unimodular and so a polynomial matrix). Therefore, $S(\lambda) = C(\lambda)U(\lambda)^{-1}$ is defined in Σ . This implies that we can write $R(\lambda_0) = S(\lambda_0)T(\lambda_0)$ for each $\lambda_0 \in \Sigma$, which in turns implies that $S(\lambda_0)$ is invertible because $R(\lambda_0)$ has full row rank.

Part (c). It follows directly from [38, Main Theorem, part 4].

Part (d). From [38, Main Theorem, part 4], we have that

$$\deg(\operatorname{row}_{i}(R(\lambda))) = \max_{1 \le j \le p} \left(\deg(s_{ij}(\lambda)) + \deg(\operatorname{row}_{j}(T(\lambda))) \right) \le 1, \quad \text{for } 1 \le i \le p,$$
(4.11)

where $\operatorname{row}_i(R(\lambda))$ denotes the *i*th row of $R(\lambda)$ and the maximum is taken over the nonzero entries $s_{ij}(\lambda)$ of $S(\lambda)$. Since all the rows of $T(\lambda)$ are different from zero, (4.11) implies that $\deg(s_{ij}(\lambda)) \leq 1$ for each nonzero entry of $S(\lambda)$. Moreover, each column of $S(\lambda)$ has at least one nonzero entry, because $S(\lambda)$ is regular, which, combined with (4.11), implies that $\deg(\operatorname{row}_i(T(\lambda))) \leq 1$, for each $j = 1, \ldots, p$.

The last concepts we need before stating and proving the main Theorem 4.3.5 are those of rational basis and dual rational bases.

Definition 4.3.3 (Rational basis). A rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ (with p < m) is said to be a rational basis if it is a basis of the rational subspace spanned by its rows, i.e., if it has full row normal rank.

Definition 4.3.4 (Dual rational bases). Two rational bases $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ and $H(\lambda) \in \mathbb{F}(\lambda)^{q \times m}$ are said to be dual if p + q = m and $R(\lambda) H(\lambda)^T = 0$.

Theorem 4.3.5. Let $\Omega \subseteq \mathbb{F}$ be nonempty. Consider a block full rank pencil

$$L(\lambda) = \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix},$$

as in (4.10), and let $N_1(\lambda)$ and $N_2(\lambda)$ be any rational bases dual to $K_1(\lambda)$ and $K_2(\lambda)$, respectively. If $K_i(\lambda)$ and $N_i(\lambda)$ have full row rank in Ω , for i = 1, 2, then $L(\lambda)$ is a linearization of the rational matrix

$$R(\lambda) = N_2(\lambda)M(\lambda)N_1(\lambda)^T$$

in Ω with empty state matrix.

Proof. In order to simplify the notation, throughout this proof we do not specify the sizes of different identity matrices and all of them are denoted by I. Let $\widetilde{K}_1(\lambda), \widetilde{K}_2(\lambda), \widetilde{N}_1(\lambda)$ and $\widetilde{N}_2(\lambda)$ be minimal bases of the row spaces of $K_1(\lambda), K_2(\lambda)$, $N_1(\lambda)$ and $N_2(\lambda)$, respectively. Then, Lemma 4.3.2 implies that there exist regular rational matrices $S_1(\lambda), S_2(\lambda), W_1(\lambda)$ and $W_2(\lambda)$ such that

$$K_i(\lambda) = S_i(\lambda)K_i(\lambda)$$
, and $S_i(\lambda)$ is invertible in Ω , for $i = 1, 2$.
 $N_i(\lambda) = W_i(\lambda)\widetilde{N}_i(\lambda)$, and $W_i(\lambda)$ is invertible in Ω , for $i = 1, 2$.

Moreover, $\widetilde{K}_1(\lambda)$, $\widetilde{K}_2(\lambda)$, $S_1(\lambda)$ and $S_2(\lambda)$ are all matrix pencils. Then, $L(\lambda)$ can be factorized as follows,

$$L(\lambda) = \begin{bmatrix} I & 0\\ 0 & S_1(\lambda) \end{bmatrix} \begin{bmatrix} M(\lambda) & \widetilde{K}_2(\lambda)^T\\ \widetilde{K}_1(\lambda) & 0 \end{bmatrix} \begin{bmatrix} I & 0\\ 0 & S_2(\lambda)^T \end{bmatrix}, \quad (4.12)$$

where the first and third factors are invertible in Ω . Note that the factor in the middle is a block minimal bases pencil (see [26, Definition 3.1]) associated with the polynomial matrix $\tilde{N}_2(\lambda)M(\lambda)\tilde{N}_1(\lambda)^T$, since the regularity of $S_i(\lambda)$ and $W_i(\lambda)$ implies that $\tilde{K}_i(\lambda)$ and $\tilde{N}_i(\lambda)$ are dual minimal bases for i = 1, 2. Then, there exist unimodular matrices $U(\lambda)$ and $V(\lambda)$ such that

$$\begin{bmatrix} M(\lambda) & \widetilde{K}_{2}(\lambda)^{T} \\ \widetilde{K}_{1}(\lambda) & 0 \end{bmatrix} = U(\lambda) \begin{bmatrix} \widetilde{N}_{2}(\lambda)M(\lambda)\widetilde{N}_{1}(\lambda)^{T} & 0 \\ 0 & I \end{bmatrix} V(\lambda)$$
$$= U(\lambda) \begin{bmatrix} W_{2}(\lambda)^{-1} & 0 \\ 0 & I \end{bmatrix} \begin{bmatrix} R(\lambda) & 0 \\ 0 & I \end{bmatrix} \begin{bmatrix} W_{1}(\lambda)^{-T} & 0 \\ 0 & I \end{bmatrix} V(\lambda),$$
(4.13)

where $U(\lambda) \operatorname{diag}(W_2(\lambda)^{-1}, I)$ and $\operatorname{diag}(W_1(\lambda)^{-T}, I)V(\lambda)$ are invertible in Ω . From combining (4.12) and (4.13), we obtain that $L(\lambda) \sim_{\Omega} \operatorname{diag}(R(\lambda), I)$.

Remark 4.3.6. Under the conditions of Theorem 4.3.5, we will say for brevity that " $L(\lambda)$ is a block full rank pencil associated with $R(\lambda)$ in Ω ". We emphasize that this "association" is not one-to-one because there are infinitely many rational bases $N_i(\lambda)$ dual to $K_i(\lambda)$. If $K_1(\lambda)$ (resp. $K_2(\lambda)$) is an empty matrix, we can take any rational matrix $N_1(\lambda) \in \mathbb{F}(\lambda)^{s_1 \times s_1}$ (resp. $N_2(\lambda) \in \mathbb{F}(\lambda)^{s_2 \times s_2}$) invertible in Ω , where s_1 (resp. s_2) is the number of columns (resp. rows) of $M(\lambda)$. The standard choices are $N_1(\lambda) = I_{s_1}$ and $N_2(\lambda) = I_{s_2}$.

In the scenario of Theorem 4.3.5, Theorem 4.2.6 guarantees that the elementary divisors of $L(\lambda)$ in Ω coincide exactly with the zero elementary divisors of $R(\lambda)$ in Ω . Moreover, it is clear from the expression $R(\lambda) = N_2(\lambda)M(\lambda)N_1(\lambda)^T$ that $R(\lambda)$ does not have poles in Ω , since the matrices $N_i(\lambda)$ must be defined in Ω but they are not defined at the poles of $R(\lambda)$. Thus, $R(\lambda)$ has only eigenvalues in Ω , and all the information about them, i.e., geometric, algebraic and partial multiplicities, is contained in $L(\lambda)$.

Next, we present sufficient conditions for a block full rank pencil to be a linearization of $R(\lambda) = N_2(\lambda)M(\lambda)N_1(\lambda)^T$ at ∞ of a certain grade g. In order to avoid cases with limited interest in applications, in Theorem 4.3.7 we assume deg $(L(\lambda)) = 1$.

Theorem 4.3.7. Consider a block full rank pencil

$$L(\lambda) = \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix},$$

as in (4.10), with deg($L(\lambda)$) = 1, and let $N_1(\lambda)$ and $N_2(\lambda)$ be rational bases dual to $K_1(\lambda)$ and $K_2(\lambda)$, respectively. If, for i = 1, 2, rev₁ $K_i(\lambda)$ has full row rank at 0, and there exists an integer number t_i such that rev_{t_i} $N_i(\lambda)$ has full row rank at 0, then $L(\lambda)$ is a linearization of the rational matrix

$$R(\lambda) = N_2(\lambda)M(\lambda)N_1(\lambda)^T$$

at ∞ of grade $1 + t_1 + t_2$ with empty state matrix.

Proof. Note that rev $L(\lambda) = \begin{bmatrix} \operatorname{rev}_1 M(\lambda) & \operatorname{rev}_1 K_2(\lambda)^T \\ \operatorname{rev}_1 K_1(\lambda) & 0 \end{bmatrix}$ is a block full rank pencil. Moreover, for i = 1, 2, $\operatorname{rev}_{t_i} N_i(\lambda)$ has full row normal rank, and $K_i(\lambda) N_i(\lambda)^T = 0$ implies $(\operatorname{rev}_1 K_i(\lambda)) (\operatorname{rev}_{t_i} N_i(\lambda))^T = 0$. Therefore, $\operatorname{rev}_{t_i} N_i(\lambda)$ is a rational basis dual to $\operatorname{rev}_1 K_i(\lambda)$. Then, Theorem 4.3.5 applied to $\operatorname{rev} L(\lambda)$ proves that $\operatorname{rev} L(\lambda)$ is a linearization at 0 of $(\operatorname{rev}_{t_2} N_2(\lambda)) (\operatorname{rev}_1 M(\lambda)) (\operatorname{rev}_{t_1} N_1(\lambda)^T) = \operatorname{rev}_{1+t_1+t_2} R(\lambda)$, with empty state matrix, which combined with Proposition 4.2.10 proves the result.

As a consequence of Theorems 4.3.5 and 4.3.7, we obtain Corollary 4.3.8. It generalizes the structure of most of the linearizations of rational approximations of NLEPs that appear in the literature in a constructive way. Moreover, it is very useful in order to characterize easily some pencils as linearizations of rational matrices when only the information about the zeros in subsets not containing poles is of interest.

Corollary 4.3.8. Let

$$R(\lambda) = (A_0 - \lambda B_0)R_0(\lambda) + (A_1 - \lambda B_1)R_1(\lambda) + \dots + (A_N - \lambda B_N)R_N(\lambda)$$

be a $p \times m$ rational matrix written in terms of some matrix pencils $A_i - \lambda B_i \in \mathbb{F}[\lambda]^{p \times n_i}$ and rational matrices $R_i(\lambda) \in \mathbb{F}(\lambda)^{n_i \times m}$. Define

$$M(\lambda) := [(A_0 - \lambda B_0) \quad (A_1 - \lambda B_1) \quad \cdots \quad (A_N - \lambda B_N)] \text{ and}$$

$$N_1(\lambda) := \begin{bmatrix} R_0(\lambda)^T & R_1(\lambda)^T & \cdots & R_N(\lambda)^T \end{bmatrix},$$

and assume that $N_1(\lambda)$ has full row normal rank. Let $L(\lambda) = \begin{bmatrix} M(\lambda) \\ K_1(\lambda) \end{bmatrix}$ be a block full rank pencil of degree 1 with only one block column and such that $K_1(\lambda)$ and $N_1(\lambda)$ are dual rational bases. Let $\Omega \subseteq \mathbb{F}$ be nonempty. Then the following statements hold:

- (a) If $K_1(\lambda)$ and $N_1(\lambda)$ have full row rank in Ω then $L(\lambda)$ is a linearization of $R(\lambda)$ in Ω with empty state matrix.
- (b) If rev₁ K₁(λ) has full row rank at 0, and there exists an integer t such that rev_t N₁(λ) has full row rank at 0, then L(λ) is a linearization of R(λ) at ∞ of grade 1 + t with empty state matrix.

Remark 4.3.9. We emphasize that in some relevant applications the rational matrices $R_i(\lambda)$ of Corollary 4.3.8 are just of the form $R_i(\lambda) = r_i(\lambda)I_m$, where $r_i(\lambda)$ are scalar rational functions, and/or most of the pencils $A_i - \lambda B_i$ are constant matrices or a linear scalar function times a constant matrix. Moreover, in some other applications a low rank structure is present in $R(\lambda)$, that is, some of the terms in $R(\lambda)$ have a rank much smaller than min $\{p, m\}$, and the corresponding rational matrices are written in the form $R_i(\lambda) = r_i(\lambda)R_i$, where $R_i \in \mathbb{F}^{n_i \times m}$ is a constant matrix with $n_i \ll m$.

In the next example, we revisit the pencil introduced in Example 4.1.4 from the perspective of the block full rank pencils. This example illustrates how the theory of block full rank pencils may simplify the analysis of the properties of important linearizations of rational matrices when one is not interested in the information about the poles.

Example 4.3.10. Consider Example 4.1.4. We partition $P(\lambda)$ as follows:

$$P(\lambda) = \begin{bmatrix} (\lambda - \sigma_1)I & I \\ & \ddots & \vdots \\ & & (\lambda - \sigma_s)I & I \\ \hline & & -B_1 & \cdots & -B_s & \lambda A_0 - B_0 \end{bmatrix} =: \begin{bmatrix} K_1(\lambda) \\ M(\lambda) \end{bmatrix}.$$

Observe that, in the above partition, we are considering a permuted version of the structure of the pencil $L(\lambda)$ in Corollary 4.3.8. Note now that $K_1(\lambda)$ has full row rank in \mathbb{C} , and

$$N_1(\lambda) := \begin{bmatrix} \frac{1}{\sigma_1 - \lambda} I & \dots & \frac{1}{\sigma_s - \lambda} I & I \end{bmatrix}$$

is a rational basis dual to $K_1(\lambda)$ with full row rank in $\Sigma := \mathbb{C} \setminus \{\sigma_1, \ldots, \sigma_s\}$. Then, by Corollary 4.3.8(a), $P(\lambda)$ is a linearization of $R(\lambda)$ in Σ with empty state matrix. Moreover, note that rev₁ $K_1(\lambda)$ and rev₀ $N_1(\lambda) = \begin{bmatrix} \lambda & \lambda \\ \lambda \sigma_1 - 1 & 1 & \lambda \\ \lambda \sigma_s - 1 & I \end{bmatrix}$ both have full row rank at 0. Thus, by Corollary 4.3.8(b), $P(\lambda)$ is a linearization of $R(\lambda)$ at ∞ of grade 1 with empty state matrix.

4.4 Block full rank linearizations of rational matrices

Block full rank pencils in the above section are local linearizations of rational matrices that contain information about their zeros. In this section, we introduce in Theorems 4.4.1 and 4.4.10 a wide family of pencils that give us information about both zeros and poles of rational matrices locally, at finite points and/or at infinity. They will be called block full rank linearizations, where we use a name similar to that of block full rank pencils for emphasizing the connection between both concepts.

4.4.1 Block full rank linearizations at finite points

In Theorem 4.4.1, we generalize Theorem 4.3.5 (or [28, Theorem 5.3]) in order to obtain local linearizations that give us not only information about the zeros but also about the poles of the corresponding rational matrix.

Theorem 4.4.1. Consider a nonconstant linear polynomial system matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) & 0\\ -C(\lambda) & M(\lambda) & K_2(\lambda)^T\\ 0 & K_1(\lambda) & 0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q)\times(n+r)}$$
(4.14)

with n > 0 and state matrix $A(\lambda)$. Let $L(\lambda) := \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix}$ be a block full rank pencil, and let $N_1(\lambda)$ and $N_2(\lambda)$ be any rational bases dual to $K_1(\lambda)$ and $K_2(\lambda)$, respectively. Let Ω be a nonempty subset of \mathbb{F} such that $K_i(\lambda)$ and $N_i(\lambda)$ have full row rank in Ω for i = 1, 2. If

$$\operatorname{rank} \begin{bmatrix} A(\lambda_0) \\ -N_2(\lambda_0)C(\lambda_0) \end{bmatrix} = \operatorname{rank} \begin{bmatrix} A(\lambda_0) & B(\lambda_0)N_1(\lambda_0)^T \end{bmatrix} = n$$
(4.15)

for all $\lambda_0 \in \Omega$ then $\mathcal{L}(\lambda)$ is a linearization of the rational matrix

$$R(\lambda) = N_2(\lambda)[M(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)]N_1(\lambda)^T$$
(4.16)

in Ω with state matrix $A(\lambda)$.

A pencil of the form (4.14) satisfying the hypotheses in Theorem 4.4.1 is called a block full rank linearization. In particular, $\mathcal{L}(\lambda)$ is said to be a block full rank linearization of $R(\lambda)$ in Ω with state matrix $A(\lambda)$.

Remark 4.4.2. The extreme case of n = 0 in the linear polynomial system matrix (4.14) was studied in Theorem 4.3.5 ([28, Theorem 5.3]). It states that the block full rank pencil $L(\lambda)$ in Theorem 4.4.1 is a linearization of the rational matrix

$$G(\lambda) = N_2(\lambda)M(\lambda)N_1(\lambda)^T$$

in Ω with empty state matrix. In this case, $L(\lambda)$ is said to be a block full rank linearization of $G(\lambda)$ in Ω with empty state matrix.

Proof of Theorem 4.4.1. In order to simplify the notation, throughout this proof we do not specify the sizes of different identity matrices and all of them are denoted by I. Let $\tilde{K}_1(\lambda), \tilde{K}_2(\lambda), \tilde{N}_1(\lambda)$ and $\tilde{N}_2(\lambda)$ be minimal bases of the row spaces of $K_1(\lambda)$, $K_2(\lambda), N_1(\lambda)$ and $N_2(\lambda)$, respectively. Then, by Lemma 4.3.2, there exist rational matrices $S_1(\lambda), S_2(\lambda), W_1(\lambda)$ and $W_2(\lambda)$ such that

$$K_i(\lambda) = S_i(\lambda)\widetilde{K}_i(\lambda)$$
, and $S_i(\lambda)$ is invertible in Ω , for $i = 1, 2$,
 $N_i(\lambda) = W_i(\lambda)\widetilde{N}_i(\lambda)$, and $W_i(\lambda)$ is invertible in Ω , for $i = 1, 2$.

Moreover, $\widetilde{K}_1(\lambda), \widetilde{K}_2(\lambda), S_1(\lambda)$ and $S_2(\lambda)$ are all matrix pencils. We consider the linear polynomial system matrix

$$\widetilde{\mathcal{L}}(\lambda) := \begin{bmatrix} A(\lambda) & B(\lambda) & 0\\ -C(\lambda) & M(\lambda) & \widetilde{K}_2(\lambda)^T\\ 0 & \widetilde{K}_1(\lambda) & 0 \end{bmatrix},$$
(4.17)

which is equivalent in Ω to $\mathcal{L}(\lambda)$, since $\begin{bmatrix} I & 0 \\ 0 & S_1(\lambda) \end{bmatrix} \widetilde{\mathcal{L}}(\lambda) \begin{bmatrix} I & 0 \\ 0 & S_2(\lambda)^T \end{bmatrix} = \mathcal{L}(\lambda)$. For i = 1, 2, there exist unimodular matrices

$$U_i(\lambda) = \begin{bmatrix} \widetilde{K}_i(\lambda) \\ \widehat{K}_i(\lambda) \end{bmatrix}$$
, and $U_i(\lambda)^{-1} = \begin{bmatrix} \widehat{N}_i(\lambda)^T & \widetilde{N}_i(\lambda)^T \end{bmatrix}$

as in [26, Theorem 2.10]. Consider now the unimodular matrices

$$V_{1}(\lambda) := \begin{bmatrix} I & 0 & 0 & 0 \\ 0 & \widehat{N}_{1}(\lambda)^{T} & \widetilde{N}_{1}(\lambda)^{T} & 0 \\ 0 & 0 & 0 & I \end{bmatrix} \begin{bmatrix} I & 0 & 0 & 0 \\ 0 & 0 & I & 0 \\ 0 & -X(\lambda) & 0 & I \end{bmatrix}, \text{ and}$$
$$V_{2}(\lambda) := \begin{bmatrix} I & 0 & 0 & 0 \\ 0 & 0 & I & -Y(\lambda) \\ 0 & 0 & 0 & I \\ 0 & I & 0 & -Z(\lambda) \end{bmatrix} \begin{bmatrix} I & 0 & 0 \\ 0 & \widehat{N}_{2}(\lambda) & 0 \\ 0 & 0 & I \end{bmatrix},$$

where

$$Z(\lambda) := \widehat{N}_2(\lambda) M(\lambda) \widehat{N}_1(\lambda)^T, X(\lambda) := \widehat{N}_2(\lambda) M(\lambda) \widetilde{N}_1(\lambda)^T, Y(\lambda) := \widetilde{N}_2(\lambda) M(\lambda) \widehat{N}_1(\lambda)^T.$$

We obtain that
$$\begin{bmatrix} & & & \\ & & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & &$$

$$V_2(\lambda)\widetilde{\mathcal{L}}(\lambda)V_1(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda)N_1(\lambda)^T & B(\lambda)N_1(\lambda)^T & 0\\ -\widetilde{N}_2(\lambda)C(\lambda) & \widetilde{N}_2(\lambda)M(\lambda)\widetilde{N}_1(\lambda)^T & 0 & 0\\ 0 & 0 & I & 0\\ -\widehat{N}_2(\lambda)C(\lambda) & 0 & 0 & I \end{bmatrix},$$

which is, in addition, unimodularly equivalent to

$$\begin{bmatrix} A(\lambda) & B(\lambda)\widetilde{N}_1(\lambda)^T & 0\\ -\widetilde{N}_2(\lambda)C(\lambda) & \widetilde{N}_2(\lambda)M(\lambda)\widetilde{N}_1(\lambda)^T & 0\\ 0 & 0 & I \end{bmatrix} := P(\lambda).$$

By condition (4.15), the polynomial matrix

$$H(\lambda) := \begin{bmatrix} A(\lambda) & B(\lambda)\widetilde{N}_1(\lambda)^T \\ -\widetilde{N}_2(\lambda)C(\lambda) & \widetilde{N}_2(\lambda)M(\lambda)\widetilde{N}_1(\lambda)^T \end{bmatrix}$$

is minimal in Ω , and its transfer function matrix is

$$\widetilde{R}(\lambda) = \widetilde{N}_2(\lambda)[M(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)]\widetilde{N}_1(\lambda)^T.$$

Moreover, $W_2(\lambda)\widetilde{R}(\lambda)W_1(\lambda)^T = R(\lambda)$, and, thus, $\widetilde{R}(\lambda)$ and $R(\lambda)$ are equivalent in Ω . Therefore, the zero elementary divisors of $H(\lambda)$ in Ω are the zero elementary divisors of $R(\lambda)$ in Ω , and the zero elementary divisors of $A(\lambda)$ in Ω are the pole elementary divisors of $R(\lambda)$ in Ω . In addition, $P(\lambda) = \begin{bmatrix} H(\lambda) & 0 \\ 0 & I \end{bmatrix}$ is unimodularly equivalent to $\widetilde{\mathcal{L}}(\lambda)$, which is equivalent in Ω to $\mathcal{L}(\lambda)$. Therefore, the zero elementary divisors of $\mathcal{L}(\lambda)$ in Ω are the zero elementary divisors of $R(\lambda)$ in Ω are the zero elementary divisors of $R(\lambda)$ in Ω . By Theorem 4.2.6, $\mathcal{L}(\lambda)$ is a linearization in Ω of $R(\lambda)$, since it is immediate to check that the rank condition in Theorem 4.2.6(a) is satisfied.

Remark 4.4.3. The linear polynomial system matrix in (4.14) generalizes the structure of the strong block minimal bases linearizations of rational matrices presented in [7, Theorem 5.11] (recall Theorem 2.5.5) from three perspectives: general pencils $B(\lambda)$ and $C(\lambda)$ are allowed, while those in [7] have a very particular structure; $A(\lambda)$ can be any regular pencil, while in [7] its coefficient in λ must be invertible; and $\begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix}$ is an arbitrary block full rank pencil (4.10), while in [7] strong block minimal bases pencils (2.12) are considered. We give more details in Subsection 4.4.3.

Remark 4.4.4. Notice that $\mathcal{L}(\lambda)$ being minimal in Ω is a necessary condition, but not sufficient, in order the rank condition (4.15) to be satisfied.

Under the conditions of Theorem 4.4.1, Theorem 4.2.6 guarantees that the elementary divisors of $\mathcal{L}(\lambda)$ in Ω are the zero elementary divisors of $R(\lambda)$ in Ω , and that the elementary divisors of $A(\lambda)$ in Ω are the pole elementary divisors of $R(\lambda)$ in Ω . We would like to emphasize the fact that Theorem 4.4.1 gives very simple conditions to determine if a pencil is a linearization (in a target set) and, in addition, to determine the rational matrices associated with it by considering rational bases dual to $K_1(\lambda)$ and $K_2(\lambda)$. Notice that this "association" is not one-to-one because there are infinitely many rational bases $N_1(\lambda)$ and $N_2(\lambda)$ dual to $K_1(\lambda)$ and $K_2(\lambda)$. Thus, Theorem 4.4.1 is important, for instance, if a linearization gets perturbed and one wants to know which rational matrix is associated with the new perturbed pencil. Then, if the perturbation is small enough that the rank conditions in Theorem 4.4.1 are still satisfied and after restoring the zero blocks of the block full rank linearization, we will get a perturbed block full rank linearization and one can then reconstruct the new rational matrix associated with it.

Remark 4.4.5. If in Theorem 4.4.1, $K_1(\lambda)$ (resp. $K_2(\lambda)$) is an empty matrix, we can take the dual rational basis $N_1(\lambda)$ (resp. $N_2(\lambda)$) as any rational matrix invertible in Ω of size the number of columns (resp. rows) of $M(\lambda)$.

Example 4.4.6. Let us see a simple example that illustrates Remark 4.4.5. For instance, for constructing a linearization of the rational matrix

$$R(\lambda) = \frac{\lambda - 2}{\lambda + 2} \begin{bmatrix} -\lambda + 3 & 1\\ \overline{\lambda^2 - 1} & \overline{\lambda(\lambda - 1)} \end{bmatrix} \text{ in the set } \Omega := \mathbb{F} - \{-1, 0, 1\},$$

we can consider $K_1(\lambda)$ and $K_2(\lambda)$ as empty matrices and dual rational bases $N_1(\lambda)^T :=$ diag $\left(\frac{1}{\lambda^2-1}, \frac{1}{(\lambda-1)\lambda}\right)$ and $N_2(\lambda) := 1$, both invertible in Ω . Then, by Theorem 4.4.1, the following linear polynomial system matrix

$$\mathcal{L}(\lambda) := \begin{bmatrix} \lambda + 2 & | -\lambda + 3 & 1 \\ \hline -\lambda + 2 & | & 0 & 0 \end{bmatrix} := \begin{bmatrix} A(\lambda) & | & B(\lambda) \\ \hline -C(\lambda) & | & M(\lambda) \end{bmatrix}$$

is a linearization of $R(\lambda)$ in Ω with state matrix $\lambda + 2$, since

$$\operatorname{rank} \begin{bmatrix} \lambda+2\\ -\lambda+2 \end{bmatrix} = \operatorname{rank} \begin{bmatrix} \lambda+2 & \frac{-\lambda+3}{\lambda^2-1} & \frac{1}{\lambda(\lambda-1)} \end{bmatrix} = 1$$

for all $\lambda \in \Omega$. Therefore, we can recover from $\mathcal{L}(\lambda)$ the pole and zero structure of $R(\lambda)$ in Ω . More precisely, -2 is the only zero of the state matrix in Ω and, thus, is the only pole of $R(\lambda)$ in Ω . Moreover, 2 is the unique zero of $\mathcal{L}(\lambda)$ in Ω and, thus, is the unique zero of $R(\lambda)$ in Ω .

Remark 4.4.7. We notice that, although the state matrix $A(\lambda)$ appears in the (1,1) block in (4.14), in practice, it can be any regular submatrix of $\mathcal{L}(\lambda)$. In particular, in some applications [47, 60] we have found pencils with the structure of block full rank linearizations of the form

$$\mathcal{L}(\lambda) = \begin{bmatrix} M(\lambda) & K_2(\lambda)^T & -C(\lambda) \\ K_1(\lambda) & 0 & 0 \\ \hline B(\lambda) & 0 & A(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(q+n)\times(r+n)}.$$
(4.18)

4.4.2 Block full rank linearizations at infinity

We now study the counterpart of Theorem 4.4.1 at infinity. First, we define the notion of degree of a rational matrix. For the scalar case, we define the degree of a rational function $r(\lambda) = \frac{n(\lambda)}{d(\lambda)}$ as

$$\deg(r(\lambda)) := \deg(n(\lambda)) - \deg(d(\lambda)).$$
(4.19)

Then, for rational matrices we consider the following definition (see, for instance, [90, p.10]).

Definition 4.4.8. Let $R(\lambda) = [r_{ij}(\lambda)] \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix with entries $r_{ij}(\lambda) = \frac{n_{ij}(\lambda)}{d_{ij}(\lambda)}$. The degree of $R(\lambda)$ is then defined as $\deg(R(\lambda)) := \max_{\substack{i=1,\dots,p\\j=1,\dots,m}} \{\deg(r_{ij}(\lambda))\}.$

Notice that this notion of degree of a rational matrix generalizes the notion of degree of a polynomial matrix. In what follows, we call the degrees of each row of $R(\lambda)$, the row degrees of $R(\lambda)$.

Lemma 4.4.9. Let $R(\lambda)$ be a rational matrix. Then there exists an integer t such that all the rows of rev_t $R(\lambda)$ are defined at 0 and are all different from zero at 0 if and only if all the row degrees of $R(\lambda)$ are equal to t.

Proof. First, we consider a rational function $r(\lambda) = \frac{a(\lambda)}{b(\lambda)}$ such that the numerator $a(\lambda)$ has degree n, and that the denominator $b(\lambda)$ has degree m. We assume that there exists an integer t for which $\operatorname{rev}_t r(0)$ is defined and is different from 0. We can write

$$\operatorname{rev}_{t} r(\lambda) = \lambda^{t+m-n} \frac{\operatorname{rev}_{n} a(\lambda)}{\operatorname{rev}_{m} b(\lambda)} := \lambda^{t+m-n} h(\lambda), \qquad (4.20)$$

where 0 is not a pole nor a zero of $h(\lambda)$ since $\operatorname{rev}_n a(0) \neq 0$ and $\operatorname{rev}_m b(0) \neq 0$. That is, $h(\lambda)$ is defined and is different from 0 at 0. Therefore, t + m - n = 0, so that $\operatorname{rev}_t r(\lambda)$ is also defined and is different from 0 at 0. Then, $t = n - m = \deg(r(\lambda))$. Now, we assume that we have a rational row vector $v(\lambda) = [r_1(\lambda) \cdots r_s(\lambda)]$ such that, for some integer t, $\operatorname{rev}_t v(0)$ is defined and is different from 0. Then, it must be $t = \max_{i=1,\dots,s} \{\deg(r_i(\lambda))\}$. That is, $t = \deg(v(\lambda))$. Finally, consider a rational matrix $R(\lambda) = [v_1(\lambda)^T \cdots v_q(\lambda)^T]^T$ where $v_i(\lambda)$ are rational row vectors. Assume that there exists an integer t such that all the rows of $\operatorname{rev}_t R(\lambda)$ are defined at 0 and are different from zero at 0. Then, for each row $v_i(\lambda)$, it must hold that $t = \deg(v_i(\lambda))$.

The converse is trivial taking into account equation (4.20) for each entry of $R(\lambda)$.

Theorem 4.4.10. Consider a nonconstant linear polynomial system matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) & 0\\ -C(\lambda) & M(\lambda) & K_2(\lambda)^T\\ 0 & K_1(\lambda) & 0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q)\times(n+r)}$$
(4.21)

with n > 0 and state matrix $A(\lambda)$. Let $L(\lambda) := \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix}$ be a block full rank pencil and let $N_1(\lambda)$ (resp. $N_2(\lambda)$) be any rational basis dual to $K_1(\lambda)$ (resp. $K_2(\lambda)$) with its row degrees all equal to an integer t_1 (resp. t_2). If rev₁ $K_i(\lambda)$ and rev_{ti} $N_i(\lambda)$ have full row rank at zero for i = 1, 2 and

$$\operatorname{rank} \begin{bmatrix} \operatorname{rev}_1 A(0) \\ -\operatorname{rev}_{t_2} N_2(0) \operatorname{rev}_1 C(0) \end{bmatrix} = \operatorname{rank} \left[\operatorname{rev}_1 A(0) \quad \operatorname{rev}_1 B(0) \operatorname{rev}_{t_1} N_1(0)^T \right] = n$$
(4.22)

then $\mathcal{L}(\lambda)$ is a linearization of the rational matrix

$$R(\lambda) = N_2(\lambda)[M(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)]N_1(\lambda)^T$$

at ∞ of grade $1 + t_1 + t_2$ with state matrix $A(\lambda)$.

A pencil of the form (4.21) satisfying the hypotheses in Theorem 4.4.10 is called block full rank linearization at infinity. In particular, $\mathcal{L}(\lambda)$ is said to be a block full rank linearization of $R(\lambda)$ at ∞ of grade $1 + t_1 + t_2$ with state matrix $A(\lambda)$. In general, a block full rank linearization is said to be strong if it is a linearization in $\mathbb{F} \cup \{\infty\}$.

Remark 4.4.11. The extreme case of n = 0 in the linear polynomial system matrix (4.21) was studied in Theorem 4.3.7 ([28, Theorem 5.5]). It states that the block full rank pencil $L(\lambda)$ in Theorem 4.4.10 is a linearization of the rational matrix

$$G(\lambda) = N_2(\lambda)M(\lambda)N_1(\lambda)^T$$

at ∞ of grade $1 + t_1 + t_2$ with empty state matrix. In this case, $L(\lambda)$ is said to be a block full rank linearization of $G(\lambda)$ at ∞ of grade $1 + t_1 + t_2$ with empty state matrix. We notice that, by Lemma 4.4.9, the integers t_1 and t_2 appearing in Theorem 4.3.7 are the row degrees of the dual bases $N_1(\lambda)$ and $N_2(\lambda)$, respectively.

Proof of Theorem 4.4.10. The result follows by applying Theorem 4.4.1 to rev₁ $\mathcal{L}(\lambda)$ at 0. Let $g := 1 + t_1 + t_2$, then rev₁ $\mathcal{L}(\lambda)$ is a linearization at 0 of

$$\operatorname{rev}_{t_2} N_2(\lambda) [\operatorname{rev}_1 M(\lambda) + \operatorname{rev}_1 C(\lambda) (\operatorname{rev}_1 A(\lambda))^{-1} \operatorname{rev}_1 B(\lambda)] \operatorname{rev}_{t_1} N_1(\lambda)^T = \operatorname{rev}_g R(\lambda).$$

Example 4.4.12. We now consider the rational matrix

$$R(\lambda) := \sum_{k=0}^{2} A_k \frac{\lambda^k}{(\lambda - \epsilon)^2} + I_n \frac{1}{\lambda} \in \mathbb{F}(\lambda)^{n \times n} \text{ and the set } \Omega := \mathbb{F} - \{\epsilon\},$$

for some $\epsilon \in \mathbb{F}$. Then we define the following linear polynomial system matrix

$$\mathcal{L}(\lambda) := \begin{bmatrix} -\lambda I_n & 0 & (\lambda - \epsilon) I_n & 0\\ 0 & A_2 & 0 & -I_n\\ (\lambda - \epsilon) I_n & 0 & \lambda A_1 + A_0 & \lambda I_n\\ \hline 0 & -I_n & \lambda I_n & 0 \end{bmatrix} =: \begin{bmatrix} A(\lambda) & B(\lambda) & 0\\ \hline -C(\lambda) & M(\lambda) & K_2(\lambda)^T\\ \hline 0 & K_1(\lambda) & 0 \end{bmatrix},$$

with state matrix $A(\lambda) = \lambda I_n$. We consider the dual rational bases $N_1(\lambda) := N_2(\lambda) := \left[\frac{\lambda I_n}{\lambda - \epsilon} \quad \frac{I_n}{\lambda - \epsilon}\right]$, which have row degrees $t_1 = t_2 = 0$. Then $R(\lambda)$ can be written as $R(\lambda) = N_2(\lambda)[M(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)]N_1(\lambda)^T$. Notice that $\operatorname{rev}_{t_i} N_i(\lambda)$ and $\operatorname{rev}_1 K_i(\lambda)$ have both full row rank at 0, for i = 1, 2, and that condition (4.22) is satisfied since $\operatorname{rev}_1 A(0) = I_n$. Thus, by Theorem 4.4.10, $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ at ∞ of grade $1 + t_1 + t_2 = 1$ with state matrix $A(\lambda)$. In addition, by Theorem 4.4.1, $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Ω , since $N_i(\lambda)$ and $K_i(\lambda)$ have both full row rank in Ω , for i = 1, 2, and condition (4.15) is satisfied in Ω . Observe that, if $R(\lambda)$ has symmetric coefficients, $\mathcal{L}(\lambda)$ preserves the symmetry.

Remark 4.4.13. If we want a linearization as in (4.21) to be a linearization at all finite and infinite points we need, besides minimality conditions, the matrices $K_1(\lambda)$ and $K_2(\lambda)$ being minimal bases with all their row degrees equal to 1. Notice that if a pencil $K(\lambda)$ has full row rank in \mathbb{F} and, in addition, rev₁ $K(\lambda)$ has full row rank at 0 then $K(\lambda)$ is a minimal basis. Conversely, if $K(\lambda)$ is a minimal basis with all its row degrees equal to one then $K(\lambda)$ has full row rank in \mathbb{F} and rev₁ $K(\lambda)$ has full row rank at 0.

4.4.3 Strong block minimal bases linearizations as block full rank linearizations

By using strong block minimal bases pencils, in [7, Theorem 5.11] (see Theorem 2.5.5) (strong) linearizations are constructed that contain the complete spectral information of rational matrices, finite and infinite, as well as the information about their minimal bases and indices [8], when the corresponding rational matrix $R(\lambda)$ is expressed in the form $R(\lambda) = D(\lambda) + C(\lambda I_n - A)^{-1}B$, where $D(\lambda)$ is the polynomial part of $R(\lambda)$ with deg $(D(\lambda)) > 1$, and $C(\lambda I_n - A)^{-1}B$ is a minimal state-space realization of the strictly proper part of $R(\lambda)$. We will see that such linearizations satisfy Theorem 4.4.1, with $\Omega = \mathbb{F}$, and Theorem 4.4.10.

88

For the construction, in [7] $L(\lambda) := \begin{bmatrix} M(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix}$ is considered to be a strong block minimal bases pencil as in (2.12) associated with $D(\lambda)$ with sharp degree, that is, $\deg(D(\lambda)) = \deg(N_2(\lambda)) + \deg(N_1(\lambda)) + 1$, where $N_1(\lambda)$ (respectively $N_2(\lambda)$) is a minimal basis dual to $K_1(\lambda)$ (respectively $K_2(\lambda)$). Then, constant matrices $\hat{K}_1 \in \mathbb{F}^{m \times (m + \hat{m})}$ and $\hat{K}_2 \in \mathbb{F}^{p \times (p + \hat{p})}$ and matrices $\hat{N}_1(\lambda)^T \in \mathbb{F}[\lambda]^{(m + \hat{m}) \times \hat{m}}$ and $\hat{N}_2(\lambda)^T \in \mathbb{F}[\lambda]^{(p + \hat{p}) \times \hat{p}}$ exist such that, for i = 1, 2,

$$U_i(\lambda) = \begin{bmatrix} K_i(\lambda) \\ \widehat{K}_i \end{bmatrix}$$
 and $U_i(\lambda)^{-1} = \begin{bmatrix} \widehat{N}_i(\lambda)^T & N_i(\lambda)^T \end{bmatrix}$

are unimodular. Finally, the following linear polynomial matrix is constructed

$$\mathcal{L}(\lambda) = \begin{bmatrix} X(\lambda I_n - A)Y & XB\widehat{K}_1 & 0\\ \hline -\widehat{K}_2^T CY & M(\lambda) & K_2(\lambda)^T\\ 0 & K_1(\lambda) & 0 \end{bmatrix},$$
(4.23)

where $X, Y \in \mathbb{F}^{n \times n}$ are any nonsingular constant matrices. With these assumptions, $\mathcal{L}(\lambda)$ is a strong linearization of $R(\lambda)$ [7, Theorem 5.11].

This result follows as a corollary of Theorems 4.4.1 and 4.4.10 as well. First, since $\widehat{K}_i N_i(\lambda)^T = I$, notice that $R(\lambda)$ can be written as in (4.16):

$$N_{2}(\lambda)[M(\lambda) + \widehat{K}_{2}^{T}CYY^{-1}(\lambda I_{n} - A)^{-1}X^{-1}XB\widehat{K}_{1}]N_{1}(\lambda)^{T} = D(\lambda) + C(\lambda I_{n} - A)^{-1}B,$$

and, in addition, conditions (4.15), with $\Omega = \mathbb{F}$, and (4.22) are satisfied. More precisely, we have that

$$\operatorname{rank} \begin{bmatrix} X(\lambda I_n - A)Y\\ -N_2(\lambda)\widehat{K}_2^T CY \end{bmatrix} = \operatorname{rank} \begin{bmatrix} X(\lambda I_n - A)Y\\ -CY \end{bmatrix} = n, \text{ and}$$
$$\operatorname{rank} \begin{bmatrix} X(\lambda I_n - A)Y \quad XB\widehat{K}_1N_1(\lambda)^T \end{bmatrix} = \operatorname{rank} \begin{bmatrix} X(\lambda I_n - A)Y \quad XB \end{bmatrix} = n$$

for all $\lambda \in \mathbb{F}$, since X and Y are nonsingular, and the realization $C(\lambda I_n - A)^{-1}B$ is minimal. Therefore, condition (4.15) is satisfied and, thus, $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in \mathbb{F} by Theorem 4.4.1. Moreover, we have that condition (4.22) is satisfied since $\operatorname{rev}_1(\lambda I_n - A)$ evaluated at 0 is just I_n . Then, by Theorem 4.4.10, $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ at infinity of grade $\deg(N_2(\lambda)) + \deg(N_1(\lambda)) + 1 = \deg(D(\lambda))$.

4.4.4 Recovery of eigenvectors

In this section, we show how to recover right and left eigenvectors of a regular rational matrix $R(\lambda)$, associated with the eigenvalues in a set Ω , from those of a block full rank linearization of $R(\lambda)$ in Ω . For that, in Theorem 4.4.15 we study the relation between their right and left nullspaces. We will use the following Lemma 4.4.14.

Lemma 4.4.14. Let $K(\lambda)$ be a rational basis and let $N(\lambda)$ be any rational basis dual to $K(\lambda)$. Let Ω be a nonempty subset of \mathbb{F} such that $K(\lambda)$ and $N(\lambda)$ have full row rank in Ω . Then there exist rational matrices of the form

$$V(\lambda) = \begin{bmatrix} K(\lambda) \\ \bar{K}(\lambda) \end{bmatrix} \quad and \quad V(\lambda)^{-1} = \begin{bmatrix} \bar{N}(\lambda)^T & N(\lambda)^T \end{bmatrix}$$
(4.24)

that are invertible in Ω .

Proof. By Lemma 4.3.2, there exist rational matrices $S(\lambda)$ and $W(\lambda)$, both invertible in Ω , such that $K(\lambda) = S(\lambda)\widetilde{K}(\lambda)$ and $N(\lambda) = W(\lambda)\widetilde{N}(\lambda)$, where $\widetilde{K}(\lambda)$ and $\widetilde{N}(\lambda)$ are minimal bases of the row spaces of $K(\lambda)$ and $N(\lambda)$, respectively. Then, by [26, Theorem 2.10], there exist unimodular matrices of the form

$$U(\lambda) = \begin{bmatrix} \widetilde{K}(\lambda) \\ \widehat{K}(\lambda) \end{bmatrix} \text{ and } U(\lambda)^{-1} = \begin{bmatrix} \widehat{N}(\lambda)^T & \widetilde{N}(\lambda)^T \end{bmatrix}.$$

Then we consider the rational matrix

$$V(\lambda) := \begin{bmatrix} S(\lambda) & 0\\ 0 & W(\lambda)^{-T} \end{bmatrix} U(\lambda) = \begin{bmatrix} K(\lambda)\\ \bar{K}(\lambda) \end{bmatrix},$$

with $\bar{K}(\lambda) := W(\lambda)^{-T} \widehat{K}(\lambda)$. We have that $V(\lambda)$ is invertible in Ω since $S(\lambda)$ and $W(\lambda)$ are invertible in Ω and $U(\lambda)$ is unimodular. Finally, we note that $V(\lambda)^{-1} = U(\lambda)^{-1} \begin{bmatrix} S(\lambda)^{-1} & 0\\ 0 & W(\lambda)^T \end{bmatrix} = \begin{bmatrix} \widehat{N}(\lambda)^T S(\lambda)^{-1} & \widetilde{N}(\lambda)^T W(\lambda)^T \end{bmatrix} = \begin{bmatrix} \overline{N}(\lambda)^T & N(\lambda)^T \end{bmatrix},$ with $\overline{N}(\lambda)^T := \widehat{N}(\lambda)^T S(\lambda)^{-1}$.

Let us now consider the rational bases $K_1(\lambda)$ and $K_2(\lambda)$ of the block full rank linearization in Theorem 4.4.1 and their dual rational bases $N_1(\lambda)$ and $N_2(\lambda)$. It then follows that, since they have full row rank in a nonempty set Ω , there exist rational matrices of the form

$$V_i(\lambda) = \begin{bmatrix} K_i(\lambda) \\ \bar{K}_i(\lambda) \end{bmatrix} \quad \text{and} \quad V_i(\lambda)^{-1} = \begin{bmatrix} \bar{N}_i(\lambda)^T & N_i(\lambda)^T \end{bmatrix}$$
(4.25)

that are invertible in Ω , for i = 1, 2, by Lemma 4.4.14. We use these matrices in the following Theorem 4.4.15.

Theorem 4.4.15. *Let*

$$\mathcal{L}(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) & 0\\ -C(\lambda) & M(\lambda) & K_2(\lambda)^T\\ 0 & K_1(\lambda) & 0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q) \times (n+r)}$$

be a block full rank linearization in a nonempty set Ω of the rational matrix

$$R(\lambda) = N_2(\lambda)[M(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)]N_1(\lambda)^T,$$

as in Theorem 4.4.1. Consider the rational matrices $\bar{N}_1(\lambda)$ and $\bar{N}_2(\lambda)$ in (4.25) and denote $S(\lambda) := M(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$. Let $\lambda_0 \in \Omega$ such that det $A(\lambda_0) \neq 0$. Then, the following statements hold:

(a) The linear map

$$E_r : \mathcal{N}_r(R(\lambda_0)) \longrightarrow \mathcal{N}_r(\mathcal{L}(\lambda_0))$$
$$x \longmapsto \begin{bmatrix} -A(\lambda_0)^{-1}B(\lambda_0)N_1(\lambda_0)^T \\ N_1(\lambda_0)^T \\ -\bar{N}_2(\lambda_0)^TS(\lambda_0)N_1(\lambda_0)^T \end{bmatrix} x$$

is a bijection between the right nullspaces over \mathbb{F} of $R(\lambda_0)$ and $\mathcal{L}(\lambda_0)$.

(b) The linear map

$$E_{\ell} : \mathcal{N}_{\ell}(R(\lambda_0)) \longrightarrow \mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$$
$$y^T \longmapsto y^T \left[N_2(\lambda_0)C(\lambda_0)A(\lambda_0)^{-1} \quad N_2(\lambda_0) \quad -N_2(\lambda_0)S(\lambda_0)\bar{N}_1(\lambda_0)^T \right]$$

is a bijection between the left nullspaces over \mathbb{F} of $R(\lambda_0)$ and $\mathcal{L}(\lambda_0)$.

Proof. We only prove (a) since (b) is analogous. We will see that E_r is a composition of maps, $E_r = G_r \circ H_r$, where G_r and H_r are linear bijections. For that, we consider the transfer function of $\mathcal{L}(\lambda)$. That is, the rational matrix

$$\widehat{R}(\lambda) = \begin{bmatrix} S(\lambda) & K_2(\lambda)^T \\ K_1(\lambda) & 0 \end{bmatrix} \in \mathbb{F}(\lambda)^{q \times r}.$$
(4.26)

By Lemma 2.4.10, the linear map

$$G_r : \mathcal{N}_r(\hat{R}(\lambda_0)) \longrightarrow \mathcal{N}_r(\mathcal{L}(\lambda_0))$$
$$\hat{x} \longmapsto \begin{bmatrix} -A(\lambda_0)^{-1} \begin{bmatrix} B(\lambda_0) & 0 \end{bmatrix} \\ I_r \end{bmatrix} \hat{x}$$

is a bijection between the right nullspaces of $\widehat{R}(\lambda_0)$ and $\mathcal{L}(\lambda_0)$, since $\widehat{R}(\lambda)$ is the transfer function of $\mathcal{L}(\lambda)$ and det $A(\lambda_0) \neq 0$. Now, we consider the linear map

$$H_r : \mathcal{N}_r(R(\lambda_0)) \longrightarrow \mathcal{N}_r(\widehat{R}(\lambda_0))$$
$$x \longmapsto \begin{bmatrix} N_1(\lambda_0)^T \\ -\bar{N}_2(\lambda_0)^T S(\lambda_0) N_1(\lambda_0)^T \end{bmatrix} x.$$

To see that H_r is well defined, let us consider the rational matrix $V_2(\lambda)^{-1} =$ $\begin{bmatrix} N_2(\lambda)^T & N_2(\lambda)^T \end{bmatrix}$ in (4.25). Then, we have that

$$\begin{bmatrix} V_2(\lambda)^{-T} & 0\\ 0 & I \end{bmatrix} \begin{bmatrix} S(\lambda) & K_2(\lambda)^T\\ K_1(\lambda) & 0 \end{bmatrix} \begin{bmatrix} N_1(\lambda)^T\\ -\bar{N}_2(\lambda)^T S(\lambda) N_1(\lambda)^T \end{bmatrix} = \begin{bmatrix} 0\\ R(\lambda)\\ 0 \end{bmatrix}.$$
 (4.27)

By (4.27), H_r is well defined since the matrix $\begin{bmatrix} V_2(\lambda)^{-T} & 0\\ 0 & I \end{bmatrix}$ is invertible in Ω , by Lemma 4.4.14. In addition, since $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Ω , we have that $\widehat{R}(\lambda)$ and diag $(R(\lambda), I_s)$ are equivalent in Ω for some s > 0 (see Definition 4.2.1). Therefore, dim $\mathcal{N}_r(R(\lambda_0)) = \dim \mathcal{N}_r(\widehat{R}(\lambda_0))$. Thus, to see that H_r is a bijection we only have to prove that H_r is injective. Assume that $\begin{bmatrix} N_1(\lambda_0)^T \\ -\bar{N}_2(\lambda_0)^T S(\lambda_0) N_1(\lambda_0)^T \end{bmatrix} x = 0$ for some $x \in \mathcal{N}_r(R(\lambda_0))$. In particular, $N_1(\lambda_0)^T x = 0$. Since $N_1(\lambda_0)^T$ has full column rank, x = 0. Therefore, H_r is a bijection. Finally, note that $E_r = G_r \circ H_r$.

Remark 4.4.16. We recall that, since the linear maps E_r and E_ℓ are bijections, one can recover a basis of the right (resp. left) nullspace of $R(\lambda_0)$ from a basis of the right (resp. left) nullspace of $\mathcal{L}(\lambda_0)$, and conversely. For instance, if $\{x_i\}_{i=1}^t$ is a basis of

$$\mathcal{N}_{r}(R(\lambda_{0})) \text{ then } \left\{ \begin{bmatrix} -A(\lambda_{0})^{-1}B(\lambda_{0})N_{1}(\lambda_{0})^{T} \\ N_{1}(\lambda_{0})^{T} \\ -\bar{N}_{2}(\lambda_{0})^{T}S(\lambda_{0})N_{1}(\lambda_{0})^{T} \end{bmatrix} x_{i} \right\}_{i=1}^{t} \text{ is a basis of } \mathcal{N}_{r}(\mathcal{L}(\lambda_{0})). \text{ And,}$$

given a basis $\{y_i\}_{i=1}^{\iota}$ of $\mathcal{N}_r(\mathcal{L}(\lambda_0))$, we can recover a scale $\mathcal{L}(\lambda_0)^{-1}B(\lambda_0)N_1(\lambda_0)^T$ account that the elements will be of the form $y_i = \begin{bmatrix} -A(\lambda_0)^{-1}B(\lambda_0)N_1(\lambda_0)^T \\ N_1(\lambda_0)^T \\ -\bar{N}_2(\lambda_0)^TS(\lambda_0)N_1(\lambda_0)^T \end{bmatrix} x_i$

for some basis $\{x_i\}_{i=1}^t$ of $\mathcal{N}_r(R(\lambda_0))$. Then, to recover $\{x_i\}_{i=1}^t$ we can use that $\bar{K}_1(\lambda_0)N_1(\lambda_0)^T x_i = x_i$. However, in practice, it can usually be recovered from the structure of $N_1(\lambda_0)^T x_i$, without computing $\bar{K}_1(\lambda_0)$.

Chapter 5

Application of the local linearization theory to linearizations of rational approximations of nonlinear eigenvalue problems

In this chapter we study in depth the pencils introduced in the influential references [47] and [60] for linearizing rational matrices obtained from approximating NonLinear Eigenvalue Problems (NLEPs). In particular, the reference [47] presents one of the first systematic approaches for solving large scale NLEPs. The results in this chapter are valid and are stated in any algebraically closed field \mathbb{F} . Note, however, that references [47] and [60] consider only the complex field \mathbb{C} and that this restriction is important in the approximation phase of the NLEP. We start by defining a NLEP [46]. Given a non-empty open set $\Omega \subseteq \mathbb{F}$ and an analytic matrix-valued function

$$\begin{array}{rccc} F: & \Omega & \to & \mathbb{F}^{n \times n} \\ & \lambda & \mapsto & F(\lambda), \end{array}$$

the NLEP consists of computing scalars $\lambda_0 \in \Omega$ (eigenvalues) and nonzero vectors $v \in \mathbb{F}^n$ (eigenvectors) such that

$$F(\lambda_0)v = 0,$$

under the regularity assumption $\det(F(\lambda)) \neq 0$. Since a direct solution of NLEPs is usually not possible, they are often solved numerically via rational approximation and by solving the corresponding REP with linearizations adapted to the structure of the obtained rational matrix. The results in this chapter appear in [28] and [29].

5.1 NLEIGS pencils

For brevity of exposition, and also for recognizing the key contribution of [47], we will call *NLEIGS pencils* to the pencils introduced in [47]. The main goal of this section is to replace the vague usage of the word "linearization" in [47] by a number of rigorous results on NLEIGS pencils which, combined with the results in Sections 4.2 and 4.3, establish the precise properties enjoyed with respect to eigenvalues (and poles) of the NLEIGS pencils. We remark that NLEIGS pencils were the initial motivation for developing the theory of local linearizations, since they are not linearizations of the corresponding rational matrix according to the definitions in [2, 7]. The approach in [47] consists essentially of three steps:

- (1) The matrix defining the NLEP is approximated by a rational matrix $Q_N(\lambda)$ via Hermite's interpolation in a certain compact target set $\Sigma \subset \mathbb{C}$ where the eigenvalues of interest are located.
- (2) The obtained rational matrix is linearized by using a certain pencil $L_N(\lambda)$.
- (3) A highly structured rational Krylov method is applied to the pencil to compute the eigenvalues of $Q_N(\lambda)$ in Σ .

Since we are interested in rational matrices and their linearizations, all the delicate details about how the rational approximations $Q_N(\lambda)$ are constructed are omitted. Such details can be found in [47]. Moreover, although [47] deals with regular rational matrices $Q_N(\lambda)$, we will not impose such condition initially in our developments.

Reference [47] uses two families of rational matrices, and corresponding pencils, depending on whether or not a certain low rank structure is present in the original NLEP. We will refer to them as the *NLEIGS basic problem* and the *NLEIGS low rank* structured problem, respectively. The NLEIGS pencils corresponding to each of these two cases will be studied from two perspectives giving rise to the four subsections included in this section. These two perspectives are considering NLEIGS pencils as block full rank pencils and, thus, as linearizations with empty state matrices, and considering them as polynomial system matrices with transfer function matrices equivalent to $Q_N(\lambda)$ everywhere except at a point ξ_N . Both perspectives allow us to state in a rigorous sense that NLEIGS pencils are linearizations of $Q_N(\lambda)$, but the one based on block full rank pencils is much simpler, does not require any hypothesis and covers fully the applications of interest in [47]. In contrast, the polynomial system matrix perspective provides more information on $Q_N(\lambda)$ but at the cost of extra hypotheses which are not imposed in [47] and that require considerable effort to check.

The families of rational matrices considered in [47] are defined in terms of the following parameters: a list of nodes $(\sigma_0, \sigma_1, \ldots, \sigma_{N-1})$ in \mathbb{F} , a list of nonzero poles $(\xi_1, \xi_2, \ldots, \xi_N)$ in $\mathbb{F} \cup \{\infty\}$, and a list of nonzero scaling parameters $(\beta_0, \beta_1, \ldots, \beta_N)$ in \mathbb{F} . It is important to bear in mind that [47] assumes that the poles are all distinct
from the nodes. However, we do not assume such property, except in a few results where it will be explicitly stated. With these parameters, the following sequence of rational scalar functions is defined:

$$b_0(\lambda) = \frac{1}{\beta_0}, \quad b_i(\lambda) = \frac{1}{\beta_0} \prod_{k=1}^i \frac{\lambda - \sigma_{k-1}}{\beta_k (1 - \lambda/\xi_k)}, \quad i = 1, \dots, N.$$
 (5.1)

Let us now define the linear scalar functions

$$g_i(\lambda) := \beta_i \left(1 - \lambda/\xi_i\right), \quad \text{and} \quad h_j(\lambda) := \lambda - \sigma_j,$$

$$(5.2)$$

for i = 1, ..., N, and j = 0, ..., N - 1. Then, the rational functions $b_i(\lambda)$ satisfy the simple recursion

$$g_{j+1}(\lambda) b_{j+1}(\lambda) = h_j(\lambda) b_j(\lambda), \quad j = 0, 1, \dots, N-1,$$

which will be useful in the sequel. Note that the rational functions $b_i(\lambda)$ could not be proper, since for any infinite pole $\xi_i = \infty$ the corresponding factor $1 - \lambda/\xi_i$ is just equal to 1, and, therefore, $b_i(\lambda)$ has a nonconstant polynomial part.

With all this information, we are in the position of introducing the first family of rational matrices considered in [47], whose elements are defined as

$$Q_N(\lambda) = b_0(\lambda)D_0 + b_1(\lambda)D_1 + \dots + b_N(\lambda)D_N \in \mathbb{F}(\lambda)^{m \times m},$$
(5.3)

where $D_0, \ldots, D_N \in \mathbb{F}^{m \times m}$ are constant matrices. In this section, all the parameters that allow us to define the considered family of rational matrices are arbitrary. However, in [47] these parameters are carefully chosen in such a way that $Q_N(\lambda)$ approximates satisfactorily the matrix defining the NLEP to be solved in a target set $\Sigma \subset \mathbb{F}$. In this scenario, it is important to stress that the poles (ξ_1, \ldots, ξ_N) are always chosen outside Σ [47, p. A2852], which implies that all the zeros of $Q_N(\lambda)$ located in Σ are eigenvalues of $Q_N(\lambda)$. Thus, the REP associated with $Q_N(\lambda)$ is an explicit example of a problem in which the poles are located outside the region of interest and, then, it is not needed to compute them. Note, however, the following subtlety: though it is clear that the finite poles of $Q_N(\lambda)$ are included in the list (ξ_1,\ldots,ξ_N) , we can construct examples of matrices as in (5.3) for which some of the finite numbers in (ξ_1, \ldots, ξ_N) are not poles due to some cancellations. Thus, all the finite numbers in (ξ_1, \ldots, ξ_N) are not necessarily finite poles of $Q_N(\lambda)$ and, even more, the partial multiplicities of such poles are not immediately visible from (5.3). Despite these comments, we will call the numbers (ξ_1, \ldots, ξ_N) poles, following the usage in |47|.

5.1.0.1 The NLEIGS basic problem from the point of view of block full rank pencils

In order to solve the REP $Q_N(\lambda) y = 0$, the authors of [47] solve the generalized eigenvalue problem corresponding to the pencil

$$L_N(\lambda) = \begin{bmatrix} M_N(\lambda) \\ K_N(\lambda) \end{bmatrix}, \qquad (5.4)$$

where

$$M_{N}(\lambda) := \left[\begin{array}{ccc} \frac{g_{N}(\lambda)}{\beta_{N}} D_{0} & \frac{g_{N}(\lambda)}{\beta_{N}} D_{1} & \cdots & \frac{g_{N}(\lambda)}{\beta_{N}} D_{N-2} & \frac{g_{N}(\lambda)}{\beta_{N}} D_{N-1} + \frac{h_{N-1}(\lambda)}{\beta_{N}} D_{N} \end{array}\right],$$
$$K_{N}(\lambda) := \left[\begin{array}{ccc} -h_{0}(\lambda) & g_{1}(\lambda) & & \\ & -h_{1}(\lambda) & g_{2}(\lambda) & & \\ & & \ddots & \ddots & \\ & & & -h_{N-2}(\lambda) & g_{N-1}(\lambda) \end{array}\right] \otimes I_{m}.$$

In [47] the use of $L_N(\lambda)$ for solving the REP associated to $Q_N(\lambda)$ is supported by [47, Theorem 3.2], which states that $L_N(\lambda)$ is a strong linearization of the rational matrix $Q_N(\lambda)$ without specifying the exact meaning of "strong linearization" in this rational context. Moreover, the proof of [47, Theorem 3.2] consists of a reference to [5, Theorem 3.1], which is a paper dealing with strong linearizations of *polynomial* matrices in the classical sense of [44]. However, as a consequence of the results in Section 4.3, it is very easy to prove that $L_N(\lambda)$ is always a linearization of $Q_N(\lambda)$ in a set including the region of interest in [47], as well as at infinity. This is proved in Theorem 5.1.1, where the nomenclature introduced in Remark 4.3.6 is used.

Theorem 5.1.1. Let $Q_N(\lambda)$ be the rational matrix in (5.3) and $L_N(\lambda)$ be the pencil in (5.4). Let \mathcal{P}_N and i_N be, respectively, the set of finite poles and the number of infinite poles in the list $(\xi_1, \xi_2, \ldots, \xi_N)$. Then, the following statements hold:

- (a) $L_N(\lambda)$ partitioned as in (5.4) is a block full rank pencil with only one block column associated with $Q_N(\lambda)$ in $\mathbb{F} \setminus \mathcal{P}_N$ and $L_N(\lambda)$ is a linearization of $Q_N(\lambda)$ in $\mathbb{F} \setminus \mathcal{P}_N$ with empty state matrix.
- (b) $L_N(\lambda)$ is a linearization of $Q_N(\lambda)$ at ∞ of grade i_N with empty state matrix.

Proof. It is immediate to check that

$$N_N(\lambda) := \frac{1}{1 - \frac{\lambda}{\xi_N}} \begin{bmatrix} b_0(\lambda) & b_1(\lambda) & \cdots & b_{N-1}(\lambda) \end{bmatrix} \otimes I_m$$
(5.5)

is a rational basis dual to $K_N(\lambda)$. Note also that $K_N(\lambda)$ and $N_N(\lambda)$ have both full row rank in $\mathbb{F} \setminus \mathcal{P}_N$. In addition, an easy direct computation proves $M_N(\lambda)N_N(\lambda)^T =$ $Q_N(\lambda)$. Thus, part (a) follows from Theorem 4.4.1. Observe that (a) can also be proved from Corollary 4.3.8, since the structures of $Q_N(\lambda)$, $L_N(\lambda)$ and $N_N(\lambda)$ are particular cases of those described in that corollary.

In order to prove part (b), note first that $\operatorname{rev}_1 K_N(\lambda)$ has full row rank at 0. We now consider the rational matrix $\operatorname{rev}_{i_N-1} N_N(\lambda) = \lambda^{i_N-1} N_N\left(\frac{1}{\lambda}\right)$, which is of the form $\lambda^{i_N-1} N_N\left(\frac{1}{\lambda}\right) = \left[* \cdots * \frac{\lambda}{\lambda - 1/\xi_N} \lambda^{i_N-1} b_{N-1}\left(\frac{1}{\lambda}\right) I_m \right]$, where the entries * are defined at 0. Denote by i_{N-1} the number of infinite poles in the list $(\xi_1, \xi_2, \ldots, \xi_{N-1})$. Then, $b_{N-1}\left(\frac{1}{\lambda}\right) = \frac{1}{\lambda^{i_{N-1}}}c(\lambda)$, for a certain rational function $c(\lambda)$ with $c(0) \neq 0$. Thus, we obtain that $\operatorname{rev}_{i_N-1} N_N(\lambda)$ has full row rank at 0, taking into account that $i_{N-1} = i_N$ if $\xi_N \neq \infty$, and $i_{N-1} = i_N - 1$ if $\xi_N = \infty$. Then, part (b) follows from Theorem 4.3.7.

Combining Theorems 5.1.1 and 4.2.6, we get that $L_N(\lambda)$ contains all the information about the finite eigenvalues of $Q_N(\lambda)$ in $\mathbb{F} \setminus \mathcal{P}_N$, including all type of multiplicities (algebraic, geometric and partial). Moreover, Proposition 4.2.15 allows us to recover the complete pole-zero structure of $Q_N(\lambda)$ at ∞ from the eigenvalue structure at 0 of rev $L_N(\lambda)$, just by noting that, in this case, t = 0 in Proposition 4.2.15 since we are taking an empty state matrix. We stress that all these results hold for *any* rational matrix $Q_N(\lambda)$ either regular or singular. However, no information is provided on the finite poles of $Q_N(\lambda)$, and some of them could also be zeros. As explained above, this is not an issue in [47], since \mathcal{P}_N is outside the target set Σ . Nevertheless, at the cost of imposing extra hypotheses, we will solve this problem in Section 5.1.0.2 for completeness and also because it is of interest for the theory of REPs.

5.1.0.2 The NLEIGS basic problem from the point of view of polynomial system matrices

As discussed previously, the approach presented in Section 5.1.0.1 to the NLEIGS pencil $L_N(\lambda)$ in (5.4) considers $L_N(\lambda)$ as a linearization with empty state matrix and, thus, it does not provide any information on the finite poles of $Q_N(\lambda)$. In order to get this information, we need to identify a convenient square regular submatrix $A_N(\lambda)$ of $L_N(\lambda)$ that may be used as state matrix. The block structure of $L_N(\lambda)$ makes it impossible to find such a matrix $A_N(\lambda)$ in a way that it includes the information of all the potential poles (ξ_1, \ldots, ξ_N) . This is related with the comment included in [47, p. A2849] on the fact that ξ_N plays a special role and that it is convenient to choose $\xi_N = \infty$. In what follows we will *not* assume that $\xi_N = \infty$, though the obtained results are simpler and stronger under such assumption, but we will focus on getting information on the finite poles in $(\xi_1, \ldots, \xi_{N-1})$. With this spirit, we consider the following partition of $L_N(\lambda)$ in (5.4), where $A_N(\lambda)$ will play the role of the state matrix,

$$L_N(\lambda) =: \begin{bmatrix} D_N(\lambda) & -C_N(\lambda) \\ \hline B_N(\lambda) & A_N(\lambda) \end{bmatrix}, \text{ where } D_N(\lambda) := \left(1 - \frac{\lambda}{\xi_N}\right) D_0, \quad (5.6)$$

and the rest of the blocks are easily described from the blocks in (5.4). With this partition, the next technical lemma reveals the transfer function matrix of $L_N(\lambda)$ and establishes necessary and sufficient conditions for $L_N(\lambda)$ to be minimal in the whole field \mathbb{F} . By definition, $L_N(\lambda)$ is minimal in \mathbb{F} if $\begin{bmatrix} B_N(\lambda_0) & A_N(\lambda_0) \end{bmatrix} \in$ $\mathbb{F}^{m(N-1)\times mN}$ and $\begin{bmatrix} -C_N(\lambda_0)^T & A_N(\lambda_0)^T \end{bmatrix}^T \in \mathbb{F}^{mN\times m(N-1)}$ have, respectively, full row and column rank for all $\lambda_0 \in \mathbb{F}$. The conditions in Lemma 5.1.2(b) require to evaluate the rational matrix $R_N(\lambda)$ of size $m \times m$, which for practical problems is much smaller than $m(N-1) \times mN$.

Lemma 5.1.2. Let us consider the pencil $L_N(\lambda)$ in (5.4) as a polynomial system matrix with state matrix $A_N(\lambda)$, where $A_N(\lambda)$ is defined through the partition (5.6), and let $Q_N(\lambda)$ be the rational matrix in (5.3). Then the following statements hold:

- (a) The transfer function matrix of $L_N(\lambda)$ is $\beta_0\left(1-\frac{\lambda}{\xi_N}\right)Q_N(\lambda)$.
- (b) Let us define the rational matrix $R_N(\lambda) := (Q_N(\lambda) b_0(\lambda)D_0)/b_N(\lambda)$, whose explicit expression is

$$R_N(\lambda) = \sum_{j=1}^{N-1} \left(\prod_{k=j+1}^N \frac{g_k(\lambda)}{h_{k-1}(\lambda)} \right) D_j + D_N \in \mathbb{F}(\lambda)^{m \times m}, \tag{5.7}$$

let \mathcal{P}_{N-1} be the set of finite poles in the list $(\xi_1, \xi_2, \ldots, \xi_{N-1})$, and assume $\xi_i \neq \sigma_j, 1 \leq i \leq N, 0 \leq j \leq N-1$. Then, $L_N(\lambda)$ is minimal in \mathbb{F} if and only if the matrix $R_N(\xi_k) \in \mathbb{F}^{m \times m}$ is nonsingular for all $\xi_k \in \mathcal{P}_{N-1}$.

Proof. Let us consider $L_N(\lambda)$ partitioned as in (5.6) and as a polynomial system matrix with state matrix $A_N(\lambda)$. The computation of the transfer function matrix of $L_N(\lambda)$ is easy because $B_N(\lambda) = \begin{bmatrix} -h_0(\lambda)I_m & 0 & \cdots & 0 \end{bmatrix}^T$, which implies that only the first block column of $A_N(\lambda)^{-1}$ is needed. It is immediate to check that $\frac{1}{b_1(\lambda)g_1(\lambda)} \begin{bmatrix} b_1(\lambda) & \cdots & b_{N-1}(\lambda) \end{bmatrix}^T \otimes I_m$ is that first block column. The rest of the proof of part (a) is just an elementary algebraic manipulation. For part (b), observe first that $\xi_i \neq \sigma_j$, $1 \leq i \leq N$ and $0 \leq j \leq N - 1$, implies that $\begin{bmatrix} B_N(\lambda_0) & A_N(\lambda_0) \end{bmatrix}$ has full row rank for any $\lambda_0 \in \mathbb{F}$. On the other hand, if we define

$$Z_N(\lambda) := \begin{bmatrix} -C_N(\lambda) \\ A_N(\lambda) \end{bmatrix}, \qquad (5.8)$$

then $Z_N(\lambda_0)$ has full column rank for every $\lambda_0 \in \mathbb{F} \setminus \mathcal{P}_{N-1}$, because $A_N(\lambda_0)$ is invertible in $\mathbb{F} \setminus \mathcal{P}_{N-1}$. Therefore, combining the discussion above with Definition

98

5.1. NLEIGS PENCILS

4.1.1, we obtain that $L_N(\lambda)$ is minimal in \mathbb{F} if and only if $Z_N(\xi_k)$ has full column rank for every $\xi_k \in \mathcal{P}_{N-1}$. The rest of the proof proceeds as follows: we will find a rational matrix $S_N(\lambda)$ such that is equivalent to $Z_N(\lambda)$ in \mathcal{P}_{N-1} and has a simple structure that allows us to see that $S_N(\xi_k)$ (and, so, $Z_N(\xi_k)$) has full column rank for every $\xi_k \in \mathcal{P}_{N-1}$ if and only if $R_N(\xi_k)$ is invertible for every $\xi_k \in \mathcal{P}_{N-1}$, where $R_N(\lambda)$ is the rational matrix in (5.7). For brevity, we use the notation $g_i := g_i(\lambda)$ and $h_i := h_i(\lambda)$ for the scalar functions in (5.2). In addition, $Z_N(\lambda)$ in (5.8) is partitioned as

$$Z_N(\lambda) =: \begin{bmatrix} Z_{11}(\lambda) & Z_{12}(\lambda) \\ Z_{21}(\lambda) & Z_{22}(\lambda) \end{bmatrix},$$
(5.9)

where

$$Z_{11}(\lambda) = \begin{bmatrix} \frac{g_N}{\beta_N} D_1 & \frac{g_N}{\beta_N} D_2 & \cdots & \cdots & \frac{g_N}{\beta_N} D_{N-2} \\ g_1 I_m & 0 & \cdots & \cdots & 0 \end{bmatrix}, \quad Z_{12}(\lambda) = \begin{bmatrix} \frac{g_N}{\beta_N} D_{N-1} + \frac{h_{N-1}}{\beta_N} D_N \\ 0 \end{bmatrix},$$
$$Z_{21}(\lambda) = \begin{bmatrix} -h_1 & g_2 & & & \\ & -h_2 & g_3 & & \\ & & \ddots & \ddots & \\ & & & -h_{N-3} & g_{N-2} \\ & & & & & -h_{N-2} \end{bmatrix} \otimes I_m, \quad Z_{22}(\lambda) = \begin{bmatrix} 0 \\ \vdots \\ \vdots \\ 0 \\ g_{N-1} I_m \end{bmatrix}.$$

Note that the matrix $Z_{21}(\lambda)$ is invertible in \mathcal{P}_{N-1} and that the last block column of $Z_{21}(\lambda)^{-1}$ is

$$Y_{22}(\lambda) := -\left[\frac{1}{h_1}\prod_{i=2}^{N-2}\frac{g_i}{h_i}, \frac{1}{h_2}\prod_{i=3}^{N-2}\frac{g_i}{h_i}, \cdots, \frac{1}{h_{N-3}}\frac{g_{N-2}}{h_{N-2}}, \frac{1}{h_{N-2}}\right]^T \otimes I_m.$$
(5.10)

Next, a sequence of equivalence transformations in \mathcal{P}_{N-1} are applied to $Z_N(\lambda)$. Such transformations are described by using the notation in (5.9) and (5.10), and the first one is

$$Y_N(\lambda) := \left[\begin{array}{c|c} I_{2m} & 0 \\ \hline 0 & Z_{21}(\lambda)^{-1} \end{array} \right] Z_N(\lambda) = \left[\begin{array}{c|c} Z_{11}(\lambda) & Z_{12}(\lambda) \\ \hline I_{(N-2)m} & g_{N-1}Y_{22}(\lambda) \end{array} \right].$$

The second transformation is designed to turn zero the second block row of $Z_{11}(\lambda)$ as follows

$$W_N(\lambda) := \operatorname{diag} \left(I_m, \begin{bmatrix} I_m & -g_1 I_m \\ 0 & I_m \end{bmatrix}, I_{(N-3)m} \right) Y_N(\lambda)$$
$$= \begin{bmatrix} \frac{g_N}{\beta_N} D_1 & \cdots & \frac{g_N}{\beta_N} D_{N-2} \\ 0 & \cdots & 0 \\ \hline I_{(N-2)m} & 0 \end{bmatrix} \begin{bmatrix} \frac{g_N}{\beta_N} D_{N-1} + \frac{h_{N-1}}{\beta_N} D_N \\ \left(\prod_{i=1}^{N-2} \frac{g_i}{h_i} \right) g_{N-1} I_m \\ g_{N-1} Y_{22}(\lambda) \end{bmatrix}$$

The third transformation turns zero the block $g_{N-1}Y_{22}(\lambda)$ of $W_N(\lambda)$ and performs a convenient scalar multiplication in its first block row. Such transformation is

$$\begin{split} X_N(\lambda) &:= \begin{bmatrix} \frac{\beta_N}{h_{N-1}} I_m & 0\\ 0 & I_{(N-1)m} \end{bmatrix} W_N(\lambda) \begin{bmatrix} I_{(N-2)m} & -g_{N-1} Y_{22}(\lambda) \\ 0 & I_m \end{bmatrix} \\ &= \begin{bmatrix} \frac{g_N}{h_{N-1}} D_1 & \cdots & \frac{g_N}{h_{N-1}} D_{N-2} & R_N(\lambda) \\ 0 & \cdots & 0 & \left(\prod_{i=1}^{N-2} \frac{g_i}{h_i} \right) g_{N-1} I_m \\ \hline & I_{(N-2)m} & 0 \end{bmatrix}, \end{split}$$

where $R_N(\lambda)$ is the rational matrix in (5.7). The last transformation makes zero the first N-2 blocks of size $m \times m$ in the first block row of $X_N(\lambda)$ and yields the announced matrix $S_N(\lambda)$ equivalent to $Z_N(\lambda)$ in \mathcal{P}_{N-1} . More precisely,

$$S_N(\lambda) := \begin{bmatrix} I_m & 0 & | & -\frac{g_N}{h_{N-1}} D_1 & \cdots & -\frac{g_N}{h_{N-1}} D_{N-2} \\ 0 & I_m & 0 & \cdots & 0 \\ \hline 0 & | & I_{(N-2)m} \end{bmatrix} X_N(\lambda)$$
$$= \begin{bmatrix} 0 & \cdots & 0 & | & R_N(\lambda) \\ 0 & \cdots & 0 & | & \left(\prod_{i=1}^{N-2} \frac{g_i}{h_i}\right) g_{N-1} I_m \\ \hline & I_{(N-2)m} & | & 0 \end{bmatrix}.$$

The block $H(\lambda) := \left(\prod_{i=1}^{N-2} \frac{g_i}{h_i}\right) g_{N-1} I_m$ of $S_N(\lambda)$ satisfies $H(\xi_k) = 0$ for all $\xi_k \in \mathcal{P}_{N-1}$. Therefore, $S_N(\xi_k)$ (and, so, $Z_N(\xi_k)$) has full column rank for every $\xi_k \in \mathcal{P}_{N-1}$ if and only if $R_N(\xi_k)$ is invertible for all $\xi_k \in \mathcal{P}_{N-1}$, and the result is proved.

The constant matrix $A_N(\lambda_0)$ is invertible for any $\lambda_0 \in \mathbb{F} \setminus \mathcal{P}_{N-1}$ and, so, $L_N(\lambda)$ is minimal in $\mathbb{F} \setminus \mathcal{P}_{N-1}$. Combining this with the fact that $Q_N(\lambda)$ and $\beta_0\left(1-\frac{\lambda}{\xi_N}\right)Q_N(\lambda)$ are equivalent in \mathbb{F} if $\xi_N = \infty$ or in $\mathbb{F} \setminus \{\xi_N\}$ if ξ_N is finite, we immediately obtain from Definition 4.2.1 that $L_N(\lambda)$ is a linearization of $Q_N(\lambda)$ with state matrix $A_N(\lambda)$ in $\mathbb{F} \setminus \mathcal{P}_N$, which is a result analogous to Theorem 5.1.1(a). This approach, of course, does not give any information on the finite poles of $Q_N(\lambda)$, because the finite eigenvalues of $A_N(\lambda)$ coincide with \mathcal{P}_{N-1} . Such information is obtained from the next result, which is the main result of this section and is a corollary of Lemma 5.1.2.

Theorem 5.1.3. Let $Q_N(\lambda)$ be the rational matrix in (5.3), $L_N(\lambda)$ be the pencil in (5.4), $A_N(\lambda)$ be the submatrix of $L_N(\lambda)$ in (5.6), and $R_N(\lambda)$ be the rational matrix in (5.7). Consider \mathcal{P}_{N-1} the set of finite poles in the list $(\xi_1, \xi_2, \ldots, \xi_{N-1})$, and assume $\xi_i \neq \sigma_j, 1 \leq i \leq N, 0 \leq j \leq N-1$. If $R_N(\xi_k) \in \mathbb{F}^{m \times m}$ is nonsingular for every $\xi_k \in \mathcal{P}_{N-1}$, then $L_N(\lambda)$ is a linearization of $Q_N(\lambda)$ with state matrix $A_N(\lambda)$ in \mathbb{F} , if $\xi_N = \infty$, or in $\mathbb{F} \setminus \{\xi_N\}$, if ξ_N is finite.

Proof. Under the hypotheses of Theorem 5.1.3, $L_N(\lambda)$ is minimal in \mathbb{F} . Moreover, its transfer function matrix, i.e., $\beta_0 \left(1 - \frac{\lambda}{\xi_N}\right) Q_N(\lambda)$ is equivalent to $Q_N(\lambda)$ in \mathbb{F} , if $\xi_N = \infty$, or in $\mathbb{F} \setminus \{\xi_N\}$, if ξ_N is finite.

We emphasize that the hypotheses that the constant matrices $R_N(\xi_k)$ in Theorem 5.1.3 are nonsingular are not mentioned at all in [47], but, fortunately, are generic, in the sense that they are satisfied by almost all regular rational matrices $Q_N(\lambda)$ expressed as in (5.3).

Remark 5.1.4. Under the conditions of Theorem 5.1.3, the pole elementary divisors of $Q_N(\lambda)$ in \mathbb{F} , if $\xi_N = \infty$, or in $\mathbb{F} \setminus \{\xi_N\}$, if ξ_N is finite, are the elementary divisors of $A_N(\lambda)$, as a consequence of Theorem 4.2.6. These elementary divisors can be easily determined as follows: first express $A_N(\lambda) = A_N(\lambda) \otimes I_m$; second note that if $\widehat{S}_N(\lambda)$ is the Smith form of $\widehat{A}_N(\lambda)$, then $\widehat{S}_N(\lambda) \otimes I_m$ is the Smith form of $A_N(\lambda)$; third, use the fact that $\xi_i \neq \sigma_j$, $1 \leq i \leq N$, $0 \leq j \leq N-1$, to prove that the greatest common divisor of all $(N-2) \times (N-2)$ minors of $\widehat{A}_N(\lambda)$ is equal to 1, which implies, according to [39, Ch. VI], that there is only one invariant polynomial of $\widehat{S}_N(\lambda)$ different from 1 and that is equal to $p(\lambda) = c (1 - \lambda/\xi_1) \cdots (1 - \lambda/\xi_{N-1}),$ where $c \in \mathbb{F}$ is a constant that makes $p(\lambda)$ monic. Finally, we get that $A_N(\lambda)$ has m invariant polynomials different from 1 all equal to $p(\lambda)$. This allows us to obtain easily the finite elementary divisors of $A_N(\lambda)$ and, thus, the finite pole elementary divisors of $Q_N(\lambda)$ (in \mathbb{F} if $\xi_N = \infty$, or in $\mathbb{F} \setminus \{\xi_N\}$ if ξ_N is finite). In particular, they are of the form $(\lambda - \xi_i)^{\nu_i}$ and, in order to obtain the partial multiplicities ν_i , we have to take into account possible repetitions in $(\xi_1, \ldots, \xi_{N-1})$. Observe that the infinite ξ_i for $i = 1, \ldots, N-1$ do not contribute at all to the finite pole elementary divisors of $Q_N(\lambda)$. Moreover, if $\xi_N = \infty$, then we can state the compact and simple result that the *m* denominators of the global Smith–McMillan form of $Q_N(\lambda)$ are all equal to $p(\lambda)$. However, with this choice of state matrix, there is no way of obtaining information on the pole structure of ξ_N when it is finite. This is the reason why, even if $L_N(\lambda)$ is minimal in \mathbb{F} , $L_N(\lambda)$ is not a linearization of $Q_N(\lambda)$ in \mathbb{F} .

5.1.1 The NLEIGS low rank problem

The second family of rational matrices considered in [47] comes from approximating NLEPs, $A(\lambda)x = 0$, such that the associated matrix $A(\lambda)$ is the sum of a polynomial matrix plus a matrix of the form $\sum_{i=1}^{n} C_i f_i(\lambda)$, where the constant matrices C_i have much smaller rank than the size of $A(\lambda)$ and $f_i(\lambda)$ are scalar nonlinear functions of λ . This type of NLEPs arise in several applications [46] and are approximated in [47, eq. (6.2)] by a family of rational matrices of the form

$$\widetilde{Q}_N(\lambda) = \sum_{i=0}^p b_i(\lambda) \, \widetilde{D}_i + \sum_{i=p+1}^N b_i(\lambda) \, \widetilde{L}_i \, \widetilde{U}^T \in \mathbb{F}(\lambda)^{m \times m}, \tag{5.11}$$

where $b_0(\lambda), \ldots, b_N(\lambda)$ are the scalar rational functions in (5.1), $\widetilde{D}_0, \ldots, \widetilde{D}_p \in \mathbb{F}^{m \times m}$, $\widetilde{L}_{p+1}, \ldots, \widetilde{L}_N \in \mathbb{F}^{m \times r}$ and $\widetilde{U} \in \mathbb{F}^{m \times r}$ are constant matrices, and $r \ll m$. For the functions in (5.2), let us consider the simpler notation $h_i := h_i(\lambda)$ and $g_i := g_i(\lambda)$.

5.1.1.1 The NLEIGS low rank problem from the point of view of block full rank pencils

In order to solve the REP $\tilde{Q}_N(\lambda)y = 0$ efficiently by taking advantage of the low rank structure of $\tilde{Q}_N(\lambda)$, the following pencil is introduced in [47, Sec. 6.4]:

$$\widetilde{L}_N(\lambda) = \begin{bmatrix} \widetilde{M}_N(\lambda) \\ \widetilde{K}_N(\lambda) \end{bmatrix}, \qquad (5.12)$$

where

$$\widetilde{M}_{N}(\lambda) := \begin{bmatrix} \frac{g_{N}}{\beta_{N}} \widetilde{D}_{0} & \frac{g_{N}}{\beta_{N}} \widetilde{D}_{1} & \cdots & \frac{g_{N}}{\beta_{N}} \widetilde{D}_{p} & \frac{g_{N}}{\beta_{N}} \widetilde{L}_{p+1} & \cdots & \frac{g_{N}}{\beta_{N}} \widetilde{L}_{N-2} & \frac{g_{N}}{\beta_{N}} \widetilde{L}_{N-1} + \frac{h_{N-1}}{\beta_{N}} \widetilde{L}_{N} \end{bmatrix},$$

$$\widetilde{K}_{N}(\lambda) := \begin{bmatrix} -h_{0}I_{m} & g_{1}I_{m} & & & & \\ & \ddots & \ddots & & \\ & & -h_{p-1}I_{m} & g_{p}I_{m} & & & \\ & & -h_{p}\widetilde{U}^{T} & g_{p+1}I_{r} & & & \\ & & & -h_{p+1}I_{r} & g_{p+2}I_{r} & & \\ & & & \ddots & \ddots & \\ & & & -h_{N-2}I_{r} & g_{N-1}I_{r} \end{bmatrix}$$

A result analogous to Theorem 5.1.1 can be proved for the pencil $\widetilde{L}_N(\lambda)$ and the matrix $\widetilde{Q}_N(\lambda)$. This is accomplished in Theorem 5.1.5. We remark, nevertheless, that the result concerning the linearizations at ∞ is weaker in Theorem 5.1.5 than in Theorem 5.1.1. This is an unavoidable consequence of the used approach and the low rank structure of $\widetilde{Q}_N(\lambda)$.

Theorem 5.1.5. Let $\widetilde{Q}_N(\lambda)$ be the rational matrix in (5.11) and $\widetilde{L}_N(\lambda)$ be the pencil in (5.12). Let \mathcal{P}_N and i_N be, respectively, the set of finite poles and the number of infinite poles in the list $(\xi_1, \xi_2, \ldots, \xi_N)$. Then, the following statements hold:

- (a) $\widetilde{L}_N(\lambda)$ partitioned as in (5.12) is a block full rank pencil with only one block column associated with $\widetilde{Q}_N(\lambda)$ in $\mathbb{F} \setminus \mathcal{P}_N$ and $\widetilde{L}_N(\lambda)$ is a linearization of $\widetilde{Q}_N(\lambda)$ in $\mathbb{F} \setminus \mathcal{P}_N$ with empty state matrix.
- (b) If, in addition, the poles $\xi_{p+1}, \xi_{p+2}, \ldots, \xi_{N-1}$ are all finite, then $\widetilde{L}_N(\lambda)$ is a linearization of $\widetilde{Q}_N(\lambda)$ at ∞ of grade i_N with empty state matrix.

Proof. The proof is similar to that of Theorem 5.1.1 with some differences coming from the presence of the low rank term in $\widetilde{Q}_N(\lambda)$. It is immediate to check that

$$\widetilde{N}_{N}(\lambda) = \frac{1}{1 - \frac{\lambda}{\xi_{N}}} \left[b_{0}(\lambda)I_{m} \cdots b_{p}(\lambda)I_{m} \quad b_{p+1}(\lambda)\widetilde{U} \cdots b_{N-1}(\lambda)\widetilde{U} \right]$$
(5.13)

is a rational basis dual to $\widetilde{K}_N(\lambda)$, that $\widetilde{K}_N(\lambda)$ and $\widetilde{N}_N(\lambda)$ have both full row rank in $\mathbb{F} \setminus \mathcal{P}_N$ and that $\widetilde{M}_N(\lambda)\widetilde{N}_N(\lambda)^T = \widetilde{Q}_N(\lambda)$. Thus, part (a) follows from Theorem 4.4.1.

In order to prove part (b), note first that $\operatorname{rev}_1 \widetilde{K}_N(\lambda)$ has full row rank at 0 as a consequence of the fact that the poles $\xi_{p+1}, \xi_{p+2}, \ldots, \xi_{N-1}$ are all finite. We now consider the rational matrix $\operatorname{rev}_{i_N-1} \widetilde{N}_N(\lambda) = \lambda^{i_N-1} \widetilde{N}_N(\frac{1}{\lambda})$, which is of the form

$$\lambda^{i_N-1}\widetilde{N}_N\left(\frac{1}{\lambda}\right) = \left[* \cdots * \frac{\lambda}{\lambda-1/\xi_N} \lambda^{i_N-1} b_p\left(\frac{1}{\lambda}\right) I_m * \cdots * \right],$$

where the entries * are defined at 0. Denote by i_p the number of infinite poles in the list $(\xi_1, \xi_2, \ldots, \xi_p)$. Then, $b_p\left(\frac{1}{\lambda}\right) = \frac{1}{\lambda^{i_p}}\tilde{c}(\lambda)$ for a certain rational function $\tilde{c}(\lambda)$ with $\tilde{c}(0) \neq 0$. Taking into account that the poles $\xi_{p+1}, \xi_{p+2}, \ldots, \xi_{N-1}$ are all finite, we have that $i_p = i_N$ if $\xi_N \neq \infty$, and $i_p = i_N - 1$ if $\xi_N = \infty$. Therefore, $\operatorname{rev}_{i_N-1} \widetilde{N}_N(\lambda)$ has full row rank at 0 because $\tilde{c}(0) \neq 0$. Thus, part (b) follows from Theorem 4.3.7.

A discussion similar to the one in the last paragraph of Section 5.1.0.1 can be developed on the basis of Theorem 5.1.5. The details are omitted for brevity. The open problem corresponding to the information of the finite poles will be solved in Section 5.1.1.2.

5.1.1.2 The NLEIGS low rank problem from the point of view of polynomial system matrices

The results in this section are the counterpart for $\widetilde{Q}_N(\lambda)$ in (5.11) and $\widetilde{L}_N(\lambda)$ in (5.12) of those presented in Section 5.1.0.2 for $Q_N(\lambda)$ and $L_N(\lambda)$. For this purpose, we consider the following partition of $\widetilde{L}_N(\lambda)$ in (5.12):

$$\begin{bmatrix} \frac{g_N}{\beta_N} \widetilde{D}_0 & \frac{g_N}{\beta_N} \widetilde{D}_1 & \cdots & \frac{g_N}{\beta_N} \widetilde{D}_p & \frac{g_N}{\beta_N} \widetilde{L}_{p+1} & \cdots & \frac{g_N}{\beta_N} \widetilde{L}_{N-2} & \frac{g_N}{\beta_N} \widetilde{L}_{N-1} + \frac{h_{N-1}}{\beta_N} \widetilde{L}_N \\ -h_0 I_m & g_1 I_m & & & & & \\ & \ddots & \ddots & & & & \\ & & -h_p \widetilde{U}^T & g_{p+1} I_r & & & \\ & & -h_{p+1} I_r & g_{p+2} I_r & & & \\ & & \ddots & \ddots & & \\ & & & -h_{N-2} I_r & g_{N-1} I_r \end{bmatrix} \\ =: \begin{bmatrix} \frac{\widetilde{M}(\lambda) & -\widetilde{C}_N(\lambda)}{\widetilde{K}(\lambda) & 0} \\ \frac{\widetilde{K}(\lambda) & 0}{\widetilde{B}_N(\lambda) & \widetilde{A}_N(\lambda)} \end{bmatrix},$$

$$(5.14)$$

where $\widetilde{A}_N(\lambda)$ will play the role of the state matrix, and the rational basis

$$\widetilde{N}(\lambda) := \frac{1}{1 - \frac{\lambda}{\xi_N}} \begin{bmatrix} b_0(\lambda) & b_1(\lambda) & \cdots & b_p(\lambda) \end{bmatrix} \otimes I_m$$
(5.15)

dual to $\widetilde{K}(\lambda)$. The next lemma is the counterpart of Lemma 5.1.2.

Lemma 5.1.6. Let us consider the pencil $\widetilde{L}_N(\lambda)$ in (5.12) as a polynomial system matrix with state matrix $\widetilde{A}_N(\lambda)$, where $\widetilde{A}_N(\lambda)$ is defined through the partition (5.14), and let $\widetilde{Q}_N(\lambda)$ be the rational matrix in (5.11). Then the following statements hold:

- (a) $\widetilde{L}_N(\lambda)$ is a block full rank linearization of $\widetilde{Q}_N(\lambda)$ in the sets where simultaneously $\widetilde{L}_N(\lambda)$ is minimal and $\widetilde{N}(\lambda)$ has full row rank.
- (b) Let us define the rational matrix

$$\widetilde{R}_N(\lambda) = \sum_{j=p+1}^{N-1} \left(\prod_{k=j+1}^N \frac{g_k(\lambda)}{h_{k-1}(\lambda)} \right) \widetilde{L}_j + \widetilde{L}_N \in \mathbb{F}(\lambda)^{m \times r},$$
(5.16)

let $\widetilde{\mathcal{P}}_{N-1}$ be the set of finite poles in the list $(\xi_{p+1}, \xi_{p+2}, \dots, \xi_{N-1})$. Assume that rank $\widetilde{U} = r$ and $\xi_i \neq \sigma_j$, $1 \leq i \leq N$, $0 \leq j \leq N-1$. Then, $\widetilde{L}_N(\lambda)$ is minimal in \mathbb{F} if and only if the matrix $\widetilde{R}_N(\xi_k) \in \mathbb{F}^{m \times r}$ has full column rank for all $\xi_k \in \widetilde{\mathcal{P}}_{N-1}$.

Proof. For part (a), we only have to check that

$$[\widetilde{M}(\lambda) + \widetilde{C}_N(\lambda)\widetilde{A}_N(\lambda)^{-1}\widetilde{B}_N(\lambda)]\widetilde{N}(\lambda)^T = \widetilde{Q}_N(\lambda).$$

For that, we take into account that the first block column of $\widetilde{A}_N(\lambda)^{-1}$ is equal to $\frac{1}{b_{p+1}(\lambda)g_{p+1}} \begin{bmatrix} b_{p+1}(\lambda) & \cdots & b_{N-1}(\lambda) \end{bmatrix}^T \otimes I_r$. The proof of part (b) is analogous to that of Lemma 5.1.2(b).

As a corollary of Lemma 5.1.6, we state Theorem 5.1.7 which is the main result in this section.

Theorem 5.1.7. Let $\widetilde{Q}_N(\lambda)$ be the rational matrix in (5.11), $\widetilde{L}_N(\lambda)$ be the pencil in (5.12), $\widetilde{A}_N(\lambda)$ be the submatrix of $\widetilde{L}_N(\lambda)$ in (5.14), and $\widetilde{R}_N(\lambda)$ be the rational matrix in (5.16). Consider $\widetilde{\mathcal{P}}_{N-1}$ the set of finite poles in the list $(\xi_{p+1}, \xi_2, \ldots, \xi_{N-1})$. If $\xi_1 = \cdots = \xi_p = \infty$, rank $\widetilde{U} = r$, $\xi_i \neq \sigma_j$, $1 \leq i \leq N$, $0 \leq j \leq N-1$, and $\widetilde{R}_N(\xi_k) \in \mathbb{F}^{m \times r}$ has full column rank for every $\xi_k \in \widetilde{\mathcal{P}}_{N-1}$, then $\widetilde{L}_N(\lambda)$ is a block full rank linearization of $\widetilde{Q}_N(\lambda)$ with state matrix $\widetilde{A}_N(\lambda)$ in \mathbb{F} , if $\xi_N = \infty$, or in $\mathbb{F} \setminus \{\xi_N\}$, if ξ_N is finite. *Proof.* First, notice that if $\xi_1 = \cdots = \xi_p = \infty$ then the dual rational basis $\widetilde{N}(\lambda)$ in (5.15) has full row rank in \mathbb{F} , if $\xi_N = \infty$, or in $\mathbb{F} \setminus \{\xi_N\}$, if ξ_N is finite. In addition, the rank condition (4.15) is satisfied in \mathbb{F} if $\widetilde{L}_N(\lambda)$ is minimal in \mathbb{F} and $\xi_1 = \cdots = \xi_p = \infty$. By Lemma 5.1.6 and Theorem 4.4.1 the results follows.

Remark 5.1.8. Note that the hypothesis $\xi_1 = \cdots = \xi_p = \infty$ implies that the "nolow rank" term $\sum_{i=0}^{p} b_i(\lambda) \widetilde{D}_i$ of $\widetilde{Q}_N(\lambda)$ is a polynomial matrix, as often happens in NLEPs [47].

5.2 Linearizations for AAA rational approximations of NLEPs

In this section we study the precise properties of the linearizations for rational approximations of NLEPs in [60]. We will see that they satisfy Theorem 4.4.1 in a particular subset of \mathbb{F} and in the whole field \mathbb{F} under mild conditions.

An approach to obtain an automatic rational approximation for the NLEP in a region Ω is given in [60]. The authors of [60] consider the $n \times n$ nonlinear matrix $F(\lambda)$ of the NLEP written in the form

$$F(\lambda) = Q(\lambda) + \sum_{i=1}^{s} (C_i - \lambda D_i) g_i(\lambda)$$
(5.17)

with $Q(\lambda)$ a polynomial matrix, C_i and D_i $n \times n$ constant matrices, and $g_i(\lambda)$ nonlinear scalar functions. Then, a CORK linearization of $Q(\lambda)$ of those introduced in [81] is considered, and the functions $g_i(\lambda)$ are approximated by rational functions employing the adaptive Antoulas–Anderson (AAA) algorithm [73], or its set-valued generalization presented in [60]. We recall the definition of CORK linearization as given in [60].

Definition 5.2.1. Let $G(\lambda)$ be an $n \times n$ rational matrix written as

$$G(\lambda) = \sum_{i=0}^{k-1} (A_i - \lambda B_i) f_i(\lambda), \qquad (5.18)$$

where $f_i(\lambda)$ are scalar rational functions with $f_0(\lambda) \equiv 1$, and A_i , B_i are $n \times n$ constant matrices. Define

$$f(\lambda) := [f_0(\lambda) \quad \cdots \quad f_{k-1}(\lambda)]^T,$$

and assume that the rational functions $f_i(\lambda)$ satisfy a linear relation

$$(X - \lambda Y)f(\lambda) = 0, \qquad (5.19)$$

where $\operatorname{rank}(X - \lambda Y) = k - 1$ for all $\lambda \in \mathbb{F}$, and $X - \lambda Y$ has size $(k - 1) \times k$. Then the matrix pencil

$$\mathcal{L}_G(\lambda) = \begin{bmatrix} A_0 - \lambda B_0 & \cdots & A_{k-1} - \lambda B_{k-1} \\ (X - \lambda Y) \otimes I_n \end{bmatrix}$$

is called a CORK linearization of $G(\lambda)$.

If the vector $f(\lambda)$ is polynomial then $G(\lambda)$ in (5.18) is a polynomial matrix and $\mathcal{L}_G(\lambda)$ is a linearization of $G(\lambda)$ in \mathbb{F} , in particular, $\mathcal{L}_G(\lambda)$ is a block full rank linearization of $G(\lambda)$ in \mathbb{F} with empty state matrix. However, if $f(\lambda)$ is a rational vector then $\mathcal{L}_G(\lambda)$ is not, in general, a linearization in the sense of [7], that is, in \mathbb{F} but it is a linearization in a local sense [28]. More precisely, it is a block full rank linearization in all the subsets where the rational vector $f(\lambda)$ is defined, i.e., has no poles. Such result is stated in the next theorem.

Theorem 5.2.2. Let Ω be a nonempty subset of \mathbb{F} where the rational vector $f(\lambda)$ in (5.19) is defined. Then a CORK linearization $\mathcal{L}_G(\lambda)$ of a rational matrix $G(\lambda)$ as in (5.18) is a block full rank linearization of $G(\lambda)$ in Ω with empty state matrix.

Proof. Notice that, by taking $M(\lambda) := \begin{bmatrix} A_0 - \lambda B_0 & \cdots & A_{k-1} - \lambda B_{k-1} \end{bmatrix}$, $K_1(\lambda) := (X - \lambda Y) \otimes I_n$, and $K_2(\lambda)$ empty, $\mathcal{L}_G(\lambda)$ is a block full rank pencil. Moreover, $f(\lambda)^T$ is a rational basis dual to $X - \lambda Y$. Then we apply Remark 4.4.2 with $N_1(\lambda) := f(\lambda)^T \otimes I_n$ and $N_2(\lambda) = I_n$.

Once CORK linearizations and some of their properties have been revised, we recall the AAA approximation of scalar functions. A given nonlinear function $g : \mathbb{F} \longrightarrow \mathbb{F}$ is approximated in [60] on a set $\Sigma \subset \mathbb{F}$ by a rational function $r(\lambda)$ in barycentric form, that is,

$$r(\lambda) = \sum_{j=1}^{m} \frac{g(z_j)w_j}{\lambda - z_j} \Big/ \sum_{j=1}^{m} \frac{w_j}{\lambda - z_j},$$
(5.20)

where z_1, \ldots, z_m are distinct support points and w_1, \ldots, w_m are nonzero weights, that can be automatically chosen as explained in [73]. In this case, $\lim_{\lambda \to z_j} r(\lambda) = g(z_j)$. It is shown in [60, Proposition 2.1] that $r(\lambda)$ in (5.20) can be written as

$$[g(z_1)w_1 \cdots g(z_m)w_m] \begin{bmatrix} w_1 & w_2 & \cdots & w_{m-1} & w_m \\ \lambda - z_1 & z_2 - \lambda & & & \\ & \lambda - z_2 & \ddots & & \\ & & \ddots & z_{m-1} - \lambda \\ & & & \lambda - z_{m-1} & z_m - \lambda \end{bmatrix}^{-1} \begin{bmatrix} 1 \\ 0 \\ \vdots \\ 0 \end{bmatrix}.$$

That is, $r(\lambda)$ can be written as a generalized state-space realization. Then, the pencil

$$P(\lambda) := \begin{bmatrix} w_1 & w_2 & \cdots & w_{m-1} & w_m & | & -1 \\ \lambda - z_1 & z_2 - \lambda & & & 0 \\ & \lambda - z_2 & \ddots & & & \vdots \\ & & \ddots & z_{m-1} - \lambda & & \vdots \\ & & & \lambda - z_{m-1} & z_m - \lambda & 0 \\ \hline g(z_1)w_1 & g(z_2)w_2 & \cdots & g(z_{m-1})w_{m-1} & g(z_m)w_m & 0 \end{bmatrix} := \begin{bmatrix} E - \lambda F & | & -b \\ \hline a^T & | & 0 \end{bmatrix}$$
(5.21)

is a linear polynomial system matrix of $r(\lambda)$ (i.e., with transfer function $r(\lambda)$) by considering $E - \lambda F$ as state matrix. In order to know what pole and zero information of $r(\lambda)$ we can obtain from this realization, we consider in Proposition 5.2.4 the polynomial system matrix $P(\lambda)$ and we study its minimality. First, we prove Lemma 5.2.3. In both Proposition 5.2.4 and Lemma 5.2.3, we consider $r(\lambda)$ written as the following quotient of polynomials

$$r(\lambda) := \frac{p(\lambda)}{q(\lambda)},\tag{5.22}$$

where

$$p(\lambda) := \left(\prod_{j=1}^{m} (\lambda - z_j)\right) \left(\sum_{j=1}^{m} \frac{g(z_j)w_j}{\lambda - z_j}\right) \text{ and } q(\lambda) := \left(\prod_{j=1}^{m} (\lambda - z_j)\right) \left(\sum_{j=1}^{m} \frac{w_j}{\lambda - z_j}\right).$$

Note that the representation of the rational function (5.22) might not be irreducible. We will see that the irreducibility of (5.22) is the key property for the minimality of $P(\lambda)$.

Lemma 5.2.3. The pencil $E - \lambda F$ in (5.21) is a strong block minimal bases pencil associated with the polynomial $q(\lambda)$ in (5.22).

Proof. We set

$$E - \lambda F = \begin{bmatrix} \frac{w_1 & w_2 & \cdots & w_{m-1} & w_m}{\lambda - z_1 & z_2 - \lambda} & & \\ & \lambda - z_2 & \ddots & & \\ & & \ddots & z_{m-1} - \lambda & \\ & & & \lambda - z_{m-1} & z_m - \lambda \end{bmatrix} = :\begin{bmatrix} M \\ K(\lambda) \end{bmatrix}.$$
(5.23)

Since z_1, \ldots, z_m are distinct, $K(\lambda_0)$ has full row rank for all $\lambda_0 \in \mathbb{F}$. In addition,

 $K(\lambda)$ is row reduced because its highest row degree coefficient matrix

$$K_{hr} = \begin{bmatrix} 1 & -1 & 0 & & \\ & 1 & -1 & 0 & & \\ & & \ddots & \ddots & \ddots & \\ & & & 1 & -1 & 0 \\ & & & & 1 & -1 \end{bmatrix}$$

has full row rank. We conclude that $K(\lambda)$ is a minimal basis. Let us denote

$$N(\lambda) := \prod_{j=1}^{m} (\lambda - z_j) \begin{bmatrix} \frac{1}{\lambda - z_1} & \cdots & \frac{1}{\lambda - z_m} \end{bmatrix}.$$
 (5.24)

Then, it is not difficult to prove that $N(\lambda)$ is also a minimal basis, taking again into account that z_1, \ldots, z_m are distinct. Moreover, since $K(\lambda)N(\lambda)^T = 0$ and $\begin{bmatrix} K(\lambda) \\ N(\lambda) \end{bmatrix}$ is a square matrix, we have that $K(\lambda)$ and $N(\lambda)$ are dual minimal bases. In addition, all the row degrees of $K(\lambda)$ are equal to 1 and the row degree of $N(\lambda)$ is equal to m-1. Hence, $E - \lambda F$ is a strong block minimal bases pencil associated with the polynomial matrix $MN(\lambda)^T = q(\lambda)$.

 $E - \lambda F$ being a strong block minimal bases pencil associated with the polynomial $q(\lambda)$ implies that $E - \lambda F$ is a (strong) linearization of $q(\lambda)$ and, in particular, that the determinant of $E - \lambda F$ is equal to $q(\lambda)$ up to a scalar multiple. This fact is used to prove the following result.

Proposition 5.2.4. Consider the rational function $r(\lambda)$ in (5.22) and its linear polynomial system matrix $P(\lambda)$ in (5.21). Then, $P(\lambda)$ is not minimal at $\lambda_0 \in \mathbb{F}$ if and only if λ_0 is a zero of both polynomials $p(\lambda)$ and $q(\lambda)$.

Proof. Consider
$$P(\lambda) = \begin{bmatrix} E - \lambda F & | & -b \\ \hline a^T & | & 0 \end{bmatrix}$$
 as in (5.21). By Lemma 5.2.3,
$$\det(E - \lambda F) = \alpha q(\lambda) \text{ with } \alpha \neq 0.$$
(5.25)

In addition, since the Schur complement of $E - \lambda F$ in $P(\lambda)$ is $r(\lambda)$, we have that

$$\det(P(\lambda)) = \det(E - \lambda F)r(\lambda) = \alpha q(\lambda)\frac{p(\lambda)}{q(\lambda)} = \alpha p(\lambda).$$
(5.26)

Now, assume that λ_0 is a zero of both polynomials $p(\lambda)$ and $q(\lambda)$. That is, we can cancel out at least one factor of the form $(\lambda - \lambda_0)$ in both numerator and denominator of $r(\lambda)$. Then, the algebraic multiplicity of λ_0 as a zero of $r(\lambda)$ is not the same as the algebraic multiplicity of λ_0 as a zero of $P(\lambda)$. Therefore,

 $P(\lambda)$ is not minimal at λ_0 . For the converse, assume that $P(\lambda)$ is not minimal at λ_0 . Then, rank $\begin{bmatrix} E - \lambda_0 F \\ a^T \end{bmatrix} < m$, since rank $\begin{bmatrix} E - \lambda_0 F \\ -b \end{bmatrix} = m$. Actually, rank $\begin{bmatrix} E - \lambda_0 F \\ a^T \end{bmatrix} = m - 1$, as the sub-matrix

$$K(\lambda_0) = \begin{bmatrix} \lambda_0 - z_1 & z_2 - \lambda_0 & & & \\ & \lambda_0 - z_2 & \ddots & & \\ & & \ddots & z_{m-1} - \lambda_0 & \\ & & & \lambda_0 - z_{m-1} & z_m - \lambda_0 \end{bmatrix}$$

contains a non-zero minor of order m-1 for all $\lambda_0 \in \mathbb{F}$, where $K(\lambda)$ is the matrix appearing in (5.23). Therefore, by using the notation $\mathcal{N}_r(\cdot)$ for the right nullspace, $\dim \mathcal{N}_r\left(\begin{bmatrix} E - \lambda_0 F \\ a^T \end{bmatrix}\right) = 1$ and $\dim \mathcal{N}_r(K(\lambda_0)) = 1$. Then, $\mathcal{N}_r\left(\begin{bmatrix} E - \lambda_0 F \\ a^T \end{bmatrix}\right) =$ $\mathcal{N}_r(K(\lambda_0))$, since $\mathcal{N}_r\left(\begin{bmatrix} E - \lambda_0 F \\ a^T \end{bmatrix}\right) \subseteq \mathcal{N}_r(K(\lambda_0))$ and both have the same dimension. Actually, $\mathcal{N}_r(K(\lambda_0)) = \operatorname{Span}\{N(\lambda_0)^T\}$, where $N(\lambda)$ is the polynomial matrix in (5.24). Hence, $\begin{bmatrix} E - \lambda_0 F \\ a^T \end{bmatrix} N(\lambda_0)^T = 0$ and, therefore, $\begin{bmatrix} w_1 & w_2 & \cdots & w_m \end{bmatrix} N(\lambda_0)^T =$ 0 and $\begin{bmatrix} g(z_1)w_1 & g(z_2)w_2 & \cdots & g(z_m)w_m \end{bmatrix} N(\lambda_0)^T = 0$. That is, λ_0 is a root of both $q(\lambda)$ and $p(\lambda)$.

With these tools at hand, we go back to the original NLEP. Let $F(\lambda)$ be the nonlinear matrix function in (5.17). Then, each function $g_i(\lambda)$ is approximated in [60] on a set $\Sigma \subset \mathbb{F}$ by a rational function $r_i(\lambda)$ as in (5.20), i.e.,

$$g_i(\lambda) \approx r_i(\lambda) = \sum_{j=1}^{\ell_i} \frac{g_i(z_j^i)w_j^i}{\lambda - z_j^i} \Big/ \sum_{j=1}^{\ell_i} \frac{w_j^i}{\lambda - z_j^i},$$

where ℓ_i is the number of support points z_j^i and weights w_j^i for each $i = 1, \ldots, s$. For that, one can use the AAA algorithm on each function $g_i(\lambda)$ separately [73], or one can use the set-valued AAA algorithm in [60, Section 2.2], so that the rational approximants $r_i(\lambda)$ are all constructed simultaneously and sharing the same support points $z_j^i := z_j$ and weights $w_j^i := w_j$. By using any of the two approaches above, the following approximation of $F(\lambda)$ on Σ is obtained:

$$F(\lambda) \approx R(\lambda) := Q(\lambda) + \sum_{i=1}^{s} (C_i - \lambda D_i) r_i(\lambda).$$
(5.27)

Next, the polynomial matrix $Q(\lambda)$ is expressed in the form of (5.18), i.e., $Q(\lambda) := \sum_{i=0}^{k-1} (A_i - \lambda B_i) f_i(\lambda)$, assuming the functions $f_i(\lambda)$ are polynomials, with $f_0(\lambda) = 1$, and each $r_i(\lambda)$ is written in generalized state-space form, that is,

$$R(\lambda) = \sum_{i=0}^{k-1} (A_i - \lambda B_i) f_i(\lambda) + \sum_{i=1}^{s} (C_i - \lambda D_i) a_i^T (E_i - \lambda F_i)^{-1} b_i,$$
(5.28)

with $a_i = \begin{bmatrix} g_i(z_1^i)w_1^i & \cdots & g_i(z_{\ell_i}^i)w_{\ell_i}^i \end{bmatrix}^T \in \mathbb{F}^{\ell_i}, b_i = \begin{bmatrix} 1 & 0 & \cdots & 0 \end{bmatrix}^T \in \mathbb{F}^{\ell_i} \text{ and } \ell_i \times \ell_i$ matrices

$$E_{i} = \begin{bmatrix} w_{1}^{i} & w_{2}^{i} & \cdots & w_{\ell_{i}-1}^{i} & w_{\ell_{i}}^{i} \\ -z_{1}^{i} & z_{2}^{i} & & & \\ & -z_{2}^{i} & \ddots & & \\ & & \ddots & z_{\ell_{i}-1}^{i} & \\ & & & -z_{\ell_{i}-1}^{i} & z_{\ell_{i}}^{i} \end{bmatrix} \text{ and } F_{i} = \begin{bmatrix} 0 & 0 & \cdots & 0 & 0 \\ -1 & 1 & & & \\ & -1 & \ddots & & \\ & & -1 & \ddots & \\ & & & \ddots & 1 \\ & & & & -1 & 1 \end{bmatrix}.$$

The linearization constructed in [60] for $R(\lambda)$ is the following.

Definition 5.2.5. [60, Definition 3.2](CORK linearization for AAA rational approximation) Let $R(\lambda)$ be a rational matrix as in (5.28). Consider $b := [b_1^T \cdots b_s^T]^T$ and $E - \lambda F := \text{diag}(E_1 - \lambda F_1, \dots, E_s - \lambda F_s)$. Then a CORK linearization for $R(\lambda)$ is

$$\mathcal{L}_{R}(\lambda) = \begin{bmatrix} A_{0} - \lambda B_{0} & \cdots & A_{k-1} - \lambda B_{k-1} & a_{1}^{T} \otimes (C_{1} - \lambda D_{1}) & \cdots & a_{s}^{T} \otimes (C_{s} - \lambda D_{s}) \\ \hline (X - \lambda Y) \otimes I_{n} & 0 & \\ \hline -b \otimes I_{n} & 0 & (E - \lambda F) \otimes I_{n} \end{bmatrix}$$
where $\begin{bmatrix} A_{0} - \lambda B_{0} & \cdots & A_{k-1} - \lambda B_{k-1} \\ \hline (X - \lambda Y) \otimes I_{n} \end{bmatrix}$ is any CORK linearization of $Q(\lambda)$.

In particular, for the set-valued AAA approximation, all the matrices $E_i - \lambda F_i$ in (5.28) are the same for all *i*, as well as all the vectors b_i . Then, in [60, Remark 3.3] the following CORK linearization for AAA rational approximation is considered

$$\mathcal{L}_{R}^{sv}(\lambda) = \begin{bmatrix} A_{0} - \lambda B_{0} & \cdots & A_{k-1} - \lambda B_{k-1} \\ \vdots \\ (X - \lambda Y) \otimes I_{n} & 0 \\ \hline -b_{1} \otimes I_{n} & 0 & (E_{1} - \lambda F_{1}) \otimes I_{n} \end{bmatrix}$$
(5.29)

Notice that $\mathcal{L}_{R}^{sv}(\lambda)$ has size $(kn + \ell_{1}n) \times (kn + \ell_{1}n)$ whereas $\mathcal{L}_{R}(\lambda)$ in Definition 5.2.5 has size $(kn + \sum_{i=1}^{s} \ell_{i}n) \times (kn + \sum_{i=1}^{s} \ell_{i}n)$.

By using Theorem 4.4.1, we study in Theorem 5.2.6 the structure of $\mathcal{L}_R(\lambda)$ and $\mathcal{L}_R^{sv}(\lambda)$ as linearizations of $R(\lambda)$.

Theorem 5.2.6. Let $R(\lambda)$ be a rational matrix as in (5.28), and let $\mathcal{L}_R(\lambda)$ (resp., $\mathcal{L}_R^{sv}(\lambda)$) be the matrix pencil in Definition 5.2.5 (resp., in (5.29)). Let $\Omega \subseteq \mathbb{F}$ be nonempty. If $\mathcal{L}_R(\lambda)$ (resp., $\mathcal{L}_R^{sv}(\lambda)$), viewed as a polynomial system matrix with state matrix $(E - \lambda F) \otimes I_n$ (resp., $(E_1 - \lambda F_1) \otimes I_n$), is minimal in Ω then $\mathcal{L}_R(\lambda)$ (resp., $\mathcal{L}_R^{sv}(\lambda)$) is a block full rank linearization of $R(\lambda)$ in Ω with state matrix $(E - \lambda F) \otimes I_n$ (resp., $(E_1 - \lambda F_1) \otimes I_n$).

Proof. Set $M(\lambda) := \begin{bmatrix} A_0 - \lambda B_0 & \cdots & A_{k-1} - \lambda B_{k-1} \end{bmatrix}$, $C(\lambda) := -\begin{bmatrix} a_1^T \otimes (C_1 - \lambda D_1) & \cdots & a_s^T \otimes (C_s - \lambda D_s) \end{bmatrix}$, $B := \begin{bmatrix} -b \otimes I_n & 0 \end{bmatrix}$, $A(\lambda) := (E - \lambda F) \otimes I_n$, $K_1(\lambda) := (X - \lambda Y) \otimes I_n$, $N_1(\lambda) := (f(\lambda) \otimes I_n)^T$, and $K_2(\lambda)$ empty. $\mathcal{L}_R(\lambda)$ being minimal in Ω implies that condition (4.15) is satisfied in Ω since $BN_1(\lambda)^T = -b \otimes I_n$ because $f_0(\lambda) = 1$. Then, by Theorem 4.4.1, $\mathcal{L}_R(\lambda)$ is a linearization of $[M(\lambda) + C(\lambda)A(\lambda)^{-1}B](f(\lambda) \otimes I_n) = R(\lambda)$ in Ω with state matrix $A(\lambda)$.

Remark 5.2.7. Theorem 5.2.6 also holds if $f(\lambda)$ is rational but, in such a case, we need the extra hypothesis of $f(\lambda)$ being defined in Ω .

According to Theorem 5.2.6, we need minimality on $\mathcal{L}_R(\lambda)$ to be a linearization of the rational matrix $R(\lambda)$. In the following Theorem 5.2.8, we give sufficient mild conditions for $\mathcal{L}_R(\lambda)$ to be minimal in \mathbb{F} in the case the rational approximants $r_i(\lambda)$ do not share the same support points and weights. That is, when the functions $g_i(\lambda)$ are approximated employing the adaptive Antoulas–Anderson (AAA) algorithm in [73] separately. For the set-valued AAA approximation [60], i.e., the rational approximants $r_i(\lambda)$ sharing the same support points and weights, the authors in [60] consider the pencil in (5.29), and we state minimality conditions for it in Theorem 5.2.9.

Theorem 5.2.8. Assume that, for i = 1, ..., s, the rational functions $r_i(\lambda)$ in (5.27) are represented as in (5.22) and that this representation is irreducible. Let $\mathcal{L}_R(\lambda)$ be the matrix pencil in Definition 5.2.5. If the pencils $C_i - \lambda D_i$ and $E_i - \lambda F_i$ are regular for i = 1, ..., s and the following conditions hold

- (a) $C_i \lambda D_i$ and $E_i \lambda F_i$ have no finite eigenvalues in common for i = 1, ..., s, and
- (b) $E_i \lambda F_i$ and $E_j \lambda F_j$ with $i \neq j$ have no finite eigenvalues in common for $i, j = 1, \ldots, s$,

then $\mathcal{L}_R(\lambda)$, viewed as a polynomial system matrix with state matrix $(E - \lambda F) \otimes I_n$, is minimal in \mathbb{F} .

Proof. Assume first that s = 1. Then notice that $\mathcal{L}_R(\lambda)$ is minimal in \mathbb{C} if the pencil

$$S(\lambda) := \begin{bmatrix} 0 & a_1^T \otimes (C_1 - \lambda D_1) \\ -b_1 \otimes I_n & (E_1 - \lambda F_1) \otimes I_n \end{bmatrix},$$

considered as a polynomial system matrix with state matrix $(E_1 - \lambda F_1) \otimes I_n$, is minimal in \mathbb{F} . Since $r_1(\lambda)$ is irreducible, we have, by Proposition 5.2.4, that the submatrix $\begin{bmatrix} -b_1 \otimes I_n & | & (E_1 - \lambda F_1) \otimes I_n \end{bmatrix}$ has full row rank for all $\lambda \in \mathbb{F}$. Then we only have to prove that the submatrix $H(\lambda) := \begin{bmatrix} a_1^T \otimes (C_1 - \lambda D_1) \\ \hline (E_1 - \lambda F_1) \otimes I_n \end{bmatrix}$ has full column rank for all $\lambda \in \mathbb{F}$. By contradiction, assume that $H(\lambda_0)$ has no full column rank for some $\lambda_0 \in \mathbb{F}$. Notice that, in such a case, λ_0 must be an eigenvalue of $E_1 - \lambda F_1$ since, otherwise, $H(\lambda_0)$ would have full column rank. In addition, there exists a nonzero vector x such that $H(\lambda_0)x = 0$. Now we write

$$H(\lambda_0)x = \begin{bmatrix} C_1 - \lambda_0 D_1 & 0\\ 0 & I_{\ell_1 n} \end{bmatrix} \begin{bmatrix} a_1^T \otimes I_n \\ (E_1 - \lambda_0 F_1) \otimes I_n \end{bmatrix} x = 0,$$
(5.30)

and define the vector $\begin{bmatrix} y_1 \\ y_2 \end{bmatrix} := \begin{bmatrix} a_1^T \otimes I_n \\ (E_1 - \lambda_0 F_1) \otimes I_n \end{bmatrix} x$, which is nonzero since $x \neq 0$ and the matrix $\begin{bmatrix} a_1^T \otimes I_n \\ (E_1 - \lambda_0 F_1) \otimes I_n \end{bmatrix}$ has full column rank by Proposition 5.2.4.

Moreover, by (5.30), we have that $y_2 = 0$ and, thus, $(C_1 - \lambda_0 D_1)y_1 = 0$ with $y_1 \neq 0$. Therefore, λ_0 is an eigenvalue of $C_1 - \lambda D_1$, which is a contradiction by condition (a). Finally, if s > 1 we have to take into account condition (b) and the result follows.

For the set-valued AAA approximation, the minimality conditions on $\mathcal{L}_{R}^{sv}(\lambda)$ are milder. We state them in the following result that is a corollary of Theorem 5.2.8.

Theorem 5.2.9. Assume that, for i = 1, ..., s, the rational functions $r_i(\lambda)$ in (5.27) are represented as in (5.22) and that this representation is irreducible. Let $\mathcal{L}_R^{sv}(\lambda)$

be the matrix pencil in (5.29). If the pencils $\sum_{i=1}^{s} a_i^T \otimes (C_i - \lambda D_i)$ and $E_1 - \lambda F_1$ are regular and have no finite eigenvalues in common then $\mathcal{L}_R^{sv}(\lambda)$, viewed as a polynomial system matrix with state matrix $(E_1 - \lambda F_1) \otimes I_n$, is minimal in \mathbb{F} .

Remark 5.2.10. It is clear that if we consider the set $\Omega := \{\lambda \in \mathbb{F} : E - \lambda F \text{ is invertible}\}$, then $\mathcal{L}_R(\lambda)$ is minimal in Ω and, by Theorem 5.2.6, $\mathcal{L}_R(\lambda)$ is a block full rank linearization of $R(\lambda)$ in Ω , with state matrix $(E - \lambda F) \otimes I_n$. However, in such a case we do not obtain any information about the poles of $R(\lambda)$ since they do not belong to Ω . For this particular choice of the set Ω , the fact that $\mathcal{L}_R(\lambda)$ is a linearization of $R(\lambda)$ in Ω can also be proved by considering $\mathcal{L}_R(\lambda)$ as a block full rank pencil of the form

$$\mathcal{L}_R(\lambda) := \left[\frac{M(\lambda)}{K_1(\lambda)} \right],$$

with

$$M(\lambda) := \begin{bmatrix} A_0 - \lambda B_0 & \cdots & A_{k-1} - \lambda B_{k-1} & a_1^T \otimes (C_1 - \lambda D_1) & \cdots & a_s^T \otimes (C_s - \lambda D_s) \end{bmatrix},$$

and by applying Remark 4.4.2. For that, write $R(\lambda)$ as

$$R(\lambda) = \sum_{i=0}^{k-1} (A_i - \lambda B_i) (f_i(\lambda) \otimes I_n) + \sum_{i=1}^s [a_i^T \otimes (C_i - \lambda D_i)] (R_i(\lambda) \otimes I_n),$$

with $R_i(\lambda) := (E_i - \lambda F_i)^{-1} b_i$, and consider the dual rational basis of $K_1(\lambda)$

$$N_1(\lambda) := [f_0(\lambda) \quad \cdots \quad f_{k-1}(\lambda) \quad | \quad R_1(\lambda)^T \quad \cdots \quad R_s(\lambda)^T] \otimes I_n.$$

Then, $\mathcal{L}_R(\lambda)$ is a linearization of $R(\lambda)$ in Ω with empty state matrix. On the other hand, if $\mathcal{L}_R(\lambda)$ (considering the partition with state matrix $(E - \lambda F) \otimes I_n$) were minimal at those $\lambda_0 \in \mathbb{F}$ such that $E - \lambda_0 F$ is singular then $\mathcal{L}_R(\lambda)$ would be a linearization of $R(\lambda)$ in \mathbb{F} with state matrix $(E - \lambda F) \otimes I_n$. That means that the zeros of $\mathcal{L}_R(\lambda)$ would be the zeros of $R(\lambda)$, and the zeros of $(E - \lambda F) \otimes I_n$ would be the poles of $R(\lambda)$, together with their partial multiplicities. This happens, for instance, under the conditions of Theorem 5.2.8.

Remark 5.2.11. In Remark 5.2.10, we consider $\mathcal{L}_R(\lambda)$ from two different points of view: as a block full rank pencil, $\begin{bmatrix} M(\lambda) \\ K_1(\lambda) \end{bmatrix}$, and as a polynomial system matrix with state matrix $(E - \lambda F) \otimes I_n$. In the former case, $\mathcal{L}_R(\lambda)$ is not in general a linearization at infinity of $R(\lambda)$ since rev₁ $K_1(\lambda)$ does not have full row rank at 0. In particular, Theorem 4.3.7 can not be applied and there is not always an integer g such that rev₁ $\mathcal{L}_R(\lambda)$ is equivalent at 0 to diag(rev_g $R(\lambda), I_{k(n-1)+\sum_{i=1}^{s} \ell_i n})$. It is not difficult to construct examples where such a g does not exist. In the latter case, $\mathcal{L}_R(\lambda)$ is not a linearization at infinity since rev₁ $\mathcal{L}_R(\lambda)$ is not minimal at 0. Both cases are due to the fact that the matrix rev₁ $(E - \lambda F)$ does not have full row rank at zero since F is singular.

5.2.1 Low-rank structure

Low-rank structures are exploited in [60] for constructing smaller linearizations that allow more efficient computations. In particular, a trimmed linearization is constructed if the matrix coefficients $C_i - \lambda D_i$ in (5.17) have low rank. For this purpose, write

$$C_i - \lambda D_i = (\widetilde{C}_i - \lambda \widetilde{D}_i) \widetilde{Z}_i^*, \qquad (5.31)$$

with $\widetilde{C}_i, \widetilde{D}_i, \widetilde{Z}_i \in \mathbb{F}^{n \times k_i}$, and $\widetilde{Z}_i^* \widetilde{Z}_i = I_{k_i}$. In several applied problems this type of structure appears with $k_i \ll n$ [47, 60]. By using the expression (5.31) for the matrix

coefficients, the matrix $R(\lambda)$ in (5.28) can be written as:

$$R(\lambda) = \sum_{i=0}^{k-1} (A_i - \lambda B_i) f_i(\lambda) + \sum_{i=1}^s (\widetilde{C}_i - \lambda \widetilde{D}_i) \widetilde{Z}_i^* a_i^T (E_i - \lambda F_i)^{-1} b_i$$

$$= \sum_{i=0}^{k-1} (A_i - \lambda B_i) (f_i(\lambda) \otimes I_n) + \sum_{i=1}^s [a_i^T \otimes (\widetilde{C}_i - \lambda \widetilde{D}_i)] ((E_i - \lambda F_i)^{-1} b_i \otimes I_{k_i}) \widetilde{Z}_i^*.$$
(5.32)

Then, the trimmed linearization $\widetilde{\mathcal{L}}_R(\lambda)$ for $R(\lambda)$ constructed in [60] is the following.

Definition 5.2.12. [60](Trimmed CORK linearization for AAA rational approximation) Let $R(\lambda)$ be a rational matrix as in (5.32). Consider the matrices

$$Z := \begin{bmatrix} -\widetilde{Z}_1(b_1^* \otimes I_{k_1}) & \cdots & -\widetilde{Z}_s(b_s^* \otimes I_{k_s}) \end{bmatrix},$$

$$E := \operatorname{diag}(E_1 \otimes I_{k_1}, \dots, E_s \otimes I_{k_s}), and$$

$$F := \operatorname{diag}(F_1 \otimes I_{k_1}, \dots, F_s \otimes I_{k_s}).$$

Then a trimmed CORK linearization for $R(\lambda)$ is

$$\widetilde{\mathcal{L}}_{R}(\lambda) = \begin{bmatrix} A_{0} - \lambda B_{0} & \cdots & A_{k-1} - \lambda B_{k-1} & a_{1}^{T} \otimes (\widetilde{C}_{1} - \lambda \widetilde{D}_{1}) & \cdots & a_{s}^{T} \otimes (\widetilde{C}_{s} - \lambda \widetilde{D}_{s}) \\ \hline (X - \lambda Y) \otimes I_{n} & 0 \\ \hline Z^{*} & 0 & E - \lambda F \end{bmatrix}$$
where
$$\begin{bmatrix} A_{0} - \lambda B_{0} & \cdots & A_{k-1} - \lambda B_{k-1} \\ \hline (X - \lambda Y) \otimes I_{n} \end{bmatrix}$$
 is any CORK linearization of $Q(\lambda)$.

Notice that the linearization $\mathcal{L}_R(\lambda)$ in Definition 5.2.5 has size $(kn + \sum_{i=1}^s \ell_i n) \times (kn + \sum_{i=1}^s \ell_i n)$ whereas the trimmed pencil $\widetilde{\mathcal{L}}_R(\lambda)$ in Definition 5.2.12 has size $(kn + \sum_{i=1}^s \ell_i k_i) \times (kn + \sum_{i=1}^s \ell_i k_i)$ with $k_i \ll n$ in several applications.

Analogous to what we did in Theorem 5.2.6, we study in Theorem 5.2.13 the structure of $\widetilde{\mathcal{L}}_R(\lambda)$ as linearization of $R(\lambda)$. The proof is omitted since it is analogous to that of Theorem 5.2.6.

Theorem 5.2.13. Let $R(\lambda)$ be a rational matrix as in (5.32), and let $\widetilde{\mathcal{L}}_R(\lambda)$ be the matrix pencil in Definition 5.2.12. Let $\Omega \subseteq \mathbb{F}$ be nonempty. If $\widetilde{\mathcal{L}}_R(\lambda)$, viewed as a polynomial system matrix with state matrix $E - \lambda F$, is minimal in Ω then $\widetilde{\mathcal{L}}_R(\lambda)$ is a block full rank linearization of $R(\lambda)$ in Ω with state matrix $E - \lambda F$.

Remark 5.2.14. As we discussed in Remark 5.2.11 for the matrix pencil $\mathcal{L}_R(\lambda)$, the trimmed CORK linearization $\widetilde{\mathcal{L}}_R(\lambda)$ is not in general a linearization at infinity of $R(\lambda)$ either. The reason is that, in this case, the matrix F is also singular and rev₁($E - \lambda F$) has not full row rank at zero.

Chapter 6

Linearizations of arbitrary rational transfer functions

In this chapter, we construct a family of linearizations of rational matrices in the sense of the definitions introduced in Chapter 4. For that purpose, rational matrices $R(\lambda)$ are written as transfer function matrices from general realizations. That is, of the form

$$R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda), \qquad (6.1)$$

where $D(\lambda)$, $C(\lambda)$, $B(\lambda)$ and $A(\lambda)$ are arbitrary polynomial matrices of appropriate sizes. For any rational matrix $R(\lambda)$, a representation as in (6.1) always exists and is not unique. The new linearizations are constructed from linearizations of the polynomial matrices $D(\lambda)$ and $A(\lambda)$, where each of them can be represented in terms of any polynomial basis. In particular, the notion of (strong) block minimal bases pencil (recall Definition 2.5.1) will be our main tool for building the linearizations. We will only consider the so-called degenerate (strong) block minimal bases pencils, that is, (strong) block minimal bases pencils of the form

$$L(\lambda) = \begin{bmatrix} M(\lambda) \\ K(\lambda) \end{bmatrix}, \tag{6.2}$$

where $K(\lambda)$ is a minimal basis. We know by Theorem 2.5.2 that the linear polynomial matrix in (6.2) is a (strong) linearization of the polynomial matrix

$$P(\lambda) = M(\lambda)N(\lambda)^T$$

considered as a polynomial of grade $1 + \deg N(\lambda)$, where $N(\lambda)$ is a minimal basis dual to $K(\lambda)$ (recall Definition 2.3.4). If $d := \deg N(\lambda) + 1$ and $P(\lambda)$ is of size $m \times n$, then $M(\lambda)$ and $K(\lambda)$ in (6.2) are, respectively, $m \times dn$ and $(d-1)n \times dn$ linear polynomial matrices [26].

Unlike other lineariations for rational matrices, as those in [27] (Chapter 3) or in [2, 7, 79], the construction of the linearizations in this chapter do not require neither

to write the corresponding rational matrix $R(\lambda)$ as the sum of its polynomial part and its strictly proper part nor to express the strictly proper rational part in statespace form. We finish the chapter by showing how to recover eigenvectors, when the rational matrix $R(\lambda)$ is regular, and minimal bases and minimal indices, when $R(\lambda)$ is singular, from those of their linearizations in this family. All the results in this chapter appear in [75].

6.1 Linearizations in a set

Throughout this chapter, \mathbb{F} denotes any algebraically closed field. Theorem 6.1.2 is the main result in this section, where we construct (local) linearizations for rational matrices that are represented with general realizations as in (6.1). To prove Theorem 6.1.2, we will use Lemma 6.1.1, whose simple proof is omitted.

Lemma 6.1.1. A polynomial matrix of the form

$$\begin{bmatrix} X(\lambda) & A(\lambda) & Y(\lambda) & B(\lambda) \\ I_s & 0 & 0 & 0 \\ Z(\lambda) & -C(\lambda) & W(\lambda) & D(\lambda) \\ 0 & 0 & I_t & 0 \end{bmatrix}$$

is unimodularly equivalent to diag $\left(\begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix}, I_{s+t} \right)$.

Theorem 6.1.2. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$, and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Let

$$L_A(\lambda) = \begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix} \quad and \quad L_D(\lambda) = \begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$$
(6.3)

be block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively. Let $N_A(\lambda)$ and $N_D(\lambda)$ be minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively. Consider linear polynomial matrices $M_C(\lambda)$ and $M_B(\lambda)$ such that

$$M_C(\lambda)N_A(\lambda)^T = C(\lambda) \quad and \quad M_B(\lambda)N_D(\lambda)^T = B(\lambda),$$
 (6.4)

and the linear polynomial system matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix},$$
(6.5)

with state matrix $L_A(\lambda)$. If the matrices

$$\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix} \quad and \quad \begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix}$$
(6.6)

have no eigenvalues in a nonempty set $\Omega \subseteq \mathbb{F}$, then $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Ω .

Remark 6.1.3. Before giving the proof of Theorem 6.1.2, we recall that there exist unimodular polynomial matrices of the form

$$U_i(\lambda) = \begin{bmatrix} K_i(\lambda) \\ \widehat{K}_i(\lambda) \end{bmatrix} \quad \text{and} \quad U_i(\lambda)^{-1} = \begin{bmatrix} \widehat{N}_i(\lambda)^T & N_i(\lambda)^T \end{bmatrix},$$
(6.7)

for $i \in \{A, D\}$; see [26, Theorem 2.10].

Proof. Throughout the proof, we use the notation $\rho_A := \deg N_A(\lambda)$ and $\rho_D := \deg N_D(\lambda)$.

To prove that $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Ω , we will use the spectral characterization in Theorem 4.2.6 of linearizations in a set.

Let $P(\lambda) := \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix}$ be a polynomial system matrix of $R(\lambda)$. Notice that $P(\lambda)$ is minimal in Ω by hypothesis. First, we have

pole elem. div. of
$$R(\lambda)$$
 in Ω =elem. div. of $A(\lambda)$ in Ω =elem. div. of $L_A(\lambda)$ in Ω ,

since $P(\lambda)$ is minimal in Ω and $L_A(\lambda)$ is a linearization of $A(\lambda)$. Hence, the pole elementary divisors of $R(\lambda)$ in Ω are equal to the elementary divisors of $L_A(\lambda)$ in Ω .

Second, we consider Remark 6.1.3 and notice that $K_A(\lambda)\widehat{N}_A(\lambda)^T = I_{n\rho_A}$ and $K_D(\lambda)\widehat{N}_D(\lambda)^T = I_{m\rho_D}$, as this will be important in what follows. Then, multiplying $\mathcal{L}(\lambda)$ on the right by the unimodular matrix $U(\lambda) = \text{diag}(U_A(\lambda)^{-1}, U_D(\lambda)^{-1})$, we get

$$\mathcal{L}(\lambda)U(\lambda) = \begin{bmatrix} X_{AA}(\lambda) & A(\lambda) & X_{BD}(\lambda) & B(\lambda) \\ I_{n\rho_A} & 0 & 0 & 0 \\ \hline -X_{CA}(\lambda) & -C(\lambda) & X_{DD}(\lambda) & D(\lambda) \\ 0 & 0 & I_{m\rho_D} & 0 \end{bmatrix},$$
(6.8)

where

$$X_{ij}(\lambda) := M_i(\lambda) \widehat{N}_j(\lambda)^T, \qquad (6.9)$$

with $i \in \{A, B, C, D\}$, and j := A if $i \in \{A, C\}$ and j := D if $i \in \{B, D\}$. Since the matrices in (6.6) have no eigenvalues in Ω , we have that $\mathcal{L}(\lambda)U(\lambda)$ is minimal in Ω and, therefore, $\mathcal{L}(\lambda)$ is minimal in Ω . By Lemma 6.1.1, $\mathcal{L}(\lambda)U(\lambda)$ is unimodularly

equivalent to diag $(P(\lambda), I_{n\rho_A+m\rho_D})$. Hence, $\mathcal{L}(\lambda)$ is a linearization of the polynomial system matrix $P(\lambda)$. As a consequence of this, we have

zero elem. div. of
$$R(\lambda)$$
 in Ω =elem. div. of $P(\lambda)$ in Ω
=elem. div. of $\mathcal{L}(\lambda)$ in Ω ,

since $P(\lambda)$ is minimal in Ω and $\mathcal{L}(\lambda)$ is a linearization of $P(\lambda)$. Therefore, the zero elementary divisors of $R(\lambda)$ in Ω are equal to the elementary divisors of $P(\lambda)$ in Ω .

Since $\mathcal{L}(\lambda)$ is of size $(n + p + s) \times (n + m + s)$, where $s = n\rho_A + m\rho_D$, to finish the proof, it suffices to notice that

nrank
$$\mathcal{L}(\lambda) = \operatorname{nrank} P(\lambda) + n\rho_A + m\rho_D$$
 (by (6.8))
= nrank $R(\lambda) + n + n\rho_A + m\rho_D$ (by (2.9)).

By Theorem 4.2.6, we conclude that $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in Ω .

Remark 6.1.4. Notice that, in Theorem 6.1.2, we assume that the polynomial matrices $A(\lambda)$ and $C(\lambda)$ can be written in terms of the same dual minimal basis $N_A(\lambda)$. That is, that there exist pencils $M_A(\lambda)$ and $M_C(\lambda)$ such that $M_A(\lambda)N_A(\lambda)^T = A(\lambda)$ and $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$, respectively. Such pencils always exist if we consider $A(\lambda)$ and $C(\lambda)$ as polynomial matrices with the same grade g, for instance,

$$g = \max\{\deg A(\lambda), \deg C(\lambda)\},\$$

see, for example, [31, Lemma 5.2]. A similar remark applies to the matrix polynomials $B(\lambda)$ and $D(\lambda)$. We can see an example in Example 6.1.5.

In Example 6.1.5, we show how to use Theorem 6.1.2 to construct linearizations of a rational matrix of the form (6.1). For simplicity, we assume that the polynomial matrices $A(\lambda)$, $B(\lambda)$, $C(\lambda)$ and $D(\lambda)$ are expressed in the monomial basis. But we emphasize that the construction can be easily adapted to many other polynomial bases (Chebyshev, Lagrange, Newton, etc); see Section 6.5 for an example in the Chebyshev basis.

Example 6.1.5. Let us consider a rational matrix of the form

$$R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda) = D_{3}\lambda^{3} + D_{2}\lambda^{2} + D_{1}\lambda + D_{0} + (C_{1}\lambda + C_{0})(A_{3}\lambda^{3} + A_{2}\lambda^{2} + A_{1}\lambda + A_{0})^{-1}(B_{2}\lambda^{2} + B_{1}\lambda + B_{0}),$$

where $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ is regular, and $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. We will use block Kronecker pencils [26, Section 4], which are particular cases of block minimal bases pencils, to construct the linearizations $L_A(\lambda)$ and $L_D(\lambda)$

in Theorem 6.1.2. We recall that the construction of block Kronecker pencils involves a pair of dual minimal bases of the form

$$K(\lambda) = \begin{bmatrix} -I_s & I_s \lambda & 0 & \cdots & 0 \\ 0 & -I_s & I_s \lambda & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & 0 \\ 0 & \cdots & 0 & -I_s & I_s \lambda \end{bmatrix} \quad and \quad N(\lambda)^T = \begin{bmatrix} I_s \lambda^{d-1} \\ \vdots \\ I_s \lambda \\ I_s \end{bmatrix}$$

To construct the linearization $L_A(\lambda)$ and the linear polynomial matrix $M_C(\lambda)$, we need to see both $A(\lambda)$ and $C(\lambda)$ as polynomial matrices of grade

 $\max\{\deg A(\lambda), \deg C(\lambda)\} = \max\{3, 1\} = 3.$

Then, we can use, for example,

$$L_A(\lambda) := \begin{bmatrix} A_3\lambda + A_2 & A_1 & A_0 \\ -I_n & I_n\lambda & 0 \\ 0 & -I_n & I_n\lambda \end{bmatrix} \quad and \quad M_C(\lambda) := \begin{bmatrix} 0 & C_1 & C_0 \end{bmatrix},$$

with $N_A(\lambda) := \begin{bmatrix} I_n \lambda^2 & I_n \lambda & I_n \end{bmatrix}$. Similarly, to construct the linearization $L_D(\lambda)$ and the linear polynomial matrix $M_B(\lambda)$, we need to see both $D(\lambda)$ and $B(\lambda)$ as polynomial matrices of grade

$$\max\{\deg D(\lambda), \deg B(\lambda)\} = \max\{3, 2\} = 3.$$

Then, we can use, for instance,

$$L_D(\lambda) := \begin{bmatrix} D_3\lambda + D_2 & D_1 & D_0 \\ -I_m & I_m\lambda & 0 \\ 0 & -I_m & I_m\lambda \end{bmatrix} \quad and \quad M_B(\lambda) := \begin{bmatrix} 0 & B_2\lambda + B_1 & B_0 \end{bmatrix},$$

with $N_D(\lambda) := \begin{bmatrix} I_m \lambda^2 & I_m \lambda & I_m \end{bmatrix}$. Then the linear polynomial system matrix $\mathcal{L}(\lambda)$ in Theorem 6.1.2 is

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_3\lambda + A_2 & A_1 & A_0 & 0 & B_2\lambda + B_1 & B_0 \\ -I_n & I_n\lambda & 0 & 0 & 0 \\ 0 & -I_n & I_n\lambda & 0 & 0 & 0 \\ \hline 0 & -C_1 & -C_0 & D_3\lambda + D_2 & D_1 & D_0 \\ 0 & 0 & 0 & -I_m & I_m\lambda & 0 \\ 0 & 0 & 0 & 0 & -I_m & I_m\lambda \end{bmatrix}.$$

Theorem 6.1.2 guarantees that $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ wherever the polynomial matrices $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix}$ and $\begin{bmatrix} A(\lambda) & B(\lambda) \end{bmatrix}$ do not have eigenvalues.

6.2 Linearizations at infinity

Theorem 6.2.1 is the main result of this section. It shows that, under some mild conditions, the local linearizations introduced in Section 6.1 are also linearizations at infinity of the rational matrix $R(\lambda)$.

Theorem 6.2.1. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$, and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Let

$$L_A(\lambda) = \begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix} \quad and \quad L_D(\lambda) = \begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$$

be strong block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively. Let $N_A(\lambda)$ and $N_D(\lambda)$ be minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively, and denote $\rho_A := \deg N_A(\lambda)$, $\rho_D := \deg N_D(\lambda)$. Consider linear polynomial matrices $M_C(\lambda)$ and $M_B(\lambda)$ such that

$$M_C(\lambda)N_A(\lambda)^T = C(\lambda)$$
 and $M_B(\lambda)N_D(\lambda)^T = B(\lambda)$,

and the linear polynomial system matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix},$$
(6.10)

with state matrix $L_A(\lambda)$. If the matrices

$$\begin{bmatrix} \operatorname{rev}_{\rho_A+1} A(\lambda) \\ \operatorname{rev}_{\rho_A+1} C(\lambda) \end{bmatrix} \quad and \quad \begin{bmatrix} \operatorname{rev}_{\rho_A+1} A(\lambda) & \operatorname{rev}_{\rho_D+1} B(\lambda) \end{bmatrix}$$
(6.11)

have no eigenvalues at 0, then $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ at infinity of grade $\rho_D + 1$.

Remark 6.2.2. Since $L_A(\lambda)$ and $L_D(\lambda)$ are strong block minimal bases pencils, we recall that there exist unimodular polynomial matrices of the form

$$\widetilde{U}_i(\lambda) = \begin{bmatrix} \operatorname{rev}_1 K_i(\lambda) \\ \widetilde{K}_i(\lambda) \end{bmatrix} \quad \text{and} \quad \widetilde{U}_i(\lambda)^{-1} = \begin{bmatrix} \widetilde{N}_i(\lambda)^T & \operatorname{rev}_{\rho_i} N_i(\lambda)^T \end{bmatrix}, \quad (6.12)$$

for $i \in \{A, D\}$; see [26, Theorem 2.10].

Proof. To prove that $\mathcal{L}(\lambda)$ is a linearization at infinity of grade g of $R(\lambda)$, we will use the spectral characterization in Theorem 4.2.14 of linearizations at infinity.

Let $g := \rho_D + 1$. Let us consider the polynomial system matrix

$$\widetilde{P}(\lambda) := \begin{bmatrix} \operatorname{rev}_{\rho_A+1} A(\lambda) & \operatorname{rev}_{\rho_D+1} B(\lambda) \\ -\operatorname{rev}_{\rho_A+1} C(\lambda) & \operatorname{rev}_{\rho_D+1} D(\lambda) \end{bmatrix},$$
(6.13)

with state matrix $\operatorname{rev}_{\rho_A+1} A(\lambda)$. We observe that the transfer function of the polynomial system matrix (6.13) is

$$\operatorname{rev}_{\rho_D+1} D(\lambda) + \operatorname{rev}_{\rho_A+1} C(\lambda) \left(\operatorname{rev}_{\rho_A+1} A(\lambda)\right)^{-1} \operatorname{rev}_{\rho_D+1} B(\lambda) = \operatorname{rev}_g R(\lambda).$$

Hence, we have

pole elem. div. of
$$\operatorname{rev}_g R(\lambda)$$
 at 0 =elem. div. of $\operatorname{rev}_{\rho_A+1} A(\lambda)$ at 0 =elem. div. of $\operatorname{rev}_1 L_A(\lambda)$ at 0 ,

since $\tilde{P}(\lambda)$ is minimal at 0 and $L_A(\lambda)$ is a strong linearization of $A(\lambda)$. Thus, the pole elementary divisors of $\operatorname{rev}_g R(\lambda)$ at 0 are equal to the elementary divisors of $\operatorname{rev}_1 L_A(\lambda)$ at 0.

Consider the unimodular matrix $\widetilde{U}(\lambda) = \operatorname{diag}(\widetilde{U}_A(\lambda)^{-1}, \widetilde{U}_D(\lambda)^{-1})$, where $\widetilde{U}_A(\lambda)^{-1}$ and $\widetilde{U}_D(\lambda)^{-1}$ are defined in (6.12). Multiplying rev₁ $\mathcal{L}(\lambda)$ on the right by $\widetilde{U}(\lambda)$, we get

$$\operatorname{rev}_{1} \mathcal{L}(\lambda) \widetilde{U}(\lambda) = \begin{bmatrix} \ast & \operatorname{rev}_{\rho_{A}+1} A(\lambda) & \ast & \operatorname{rev}_{\rho_{D}+1} B(\lambda) \\ I_{n\rho_{A}} & 0 & 0 & 0 \\ & & I_{n\rho_{A}} & 0 & 0 \\ & & & \operatorname{rev}_{\rho_{A}+1} D(\lambda) \\ 0 & 0 & I_{m\rho_{D}} & 0 \end{bmatrix}, \quad (6.14)$$

where * denotes polynomial matrices that are not important for the argument. Since the matrices in (6.11) have no eigenvalues at 0, we have that rev₁ $\mathcal{L}(\lambda) \widetilde{U}(\lambda)$ is minimal at 0 and, therefore, rev₁ $\mathcal{L}(\lambda)$ is minimal at 0. By Lemma 6.1.1, the matrix polynomial rev₁ $\mathcal{L}(\lambda) \widetilde{U}(\lambda)$ is unimodularly equivalent to diag($\widetilde{P}(\lambda), I_{n\rho_A+m\rho_D}$). Hence, rev₁ $\mathcal{L}(\lambda)$ is a linearization of $\widetilde{P}(\lambda)$ at 0. Thus, we have

zero elem. div. of
$$\operatorname{rev}_g R(\lambda)$$
 at 0 =elem. div. of $P(\lambda)$ at 0 =elem. div. of $\operatorname{rev}_1 \mathcal{L}(\lambda)$ at 0 ,

since $\widetilde{P}(\lambda)$ is minimal at 0 and rev₁ $\mathcal{L}(\lambda)$ is a linearization of $\widetilde{P}(\lambda)$. Therefore, the zero elementary divisors of rev_g $R(\lambda)$ at 0 are equal to the elementary divisors of rev₁ $\mathcal{L}(\lambda)$ at 0.

To finish the proof, it suffices to notice that

nrank rev₁
$$\mathcal{L}(\lambda) = \operatorname{nrank} P(\lambda) + n\rho_A + m\rho_D$$
 (by (6.14))

$$= \operatorname{nrank} \operatorname{rev}_g R(\lambda) + n + n\rho_A + m\rho_D \qquad (by (2.9)).$$

Conclusively, by Theorem 4.2.14, the linear polynomial system matrix $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ at infinity of grade $g = \rho_D + 1$.

We conclude this section by noting that the linearization $\mathcal{L}(\lambda)$ constructed in Theorem 6.1.2 contains the spectral information of the rational matrix $R(\lambda)$ in a nonempty set Ω , whenever certain minimality conditions are satisfied. More precisely, if the matrix polynomials in (6.6) have no eigenvalues in Ω then the zeros in Ω of $R(\lambda)$ are the eigenvalues in Ω of $\mathcal{L}(\lambda)$, and the poles in Ω of $R(\lambda)$ are the eigenvalues in Ω of the block minimal bases pencil $L_A(\lambda)$. If $\Omega = \mathbb{F}$ then we can recover the complete information about finite poles and zeros of $R(\lambda)$ from $\mathcal{L}(\lambda)$. If, in addition, the matrix polynomials (6.11) have no eigenvalues at zero, then $\mathcal{L}(\lambda)$ is a strong linearization of grade $\rho_D + 1$ and, hence, by Proposition 4.2.15, we can also recover the complete zero and pole information of $R(\lambda)$ at infinity.

6.3 An illustrative example

Let us consider an $m \times m$ rational matrix of the form

$$R(\lambda) = D(\lambda) + f_1(\lambda)K_1 + f_2(\lambda)K_2, \qquad (6.15)$$

where $D(\lambda)$ is a polynomial matrix of degree 2, $f_1(\lambda) = \frac{(\lambda^2+1)(\lambda+2)}{\lambda^2-\lambda-2}$, $f_2(\lambda) = \frac{(\lambda^2-1)\lambda^2}{\lambda+2}$, and K_1 and K_2 are constant matrices having ranks r_1 and r_2 , respectively. For i = 1, 2, we can write $K_i = c_i b_i^T$, for some c_i and b_i both of size $m \times r_i$. Then, a realization of $R(\lambda)$ as in (6.1) is

$$R(\lambda) = D(\lambda) + \underbrace{\left[c_{1}(\lambda^{2}+1) \quad c_{2}(\lambda^{2}-1)\right]}_{:=C(\lambda)} \underbrace{\left[\begin{matrix}I_{r_{1}}(\lambda^{2}-\lambda-2) & 0\\ 0 & I_{r_{2}}(\lambda+2)\end{matrix}\right]^{-1}}_{:=A(\lambda)^{-1}} \underbrace{\left[\begin{matrix}b_{1}^{T}(\lambda+2)\\ b_{2}^{T}\lambda^{2}\end{matrix}\right]}_{:=B(\lambda)}$$

Notice that the polynomial matrices $A(\lambda)$, $B(\lambda)$ and $C(\lambda)$ have degree 2. Hence, we can write $A(\lambda) := A_2\lambda^2 + A_1\lambda + A_0$, $B(\lambda) := B_2\lambda^2 + B_1\lambda + B_0$, $C(\lambda) := C_2\lambda^2 + C_1\lambda + C_0$ and $D(\lambda) := D_2\lambda^2 + D_1\lambda + D_0$. Then, we set $r := r_1 + r_2$, and we define the linear polynomial system matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_2\lambda + A_1 & A_0 & B_2\lambda + B_1 & B_0 \\ -I_r & I_r\lambda & 0 & 0 \\ \hline -C_2\lambda - C_1 & -C_0 & D_2\lambda + D_1 & D_0 \\ 0 & 0 & -I_m & I_m\lambda \end{bmatrix}$$

It can be proved that condition (6.6) is satisfied for all $\lambda_0 \in \mathbb{F}$ and that condition (6.11) is also satisfied. Thus, by Theorems 6.1.2 and 6.2.1, $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$ in \mathbb{F} and also a linearization of $R(\lambda)$ at infinity of grade 2.

Notice that the rational functions $f_1(\lambda)$ and $f_2(\lambda)$ in (6.15) are not (strictly) proper. Nevertheless, we have been able to construct a strong linearization in the sense of [28] for $R(\lambda)$ without decomposing $R(\lambda)$ as the sum of its polynomial part and its strictly proper rational part and without considering a minimal state space realalization of the strictly proper part, in contrast to the methods appearing in [2, 7, 27, 79]. We emphasize that, in order to construct linearizations of rational matrices, we must take into account that a realization as in (6.1) is not unique and that the ideal goal is to consider a realization easy to build from the original expression of the corresponding rational matrix without performing many computations.

6.4 Recovery of eigenvectors, minimal bases and minimal indices

In this section, we show how to recover the elements in the right and left nullspaces of a rational matrix $R(\lambda)$ from those in the right and left nullspaces of a linearization $\mathcal{L}(\lambda)$ as in Theorem 6.1.2, as well as minimal bases and minimal indices, assuming $R(\lambda)$ is singular. In addition, we show how to recover right and left eigenvectors in the regular case. To do this, we consider the transfer function matrix $\hat{R}(\lambda)$ of $\mathcal{L}(\lambda)$ and a polynomial system matrix $P(\lambda)$ of $R(\lambda)$, and we study the relation between the elements in their right and left nullspaces.

6.4.1 One-sided factorizations

Theorem 6.4.1 is the only result in this subsection. It establishes a relation, in terms of one-side factorizations, between a rational matrix $R(\lambda)$ and the transfer function matrix $\widehat{R}(\lambda)$ of a linearization for $R(\lambda)$ as in Theorem 6.1.2.

Theorem 6.4.1. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Consider the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix},$$

as in Theorem 6.1.2, where $\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$ are block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively, with $N_A(\lambda)$ and $N_D(\lambda)$ being minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively, and $M_C(\lambda)$ and $M_B(\lambda)$ being linear polynomial matrices such that $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$ and $M_B(\lambda)N_D(\lambda)^T = B(\lambda)$, respectively. Let $\widehat{R}(\lambda)$ be the transfer function matrix of $\mathcal{L}(\lambda)$. Then, we have the following one-sided factorizations

$$\widehat{R}(\lambda)N_D(\lambda)^T = \begin{bmatrix} R(\lambda) \\ 0 \end{bmatrix}, \qquad (6.16)$$

and

$$\begin{bmatrix} I_p & -M_R(\lambda)\widehat{N}_D(\lambda)^T \end{bmatrix} \widehat{R}(\lambda) = R(\lambda)\widehat{K}_D(\lambda), \qquad (6.17)$$

where $M_R(\lambda) := M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda)$, and $\widehat{K}_D(\lambda)$ and $\widehat{N}_D(\lambda)$ are defined in (6.7).

Proof. Notice that the matrix pencil $L_A(\lambda) = \begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ is regular since $L_A(\lambda)$ is a linearization of $A(\lambda)$ and the polynomial matrix $A(\lambda)$ is regular. Then, from $L_A(\lambda)N_A(\lambda)^T = \begin{bmatrix} A(\lambda) \\ 0 \end{bmatrix}$, we obtain

$$\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}^{-1} \begin{bmatrix} I_n \\ 0 \end{bmatrix} = N_A(\lambda)^T A(\lambda)^{-1}.$$
 (6.18)

Hence, the transfer function matrix of $\mathcal{L}(\lambda)$ is given by

$$\widehat{R}(\lambda) = \begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix} + \begin{bmatrix} M_C(\lambda) \\ 0 \end{bmatrix} \begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}^{-1} \begin{bmatrix} M_B(\lambda) \\ 0 \end{bmatrix} = \begin{bmatrix} M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda) \\ K_D(\lambda) \end{bmatrix},$$
(6.19)

where we have used $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$.

Multiplying (6.19) on the right by $N_D(\lambda)^T$ yields

$$\widehat{R}(\lambda)N_D(\lambda)^T = \begin{bmatrix} D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda) \\ 0 \end{bmatrix} = \begin{bmatrix} R(\lambda) \\ 0 \end{bmatrix},$$

where we have used $M_D(\lambda)N_D(\lambda)^T = D(\lambda)$ and $M_B(\lambda)N_D(\lambda)^T = B(\lambda)$. This establishes the right-sided factorization (6.16).

Multiplying (6.19) on the left by $\left[I_p - \left(M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda)\right)\widehat{N}_D(\lambda)^T\right]$ gives

$$\begin{bmatrix} I_p & -(M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda))\,\hat{N}_D(\lambda)^T \end{bmatrix} \hat{R}(\lambda) = \\ & (M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda))(I_{(\rho_D+1)m} - \hat{N}_D(\lambda)^T K_D(\lambda)) = \\ & (M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda))N_D(\lambda)^T \hat{K}_D(\lambda) = R(\lambda)\hat{K}_D(\lambda), \end{bmatrix}$$

where $\rho_D = \deg N_D(\lambda)$. This establishes the left-sided factorization (6.17).

6.4.2 Recovery of minimal bases and minimal indices

In this section, we assume that the rational matrix $R(\lambda)$ is singular and show how to recover the right and left minimal indices and minimal bases of $R(\lambda)$ from those of a linearization $\mathcal{L}(\lambda)$ of $R(\lambda)$ as in Theorem 6.1.2.

We begin with Lemma 6.4.2, which establishes a bijection between the nullspaces of $R(\lambda)$ and the transfer function matrix of $\mathcal{L}(\lambda)$.

Lemma 6.4.2. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Consider the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix},$$

as in Theorem 6.1.2, where $\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$ are block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively, with $N_A(\lambda)$ and $N_D(\lambda)$ being minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively, and $M_C(\lambda)$ and $M_B(\lambda)$ being linear polynomial matrices such that $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$ and $M_B(\lambda)N_D(\lambda)^T = B(\lambda)$, respectively. Let $\widehat{R}(\lambda)$ be the transfer function matrix of $\mathcal{L}(\lambda)$. Then the following statements hold:

(a) The linear map

$$M_r : \mathcal{N}_r(R) \longrightarrow \mathcal{N}_r(R)$$
$$x(\lambda) \longmapsto \widehat{x}(\lambda) := N_D(\lambda)^T x(\lambda)$$

is a bijection between the right nullspaces of $R(\lambda)$ and $\widehat{R}(\lambda)$.

(b) The linear map

$$M_{\ell} : \mathcal{N}_{\ell}(R) \longrightarrow \mathcal{N}_{\ell}(R)$$
$$y(\lambda)^{T} \longmapsto \widehat{y}(\lambda)^{T} := y(\lambda)^{T} \left[I_{p} - M_{R}(\lambda) \widehat{N}_{D}(\lambda)^{T} \right]$$

is a bijection between the left nullspaces of $R(\lambda)$ and $\widehat{R}(\lambda)$, where $M_R(\lambda) := M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda)$ and $\widehat{N}_D(\lambda)$ is defined in (6.7).

Proof. We will prove part (a). Part (b) can be proved analogously.

That the map M_r is well-defined from the right nullspace of $R(\lambda)$ to the right nullspace $\widehat{R}(\lambda)$ follows immediately from the right-sided factorization (6.16). Notice

that M_r is, in addition, linear. Moreover, by Theorem 6.1.2, $\mathcal{L}(\lambda)$ is a linearization of $R(\lambda)$, at least, at some point $\lambda_0 \in \mathbb{F}$. Indeed, since $A(\lambda)$ is regular, there exists $\lambda_0 \in \mathbb{F}$ such that $A(\lambda_0)$ is invertible. This implies that the realization $D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$ is minimal at λ_0 . Hence, we have dim $\mathcal{N}_r(R) = \dim \mathcal{N}_r(\widehat{R})$. Thus, to show that M_r is a bijection, if suffices to show that it is injective. So, suppose $\widehat{y}(\lambda) = N_D(\lambda)^T y(\lambda) = 0$. Since $N_D(\lambda)$ is a minimal basis, $N_D(\lambda)^T$ has full column rank. Hence, $N_D(\lambda)^T y(\lambda) = 0$ implies $y(\lambda) = 0$. This establishes the injectivity of the linear map M_r .

Remark 6.4.3. Since the maps in Lemma 6.4.2 are bijections, they preserve linear independence and allow us to recover bases of the right (resp. left) nullspace of $R(\lambda)$ from bases of the right (resp. left) nullspace of $\widehat{R}(\lambda)$, and conversely.

By combining Lemmas 2.4.5 and 6.4.2, we obtain Theorem 6.4.4, which establishes a bijection between the nullspaces of $R(\lambda)$ and $\mathcal{L}(\lambda)$.

Theorem 6.4.4. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Consider the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix},$$

as in Theorem 6.1.2, where $\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$ are block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively, with $N_A(\lambda)$ and $N_D(\lambda)$ being minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively, and $M_C(\lambda)$ and $M_B(\lambda)$ being linear polynomial matrices such that $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$ and $M_B(\lambda)N_D(\lambda)^T = B(\lambda)$, respectively. Let $\widehat{R}(\lambda)$ be the transfer function matrix of $\mathcal{L}(\lambda)$. Then the following statements hold:

(a) The linear map

$$F_r : \mathcal{N}_r(R) \longrightarrow \mathcal{N}_r(\mathcal{L})$$
$$x(\lambda) \longmapsto \widetilde{x}(\lambda) := \begin{bmatrix} -N_A(\lambda)^T A(\lambda)^{-1} B(\lambda) \\ N_D(\lambda)^T \end{bmatrix} x(\lambda)$$

is a bijection between the right nullspaces of $R(\lambda)$ and $\mathcal{L}(\lambda)$.

(b) The linear map

$$F_{\ell} : \mathcal{N}_{\ell}(R) \longrightarrow \mathcal{N}_{\ell}(\mathcal{L})$$

$$y(\lambda)^T \longmapsto \widetilde{y}(\lambda)^T := y(\lambda)^T \begin{bmatrix} M_C(\lambda) L_A(\lambda)^{-1} & I_p & -M_R(\lambda) \widehat{N}_D(\lambda)^T \end{bmatrix}$$

is a bijection between the left nullspaces of $R(\lambda)$ and $\mathcal{L}(\lambda)$, where $L_A(\lambda) := \begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$, $M_R(\lambda) := M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda)$ and $\widehat{N}_D(\lambda)$ is defined in (6.7).

Proof. We will prove part (a). Part (b) can be proved analogously.

Consider the linear bijections T_r and M_r in Lemma 2.4.5 and Lemma 6.4.2, respectively. Then, F_r is the composition $F_r = T_r \circ M_r$. Indeed, we have

$$F_r : \mathcal{N}_r(R) \longrightarrow \mathcal{N}_r(\widehat{R}) \longrightarrow \mathcal{N}_r(\mathcal{L})$$
$$x(\lambda) \longmapsto N_D(\lambda)^T x(\lambda) \longmapsto \begin{bmatrix} -\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}^{-1} \begin{bmatrix} M_B(\lambda) \\ 0 \\ N_D(\lambda)^T x(\lambda) \end{bmatrix} N_D(\lambda)^T x(\lambda) \end{bmatrix} = \begin{bmatrix} -N_A(\lambda)^T A(\lambda)^{-1} B(\lambda) x(\lambda) \\ N_D(\lambda)^T x(\lambda) \end{bmatrix},$$

where we have used (6.18) and $M_B(\lambda)N_D(\lambda)^T = B(\lambda)$.

Lemma 6.4.5 will allow us to prove that, under some minimality conditions, the linear polynomial system matrix $\mathcal{L}(\lambda)$ and its transfer function $\widehat{R}(\lambda)$ have the same right and left minimal indices.

Lemma 6.4.5. Consider a linear polynomial system matrix

$$L(\lambda) = \begin{bmatrix} A_1\lambda + A_0 & B_1\lambda + B_0 \\ \hline -(C_1\lambda + C_0) & D_1\lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q)\times(n+r)}$$

with state matrix $A_1\lambda + A_0 \in \mathbb{F}[\lambda]^{n \times n}$ and transfer function matrix $T(\lambda)$. Then the following statements hold:

- (a) If $u(\lambda) = \begin{bmatrix} y(\lambda) \\ x(\lambda) \end{bmatrix} \in \mathcal{N}_r(L)$, then $x(\lambda) \in \mathcal{N}_r(T)$. In addition, if $u(\lambda)$ is a polynomial vector and rank $\begin{bmatrix} A_1 \\ C_1 \end{bmatrix} = n$, then $\deg u(\lambda) = \deg x(\lambda)$.
- (b) If $v(\lambda)^T = \begin{bmatrix} \tilde{y}(\lambda)^T & \tilde{x}(\lambda)^T \end{bmatrix} \in \mathcal{N}_{\ell}(L)$, then $\tilde{x}(\lambda)^T \in \mathcal{N}_{\ell}(T)$. In addition, if $v(\lambda)$ is a polynomial vector and rank $\begin{bmatrix} A_1 & B_1 \end{bmatrix} = n$, then $\deg v(\lambda) = \deg \tilde{x}(\lambda)$.

Proof. The proof follows the same lines as the proof of [18, Theorem 6.8]. We only prove part (a) since part (b) follows from a similar argument. From Lemma 2.4.5, we obtain that if $u(\lambda) \in \mathcal{N}_r(L)$, then the vector $u(\lambda)$ must be of the form

$$u(\lambda) = \begin{bmatrix} y(\lambda) \\ x(\lambda) \end{bmatrix} = \begin{bmatrix} -(A_1\lambda + A_0)^{-1}(B_1\lambda + B_0)x(\lambda) \\ x(\lambda) \end{bmatrix},$$

for some $x(\lambda) \in \mathcal{N}_r(T)$, as we wanted to show.

For proving that $\deg u(\lambda) = \deg x(\lambda)$, we will show that $\deg y(\lambda) \leq \deg x(\lambda)$ by contradiction. Let us assume that $\ell := \deg y(\lambda) > \deg x(\lambda)$. Then, the vector $u(\lambda)$ must be of the form

$$u(\lambda) = \begin{bmatrix} y_{\ell} \\ 0 \end{bmatrix} \lambda^{\ell} + \text{ lower degree terms, with } y_{\ell} \neq 0.$$

Since $u(\lambda) \in \mathcal{N}_r(L)$, we have

$$(A_1\lambda + A_0)y(\lambda) + (B_1\lambda + B_0)x(\lambda) = 0,$$

$$(C_1\lambda + C_0)y(\lambda) - (D_1\lambda + D_0)x(\lambda) = 0.$$

Considering the highest degree terms in the left hand side of the two equations above, we obtain

$$\begin{bmatrix} A_1 \\ C_1 \end{bmatrix} y_\ell = 0.$$

Since the matrix $\begin{bmatrix} A_1 \\ C_1 \end{bmatrix}$ has full column rank by assumption, we get $y_{\ell} = 0$, which contradicts our original hypothesis.

Remark 6.4.6. We emphasize that

$$\operatorname{rank} \begin{bmatrix} A_1 \\ C_1 \end{bmatrix} = \operatorname{rank} \begin{bmatrix} A_1 & B_1 \end{bmatrix} = n \tag{6.20}$$

is the condition for a linear polynomial system matrix to be minimal at infinity in the sense of [28], which is also a necessary condition for a linear polynomial system matrix to be a linearization at infinity [28]. In Chapter 7 (or [32]) there is a procedure to reduce any linear polynomial system matrix to one satisfying condition (6.20).

Remark 6.4.7. We notice that Lemma 6.4.5 extends [18, Theorem 6.1]. But, while [18, Theorem 6.1] assumes the invertibility of A_1 , we use the more general condition (6.20).

For completeness, in Lemma 6.4.8 we recall [92, Lemma 2] and different versions of it that can be analogously proved.

Lemma 6.4.8. Let $\begin{bmatrix} X_1 & X_2 \end{bmatrix} \begin{bmatrix} Y_1 \\ Y_2 \end{bmatrix} = 0.$

- (a) Assume that $\begin{bmatrix} Y_1 \\ Y_2 \end{bmatrix}$ has full column rank.
 - (a1) If X_1 has full column rank, then Y_2 also has full column rank.
 - (a2) If X_2 has full column rank, then Y_1 also has full column rank.

(b) Assume that $\begin{bmatrix} X_1 & X_2 \end{bmatrix}$ has full row rank.

- (b1) If Y_1 has full row rank, then X_2 also has full row rank.
- (b2) If Y_2 has full row rank, then X_1 also has full row rank.

We are finally ready to state and prove the main results of this section, Theorems 6.4.9 and 6.4.11. These theorems show how to recover right and left minimal bases and minimal indices of rational matrices from those of their linearizations in Theorem 6.1.2.

Theorem 6.4.9 (Right minimal bases and minimal indices). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Consider the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n(1+\rho_A)+p+m\rho_D)\times(n(\rho_A+1)+m(\rho_D+1))},$$

as in Theorem 6.1.2, where $\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$ are strong block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively, with $N_A(\lambda)$ and $N_D(\lambda)$ being minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively, and $M_C(\lambda)$ and $M_B(\lambda)$ being linear polynomial matrices such that $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$ and $M_B(\lambda)N_D(\lambda)^T =$ $B(\lambda)$, respectively. Define $\rho_i := \deg N_i(\lambda)$, for $i \in \{A, D\}$, and let $\widehat{R}(\lambda)$ be the transfer function matrix of $\mathcal{L}(\lambda)$. If

$$\operatorname{rank} \begin{bmatrix} A(\lambda_0) \\ C(\lambda_0) \end{bmatrix} = n \quad \text{for all} \quad \lambda_0 \in \mathbb{F}, \quad and \tag{6.21}$$

$$\operatorname{rank} \begin{bmatrix} \operatorname{rev}_{\rho_A+1} A(0) \\ \operatorname{rev}_{\rho_A+1} C(0) \end{bmatrix} = n, \tag{6.22}$$

then the following statements hold:

- (a) If $\left\{ \begin{bmatrix} y_i(\lambda) \\ x_i(\lambda) \end{bmatrix} \right\}_{i=1}^s$ is a right minimal basis of $\mathcal{L}(\lambda)$, where $x_i(\lambda) \in \mathbb{F}[\lambda]^{(\rho_D+1)m}$, then $\{x_i(\lambda)\}_{i=1}^s$ is a right minimal basis of $\widehat{R}(\lambda)$, and there exists a right minimal basis $\{u_i(\lambda)\}_{i=1}^s$ of $R(\lambda)$ such that $x_i(\lambda) = N_D(\lambda)^T u_i(\lambda)$, for $i = 1, \ldots, s$.
- (b) If $\epsilon_1 \leq \cdots \leq \epsilon_s$ are the right minimal indices of $\mathcal{L}(\lambda)$, then $\epsilon_1 \leq \cdots \leq \epsilon_s$ are the right minimal indices of $\widehat{R}(\lambda)$, and $\epsilon_1 \rho_D \leq \cdots \leq \epsilon_s \rho_D$ are the right minimal indices of $R(\lambda)$.

Proof. See Appendix A.

Remark 6.4.10. We recall that the polynomial matrix $\widehat{K}_D(\lambda)$ in (6.7) is the left polynomial matrix inverse of $N_D(\lambda)^T$. Hence, from the right minimal basis $\{x_i(\lambda)\}_{i=1}^s$ of $\widehat{R}(\lambda)$ in part (a) of Theorem 6.4.9, we can recover a right minimal basis of the rational matrix $R(\lambda)$ as $\{u_i(\lambda)\}_{i=1}^s = \{\widehat{K}_D(\lambda)x_i(\lambda)\}_{i=1}^s$.

Theorem 6.4.11 (Left minimal bases and minimal indices). Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Consider the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix} \in \mathbb{F}[\lambda]^{(n(1+\rho_A)+p+m\rho_D)\times(n(\rho_A+1)+m(\rho_D+1))},$$

as in Theorem 6.1.2, where $\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$ are strong block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively, with $N_A(\lambda)$ and $N_D(\lambda)$ being minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively, and $M_C(\lambda)$ and $M_B(\lambda)$ being linear polynomial matrices such that $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$ and $M_B(\lambda)N_D(\lambda)^T =$ $B(\lambda)$, respectively. Define $\rho_i := \deg N_i(\lambda)$, for $i \in \{A, D\}$, and let $\widehat{R}(\lambda)$ be the transfer function matrix of $\mathcal{L}(\lambda)$. If

$$\operatorname{rank} \begin{bmatrix} A(\lambda_0) & B(\lambda_0) \end{bmatrix} = n \quad \text{for all} \quad \lambda_0 \in \mathbb{F}, \quad and \tag{6.23}$$

$$\operatorname{rank}\left[\operatorname{rev}_{\rho_A+1} A(0) \quad \operatorname{rev}_{\rho_D+1} B(0)\right] = n, \tag{6.24}$$

then the following statements hold:

- (a) If $\{z_i(\lambda)^T\}_{i=1}^t$ is a left minimal basis of $\mathcal{L}(\lambda)$, then $z_i(\lambda)^T = \begin{bmatrix} y_i(\lambda)^T & x_i(\lambda)^T \end{bmatrix}$, for $i = 1, \ldots, s$, for some left minimal basis $\{x_i(\lambda)^T\}_{i=1}^s$ of $\widehat{R}(\lambda)$, and $x_i(\lambda)^T = \begin{bmatrix} u_i(\lambda)^T & w_i(\lambda)^T \end{bmatrix}$, for $i = 1, \ldots, s$, for some left minimal basis $\{u_i(\lambda)^T\}_{i=1}^s$ of $R(\lambda)$.
- (b) If $\eta_1 \leq \cdots \leq \eta_t$ are the left minimal indices of $\mathcal{L}(\lambda)$, then $\eta_1 \leq \cdots \leq \eta_t$ are the left minimal indices of $\widehat{R}(\lambda)$ and $R(\lambda)$.

Proof. See Appendix B.

6.4.3 Recovery of eigenvectors

In this section, we assume that the rational matrix $R(\lambda)$ is regular and show how to recover right and left eigenvectors of $R(\lambda)$ from those of a linearization $\mathcal{L}(\lambda)$ of $R(\lambda)$ as in Theorem 6.1.2. In Proposition 6.4.12 we state, without proof, analogous results to those of Theorem 6.4.4 but for the right and left nullspaces of $R(\lambda)$ evaluated at a particular value λ_0 .
Proposition 6.4.12. Let $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ be a rational matrix expressed in the form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, for some regular polynomial matrix $A(\lambda) \in \mathbb{F}[\lambda]^{n \times n}$ and polynomial matrices $B(\lambda) \in \mathbb{F}[\lambda]^{n \times m}$, $C(\lambda) \in \mathbb{F}[\lambda]^{p \times n}$ and $D(\lambda) \in \mathbb{F}[\lambda]^{p \times m}$. Consider the linear polynomial matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix},$$

as in Theorem 6.1.2, where $\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$ are block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively, with $N_A(\lambda)$ and $N_D(\lambda)$ being minimal bases dual to $K_A(\lambda)$ and $K_D(\lambda)$, respectively, and $M_C(\lambda)$ and $M_B(\lambda)$ being linear polynomial matrices such that $M_C(\lambda)N_A(\lambda)^T = C(\lambda)$ and $M_B(\lambda)N_D(\lambda)^T = B(\lambda)$, respectively. Let $\lambda_0 \in \mathbb{F}$ such that det $A(\lambda_0) \neq 0$. Then the following statements hold:

(a) The linear map

$$F_r : \mathcal{N}_r(R(\lambda_0)) \longrightarrow \mathcal{N}_r(\mathcal{L}(\lambda_0))$$
$$x \longmapsto \widetilde{x} := \begin{bmatrix} -N_A(\lambda_0)^T A(\lambda_0)^{-1} B(\lambda_0) \\ N_D(\lambda_0)^T \end{bmatrix} x$$

is a bijection between the right nullspaces over \mathbb{F} of $R(\lambda_0)$ and $\mathcal{L}(\lambda_0)$.

(b) The linear map

$$F_{\ell} : \mathcal{N}_{\ell}(R(\lambda_0)) \longrightarrow \mathcal{N}_{\ell}(\mathcal{L}(\lambda_0))$$
$$y^T \longmapsto \widetilde{y}^T := y^T \begin{bmatrix} M_C(\lambda_0) L_A(\lambda_0)^{-1} & I_p & -M_R(\lambda_0) \widehat{N}_D(\lambda_0)^T \end{bmatrix}$$

is a bijection between the left nullspaces over \mathbb{F} of $R(\lambda_0)$ and $\mathcal{L}(\lambda_0)$, where $L_A(\lambda) := \begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$, $M_R(\lambda) := M_D(\lambda) + C(\lambda)A(\lambda)^{-1}M_B(\lambda)$ and $\widehat{N}_D(\lambda)$ is defined in (6.7).

Remark 6.4.13. Let $\mathcal{L}(\lambda)$ be as in Proposition 6.4.12, let $\lambda_0 \in \mathbb{F}$ be an eigenvalue of $\mathcal{L}(\lambda)$ such that det $A(\lambda_0) \neq 0$, and let \tilde{x} and \tilde{y}^T be, respectively, right and left eigenvectors of $\mathcal{L}(\lambda)$ with eigenvalue λ_0 .

By Proposition 6.4.12, the vector \tilde{y}^T must be of the form

$$\widetilde{y}^T = y^T \begin{bmatrix} M_C(\lambda_0) L_A(\lambda_0)^{-1} & I_p & -M_R(\lambda_0) \widehat{N}_D(\lambda_0)^T \end{bmatrix},$$

for some left eigenvector y^T of $R(\lambda)$ with eigenvalue λ_0 . Hence, one can readily recover a left eigenvector y^T of $R(\lambda)$ from the middle block of \tilde{y}^T . Furthermore, from Proposition 6.4.12, we get that \tilde{x} must be of the form

$$\widetilde{x} = \begin{bmatrix} -N_A(\lambda_0)^T A(\lambda_0)^{-1} B(\lambda_0) \\ N_D(\lambda_0)^T \end{bmatrix} x,$$

for some right eigenvector x of $R(\lambda)$ with eigenvalue λ_0 . Since the polynomial matrix $\widehat{K}_D(\lambda)$ in (6.7) satisfies $\widehat{K}_D(\lambda)N_D(\lambda)^T = I_m$ for all $\lambda \in \mathbb{F}$, we have $\widehat{K}_D(\lambda_0)N_D(\lambda_0)^T x = x$. Thus, one can also recover a right eigenvector of $R(\lambda)$ from the right eigenvector \widetilde{x} of $\mathcal{L}(\lambda)$.

6.5 Application to scalar rational equations

In this section, we show by example how the theory developed in this chapter can be used for solving (scalar) rational equations of the form

$$\frac{c(\lambda)}{a(\lambda)} = -\frac{d(\lambda)}{b(\lambda)},\tag{6.25}$$

where $a(\lambda)$, $b(\lambda)$, $c(\lambda)$ and $d(\lambda)$ are nonzero scalar polynomials, and where the numerators and the denominators of each rational function can be expressed in terms of different polynomial bases. For instance, let us assume that the polynomials $a(\lambda)$ and $c(\lambda)$ are written in terms of the monomial basis, that is,

$$a(\lambda) = \sum_{i=0}^{n} a_i \lambda^i$$
 and $c(\lambda) = \sum_{i=0}^{n} c_i \lambda^i$,

with $n = \max\{\deg a(\lambda), \deg c(\lambda)\}$, and that the polynomials $b(\lambda)$ and $d(\lambda)$ are written in terms of Chebyshev polynomials of the first kind $\{\phi_j(\lambda)\}_{j=0}^{\infty}$, that is,

$$b(\lambda) = \sum_{i=0}^{m} b_i \phi_i(\lambda)$$
 and $d(\lambda) = \sum_{i=0}^{m} d_i \phi_i(\lambda)$,

with $m = \max\{\deg b(\lambda), \deg d(\lambda)\}$. We recall that the Chebyshev basis $\{\phi_j(\lambda)\}_{j=0}^{\infty}$ satisfies the three-term recurrence relation:

$$\frac{1}{2}\phi_{j+1}(\lambda) = \lambda\phi_j(\lambda) - \frac{1}{2}\phi_{j-1}(\lambda) \quad j \ge 1$$
(6.26)

where $\phi_0(\lambda) = 1$ and $\phi_1(\lambda) = \lambda$.

Notice that, outside the set of the roots of $b(\lambda)$, that is, in $\Omega := \mathbb{C} \setminus \{\lambda_0 \in \mathbb{C} : b(\lambda_0) = 0\}$, equation (6.25) is equivalent to the equation

$$r(\lambda) := d(\lambda) + c(\lambda)a(\lambda)^{-1}b(\lambda) = 0.$$
(6.27)

For computing the roots of (6.27), that is, the zeros that are not poles, we consider a linear polynomial system matrix of the form

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_a(\lambda) & M_b(\lambda) \\ K_a(\lambda) & 0 \\ \hline -M_c(\lambda) & M_d(\lambda) \\ 0 & K_d(\lambda) \end{bmatrix},$$

where

$$M_{a}(\lambda) := \begin{bmatrix} a_{n}\lambda + a_{n-1} & a_{n-2} & a_{n-3} & \cdots & a_{1} & a_{0} \end{bmatrix},$$

$$M_{c}(\lambda) := \begin{bmatrix} c_{n}\lambda + c_{n-1} & c_{n-2} & c_{n-3} & \cdots & c_{1} & c_{0} \end{bmatrix},$$

and

$$K_{a}(\lambda) := \begin{bmatrix} -1 & \lambda & 0 & \cdots & 0 \\ 0 & -1 & \lambda & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & 0 \\ 0 & \cdots & 0 & -1 & \lambda \end{bmatrix} \quad \text{and} \quad N_{a}(\lambda)^{T} = \begin{bmatrix} \lambda^{n-1} \\ \vdots \\ \lambda \\ 1 \end{bmatrix}$$

is a pair of dual minimal bases, and

$$M_b(\lambda) := \begin{bmatrix} 2b_m \lambda + b_{m-1} & b_{m-2} - b_m & b_{m-3} & \cdots & b_1 & b_0 \end{bmatrix}, M_d(\lambda) := \begin{bmatrix} 2d_m \lambda + d_{m-1} & d_{m-2} - d_m & d_{m-3} & \cdots & d_1 & d_0 \end{bmatrix},$$

and, by (6.26),

$$K_d(\lambda) = \begin{bmatrix} -\frac{1}{2} & \lambda & -\frac{1}{2} & 0 & \cdots & 0\\ 0 & -\frac{1}{2} & \lambda & -\frac{1}{2} & \ddots & \vdots\\ \vdots & \ddots & \ddots & \ddots & \ddots & 0\\ \vdots & \ddots & -\frac{1}{2} & \lambda & -\frac{1}{2}\\ 0 & \cdots & \cdots & 0 & -1 & \lambda \end{bmatrix} \quad \text{and} \quad N_d(\lambda)^T = \begin{bmatrix} \phi_{m-1}(\lambda)\\ \phi_{m-2}(\lambda)\\ \vdots\\ \phi_{1}(\lambda)\\ \phi_{0}(\lambda) \end{bmatrix}$$

is another pair of dual minimal bases. Observe that $a(\lambda) = M_a(\lambda)N_a(\lambda)^T$, $c(\lambda) = M_c(\lambda)N_a(\lambda)^T$, $b(\lambda) = M_b(\lambda)N_d(\lambda)^T$ and $d(\lambda) = M_d(\lambda)N_d(\lambda)^T$.

It is immediate that the matrices

$$\begin{bmatrix} a(\lambda_0) \\ c(\lambda_0) \end{bmatrix} \quad \text{and} \quad \begin{bmatrix} a(\lambda_0) & b(\lambda_0) \end{bmatrix}$$

have full rank (equal to 1) at every $\lambda_0 \in \Omega$ that is not a root of $a(\lambda)$ and $c(\lambda)$ simultaneously. Hence, if $\frac{c(\lambda)}{a(\lambda)}$ is irreducible, i.e., $a(\lambda)$ and $c(\lambda)$ do not have roots in common, then, by Theorem 6.1.2, $\mathcal{L}(\lambda)$ is a linearization of $r(\lambda)$ in Ω . Therefore, the zeros of $\mathcal{L}(\lambda)$ in Ω are the zeros of $r(\lambda)$ in Ω .

The idea of transforming the rational problem (6.25) into an eigenvalue problem is not new [77]. An algorithm based on the Ehrlich-Aberth iteration that uses this approach can be found in [76].

134 CHAPTER 6. LINEARIZATIONS FROM GENERAL REPRESENTATIONS

Chapter 7

Strongly minimal linearizations of rational transfer functions

In this chapter we introduce the conditions for a linear polynomial system matrix

$$L(\lambda) := \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix} := \begin{bmatrix} \lambda A_1 - A_0 & B_0 - \lambda B_1 \\ \lambda C_1 - C_0 & \lambda D_1 - D_0 \end{bmatrix},$$
(7.1)

where $A(\lambda)$ is assumed regular, to be strongly minimal and prove that the strong minimality conditions imply the strong irreducibility conditions in [91]. We remark that, although the notions of irreducible or minimal polynomial system matrix refer to the same conditions in Definition 2.4.2, the conditions for a polynomial system matrix to be strongly irreducible or strongly minimal are different in general.

Recall that a strongly minimal polynomial system matrix, introduced in Chapter 4 (Definition 4.1.15), contains the complete finite and infinite pole and zero structures of its transfer function matrix

$$R(\lambda) := D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda).$$

We will see that strongly minimal linear polynomial system matrices preserve the minimal indices of their transfer function matrices $R(\lambda)$ as well, when $R(\lambda)$ is singular. Then, the pole structure, zero structure and null space structure of $R(\lambda)$ can be computed with the staircase algorithm and the QZ algorithm applied to $A(\lambda)$ and $L(\lambda)$. We will also show that when the strong minimality conditions are not satisfied, we can reduce the linear system matrix to one where they are satisfied, and this without modifying the transfer function matrix. Such a procedure was already derived in [87], but only for linear system matrices that were already minimal at finite points. In this chapter we extend this to arbitrary linear system matrices.

For the particular case of proper rational matrices $R_p(\lambda)$, it is known that they can be written in state-space form as $R_p(\lambda) := D + C(\lambda I - A)^{-1}B$. Then, for the corresponding state-space linear system matrix

$$L_p(\lambda) := \begin{bmatrix} \lambda I - A & -B \\ C & D \end{bmatrix},$$

there are algorithms available in the literature to derive a minimal state-space linear system matrix from a non-minimal one, and these algorithms are based on unitary transformations only [84]. When allowing generalized state-space linear system matrices, then all the rational matrices can be realized by linear system matrices of the form

$$L_g(\lambda) := \begin{bmatrix} \lambda E - A & -B \\ C & D \end{bmatrix},$$
(7.2)

where the matrix E is allowed to be singular. Moreover, when the pencils

$$\begin{bmatrix} \lambda E - A & -B \end{bmatrix}, \begin{bmatrix} \lambda E - A \\ C \end{bmatrix}$$
 (7.3)

have, respectively, full row rank and column rank for all finite λ , then we retrieve the irreducibility or minimality conditions of Rosenbrock in (2.4.2), which imply that the finite poles of $R(\lambda) := D + C(\lambda E - A)^{-1}B$ are the finite eigenvalues of $\lambda E - A$ and the finite zeros of $R(\lambda)$ are the finite eigenvalues of $L_a(\lambda)$. It was shown in [92] that when imposing also the conditions that the pencil in (7.2) is strongly irreducible, meaning that the matrices in (7.3) have full row rank for all finite and infinite λ , then also the infinite pole and zero structure of $R(\lambda)$ can be retrieved from the infinite structure of $\lambda E - A$ and $L_q(\lambda)$, respectively, and that the left and right minimal indices of $R(\lambda)$ and $L_q(\lambda)$ are also the same. Moreover, a reduction procedure to derive a strongly irreducible generalized state-space linear system from a reducible one was also given in [84], and it is also based on unitary transformations only. In [91] these results were then extended to arbitrary polynomial system matrices, but the procedure required irreducibility tests that were more involved. In this chapter, we will show that these conditions can be simplified when the polynomial system matrix is linear. In addition, the reduction procedure presented in this chapter to derive strongly minimal linear systems only uses unitary equivalence transformations. This implies that numerical errors performed during the reduction procedure remain bounded. Since we use unitary transformations in both the reduction procedure and the computation of the eigenstructure, this guarantees that we compute the exact eigenstructure of a perturbed linear polynomial system matrix, but where the perturbation is of the order of the machine precision. The main results in this chapter appear in [32].

Recall that in Chapter 6, we construct linear polynomial system matrices from arbitrary transfer function matrices. Then, by combining the results in this chapter with those in Chapter 6, we can get strongly minimal linearizations for any of the representations of an arbitrary rational matrix considered in Chapter 6. More precisely, given a rational matrix $R(\lambda)$, we can use Theorem 6.1.2 to construct a linear polynomial system matrix of $R(\lambda)$ and, if the obtained system matrix is not strongly minimal, we can apply the reduction procedure in this chapter to obtain a strongly minimal one.

7.1 Some preliminaries

Throughout this chapter we only consider the field of complex number \mathbb{C} . As we explain in Chapter 2, the strictly positive structural indices in the local Smith-McMillan form of a rational matrix correspond to a zero, and the strictly negative structural indices correspond to a pole. Then, the zero degree is defined as the sum of all structural indices of all zeros (infinity included), and the polar degree is the sum of all structural indices (in absolute value) of all poles (infinity included). The polar degree of a rational matrix $R(\lambda)$ is also called as the *McMillan degree*, denoted by $\delta(R)$. The following degree sum theorem was proven in [92], and relates the McMillan degree to the other structural elements of $R(\lambda)$: to the the zero degree $\delta_z(R)$, to the left nullspace degree $\delta_\ell(R)$, that is the sum of all right minimal indices.

Theorem 7.1.1. Let $R(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$. Then

$$\delta(R) := \delta_p(R) = \delta_z(R) + \delta_\ell(R) + \delta_r(R).$$

Example 7.1.2. Let us consider the 2×2 rational matrix

$$R(\lambda) = \begin{bmatrix} e_5(\lambda) & 0\\ c/\lambda & e_1(\lambda) \end{bmatrix}$$
(7.4)

where $e_5(\lambda)$ is a monic polynomial of degree 5 and $e_1(\lambda)$ is a monic polynomial of degree 1, with $e_5(0) \neq 0$ and $e_1(0) \neq 0$. If $c \neq 0$, the only poles are 0 and infinity, and the corresponding local Smith-McMillan forms for these two points are

$$\lambda_0 = 0$$
: diag $(\lambda^{-1}, \lambda^1)$, $\lambda_0 = \infty$: diag $((1/\lambda)^{-5}, (1/\lambda)^{-1})$

indicating that $\lambda_0 = 0$ is a zero as well as a pole. The other finite zeros are the six finite roots of $e_5(\lambda)$ and $e_1(\lambda)$. The polar degree and the zero degree for this example are thus both equal to 7. When c = 0, the pole and zero at $\lambda = 0$ disappear and the matrix is polynomial instead of rational. The polar and zero degree are then both equal to 6.

7.2 Strong irreducibility and minimality

In this section we recall the strong irreducibility conditions in [91] for polynomial system matrices, and we recall the notion of strong minimality. Then, we study the relation between them for the case of linear system matrices. Consider a polynomial system matrix

$$S(\lambda) := \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix},$$
(7.5)

where $A(\lambda)$ is assumed regular, of arbitrary degree d.

Definition 7.2.1. A polynomial system matrix $S(\lambda)$ as in (7.5) is said to be strongly controllable and strongly observable, respectively, if the polynomial matrices

$$\begin{bmatrix} A(\lambda) & -B(\lambda) & 0\\ C(\lambda) & D(\lambda) & -I \end{bmatrix}, \text{ and } \begin{bmatrix} A(\lambda) & -B(\lambda)\\ C(\lambda) & D(\lambda)\\ 0 & I \end{bmatrix},$$
(7.6)

have no finite or infinite zeros. If both conditions are satisfied $S(\lambda)$ is said to be strongly irreducible.

Let us now consider the transfer function matrix $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$ of the polynomial system matrix in (7.5). In such a case, we also say that the system quadruple $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ realizes $R(\lambda)$. Moreover, we say that the system quadruple is strongly irreducible if the polynomial system matrix is strongly irreducible. It was shown in [91] that the pole/zero and null space structure of $R(\lambda)$ can be retrieved from a strongly irreducible system quadruple $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ as follows.

Theorem 7.2.2. If the polynomial system matrix $S(\lambda)$ in (7.5) is strongly irreducible, then

- 1. the zero structure of $R(\lambda)$ at finite and infinite λ is the same as the zero structure of $S(\lambda)$ at finite and infinite λ ,
- 2. the pole structure of $R(\lambda)$ at finite λ is the same as the zero structure at λ of $A(\lambda)$,
- 3. the pole structure of $R(\lambda)$ at infinity is the same as the zero structure at infinity of

$$\begin{bmatrix} A(\lambda) & -B(\lambda) & 0\\ C(\lambda) & D(\lambda) & -I\\ 0 & I & 0 \end{bmatrix},$$

4. the left and right minimal indices of $R(\lambda)$ and $S(\lambda)$ are the same.

If one specializes this to the generalized state space model (7.2) one retrieves the results of [92], which are simpler and only involve the pencils $\lambda E - A$, for the finite and infinite pole structure; (7.2), for the finite and infinite zero structure and (7.3), for the definition of strongly irreducible.

We now show that the above conditions can be simplified when the system matrices are linear as in (7.1). First, we present the definition of strongly E-controllable and strongly E-observable polynomial system matrices, and based on them, we redefined the concept of strongly minimal polynomial system matrix. It is easy to see that this definition is equivalent to that in Definition 4.1.15. **Definition 7.2.3.** Let d be the degree of the polynomial system matrix $S(\lambda)$ in (7.5). $S(\lambda)$ is said to be strongly E-controllable and strongly E-observable, respectively, if the polynomial matrices

$$\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$$
, and $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix}$, (7.7)

have degree exactly d and have no finite or infinite eigenvalues. If both conditions are satisfied $S(\lambda)$ is said to be strongly minimal.

The letter E in the definition of strong E-controllability and E-observability refers to the condition of the matrices in (7.7) not having eigenvalues, finite or infinite, and emphasizes the differences with the concepts of "strong controlability, observability and irreducibility" used in [92, 91].

7.2.1 Strongly minimal linearizations

In this work we focus on linear polynomial system matrices. This means that we consider block partitioned pencils

$$L(\lambda) := \begin{bmatrix} \lambda A_1 - A_0 & B_0 - \lambda B_1 \\ \lambda C_1 - C_0 & \lambda D_1 - D_0 \end{bmatrix} =: \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix},$$
(7.8)

where $A(\lambda)$ is regular.

7.2.1.1 Finite pole and zero structure

A linear polynomial system matrix $L(\lambda)$ as in (7.8) contains the finite zero and pole structures of its transfer function matrix $R(\lambda) := D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$ provided that $L(\lambda)$ is minimal. That is, if matrices

$$\begin{bmatrix} \lambda A_1 - A_0 & -\lambda B_1 + B_0 \end{bmatrix}, \begin{bmatrix} \lambda A_1 - A_0 \\ \lambda C_1 - C_0 \end{bmatrix},$$
(7.9)

have, respectively, full row and column rank for all $\lambda_0 \in \mathbb{C}$. This is equivalent to state that the pencils in (7.9) do not have finite eigenvalues. If $L(\lambda)$ is minimal then we know by Rosenbrock [78] (recall Theorem 2.4.4) that the finite zero structure of $R(\lambda)$ is the same as the finite zero structure of $L(\lambda)$, and that the finite pole structure of $R(\lambda)$ is the same as the finite zero structure of $\lambda A_1 - A_0$.

Since dim $\mathcal{N}_r(L) = \dim \mathcal{N}_r(R)$ and dim $\mathcal{N}_\ell(L) = \dim \mathcal{N}_\ell(R)$ (recall (2.10)), we have that the number of right (resp. left) minimal indices of a minimal polynomial system matrix is equal to the number of right (resp. left) minimal indices of its transfer function matrix. However, their values may be different [92, 8].

7.2.1.2 Finite and infinite pole and zero structure

Notice that Theorem 2.4.4 does not provide information about the structure at infinity. The recovering of this structure requires the following concept introduced in Chapter 4: $L(\lambda)$ in (7.8) is minimal at infinity if the matrices

$$\begin{bmatrix} A_1 & -B_1 \end{bmatrix}$$
 and $\begin{bmatrix} A_1 \\ C_1 \end{bmatrix}$ (7.10)

have, respectively, full row and column rank. This condition is equivalent to state that the pencils in (7.9) have degree exactly 1 and do not have eigenvalues at ∞ . A linear polynomial system matrix that is minimal (at finite points) and also minimal at ∞ is strongly minimal according to Definition 7.2.3. More precisely, a linear polynomial system matrix $L(\lambda)$ as in (7.8) is strongly minimal if the pencils

$$\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$$
, and $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix}$, (7.11)

have degree exactly 1 and have no finite or infinite eigenvalues. As mentioned before, the degree 1 pencils in (7.11) do not have infinite eigenvalues if and only if the matrices in (7.10) have full row and full column rank, respectively. The ranks of the matrices in (7.10) will be also called the *ranks at infinity* of the pencils in (7.7), even in the case the matrices in (7.10) do not have full ranks.

We say that strongly minimal linear polynomial system matrices are strongly minimal linearizations of their transfer function matrices.

Definition 7.2.4. Let $R(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$ be a rational matrix. A linear polynomial system matrix $L(\lambda)$ as in (7.8) is said to be a strongly minimal linearization of $R(\lambda)$ if $L(\lambda)$ is strongly minimal and its transfer function matrix is $R(\lambda)$. Equivalently, $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ is said to be a strongly minimal linear realization of $R(\lambda)$.

We prove in Proposition 7.2.10 that, for linear polynomial system matrices, the strong irreducibility conditions hold if the strong minimality conditions are satisfied. For this, we need to recall Lemma 1 of [92], which we give here in its transposed form. Then, we prove Theorems 7.2.6 and 7.2.7, and Proposition 7.2.10 as a corollary of them.

Lemma 7.2.5. The zero structure at infinity of the pencil $\begin{bmatrix} \lambda K_1 - K_0 & | & -L_0 \end{bmatrix}$ where K_1 has full column rank, is isomorphic to the zero structure at zero of the pencil $\begin{bmatrix} K_1 - \mu K_0 & | & -L_0 \end{bmatrix}$. Moreover, if the pencil has full row normal rank, then it has no zeros at infinity, provided the constant matrix $\begin{bmatrix} K_1 & | & -L_0 \end{bmatrix}$ has full row rank.

Proof. The first part is proven in [92]. The second part is a direct consequence of the first part, when we evaluate at $\mu = 0$.

Theorem 7.2.6. The pencil

$$\begin{bmatrix} \lambda A_1 - A_0 & B_0 - \lambda B_1 & 0\\ \lambda C_1 - C_0 & \lambda D_1 - D_0 & -I \end{bmatrix},$$
(7.12)

where $\lambda A_1 - A_0$ is regular, has no zeros at infinity if the pencil

$$\begin{bmatrix} \lambda A_1 - A_0 & B_0 - \lambda B_1 \end{bmatrix}$$
(7.13)

has no eigenvalues at infinity.

Proof. Clearly the pencils in (7.12) and (7.13) have full row normal rank since $\lambda A_1 - A_0$ is regular. We can thus apply the result of Lemma 7.2.5 as follows. If we use an invertible matrix V to "compress" the columns of the coefficient of λ in the following pencil

$$\begin{bmatrix} \lambda A_1 - A_0 & B_0 - \lambda B_1 & 0\\ \lambda C_1 - C_0 & \lambda D_1 - D_0 & -I \end{bmatrix} \begin{bmatrix} V & 0\\ 0 & I \end{bmatrix} = \begin{bmatrix} \lambda K_1 - K_0 & -L_0 & 0\\ \lambda \widehat{K}_1 - \widehat{K}_0 & -\widehat{L}_0 & -I \end{bmatrix},$$

such that the matrix $\begin{bmatrix} K_1 \\ \widehat{K}_1 \end{bmatrix}$ has full column rank, then this pencil has no zeros at infinity provided the constant matrix $\begin{bmatrix} K_1 & -L_0 & 0 \\ \widehat{K}_1 & -\widehat{L}_0 & -I \end{bmatrix}$ has full row rank. But if $\begin{bmatrix} \lambda A_1 - A_0 & B_0 - \lambda B_1 \end{bmatrix}$ has no infinite eigenvalues, it follows that $\begin{bmatrix} A_1 & -B_1 \end{bmatrix}$ has full row rank. And since $\begin{bmatrix} A_1 & -B_1 \end{bmatrix} V = \begin{bmatrix} K_1 & 0 \end{bmatrix}$, K_1 must have full row rank as well (in fact, it is invertible). It then follows from Lemma 7.2.5 that the pencil in (7.12) has no zeros at infinity.

In the next theorem, we state without proof the transposed version of Theorem 7.2.6.

Theorem 7.2.7. The pencil

$$\begin{bmatrix} \lambda A_1 - A_0 & B_0 - \lambda B_1 \\ \lambda C_1 - C_0 & \lambda D_1 - D_0 \\ 0 & I \end{bmatrix},$$

where $\lambda A_1 - A_0$ is regular, has no zeros at infinity if the pencil

$$\begin{bmatrix} \lambda A_1 - A_0 \\ \lambda C_1 - C_0 \end{bmatrix}$$
(7.14)

has no eigenvalues at infinity.

By Theorems 7.2.6, 7.2.7 and [91], we have the next result for recovering the structure at infinity.

Theorem 7.2.8. Let $R(\lambda)$ be the transfer function matrix of $L(\lambda)$ in (7.8). If $L(\lambda)$ is minimal at ∞ then

- 1. the zero structure of $R(\lambda)$ at infinity is the same as the zero structure of $L(\lambda)$ at infinity, and
- 2. the polar structure of $R(\lambda)$ at infinity is the same as the zero structure of the pencil

$$\begin{bmatrix} \lambda A_1 - A_0 & -\lambda B_1 & 0\\ \lambda C_1 & \lambda D_1 & -I_m\\ 0 & I_n & 0 \end{bmatrix}$$
(7.15)

at infinity.

The polar structure of $R(\lambda)$ at ∞ can also be recovered without considering the extended pencil in (7.15). In particular, both the zero and polar structures of $R(\lambda)$ at infinity can be obtained from the eigenvalue structures of the pencils $L(\lambda)$ and $A(\lambda)$ at infinity as Theorem 7.2.9 shows (see Chapter 4 or [28]).

Theorem 7.2.9. Let $R(\lambda)$ be the transfer function matrix of $L(\lambda)$ in (7.8). Assume that $R(\lambda)$ has normal rank r. Let $e_1 \leq \cdots \leq e_s$ be the partial multiplicities of rev₁ $A(\lambda)$ at 0 and let $\tilde{e}_1 \leq \cdots \leq \tilde{e}_u$ be the partial multiplicities of rev₁ $L(\lambda)$ at 0. If $L(\lambda)$ is minimal at ∞ then the structural indices at infinity $q_1 \leq \cdots \leq q_r$ of $R(\lambda)$ are

$$(d_1, d_2, \dots, d_r) = (-e_s, -e_{s-1}, \dots, -e_1, \underbrace{0, \dots, 0}_{r-s-u}, \widetilde{e}_1, \widetilde{e}_2, \dots, \widetilde{e}_u) - (1, 1, \dots, 1).$$

Notice that if a linear system matrix $L(\lambda)$ is minimal (i.e., satisfies (7.9)) and, in addition, satisfies the conditions in (7.13) and (7.14), then it is strongly minimal. By Theorems 7.2.6 and 7.2.7, we have that these conditions imply strong irreducibility on linear system matrices. We state such result in Proposition 7.2.10.

Proposition 7.2.10. A linear system matrix as in (7.8) is strongly irreducible if it is strongly minimal.

Theorems 7.2.6 and 7.2.7 and Proposition 7.2.10 can be extended to polynomial system matrices. However, we do not state these results here since we are focusing on linear system matrices.

Strongly minimal linearizations $L(\lambda)$ of a rational matrix $R(\lambda)$ have been defined with the goal of constructing pencils that allow us to recover the complete pole and zero structures of $R(\lambda)$ through Theorems 2.4.4 and 7.2.9, or 7.2.8. Surprisingly, the condition of strong minimality implies that the minimal indices of $L(\lambda)$ and $R(\lambda)$ are the same. This is proved in Theorem 7.2.11, which, together with Theorems 2.4.4 and 7.2.9, allows us to recover the complete eigenstructure of a rational matrix from any of its strongly minimal linearizations. **Theorem 7.2.11.** Let $L(\lambda)$ be a strongly minimal linearization of a rational matrix $R(\lambda)$. Then the left and right minimal indices of $R(\lambda)$ are the same as the left and right minimal indices of $L(\lambda)$.

Proof. By Proposition 7.2.10, a strongly minimal linear polynomial system matrix is strongly irreducible. Then, by [91, Result 2], the left and right minimal indices of $R(\lambda)$ and $L(\lambda)$ are the same.

Remark 7.2.12. It follows from Theorems 2.4.4, 7.2.8 and 7.2.11 that, if $L(\lambda)$ is a strongly minimal linearization of a rational matrix $R(\lambda)$, then

$$\delta_z(R) + \delta_\ell(R) + \delta_r(R) = \delta_z(L) + \delta_\ell(L) + \delta_r(L),$$

and then from Theorem 7.1.1 that $\delta_p(R) = \delta_p(L)$. But the only pole of $L(\lambda) := \lambda L_1 + L_0$ is the point at infinity and its polar degree $\delta_p(L)$ is equal to rank L_1 [84, p. 126]. Therefore, the McMillan degree $\delta_p(R)$ of $R(\lambda)$ equals the rank of L_1 for any strongly minimal linearization of $R(\lambda)$, and no other pencils with the same zero structure and the same left and right minimal indices as $R(\lambda)$ can have a first order coefficient with smaller rank. Thus, strongly minimal linearizations are optimal in this sense.

7.2.2 Strongly minimal linearizations and their relation with other classes of linearizations

In this subsection we study the relation of strongly minimal linearizations with other definitions of strong linearizations of polynomial and rational matrices in the literature (recall Section 2.5).

For polynomial matrices, first we can combine Theorem 2.4.4 applied to a polynomial matrix $P(\lambda)$ and the equality of the number of the minimal indices of $L(\lambda)$ and $P(\lambda)$ with [22, Theorem 4.1] for proving that any minimal linear polynomial system matrix of a polynomial matrix $P(\lambda)$ is always a GLR-linearization of $P(\lambda)$. The reverse result is not true in general. Observe also that any minimal polynomial system matrix of a polynomial matrix $P(\lambda)$ must have the block $A(\lambda)$ in (7.8) unimodular, because $P(\lambda)$ does not have finite poles. However, the following example shows that strongly minimal linearizations for polynomial matrices are not, in general, GLR-strong linearizations.

Example 7.2.13. (Strongly minimal linearizations of polynomial matrices are not strong linearizations in the sense of Gohberg, Lancaster and Rodman) Consider the polynomial matrix

$$P(\lambda) = \lambda^2 \begin{bmatrix} 0 & 0 \\ 0 & 1 \end{bmatrix} + \lambda \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix} + \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}$$

and the partitioned pencil

$$L(\lambda) = \begin{bmatrix} -1 & 0 & \lambda \\ 0 & \lambda + 1 & 0 \\ \lambda & 0 & \lambda + 1 \end{bmatrix}.$$

The transfer function matrix of $L(\lambda)$ is $P(\lambda)$ and $L(\lambda)$ is minimal and minimal at infinity. Therefore, $L(\lambda)$ is a strongly minimal linearization of $P(\lambda)$ and also a GLR-linearization of $P(\lambda)$. However, $\operatorname{rev}_1 L(\lambda)$ is not unimodularly equivalent to diag($\operatorname{rev}_2 P(\lambda), 1$) and, thus, $L(\lambda)$ is not a GLR-strong linearization of $P(\lambda)$. In order to see this, observe that

$$\operatorname{rev}_2 P(\lambda) = \lambda^2 \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix} + \lambda \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix} + \begin{bmatrix} 0 & 0 \\ 0 & 1 \end{bmatrix}$$

and

$$\mathrm{rev}_1 L(\lambda) = \begin{bmatrix} -\lambda & 0 & 1 \\ 0 & \lambda + 1 & 0 \\ 1 & 0 & \lambda + 1 \end{bmatrix},$$

which makes it transparent that $\operatorname{rev}_1 L(\lambda)$ does not have eigenvalues (or zeros) at zero, while $\operatorname{rev}_2 P(\lambda)$ does. In general, it is possible to prove by using Theorem 7.2.9 that strongly minimal linearizations of polynomial matrices of degree larger than 1 with eigenvalues at infinity are not GLR-strong linearizations.

Despite of the fact of strongly mininal linearizations not being GLR-strong linearizations, strongly minimal linearizations $L(\lambda)$ of a polynomial matrix $P(\lambda)$ allow us to recover always the complete list of structural data of $P(\lambda)$, including its minimal indices. Moreover, the minimal indices of $L(\lambda)$ and $P(\lambda)$ are always the same in contrast with the minimal indices of GLR-strong linearizations whose values may be different to those of $P(\lambda)$. In addition, we will prove in Chapter 8 that strongly minimal linearizations allow us to preserve structures of polynomial matrices that cannot always be preserved by GLR-strong linearizations.

For rational matrices, the minimal linear polynomial system matrices of an arbitrary rational matrix $R(\lambda)$ are particular cases of the linearizations of $R(\lambda)$ defined in [6], which were introduced with the idea of combining the concept of minimal polynomial system matrix with the extension of GLR-linearizations from polynomial to rational matrices. However, strongly minimal linearizations are not always strong linearizations in the sense of [6] (Definition 2.5.3) since the first degree coefficients of their (1, 1)-blocks are not necessarily invertible. But, as in the polynomial case, strongly minimal linearizations of a rational matrix allow us to recover always the complete list of its structural data, including its minimal indices.

7.3 Reduction to a strongly minimal linear system matrix

In this section we give an algorithm to reduce an arbitrary linear system matrix to a strongly minimal one.

Given a linear system quadruple $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$, where $A(\lambda) \in \mathbb{C}(\lambda)^{d \times d}$, $B(\lambda) \in \mathbb{C}(\lambda)^{d \times n}, C(\lambda) \in \mathbb{C}(\lambda)^{m \times d}, D(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$ and $A(\lambda)$ is assumed to be regular, we describe first how to obtain a strongly E-controllable quadruple $\{A_c(\lambda), B_c(\lambda), C_c(\lambda), D_c(\lambda)\}$ of smaller state dimension (d-r). For that, our reduction procedure deflates finite and infinite "uncontrollable eigenvalues" by proceeding in three different steps. Then the reduction to a strongly E-observable one is dual and can be obtained by mere transposition of the system matrix and application of the first method for obtaining a strongly E-controllable system.

Step 1: We first show that there exist unitary transformations U and V that yield a decomposition of the type

$$\begin{bmatrix} U & 0 \\ 0 & I_m \end{bmatrix} \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} V & 0 \\ 0 & I_n \end{bmatrix} = \begin{bmatrix} X(\lambda)\widehat{W}_{11} & 0 & X(\lambda)W_{13} \\ \widetilde{Y}(\lambda) & \widetilde{A}(\lambda) & -\widetilde{B}(\lambda) \\ \widetilde{Z}(\lambda) & \widetilde{C}(\lambda) & D(\lambda) \end{bmatrix},$$
(7.16)

where $\widehat{W}_{11} \in \mathbb{C}^{r \times r}$ and $W_{13} \in \mathbb{C}^{r \times n}$ are constant, and \widehat{W}_{11} is invertible. This will allow us in step 2 to deflate the block $X(\lambda)$ and construct a lower order model that is strongly E-controllable. In order to prove this, we start from the generalized Schur decomposition for singular pencils (see [83])

$$U\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix} W^* = \begin{bmatrix} X(\lambda) & 0 & 0\\ Y(\lambda) & \widehat{A}(\lambda) & -\widehat{B}(\lambda) \end{bmatrix},$$
(7.17)

where $X(\lambda) \in \mathbb{C}[\lambda]^{r \times r}$ is the regular part of $\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$, $\widehat{A}(\lambda) \in \mathbb{C}[\lambda]^{(d-r) \times (d-r)}$, and $\begin{bmatrix} \widehat{A}(\lambda) & -\widehat{B}(\lambda) \end{bmatrix}$ has no finite or infinite eigenvalues anymore. The decomposition in (7.17) can be obtained by using unitary transformations U and W. If we partition U as $\begin{bmatrix} U_1 \\ U_2 \end{bmatrix}$, with $U_1 \in \mathbb{C}^{r \times d}$, then

$$U_1 \begin{bmatrix} A(\lambda) \mid -B(\lambda) \end{bmatrix} = \begin{bmatrix} X(\lambda)W_{11} & X(\lambda)W_{12} \mid X(\lambda)W_{13} \end{bmatrix},$$

where $W_{11} \in \mathbb{C}^{r \times r}$, $W_{12} \in \mathbb{C}^{r \times (d-r)}$ and $W_{13} \in \mathbb{C}^{r \times n}$ are the corresponding submatrices of W. Since $A(\lambda)$ is regular, $X(\lambda) \begin{bmatrix} W_{11} & W_{12} \end{bmatrix}$ must be full normal rank, and hence $\begin{bmatrix} W_{11} & W_{12} \end{bmatrix}$ must be full row rank as well. Therefore, there must exist a unitary matrix V such that $\begin{bmatrix} W_{11} & W_{12} \end{bmatrix} V = \begin{bmatrix} \widehat{W}_{11} & 0 \end{bmatrix}$, where \widehat{W}_{11} is invertible. Hence, we have

$$\begin{bmatrix} U & 0 \\ 0 & I_m \end{bmatrix} \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} V & 0 \\ 0 & I_n \end{bmatrix} = \begin{bmatrix} X(\lambda)\widehat{W}_{11} & 0 & X(\lambda)W_{13} \\ \widetilde{Y}(\lambda) & \widetilde{A}(\lambda) & -\widetilde{B}(\lambda) \\ \widetilde{Z}(\lambda) & \widetilde{C}(\lambda) & D(\lambda) \end{bmatrix},$$

where

$$W \begin{bmatrix} V & 0 \\ 0 & I_n \end{bmatrix} = \begin{bmatrix} \widehat{W}_{11} & 0 & W_{13} \\ \widehat{W}_{21} & \widehat{W}_{22} & W_{23} \\ \widehat{W}_{31} & \widehat{W}_{32} & W_{33} \end{bmatrix}$$

Step 2: We now define $E := -\widehat{W}_{11}^{-1}W_{13}$ and perform the following non-unitary transformation on the pencil:

$$\begin{bmatrix} X(\lambda)\widehat{W}_{11} & 0 & X(\lambda)W_{13} \\ \widetilde{Y}(\lambda) & \widetilde{A}(\lambda) & -\widetilde{B}(\lambda) \\ \widetilde{Z}(\lambda) & \widetilde{C}(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} I_r & 0 & E \\ 0 & I_{d-r} & 0 \\ 0 & 0 & I_n \end{bmatrix}$$
$$=\begin{bmatrix} X(\lambda)\widehat{W}_{11} & 0 & 0 \\ \widetilde{Y}(\lambda) & \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)E - \widetilde{B}(\lambda) \\ \widetilde{Z}(\lambda) & \widetilde{C}(\lambda) & \widetilde{Z}(\lambda)E + D(\lambda) \end{bmatrix}.$$

We have obtained an equivalent system representation in which the (1, 1)-block, $X(\lambda)\widehat{W}_{11}$, can be deflated since it does not contribute to the transfer function. We then obtain a smaller linear system pencil:

$$\left[\begin{array}{cc} \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)E - \widetilde{B}(\lambda) \\ \widetilde{C}(\lambda) & \widetilde{Z}(\lambda)E + D(\lambda) \end{array}\right],$$

that has the same transfer function. One can also perform this elimination by another unitary transformation \widetilde{W} constructed to eliminate W_{13} :

$$\begin{bmatrix} \widehat{W}_{11} & 0 & W_{13} \end{bmatrix} \begin{bmatrix} \widetilde{W}_{11} & 0 & \widetilde{W}_{13} \\ 0 & I_{d-r} & 0 \\ \widetilde{W}_{31} & 0 & \widetilde{W}_{33} \end{bmatrix} = \begin{bmatrix} I_r & 0 & 0 \end{bmatrix},$$
(7.18)

implying $\widetilde{W}_{11} = \widehat{W}_{11}^*$, $\widetilde{W}_{31} = W_{13}^*$, and $\widetilde{W}_{13} = -\widehat{W}_{11}^{-1}W_{13}\widetilde{W}_{33}$. This then yields

$$\begin{bmatrix} X(\lambda)\widehat{W}_{11} & 0 & X(\lambda)W_{13} \\ \widetilde{Y}(\lambda) & \widetilde{A}(\lambda) & -\widetilde{B}(\lambda) \\ \widetilde{Z}(\lambda) & \widetilde{C}(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} \widetilde{W}_{11} & 0 & \widetilde{W}_{13} \\ 0 & I_{d-r} & 0 \\ \widetilde{W}_{31} & 0 & \widetilde{W}_{33} \end{bmatrix}$$

$$= \left[\begin{array}{ccc} X(\lambda) & 0 & 0 \\ \widetilde{Y}(\lambda) \widetilde{W}_{11} - \widetilde{B}(\lambda) \widetilde{W}_{31} & \widetilde{A}(\lambda) & \widetilde{Y}(\lambda) \widetilde{W}_{13} - \widetilde{B}(\lambda) \widetilde{W}_{33} \\ \widetilde{Z}(\lambda) \widetilde{W}_{11} + D(\lambda) \widetilde{W}_{31} & \widetilde{C}(\lambda) & \widetilde{Z}(\lambda) \widetilde{W}_{13} + D(\lambda) \widetilde{W}_{33} \end{array} \right].$$

Notice that the new transfer function has now changed, but only by postmultiplication by the constant matrix \widetilde{W}_{33} , which moreover is invertible. This follows from

$$\left[\begin{array}{c}E\\I_n\end{array}\right]\widetilde{W}_{33}=\left[\begin{array}{c}\widetilde{W}_{13}\\\widetilde{W}_{33}\end{array}\right],$$

expressing that both matrices span the null-space of the same matrix $\begin{bmatrix} \widehat{W}_{11} & W_{13} \end{bmatrix}$ and where the right hand side matrix has full rank since it has orthonormal columns. This also implies that

$$\begin{bmatrix} \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)E - \widetilde{B}(\lambda) \\ \widetilde{C}(\lambda) & \widetilde{Z}(\lambda)E + D(\lambda) \end{bmatrix} \begin{bmatrix} I_{d-r} & 0 \\ 0 & \widetilde{W}_{33} \end{bmatrix} = \begin{bmatrix} \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)\widetilde{W}_{13} - \widetilde{B}(\lambda)\widetilde{W}_{33} \\ \widetilde{C}(\lambda) & \widetilde{Z}(\lambda)\widetilde{W}_{13} + D(\lambda)\widetilde{W}_{33} \end{bmatrix},$$

which shows that their Schur complements are related by the constant matrix \widetilde{W}_{33} .

Step 3: Finally, we show that the submatrix

$$\begin{bmatrix} \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)E - \widetilde{B}(\lambda) \end{bmatrix} \begin{bmatrix} I_{d-r} & 0\\ 0 & \widetilde{W}_{33} \end{bmatrix} = \begin{bmatrix} \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)\widetilde{W}_{13} - \widetilde{B}(\lambda)\widetilde{W}_{33} \end{bmatrix},$$

has no finite or infinite eigenvalues anymore. For this, we first point out that the following product of unitary matrices has the form given below

$$W\begin{bmatrix}V & 0\\0 & I_n\end{bmatrix}\begin{bmatrix}\widetilde{W}_{11} & 0 & \widetilde{W}_{13}\\0 & I_{d-r} & 0\\\widetilde{W}_{31} & 0 & \widetilde{W}_{33}\end{bmatrix} =: \begin{bmatrix}I_r & 0 & 0\\0 & \widetilde{V}_{22} & \widetilde{V}_{23}\\0 & \widetilde{V}_{32} & \widetilde{V}_{33}\end{bmatrix} =: \begin{bmatrix}I_r & 0\\0 & \widetilde{V}\end{bmatrix}$$

because the identity (7.18) implies that the first block column equals $\begin{bmatrix} I_r & 0 & 0 \end{bmatrix}$. This then implies the equality

$$\begin{bmatrix} X(\lambda) & 0 & 0\\ \widetilde{Y}(\lambda)\widetilde{W}_{11} - \widetilde{B}(\lambda)\widetilde{W}_{31} & \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)\widetilde{W}_{13} - \widetilde{B}(\lambda)\widetilde{W}_{33} \end{bmatrix}$$
$$= \begin{bmatrix} X(\lambda) & 0 & 0\\ Y(\lambda) & \widehat{A}(\lambda) & -\widehat{B}(\lambda) \end{bmatrix} \begin{bmatrix} I_r & 0\\ 0 & \widetilde{V} \end{bmatrix},$$

which in turn implies that $\begin{bmatrix} \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)\widetilde{W}_{13} - \widetilde{B}(\lambda)\widetilde{W}_{33} \end{bmatrix}$ has no finite or infinite eigenvalues. We thus have shown that the system matrix

$$S_{c}(\lambda) := \begin{bmatrix} A_{c}(\lambda) & -B_{c}(\lambda) \\ C_{c}(\lambda) & D_{c}(\lambda) \end{bmatrix} := \begin{bmatrix} \widetilde{A}(\lambda) & \widetilde{Y}(\lambda)\widetilde{W}_{13} - \widetilde{B}(\lambda)\widetilde{W}_{33} \\ \widetilde{C}(\lambda) & \widetilde{Z}(\lambda)\widetilde{W}_{13} + D(\lambda)\widetilde{W}_{33} \end{bmatrix}$$

is now strongly E-controllable and that its transfer function $R_c(\lambda)$ equals $R(\lambda)\widetilde{W}_{33}$, where $R(\lambda)$ is the transfer function of the original quadruple and \widetilde{W}_{33} is invertible. We summarize the result obtained by the three-step procedure above in Theorem 7.3.1, where we denote d-r by d_c , to indicate that it is the size of $A_c(\lambda)$ in the new strongly E-controllable system, and r is replaced by $d_{\overline{c}}$, so that $d = d_{\overline{c}} + d_c$.

Theorem 7.3.1. Let $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ be a linear system quadruple, with $A(\lambda) \in \mathbb{C}[\lambda]^{d \times d}$ regular, realizing the rational matrix $R(\lambda) := C(\lambda)A(\lambda)^{-1}B(\lambda) + D(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$. Then there exist unitary transformations $U, V \in \mathbb{C}^{d \times d}$ and $\widetilde{W} \in \mathbb{C}^{(d+n) \times (d+n)}$ such that the following identity holds

$$\begin{bmatrix} U & 0 \\ 0 & I_m \end{bmatrix} \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} V & 0 \\ 0 & I_n \end{bmatrix} \widetilde{W} = \begin{bmatrix} X_{\overline{c}}(\lambda) & 0 & 0 \\ Y_{\overline{c}}(\lambda) & A_c(\lambda) & -B_c(\lambda) \\ Z_{\overline{c}}(\lambda) & C_c(\lambda) & D_c(\lambda) \end{bmatrix},$$
where \widetilde{W} is of the form $\widetilde{W} := \begin{bmatrix} \widetilde{W}_{11} & 0 & \widetilde{W}_{13} \\ 0 & I_{d_c} & 0 \\ \widetilde{W}_{31} & 0 & \widetilde{W}_{33} \end{bmatrix} \in \mathbb{C}^{(d_{\overline{c}}+d_c+n)\times(d_{\overline{c}}+d_c+n)}, d_{\overline{c}}$ is

the number of (finite and infinite) eigenvalues of $\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$, and $X_{\overline{c}}(\lambda) \in \mathbb{C}[\lambda]^{d_{\overline{c}} \times d_{\overline{c}}}$ is a regular pencil. Moreover,

- a) the eigenvalues of $\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$ are the eigenvalues of $X_{\overline{c}}(\lambda)$,
- b) $\begin{bmatrix} A_c(\lambda) & -B_c(\lambda) \end{bmatrix} \in \mathbb{C}[\lambda]^{d_c \times (d_c+n)}$ has no (finite or infinite) eigenvalues,
- c) the quadruple $\{A_c(\lambda), B_c(\lambda), C_c(\lambda), D_c(\lambda)\}$ is a realization of the transfer function $R_c(\lambda) := R(\lambda)\widetilde{W}_{33}$, with $\widetilde{W}_{33} \in \mathbb{C}^{n \times n}$ invertible, and
- d) if $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix}$ has no finite or infinite eigenvalues, then $\begin{bmatrix} A_c(\lambda) \\ C_c(\lambda) \end{bmatrix}$ also has no finite or infinite eigenvalues.

Remark 7.3.2. Notice that conditions b) and d) in Theorem 7.3.1 imply that the system quadruple $\{A_c(\lambda), B_c(\lambda), C_c(\lambda), D_c(\lambda)\}$ is strongly minimal.

Proof. The decomposition and the three properties a), b) and c) were shown in the discussion above. The only part that remains to be proven is property d). This follows from the identity (7.16), which yields

$$\begin{bmatrix} U & 0 \\ 0 & I_m \end{bmatrix} \begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix} V = \begin{bmatrix} X(\lambda)\widehat{W}_{11} & 0 \\ \widetilde{Y}(\lambda) & A_c(\lambda) \\ \widetilde{Z}(\lambda) & C_c(\lambda) \end{bmatrix}.$$

This clearly implies that if $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix}$ has full rank for all λ (including infinity), then so does $\begin{bmatrix} A_c(\lambda) \\ C_c(\lambda) \end{bmatrix}$. We state below a dual theorem that constructs, from an arbitrary linear system quadruple $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$, a subsystem $\{A_o(\lambda), B_o(\lambda), C_o(\lambda), D_o(\lambda)\}$ where $\begin{bmatrix} A_o(\lambda) \\ C_o(\lambda) \end{bmatrix}$ has no finite or infinite eigenvalues. Its proof is obtained by applying the previous theorem on the transposed system $\{A^A(\lambda), C^A(\lambda), B^A(\lambda), D^A(\lambda)\}$ and then transposing back the result.

Theorem 7.3.3. Let $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ be a linear system quadruple, with $A(\lambda) \in \mathbb{C}[\lambda]^{d \times d}$ regular, realizing the rational matrix $R(\lambda) := C(\lambda)A(\lambda)^{-1}B(\lambda) + D(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$. Then there exist unitary transformations $U, V \in \mathbb{C}^{d \times d}$ and $\widetilde{W} \in \mathbb{C}^{(d+m) \times (d+m)}$ such that the following identity holds

$$\widetilde{W} \begin{bmatrix} U & 0 \\ 0 & I_m \end{bmatrix} \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} V & 0 \\ 0 & I_n \end{bmatrix} = \begin{bmatrix} X_{\overline{o}}(\lambda) & Y_{\overline{o}}(\lambda) & Z_{\overline{o}}(\lambda) \\ 0 & A_o(\lambda) & -B_o(\lambda) \\ 0 & C_o(\lambda) & D_o(\lambda) \end{bmatrix},$$

where \widetilde{W} is of the form $\widetilde{W} := \begin{bmatrix} \widetilde{W}_{11} & 0 & \widetilde{W}_{13} \\ 0 & I_{d_o} & 0 \\ \widetilde{W}_{31} & 0 & \widetilde{W}_{33} \end{bmatrix} \in \mathbb{C}^{(d_{\overline{o}}+d_o+m)\times(d_{\overline{o}}+d_o+m)}, d_{\overline{o}} \text{ is the}$

number of (finite and infinite) eigenvalues of $\begin{bmatrix} \bar{A}(\lambda) \\ C(\lambda) \end{bmatrix}$, and $X_{\bar{o}}(\lambda) \in \mathbb{C}[\lambda]^{d_{\bar{o}} \times d_{\bar{o}}}$ is a regular pencil. Moreover,

- a) the eigenvalues of $\begin{bmatrix} A(\lambda) \\ C(\lambda) \end{bmatrix}$ are the eigenvalues of $X_{\overline{o}}(\lambda)$,
- b) $\begin{bmatrix} A_o(\lambda) \\ C_o(\lambda) \end{bmatrix} \in \mathbb{C}[\lambda]^{(d_o+m) \times d_o}$ has no (finite or infinite) eigenvalues,
- c) the quadruple $\{A_o(\lambda), B_o(\lambda), C_o(\lambda), D_o(\lambda)\}$ is a realization of the transfer function $R_o(\lambda) := \widetilde{W}_{33}R(\lambda)$, with $\widetilde{W}_{33} \in \mathbb{C}^{m \times m}$ invertible, and
- d) if $\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$ has no finite or infinite eigenvalues then $\begin{bmatrix} A_o(\lambda) & -B_o(\lambda) \end{bmatrix}$ also has no finite or infinite eigenvalues.

In order to extract from the system quadruple $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ a subsystem $\{A_{co}(\lambda), B_{co}(\lambda), C_{co}(\lambda), D_{co}(\lambda)\}$ that is both strongly E-controllable and Eobservable (and hence also strongly minimal), we only need to apply the above two theorems one after the other. The resulting subsystem would then be a realization of the transfer function $R_{co} = C_{co}(\lambda)A_{co}(\lambda)^{-1}B_{co}(\lambda) + D_{co}(\lambda) = W_{\ell}R(\lambda)W_r \in \mathbb{C}(\lambda)^{m \times n}$. Since the transfer function was changed only by left and right transformations that are constant and invertible, the left and right nullspace will be transformed by these invertible transformations, but their minimal indices will be unchanged.

7.4 Computational aspects

In this section we give a more "algorithmic" description of the procedure described in Section 7.3 to reduce a given system quadruple $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ to a strongly E-controllable quadruple $\{A_c(\lambda), B_c(\lambda), C_c(\lambda), D_c(\lambda)\}$ of smaller size. We describe the essence of the three steps that were discussed in that section.

Step 1 : Compute the staircase reduction of the submatrix $\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$

$$U\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix} W^* = \begin{bmatrix} X(\lambda) & 0 & 0\\ \hline Y(\lambda) & \widehat{A}(\lambda) & -\widehat{B}(\lambda) \end{bmatrix}.$$

Step 2 : Compute the unitary matrices V and \widetilde{W} to compress the first block row of W

$$\begin{bmatrix} W_{11} & W_{12} & W_{13} \end{bmatrix} \begin{bmatrix} V & 0 \\ 0 & I_n \end{bmatrix} \begin{bmatrix} \widetilde{W}_{11} & 0 & \widetilde{W}_{13} \\ 0 & I_{d-r} & 0 \\ \widetilde{W}_{31} & 0 & \widetilde{W}_{33} \end{bmatrix} = \begin{bmatrix} I_r & 0 & 0 \end{bmatrix},$$

where V does the compression $\begin{bmatrix} W_{11} & W_{12} \end{bmatrix} V = \begin{bmatrix} \widetilde{W}_{11}^* & 0 \end{bmatrix}$ of the first two blocks and \widetilde{W} does the further reduction of the first block row to $\begin{bmatrix} I_r & 0 & 0 \end{bmatrix}$.

Step 3 : Display the uncontrollable part $X(\lambda)$ using the transformations U, V and \widetilde{W}

$$\begin{bmatrix} U & 0 \\ 0 & I_m \end{bmatrix} \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix} \begin{bmatrix} V & 0 \\ 0 & I_n \end{bmatrix} \widetilde{W} = \begin{bmatrix} X_{\overline{c}}(\lambda) & 0 & 0 \\ \times & A_c(\lambda) & -B_c(\lambda) \\ \times & C_c(\lambda) & D_c(\lambda) \end{bmatrix},$$

where we have used the notations introduced in Section 7.3, and the resulting × entries are of no interest because they do not contribute to the transfer function $R_c(\lambda) := C_c(\lambda)A_c(\lambda)^{-1}B_c(\lambda) + D_c(\lambda).$

7.5 Numerical results

We illustrate in this section that the reduction procedure presented in this chapter may improve the accuracy of computed eigenvalues, with a polynomial example and a rational one.

Example 7.5.1. We consider the 2×2 polynomial matrix

$$P(\lambda) = \begin{bmatrix} e_5(\lambda) & 0\\ 0 & e_1(\lambda) \end{bmatrix},$$

where $e_5(\lambda)$ is a polynomial of degree 5 with coefficients [9.6367e - 01 - 5.4026e - 07 2.6333e - 01 - 1.1101e - 04 - 2.9955e - 04 4.4650e - 02], ordered by descending powers of λ , and $e_1(\lambda)$ is a polynomial of degree 1 with coefficients [-2.1886e - 03 - 1.0000e + 00], that were randomly chosen. Expanding this fifth order polynomial matrix as

$$P(\lambda) = P_0 + P_1 \lambda + \dots + P_5 \lambda^5,$$

a linear system matrix $S_P(\lambda)$ of $P(\lambda)$ is given by the following 10×10 pencil:

$$S_P(\lambda) = \begin{bmatrix} I_2 & -\lambda I_2 & & P_1 \\ I_2 & -\lambda I_2 & P_2 \\ & I_2 & -\lambda I_2 & P_3 \\ & & I_2 & P_4 + \lambda P_5 \\ \hline -\lambda I_2 & & P_0 \end{bmatrix}.$$

The six finite Smith zeros of $P(\lambda)$ are clearly those of the scalar polynomials $e_1(\lambda)$ and $e_5(\lambda)$. These are also the finite zeros of $S_P(\lambda)$, since $S_P(\lambda)$ is minimal. However, $S_P(\lambda)$ is not strongly minimal if P_5 is singular and, in fact, it has 4 eigenvalues at infinity (in the sense of [44]). But in the McMillan sense, $P(\lambda)$ has no infinite zeros. The deflation procedure that we derived in this chapter precisely gets rid of the *extraneous* infinite eigenvalues of $S_P(\lambda)$. The numerical tests show that the sensitivity of the true McMillan zeros also can benefit from this.

In this example we compare the roots computed by four different methods:

- 1. computing the roots of the scalar polynomials and appending four ∞ roots,
- 2. computing the generalized eigenvalues of $S_P(\lambda)$,
- 3. computing the roots of $QS_P(\lambda)Z$ for random orthogonal matrices Q and Z,
- 4. computing the roots of the *minimal* pencil obtained by our method.

The first column are the so-called "correct" eigenvalues λ_i , corresponding to the first method, the next three columns are the corresponding errors $\delta_i^{(k)} := |\lambda_i - \hat{\lambda}_i^{(k)}|$, k = 2, 3, 4, of the above three methods¹. The *extraneous* eigenvalues that are deflated in our approach are put between brackets.

We notice that for the largest finite eigenvalue of the order of 10^2 the QZ algorithm applied to $S_P(\lambda)$ gets 14 digits of relative accuracy but, when deflating the four uncontrollable eigenvalues at ∞ , our method recovers a relative accuracy of 16 digits.

¹An error $\delta_i^{(k)}$ is NaN when it is the indeterminate form Inf – Inf. However, some of the eigenvalues at ∞ are computed as a large but finite number and, then, the corresponding error is Inf.

λ_i	$\delta_i^{(2)}$	$\delta_i^{(3)}$	$\delta_i^{(4)}$
-4.5811e-01	2.7756e-16	4.4409e-16	1.1102e-16
3.5076e-01 + 3.5785e-01i	9.5020e-16	1.1102e-16	4.0030e-16
3.5076e-01 - 3.5785e-01i	9.5020e-16	1.1102e-16	4.0030e-16
-1.2170e-01 + 6.2287e-01i	6.7589e-16	7.8945e-16	2.2248e-16
-1.2170e-01 - 6.2287e-01i	6.7589e-16	7.8945e-16	2.2248e-16
-4.5691e + 02	2.9559e-12	2.7285e-12	5.6843e-14
Inf	NaN	NaN	(Inf)
Inf	NaN	NaN	(Inf)
Inf	NaN	NaN	(Inf)
Inf	NaN	NaN	(Inf)

Table 7.1: The correct generalized λ_i and the corresponding accuracies δ_i^k for the three different calculations

Example 7.5.2. The second example is the rational matrix $R(\lambda)$ in (7.4) with c = 1.

$$R(\lambda) = \begin{bmatrix} e_5(\lambda) & 0\\ 1/\lambda & e_1(\lambda) \end{bmatrix} = P_0 + P_1\lambda + \dots + P_5\lambda^5 + \begin{bmatrix} 0 & 0\\ 1/\lambda & 0 \end{bmatrix},$$

by using the notation of the example above. In this case, $e_5(\lambda)$ has the row vector $[4.7865e-02 \ 1.4279e-04 \ 2.4361e-03 \ -1.5336e-02 \ -9.9155e-01 \ 1.1948e-01]$ as coefficients, and $e_1(\lambda)$ has the row vector $[6.5250e-03 \ 9.9997e-01]$. We consider the 12×12 linear system matrix

$$S_R(\lambda) = \begin{bmatrix} \lambda I_2 - A & & & & -B \\ & I_2 & -\lambda I_2 & & P_1 \\ & & I_2 & -\lambda I_2 & P_2 \\ & & & I_2 & -\lambda I_2 & P_3 \\ \hline & & & & I_2 & P_4 + \lambda P_5 \\ \hline C & -\lambda I_2 & & & P_0 \end{bmatrix},$$

where

$$A = \begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix}, \quad B = \begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix} \quad C = \begin{bmatrix} 0 & 0 \\ 0 & 1 \end{bmatrix}$$

is a non-minimal realization of the strictly proper rational function $1/\lambda$. In fact, the matrix A in the realization triple (A, B, C) has two eigenvalues at $\lambda = 0$, of which one is uncontrollable since $1/\lambda$ only has a pole at 0 of order 1. This is an artificial example since we could have realized the strictly proper part by using a minimal triple (A, B, C) by removing the uncontrollable eigenvalue, but this is precisely what our reduction procedure does simultaneously for finite and infinite uncontrollable eigenvalues. The quantities given in the following table are defined as in the previous example, except that we added two roots at 0 corresponding to the "exact" eigenvalues.

λ_i	$\delta_i^{(2)}$	$\delta_i^{(3)}$	$\delta_i^{(4)}$
0	0	8.1752e-09	(4.5874e-16)
0	3.6752e-18	8.1752e-09	5.3729e-16
1.2028e-01	1.8041e-16	9.7145e-17	9.7145e-17
2.1135e+00	1.7764e-15	2.6645 e- 15	1.3323e-15
-2.1404e+00	1.7764e-15	2.2204e-15	8.8818e-16
-4.8180e-02 + 2.1412e+00i	2.3216e-15	1.7990e-15	4.0614e-15
-4.8180e-02 - 2.1412e+00i	2.3216e-15	1.7990e-15	4.0614e-15
-1.5325e + 02	2.5580e-13	1.5321e-07	5.6843e-14
Inf	NaN	Inf	(Inf)
Inf	NaN	Inf	(Inf)
Inf	NaN	NaN	(NaN)
Inf	NaN	NaN	(NaN)

Table 7.2: The correct generalized λ_i and the corresponding accuracies δ_i^k for the three different calculations

In this example the QZ algorithm applied to $S_R(\lambda)$ recovers well all generalized eigenvalues. When applying the QZ algorithm to an orthogonally equivalent pencil $QS_R(\lambda)Z$ both eigenvalues at 0 gets perturbed to two roots of the order of the square root of the machine precision, which can be expected. But when deflating the uncontrollable eigenvalue at 0 part of the accuracy gets restored.

These two examples show that deflating uncontrollable eigenvalues may improve the sensitivity of the remaining eigenvalues which may improve the accuracy of their computation.

Chapter 8

Constructing strongly minimal linearizations for rational matrices from Laurent expansions: preserving structures

In this chapter we will show a procedure to construct strongly minimal linearizations (recall Definition 7.2.4) of arbitrary rational matrices $R(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$ from their Laurent expansions around the point at infinity. Namely,

$$R(\lambda) = R_d \lambda^d + \ldots + R_1 \lambda + R_0 + R_{-1} \lambda^{-1} + R_{-2} \lambda^{-2} + R_{-3} \lambda^{-3} + \ldots, \qquad (8.1)$$

which is convergent for sufficiently large $\lambda \in \mathbb{C}$. We will first pay special attention to the construction of strongly minimal linearizations for the particular case of polynomial matrices $P(\lambda)$, written in the form

$$P(\lambda) = P_d \lambda^d + \ldots + P_1 \lambda + P_0.$$

If the rational matrix $R(\lambda)$ is square, i.e., m = n, and has a particular type of self-conjugate structure then the coefficients $R_i \in \mathbb{C}^{m \times m}$ of its expansion also inherit the self-conjugate structure and the poles and zeros of $R(\lambda)$ appear in selfconjugate pairs. We will also show how to construct strongly minimal linearizations preserving the structure for both polynomial and rational matrices. All the results in this chapter appear in [34].

We will consider four types of self-conjugate rational matrices, two with respect to the real line and two with respect to the imaginary axis:

• The Hermitian and skew-Hermitian rational matrices $R(\lambda)$, with respect to the real line, satisfy

$$[R(\lambda)]^* = R(\lambda), \text{ and } [R(\lambda)]^* = -R(\lambda),$$

respectively. They have poles and zeros that are mirror images with respect to the real line \mathbb{R} , and have coefficient matrices R_i that are Hermitian (i.e. $R_i^* = R_i$) and skew-Hermitian (i.e. $R_i^* = -R_i$), respectively.

• The para-Hermitian and para-skew-Hermitian rational matrices, with respect to the imaginary axis, satisfy

 $[R(\lambda)]^* = R(-\overline{\lambda}), \text{ and } [R(\lambda)]^* = -R(-\overline{\lambda}),$

respectively. They have poles and zeros that are mirror images with respect to the imaginary line $j\mathbb{R}$, and have scaled coefficient matrices $j^i R_i$ that are Hermitian and skew-Hermitian, respectively.

Para-Hermitian and para-skew-Hermitian structures are also called alternating structures. There are equivalent definitions for real rational matrices, where all coefficient matrices R_i are real. Namely, (skew-)symmetric and para-(skew-)symmetric rational matrices. In these cases, poles and zeros appear in complex conjugate pairs as well.

We will divide the construction of strongly minimal linearizations of arbitrary and self-conjugate rational matrices in different sections. In Section 8.1, we construct strongly minimal linearizations for arbitrary polynomial matrices and, in Section 8.2, for self-conjugate polynomial matrices. In Section 8.3, we construct strongly minimal linearizations for arbitrary strictly proper rational matrices and, in Section 8.4, for self-conjugate strictly proper rational matrices. In Section 8.5 we state the results for both arbitrary and self-conjugate rational matrices, by combining the results in previous sections.

8.1 For arbitrary polynomial matrices

In this section we focus on constructing explicitly a strongly minimal linearization for any given polynomial matrix $P(\lambda) \in \mathbb{C}[\lambda]^{m \times n}$ of degree d > 1:

$$P(\lambda) := P_0 + P_1 \lambda + \dots + P_d \lambda^d.$$
(8.2)

Such a strongly minimal linearization is constructed in Theorem 8.1.2 and we will prove in Section 8.2 that it inherits the structure of $P(\lambda)$, when $P(\lambda)$ possesses any of the self-conjugate structures considered in this work. The construction uses three pencils associated with $P(\lambda)$ that have appeared before in the literature. They are described in the following paragraphs. The pencil

$$L_r(\lambda) := \begin{bmatrix} A_r(\lambda) & -B_r(\lambda) \\ \hline C_r(\lambda) & D_r(\lambda) \end{bmatrix} := \begin{bmatrix} -I_n & \lambda I_n & & 0 \\ & \ddots & \ddots & & \vdots \\ & & -I_n & \lambda I_n & 0 \\ \hline & & & -I_n & \lambda I_n \\ \hline \lambda P_d & \dots & \dots & \lambda P_2 & \lambda P_1 + P_0 \end{bmatrix}$$
(8.3)

was used in the classical reference [87]. It is easy to see that $L_r(\lambda)$ is a linear polynomial system matrix of $P(\lambda)$, since $P(\lambda) = D_r(\lambda) + C_r(\lambda)A_r(\lambda)^{-1}B_r(\lambda)$, and that it is minimal for all finite λ . For the point at ∞ , E-controllability is clearly satisfied but E-observability is only satisfied if the matrix P_d has full column rank n. Thus, $L_r(\lambda)$ is not a strongly minimal linearization of $P(\lambda)$ when P_d does not have full column rank. However, note that $L_r(\lambda)$ is always a GLR-strong linearization of $P(\lambda)$. This can be seen, for instance, by noting that if the two block rows in (8.3) are interchanged, we obtain one of the block Kronecker linearizations (with only one block column) associated to $P(\lambda)$ defined in [26, Section 4]. The pencil (8.3) has a structure similar to that of the classical first or row Frobenius companion form.

The pencil

$$L_{c}(\lambda) := \begin{bmatrix} A_{c}(\lambda) & -B_{c}(\lambda) \\ \hline C_{c}(\lambda) & D_{c}(\lambda) \end{bmatrix} := \begin{bmatrix} -I_{m} & \lambda P_{d} \\ \lambda I_{m} & \ddots & \vdots \\ & \ddots & -I_{m} & \vdots \\ \hline & \lambda I_{m} & -I_{m} & \lambda P_{2} \\ \hline 0 & \dots & 0 & \lambda I_{m} & \lambda P_{1} + P_{0} \end{bmatrix}$$
(8.4)

is in some sense "dual" to (8.3). It is also a linear polynomial system matrix of $P(\lambda)$, since $P(\lambda) = D_c(\lambda) + C_c(\lambda)A_c(\lambda)^{-1}B_c(\lambda)$. Moreover, $L_c(\lambda)$ is strongly E-observable, but not necessarily strongly E-controllable, unless P_d has full row rank. As a consequence, $L_c(\lambda)$ is a strongly minimal linearization of $P(\lambda)$ if and only if P_d has full row rank. However, $L_c(\lambda)$ is always a GLR-strong linearization of $P(\lambda)$. The pencil (8.4) has a structure similar to that of the classical second or column Frobenius companion form.

The pencil

$$L_{s}(\lambda) := \begin{bmatrix} A_{s}(\lambda) & -B_{s}(\lambda) \\ \hline C_{s}(\lambda) & D_{s}(\lambda) \end{bmatrix} := \begin{bmatrix} -P_{d} & \lambda P_{d} \\ & \ddots & \lambda P_{d} - P_{d-1} & \vdots \\ -P_{d} & \ddots & \vdots & \vdots \\ \hline -P_{d} & \lambda P_{d} - P_{d-1} & \dots & \lambda P_{3} - P_{2} & \lambda P_{2} \\ \hline \lambda P_{d} & \dots & \dots & \lambda P_{2} & \lambda P_{1} + P_{0} \end{bmatrix}$$
(8.5)

was originally proposed by Lancaster in [58, pp. 58-59] for regular polynomial matrices with P_d invertible. In this chapter, we use it for arbitrary polynomial matrices, including rectangular ones. $L_s(\lambda)$ has the advantage to preserve the Hermitian or skew-Hermitian nature of the coefficients of the linearization, if $P(\lambda)$ happens to have coefficients with such properties. The pencil $L_s(\lambda)$ has been also studied more recently in [53, 63], where it is seen as one of the pencils of the standard basis of the linear space $\mathbb{DL}(P)$ of pencils related to $P(\lambda)$. It is well known that $L_s(\lambda)$ is a GLR-strong linearization of $P(\lambda)$ if and only if P_d is invertible [21, 63]. In fact, in this case, $L_s(\lambda)$ is also a strongly minimal linearization of $P(\lambda)$ since it is strongly minimal and $P(\lambda) = D_s(\lambda) + C_s(\lambda)A_s(\lambda)^{-1}B_s(\lambda)$. However, if P_d is not invertible, $L_s(\lambda)$ is not a linearization of $P(\lambda)$ in any of the senses considered in the literature and, even more, it is not a Rosenbrock polynomial system matrix of $P(\lambda)$ since $A_s(\lambda)$ is not regular. Despite of this fact, $L_s(\lambda)$ is our starting point for constructing the strongly minimal linearization of $P(\lambda)$ of interest in this work.

The constant block Hankel matrix T defined in the next equation

$$T := \begin{bmatrix} & P_d \\ & \ddots & P_{d-1} \\ P_d & \ddots & \vdots \\ P_d & P_{d-1} & \dots & P_2 \end{bmatrix}$$
(8.6)

plays a key role in the rest of the chapter. To begin with, it allows us to obtain the following relations

$$\begin{bmatrix} A_s(\lambda) & -B_s(\lambda) \end{bmatrix} = T \begin{bmatrix} A_r(\lambda) & -B_r(\lambda) \end{bmatrix}, \begin{bmatrix} A_s(\lambda) \\ C_s(\lambda) \end{bmatrix} = \begin{bmatrix} A_c(\lambda) \\ C_c(\lambda) \end{bmatrix} T, \quad (8.7)$$

between submatrices of the pencils $L_s(\lambda)$, $L_r(\lambda)$ and $L_c(\lambda)$. The matrix T is invertible if and only if P_d is square and invertible. Otherwise, T is singular and this is the case that requires a careful analysis.

In [87], it was shown how to derive from the linear polynomial system matrix $L_r(\lambda)$ of $P(\lambda)$, a smaller linear polynomial system matrix $\hat{L}_r(\lambda)$ that is both strongly E-controllable and E-observable, and hence strongly minimal, by using only multiplications by constant unitary matrices. This was obtained by deflating the unobservable infinite eigenvalues from the pencil $L_r(\lambda)$. Moreover, the obtained pencil $\hat{L}_r(\lambda)$ allows us to recover the complete list of structural data of $P(\lambda)$. The reduction procedure in [87] has been extended to arbitrary linear polynomial system matrices of arbitrary rational matrices $R(\lambda)$ in Chapter 7, where it is proved that the obtained strongly minimal linear polynomial system matrix has as transfer function matrix $Q_1 R(\lambda) Q_2$, where Q_1 and Q_2 are constant invertible matrices. We emphasize that the procedures in [87, 32] lead to stable and efficient numerical algorithms since both are based on unitary transformations. We show in Theorem 8.1.2 that a procedure similar to that in [87] can be applied to $L_s(\lambda)$ in order to derive a strongly minimal linear polynomial system matrix $\hat{L}_s(\lambda)$ of $P(\lambda)$, despite of the fact that if P_d is not square or invertible, then $L_s(\lambda)$ is not a *Rosenbrock polynomial system matrix* since $A_s(\lambda)$ is then not regular. Moreover, we remark that the procedure in Theorem 8.1.2 is much simpler than those in [87, 32] and that, as said before, it yields a polynomial system matrix whose transfer function matrix is precisely $P(\lambda)$. Before stating and proving Theorem 8.1.2, we prove the simple auxiliary Lemma 8.1.1 and introduce some other auxiliary concepts.

A rational matrix $G(\lambda) \in \mathbb{C}(\lambda)^{p \times n}$ (with p < n) is said to be a rational basis if its rows form a basis of the rational subspace they span, i.e., if it has full row normal rank. Two rational bases $G(\lambda) \in \mathbb{C}(\lambda)^{p \times n}$ and $H(\lambda) \in \mathbb{C}(\lambda)^{q \times n}$ are said to be dual if p + q = n, and $G(\lambda) H(\lambda)^T = 0$.

Lemma 8.1.1. Let

$$S(\lambda) := \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix} \in \mathbb{C}[\lambda]^{(p+m) \times (p+n)}$$

be a polynomial system matrix, where $A(\lambda)$ is assumed to be regular. Let $H(\lambda)$ be a rational basis of the form $H(\lambda) := \begin{bmatrix} M(\lambda) & I_n \end{bmatrix}$ dual to $\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix}$, i.e., such that $\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix} H(\lambda)^T = 0$, then $\begin{bmatrix} C(\lambda) & D(\lambda) \end{bmatrix} H(\lambda)^T$ is the transfer function of $S(\lambda)$.

Proof. The equation

$$\begin{bmatrix} A(\lambda) & -B(\lambda) \end{bmatrix} \begin{bmatrix} M(\lambda)^T \\ I_n \end{bmatrix} = 0$$

implies $A(\lambda)M(\lambda)^T = B(\lambda)$ and, since $A(\lambda)$ is regular, $M(\lambda)^T = A(\lambda)^{-1}B(\lambda)$. Thus $\begin{bmatrix} C(\lambda) & D(\lambda) \end{bmatrix} H(\lambda)^T = C(\lambda)A(\lambda)^{-1}B(\lambda) + D(\lambda)$.

Theorem 8.1.2. Let $P(\lambda) \in \mathbb{C}[\lambda]^{m \times n}$ be a polynomial matrix as in (8.2). Let T be the block Hankel matrix in (8.6) and $r := \operatorname{rank} T$. Let $U = \begin{bmatrix} U_1 & U_2 \end{bmatrix}$ and $V = \begin{bmatrix} V_1 & V_2 \end{bmatrix}$ be unitary matrices that "compress" the matrix T as follows:

$$U^*TV = \begin{bmatrix} 0 & 0\\ 0 & U_2^*TV_2 \end{bmatrix} =: \begin{bmatrix} 0 & 0\\ 0 & \widehat{T} \end{bmatrix},$$
(8.8)

where \widehat{T} is of dimension $r \times r$ and invertible. Then, if $L_s(\lambda)$ is the matrix pencil in (8.5), the pencil diag $(U^*, I_m) L_s(\lambda)$ diag (V, I_n) is equal to the "compressed" pencil

$$\begin{bmatrix} 0 & 0 & 0\\ 0 & \widehat{A}_s(\lambda) & -\widehat{B}_s(\lambda)\\ \hline 0 & \widehat{C}_s(\lambda) & \widehat{D}_s(\lambda) \end{bmatrix} := \begin{bmatrix} U^*A_s(\lambda)V & -U^*B_s(\lambda)\\ \hline C_s(\lambda)V & D_s(\lambda) \end{bmatrix},$$
(8.9)

and

$$\widehat{L}_{s}(\lambda) := \begin{bmatrix} \widehat{A}_{s}(\lambda) & -\widehat{B}_{s}(\lambda) \\ \hline \widehat{C}_{s}(\lambda) & \widehat{D}_{s}(\lambda) \end{bmatrix}$$
(8.10)

is a strongly minimal linearization of $P(\lambda)$, where $\widehat{A}_s(\lambda) \in \mathbb{C}[\lambda]^{r \times r}$ is regular. In particular, $P(\lambda) = \widehat{D}_s(\lambda) + \widehat{C}_s(\lambda)\widehat{A}_s(\lambda)^{-1}\widehat{B}_s(\lambda)$.

Proof. It follows from (8.7) and the strong E-controllability of $[A_r(\lambda) - B_r(\lambda)]$ that $[A_s(\lambda) - B_s(\lambda)]$ has rank r for all λ , infinity included, and that its left null space is spanned by the rows of U_1^* . Likewise, it follows from (8.7) and the strong E-observability of $\begin{bmatrix} A_c(\lambda) \\ C_c(\lambda) \end{bmatrix}$ that $\begin{bmatrix} A_s(\lambda) \\ C_s(\lambda) \end{bmatrix}$ has rank r for all λ , infinity included and that its right null space is spanned by the columns of V_1 . This proves the compressed form (8.9).

We then prove that the $r \times r$ matrix pencil $\widehat{A}_s(\lambda)$ is regular. This follows from the identity

$$\widehat{A}_{s}(\lambda) = U_{2}^{*}TA_{r}(\lambda)V_{2}, \quad \text{where} \quad A_{r}(\lambda) = \begin{bmatrix} -I_{n} & \lambda I_{n} & & \\ & \ddots & \ddots & \\ & & -I_{n} & \lambda I_{n} \\ & & & -I_{n} \end{bmatrix}$$
(8.11)

which, for $\lambda = 0$ becomes $\widehat{A}_s(0) = -U_2^* T V_2 = -\widehat{T}$.

The fact that $[\hat{A}_s(\lambda) - \hat{B}_s(\lambda)]$ has full row rank r for all λ , ∞ included, follows from the identity

$$\begin{bmatrix} 0 & \widehat{A}_s(\lambda) & -\widehat{B}_s(\lambda) \end{bmatrix} = \widehat{T}V_2^* \begin{bmatrix} A_r(\lambda) & -B_r(\lambda) \end{bmatrix} \operatorname{diag}(V, I_n)$$

The fact that $\begin{bmatrix} \hat{A}_s(\lambda) \\ \hat{C}_s(\lambda) \end{bmatrix}$ has full column rank r for all λ , ∞ included, follows from the dual identity

$$\begin{bmatrix} 0\\ \widehat{A}_s(\lambda)\\ \widehat{C}_s(\lambda) \end{bmatrix} = \operatorname{diag}(U^*, I_m) \begin{bmatrix} A_c(\lambda)\\ C_c(\lambda) \end{bmatrix} U_2 \widehat{T}.$$

Together, these properties guarantee that $\widehat{L}_s(\lambda)$ is a strongly minimal linear polynomial system matrix. Its transfer function $\widehat{C}_s(\lambda)\widehat{A}_s(\lambda)^{-1}\widehat{B}_s(\lambda) + \widehat{D}_s(\lambda)$ can then be obtained from a particular dual basis $N(\lambda) \in \mathbb{C}(\lambda)^{n \times (r+n)}$ of $[\widehat{A}_s(\lambda) - \widehat{B}_s(\lambda)]$, by using Lemma 8.1.1. Since

$$\begin{bmatrix} 0 & 0 & | & 0 \\ 0 & \widehat{A}_{s}(\lambda) & | & -\widehat{B}_{s}(\lambda) \end{bmatrix} \operatorname{diag}(V^{*}, I_{n}) = U^{*} \begin{bmatrix} A_{s}(\lambda) & -B_{s}(\lambda) \end{bmatrix} = U^{*}T \begin{bmatrix} A_{r}(\lambda) & -B_{r}(\lambda) \end{bmatrix},$$
(8.12)

it follows that

$$\begin{bmatrix} 0 & 0 & | & 0 \\ 0 & \widehat{A}_s(\lambda) & | & -\widehat{B}_s(\lambda) \end{bmatrix} \operatorname{diag}(V^*, I_n) \begin{bmatrix} \lambda^{d-1}I_n \\ \vdots \\ \lambda I_n \\ I_n \end{bmatrix} = 0$$

and hence that

$$\begin{bmatrix} \widehat{A}_s(\lambda) & | & -\widehat{B}_s(\lambda) \end{bmatrix} \operatorname{diag}(\begin{bmatrix} 0 & I_r \end{bmatrix} V^*, I_n) \begin{bmatrix} \lambda^{d-1}I_n \\ \vdots \\ \lambda I_n \\ I_n \end{bmatrix} = 0.$$

Therefore, by setting

$$N(\lambda)^{T} := \operatorname{diag}(\begin{bmatrix} 0 & I_{r} \end{bmatrix} V^{*}, I_{n}) \begin{bmatrix} \lambda^{d-1}I_{n} \\ \vdots \\ \lambda I_{n} \\ I_{n} \end{bmatrix} \in \mathbb{C}[\lambda]^{(r+n) \times n},$$

we have that $N(\lambda)$ is a dual basis of $[\widehat{A}_s(\lambda) - \widehat{B}_s(\lambda)]$ with its rightmost block equal to I_n . By Lemma 8.1.1, and using the fact that

$$\widehat{C}_s(\lambda) \begin{bmatrix} 0 & I_r \end{bmatrix} V^* = \begin{bmatrix} 0 & \widehat{C}_s(\lambda) \end{bmatrix} V^* = C_s(\lambda),$$

we obtain that

$$\begin{bmatrix} \widehat{C}_s(\lambda) & \widehat{D}_s(\lambda) \end{bmatrix} N(\lambda)^T = \begin{bmatrix} C_s(\lambda) & D_s(\lambda) \end{bmatrix} \begin{bmatrix} \lambda^{d-1} I_n \\ \vdots \\ \lambda I_n \\ I_n \end{bmatrix} = P(\lambda)$$

is the transfer function of $\widehat{L}_s(\lambda)$.

Remark 8.1.3. Once the unitary matrices U and V and the matrix \hat{T} in (8.8) are computed, Theorem 8.1.2 yields an efficient and stable algorithm for computing the strongly minimal linear realization $\{\hat{A}_s(\lambda), \hat{B}_s(\lambda), \hat{C}_s(\lambda), \hat{D}_s(\lambda)\}$ of $P(\lambda)$. An expensive method for computing U, V and \hat{T} is to compute the SVD of T. A cheaper method is to use the complete orthogonal decomposition in [45, Sec. 5.4.2], which amounts to compute two QR factorizations. The block Hankel structure of T (which by flipping the order of the block rows becomes block Toeplitz) allows us to use the very fast and stable method in [88, Sec. IV], which makes all the computations

on $m \times n$ submatrices. The method in [88] has the additional advantage that if $r_d = \operatorname{rank}(P_d)$, then the rows of $U^*(1 : (m - r_d), 1 : m)$ and the columns of $V(1 : n, 1 : (n - r_d))$ (where we used MATLAB's notation) of the computed U and V are, respectively, unitary bases of the left and right nullspaces of P_d . Recall that these subspaces are precisely the left and right eigenspaces associated to the infinite eigenvalue of $P(\lambda)$ when $P(\lambda)$ is regular.

Remark 8.1.4. Even though the pencil $L_s(\lambda)$ in (8.5) is not a Rosenbrock polynomial system matrix and neither is a GLR-linearization of $P(\lambda)$ if P_d is rectangular or square and singular, it is easy to see, by using unimodular transformations that are well-known in the literature, that it has the same finite eigenvalues as $P(\lambda)$ with the same partial multiplicities. For this purpose, note that

$$V(\lambda)L_s(\lambda)W(\lambda) = \operatorname{diag}(-T, P(\lambda)),$$

where

$$V(\lambda) := \begin{bmatrix} 1 & & & \\ 0 & 1 & & \\ \vdots & \ddots & \ddots & \\ 0 & \cdots & 0 & 1 \\ \lambda^{d-1} & \cdots & \lambda^2 & \lambda & 1 \end{bmatrix} \otimes I_m, \quad W(\lambda) := \begin{bmatrix} 1 & \lambda & \lambda^2 & \dots & \lambda^{d-1} \\ & 1 & \lambda & \ddots & \vdots \\ & & \ddots & \ddots & \lambda^2 \\ & & & 1 & \lambda \\ & & & & 1 \end{bmatrix} \otimes I_m.$$

Since the polynomial matrices $V(\lambda)$ and $W(\lambda)$ are unimodular, and diag $(-T, P(\lambda))$ is strictly equivalent to diag $(0, I_r, P(\lambda))$, this implies that $L_s(\lambda)$ is unimodularly equivalent to diag $(0, I_r, P(\lambda))$. Therefore, $L_s(\lambda)$ and $P(\lambda)$ have the same finite eigenvalues with the same partial multiplicities. Of course, this also follows from Theorem 8.1.2 and the properties of strongly minimal linearizations studied in Section 7.2.2. However, note that $L_s(\lambda)$ is not a GLR-linearization of $P(\lambda)$ because $L_s(\lambda)$ is not unimodularly equivalent to diag $(I, P(\lambda))$.

On the other hand, if we consider the pencil $\hat{L}_s(\lambda)$ in (8.10), then Theorem 8.1.2 proves that $L_s(\lambda)$ is strictly equivalent to diag $(0, \hat{L}_s(\lambda))$. Combining the results above, we see that $\hat{L}_s(\lambda)$ and diag $(I_r, P(\lambda))$ have the same normal rank and the same finite eigenvalues and partial multiplicities, which implies that $\hat{L}_s(\lambda)$ and diag $(I_r, P(\lambda))$ are unimodularly equivalent [39] and, therefore, that $\hat{L}_s(\lambda)$ is a GLRlinearization of $P(\lambda)$. This is a particular instance of the result mentioned in Section 7.2.2 that any minimal linear polynomial system matrix of $P(\lambda)$ is a GLRlinearization of $P(\lambda)$.

8.2 For self-conjugate polynomial matrices

The main purpose of this section is to prove that the strongly minimal linearization (8.10) of the matrix polynomial $P(\lambda)$ developed in Theorem 8.1.2 inherits the structure of $P(\lambda)$ for any of the four structures considered in this work.

We start by reviewing the four structures of interest in this chapter and their properties. In these definitions, note that if $P(\lambda)$ is the complex polynomial matrix in (8.2), then

$$[P(\lambda)]^* := P_0^* + P_1^* \overline{\lambda} + \dots + P_d^* \overline{\lambda}^d.$$

The considered self-conjugate structures are:

- 1. Hermitian polynomial matrices, which are defined as those satisfying $[P(\lambda)]^* = P(\overline{\lambda})$. Equivalently, they are Hermitian for $\lambda \in \mathbb{R}$ or have Hermitian coefficients $P_i^* = P_i$. They have a set of eigenvalues Λ that is symmetric with respect to the real axis: $\Lambda = \overline{\Lambda}$.
- 2. Skew-Hermitian polynomial matrices, which are defined as those satisfying $[P(\lambda)]^* = -P(\overline{\lambda})$. Equivalently, they are skew-Hermitian for $\lambda \in \mathbb{R}$ or have skew-Hermitian coefficients $P_i^* = -P_i$. They have a set of eigenvalues Λ that is symmetric with respect to the real axis: $\Lambda = \overline{\Lambda}$.
- 3. Para-Hermitian polynomial matrices, which are those satisfying $[P(\lambda)]^* = P(-\overline{\lambda})$. Equivalently, they are Hermitian for $\lambda \in j\mathbb{R}$, i.e., for λ on the imaginary axis, or have scaled Hermitian coefficients $P_i^* = (-1)^i P_i$. They have a set of eigenvalues Λ that is symmetric with respect to the imaginary axis: $\Lambda = -\overline{\Lambda}$.
- 4. Para-skew-Hermitian polynomial matrices, which are defined as those satisfying $[P(\lambda)]^* = -P(-\overline{\lambda})$. Equivalently, they are skew-Hermitian for $\lambda \in \mathfrak{I}\mathbb{R}$ or have skew-Hermitian scaled coefficients $P_i^* = (-1)^{(i+1)}P_i$. They have a set of eigenvalues Λ that is symmetric with respect to the imaginary axis: $\Lambda = -\overline{\Lambda}$.

When the polynomial matrix $P(\lambda)$ has real coefficients P_i , these conditions become conditions on the transpose of each P_i , and the polynomial matrices are said to be symmetric, skew-symmetric, para-symmetric and para-skew-symmetric.

We first point out that the block Hankel matrix T defined in (8.6) and its compression (8.8) inherit particular properties from the self-conjugate structures defined above.

Lemma 8.2.1. Let $P(\lambda) \in \mathbb{C}[\lambda]^{m \times m}$ be a polynomial matrix as in (8.2). Let us define the scaling matrix $S := \operatorname{diag}((-1)^{(d-1)}I_m, \ldots, (-1)^2I_m, -I_m)$. Then the block Hankel matrix T in (8.6) satisfies the following equations

- 1. for Hermitian $P(\lambda)$: $P_i^* = P_i$ and $T^* = T$,
- 2. for skew-Hermitian $P(\lambda)$: $P_i^* = -P_i$ and $T^* = -T$,
- 3. for para-Hermitian $P(\lambda)$: $P_i^* = (-1)^i P_i$ and $(ST)^* = ST$,

4. for para-skew-Hermitian $P(\lambda)$: $P_i^* = (-1)^{(i+1)}P_i$ and $(ST)^* = -ST$.

The left and right transformations U and V of Theorem 8.1.2 can then be chosen as U = V in the Hermitian and skew-Hermitian cases and as U = SV in the para-Hermitian and para-skew-Hermitian cases.

Proof. The symmetries of the coefficient matrices P_i trivially yield the four types of symmetries of T mentioned in the Theorem. For the compression (8.8), we can then choose U = V in the Hermitian and skew-Hermitian cases because T is normal, and we can choose U = SV in the para-Hermitian and para-skew-Hermitian cases because ST is then normal.

This implies that in the decomposition of Theorem 8.1.2, it suffices to construct a transformation V that compresses the columns of T in order to obtain a rank rfactorization

$$U^*TV = \begin{bmatrix} 0 & 0\\ 0 & \widehat{T} \end{bmatrix} = \begin{bmatrix} 0 & 0\\ 0 & U_2^*TV_2 \end{bmatrix}$$
(8.13)

where \widehat{T} is $r \times r$ and is invertible. This then leads to the following theorem.

Theorem 8.2.2. Let $P(\lambda) \in \mathbb{C}[\lambda]^{m \times m}$ be a polynomial matrix as in (8.2), with a Hermitian, skew-Hermitian, para-Hermitian or para-skew-Hermitian structure. Let U, V be the unitary matrices appearing in (8.8), where U = V in the Hermitian and skew-Hermitian cases, and U = SV in the para-Hermitian and para-skew-Hermitian cases with $S := \operatorname{diag}((-1)^{(d-1)}I_m, \ldots, (-1)^2I_m, -I_m)$. Then the linear polynomial system matrix

$$\widehat{L}_s(\lambda) := \left[egin{array}{c|c} \widehat{A}_s(\lambda) & -\widehat{B}_s(\lambda) \ \hline \widehat{C}_s(\lambda) & \widehat{D}_s(\lambda) \end{array}
ight],$$

defined in Theorem 8.1.2, is a strongly minimal linear polynomial system matrix of $P(\lambda)$ with the same self-conjugate structure as $P(\lambda)$.

Proof. Let us denote the original pencil in (8.5) as

$$L_s(\lambda) = L_0 + \lambda L_1,$$

then we have the following properties in the four self-conjugate cases :

1. for Hermitian $P(\lambda)$,

$$L_0^* = L_0$$
, and $L_1^* = L_1$,

2. for skew-Hermitian $P(\lambda)$,

$$L_0^* = -L_0$$
, and $L_1^* = -L_1$,

3. for para-Hermitian $P(\lambda)$,

$$L_0^* \operatorname{diag}(S, I_m) = \operatorname{diag}(S, I_m) L_0$$
, and $L_1^* \operatorname{diag}(S, I_m) = -\operatorname{diag}(S, I_m) L_1$

4. for para-skew-Hermitian $P(\lambda)$,

 $L_0^* \operatorname{diag}(S, I_m) = -\operatorname{diag}(S, I_m)L_0$, and $L_1^* \operatorname{diag}(S, I_m) = \operatorname{diag}(S, I_m)L_1$.

If we choose in the first two cases U = V, and in the last two cases U = SV then we obtain for the transformed pair of matrices

$$\tilde{L}_0 := \operatorname{diag}(U^*, I_m) L_0 \operatorname{diag}(V, I_m), \text{ and } \tilde{L}_1 := \operatorname{diag}(U^*, I_m) L_1 \operatorname{diag}(V, I_m)$$

the properties

1. for Hermitian $P(\lambda)$,

$$\tilde{L}_0^* = \tilde{L}_0$$
, and $\tilde{L}_1^* = \tilde{L}_1$,

2. for skew-Hermitian $P(\lambda)$,

$$\tilde{L}_0^* = -\tilde{L}_0$$
, and $\tilde{L}_1^* = -\tilde{L}_1$,

3. for para-Hermitian $P(\lambda)$,

$$\tilde{L}_0^* = \tilde{L}_0$$
, and $\tilde{L}_1^* = -\tilde{L}_1$,

4. for para-skew-Hermitian $P(\lambda)$,

$$\tilde{L}_0^* = -\tilde{L}_0$$
, and $\tilde{L}_1^* = \tilde{L}_1$,

and moreover, their first m(d-1) - r columns and rows are zero because of (8.13). The pencil $\tilde{L}_s(\lambda)$ thus has the same self-conjugate structure as $P(\lambda)$, and so does the deflated pencil $\hat{L}_s(\lambda)$. The strong minimality follows from Theorem 8.1.2.

Remark 8.2.3. We remark that the procedure presented in Theorem 8.1.2 has an interpretation in terms of strongly minimal realizations of strictly proper rational matrices that have all its poles at 0. To see that, we apply the change of variable $\lambda = 1/\mu$ to the system matrix

$$\begin{bmatrix} \lambda E - F & \lambda G \\ \hline \lambda H & 0 \end{bmatrix} := \begin{bmatrix} \widehat{A}_s(\lambda) & -\widehat{B}_s(\lambda) \\ \hline \widehat{C}_s(\lambda) & 0 \end{bmatrix}$$

and we multiply it by μ . Then, we obtain a new linear polynomial system matrix

$$\begin{bmatrix} E - \mu F & G \\ \hline H & 0 \end{bmatrix},$$

whose transfer function matrix is

$$H(\mu F - E)^{-1}G = P_2 \,\mu^{-1} + P_3 \,\mu^{-2} + \dots + P_d \,\mu^{-(d-1)}.$$
(8.14)

It can be proved that new system is also strongly minimal. This means that the triple $\{E - \mu F, -G, H\}$ is a strongly minimal realization of the strictly proper transfer function in (8.14), which has all its poles at $\mu = 0$. Moreover, the minimal degree of $\det(E - \mu F)$ is known to be $r = \operatorname{rank} T$, since, according to (8.11), $\widehat{A}_s(0) = -F = -\widehat{T}$ is nonsingular.

We give a general procedure for the construction of strongly minimal realizations of arbitrary strictly proper rational matrices in Section 8.3.

8.3 For arbitrary strictly proper rational matrices

For strictly proper rational matrices $R_{sp}(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$, we represent them via a Laurent expansion around the point at infinity:

$$R_{sp}(\lambda) := R_{-1}\lambda^{-1} + R_{-2}\lambda^{-2} + R_{-3}\lambda^{-3} + \dots$$
(8.15)

In this section, we obtain strongly minimal linearizations for such strictly proper rational matrices by using the algorithm in [48, Section 3.4], as we explain in the sequel. Let the block Hankel matrix H and shifted block Hankel matrix H_{σ} associated with $R_{sp}(\lambda)$ be denoted as

$$H := \begin{bmatrix} R_{-1} & R_{-2} & \dots & R_{-k} \\ R_{-2} & & \ddots & R_{-k-1} \\ \vdots & & \ddots & \ddots & \vdots \\ R_{-k} & R_{-k-1} & \dots & R_{-2k+1} \end{bmatrix}, \quad H_{\sigma} := \begin{bmatrix} R_{-2} & R_{-3} & \dots & R_{-k-1} \\ R_{-3} & & \ddots & R_{-k-2} \\ \vdots & & \ddots & \ddots & \vdots \\ R_{-k-1} & R_{-k-2} & \dots & R_{-2k} \end{bmatrix},$$
(8.16)

then for sufficiently large k the rank r_f of H equals the total polar degree of the finite poles, i.e., the sum of the degrees of the denominators in the Smith McMillan form of $R_{sp}(\lambda)$ [55]. The general theory for Hankel based realizations of proper transfer function matrices in [48, Section 3.4], implies that the following Rosenbrock linear system matrix in Theorem 8.3.1 is a strongly minimal linearization for the strictly proper rational matrix $R_{sp}(\lambda)$.

Theorem 8.3.1. Let $R_{sp}(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$ be a strictly proper rational matrix as in (8.15). Let H and H_{σ} be the block Hankel matrices in (8.16) and $r_f := \operatorname{rank} H$. Let $U := \begin{bmatrix} U_1 & U_2 \end{bmatrix}$ and $V := \begin{bmatrix} V_1 & V_2 \end{bmatrix}$ be unitary matrices such that

$$U^*HV = \begin{bmatrix} \hat{H} & 0\\ 0 & 0 \end{bmatrix} = \begin{bmatrix} U_1^*HV_1 & 0\\ 0 & 0 \end{bmatrix},$$
(8.17)
where \widehat{H} is $r_f \times r_f$ and invertible. Let us now partition the matrices U_1 and V_1 as follows

$$U_1 = \begin{bmatrix} U_{11} \\ U_{21} \end{bmatrix}, \quad \text{and} \quad V_1 = \begin{bmatrix} V_{11} \\ V_{21} \end{bmatrix}, \quad (8.18)$$

where the matrices U_{11} and V_{11} have dimension $m \times r_f$ and $n \times r_f$, respectively. Then

$$L_{sp}(\lambda) := \begin{bmatrix} U_1^* H_\sigma V_1 - \lambda \widehat{H} & \widehat{H} V_{11}^* \\ \hline U_{11} \widehat{H} & 0 \end{bmatrix}$$
(8.19)

is a strongly minimal linearization for $R_{sp}(\lambda)$. In particular, $R_{sp}(\lambda) = U_{11}\widehat{H}(\lambda\widehat{H} - U_1^*H_{\sigma}V_1)^{-1}\widehat{H}V_{11}^*$.

8.4 For self-conjugate strictly proper rational matrices

If a strictly proper rational matrix $R_{sp}(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$ have one of the four selfconjugate structures considered in this chapter, we use Theorem 8.3.1 and the ideas developed in Chapter 3 (Remarks 3.5.6 and 3.7.4) to construct strongly minimal self-conjugate linearizations for $R_{sp}(\lambda)$. First notice that the block Hankel matrices H and H_{σ} in (8.16) have the following self-conjugate property depending on that of $R_{sp}(\lambda)$.

Lemma 8.4.1. Let $R_{sp}(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$ be a strictly proper rational matrix as in (8.15). Let us define the scaling matrix $S := \text{diag}(-I_m, (-1)^2 I_m, \dots, (-1)^k I_m)$. Then the block Hankel matrices H and H_{σ} in (8.16) satisfy the following equations

- 1. for Hermitian $R_{sp}(\lambda)$: $R_{-i}^* = R_{-i}$, $H^* = H$ and $H_{\sigma}^* = H_{\sigma}$,
- 2. for skew-Hermitian $R_{sp}(\lambda)$: $R^*_{-i} = -R_{-i}$, $H^* = -H$ and $H^*_{\sigma} = -H_{\sigma}$,
- 3. for para-Hermitian $R_{sp}(\lambda)$: $R^*_{-i} = (-1)^i R_{-i}$, $(SH)^* = -SH$ and $(SH_{\sigma})^* = SH_{\sigma}$,
- 4. for para-skew-Hermitian $R_{sp}(\lambda)$: $R_{-i}^* = (-1)^{(i+1)}R_{-i}$, $(SH)^* = SH$ and $(SH_{\sigma})^* = -SH_{\sigma}$.

For each of these cases, the left and right transformations U and V in (8.17) can be chosen as U = V in the Hermitian and skew-Hermitian cases and as U = SV in the para-Hermitian and para-skew-Hermitian cases.

Proof. The symmetries of the coefficient matrices R_i trivially yield the four types of symmetries of H and H_{σ} . For the rank compression (8.17), we can then choose U = V in the Hermitian and skew-Hermitian cases because H is normal, and we can choose U = SV in the para-Hermitian and para-skew-Hermitian cases because SH is then normal.

We can now use Lemma 8.4.1 and Theorem 8.3.1 to obtain the following strongly minimal linearizations for the four considered structures.

Theorem 8.4.2. Let $R_{sp}(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$ be a strictly proper rational matrix as in (8.15), and let H and H_{σ} be the associated block Hankel matrices appearing in (8.16). Let U, V be the unitary matrices appearing in (8.17), where U = V if $R_{sp}(\lambda)$ is Hermitian or skew-Hermitian, and U = SV if $R_{sp}(\lambda)$ is para-Hermitian or para-skew-Hermitian, and where $S := \text{diag}(-I_m, (-1)^2 I_m, \ldots, (-1)^k I_m)$. Finally, let U_1, V_1 be the km $\times r_f$ matrices formed by the first r_f columns of U and V, respectively, \hat{H} be the $r_f \times r_f$ defined in (8.17) and U_{11}, V_{11} be the m $\times r_f$ matrices defined in (8.18).

1. If $R_{sp}(\lambda)$ is Hermitian, then $\widehat{H}^* = \widehat{H}$, $H^*_{\sigma} = H_{\sigma}$, and

$$L_{sp}(\lambda) := \left[\begin{array}{c|c} V_1^* H_\sigma V_1 - \lambda \widehat{H} & \widehat{H} V_{11}^* \\ \hline V_{11} \widehat{H} & 0 \end{array} \right]$$

is a Hermitian strongly minimal linear polynomial system matrix of $R_{sp}(\lambda)$.

2. If $R_{sp}(\lambda)$ is skew-Hermitian, then $\widehat{H}^* = -\widehat{H}$, $H^*_{\sigma} = -H_{\sigma}$, and

$$L_{sp}(\lambda) := \left[\begin{array}{c|c} V_1^* H_\sigma V_1 - \lambda \widehat{H} & \widehat{H} V_{11}^* \\ \hline V_{11} \widehat{H} & 0 \end{array} \right]$$

is a skew-Hermitian strongly minimal linear polynomial system matrix of $R_{sp}(\lambda)$.

3. If $R_{sp}(\lambda)$ is para-Hermitian, then $\widehat{H}^* = -\widehat{H}$, $(SH_{\sigma})^* = SH_{\sigma}$, and

$$L_{sp}(\lambda) := \left[\begin{array}{c|c} V_1^* S H_\sigma V_1 - \lambda \widehat{H} & \widehat{H} V_{11}^* \\ \hline - V_{11} \widehat{H} & 0 \end{array} \right]$$

is a para-Hermitian strongly minimal linear polynomial system matrix of $R_{sp}(\lambda)$.

4. If $R_{sp}(\lambda)$ is para-skew-Hermitian, then $\widehat{H}^* = \widehat{H}$, $(SH_{\sigma})^* = -SH_{\sigma}$, and

$$L_{sp}(\lambda) := \begin{bmatrix} V_1^* S H_\sigma V_1 - \lambda \widehat{H} & \widehat{H} V_{11}^* \\ \hline - V_{11} \widehat{H} & 0 \end{bmatrix}$$

is a para-skew-Hermitian strongly minimal linear polynomial system matrix of $R_{sp}(\lambda)$.

Proof. 1. If $R_{sp}(\lambda)$ is Hermitian, then Lemma 8.4.1 and (8.17) imply that $H^* = H$, $H^*_{\sigma} = H_{\sigma}$, $U_1 = V_1$, and $U_{11} = V_{11}$. The result then follows from $\hat{H} = V_1^* H V_1 = \hat{H}^*$ and (8.19).

- 2. If $R_{sp}(\lambda)$ is skew-Hermitian, then Lemma 8.4.1 and (8.17) imply that $H^* = -H$, $H^*_{\sigma} = -H_{\sigma}$, $U_1 = V_1$, and $U_{11} = V_{11}$. The result then follows from $\widehat{H} = V_1^* H V_1 = -\widehat{H}^*$ and (8.19).
- 3. If $R_{sp}(\lambda)$ is para-Hermitian, then Lemma 8.4.1 and (8.17) imply that $(SH)^* = -SH$, $(SH_{\sigma})^* = SH_{\sigma}$, $U_1 = SV_1$, and $U_{11} = -V_{11}$. The result then follows from $\widehat{H} = V_1^*SHV_1 = -\widehat{H}^*$ and (8.19).
- 4. If $R_{sp}(\lambda)$ is para-skew-Hermitian, then Lemma 8.4.1 and (8.17) imply that $(SH)^* = SH, (SH_{\sigma})^* = -SH_{\sigma}, U_1 = SV_1$, and $U_{11} = -V_{11}$. The result then follows from $\widehat{H} = V_1^* SHV_1 = \widehat{H}^*$ and (8.19).

8.5 For arbitrary and self-conjugate rational matrices

For any given rational matrix $R(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$ we assume that we have an additive decomposition into its polynomial part $P(\lambda)$ and its strictly proper part $R_{sp}(\lambda)$ as in (2.1). That is,

$$R(\lambda) = P(\lambda) + R_{sp}(\lambda). \tag{8.20}$$

For the Laurent expansion given in (8.1), this corresponds to

$$P(\lambda) := R_0 + R_1 \lambda + \ldots + R_d \lambda^d, \quad R_{sp}(\lambda) := R_{-1} \lambda^{-1} + R_{-2} \lambda^{-2} + R_{-3} \lambda^{-3} + \ldots$$

In this section, we obtain strongly minimal linearizations for arbitrary and structured rational matrices by combining strongly minimal linearizations for both parts. The construction for the polynomial part was given in Sections 8.1 and 8.2, and for the strictly proper part in Sections 8.3 and 8.4. Although some partial results can be found in the literature [40, 27, 19], the derivation given here is new and more general.

Once we have strongly minimal linearizations for both the polynomial part and the strictly proper part of a given rational matrix, it is straightforward to construct a strongly minimal linearization for the sum, as shown below.

Theorem 8.5.1. Let $R(\lambda) \in \mathbb{C}(\lambda)^{m \times n}$ be an arbitrary rational matrix, i.e., regular or singular. Let $R(\lambda) = P(\lambda) + R_{sp}(\lambda)$ where $P(\lambda)$ is the polynomial part of $R(\lambda)$ and $R_{sp}(\lambda)$ is the strictly proper part of $R(\lambda)$. Let

$$\widehat{L}_{s}(\lambda) := \left[\begin{array}{c|c} \widehat{A}_{s}(\lambda) & -\widehat{B}_{s}(\lambda) \\ \hline \widehat{C}_{s}(\lambda) & \widehat{D}_{s}(\lambda) \end{array} \right], \quad \text{and} \quad L_{sp}(\lambda) := \left[\begin{array}{c|c} A_{sp}(\lambda) & -B_{sp}(\lambda) \\ \hline C_{sp}(\lambda) & 0 \end{array} \right], \quad (8.21)$$

be strongly minimal linear polynomial system matrices of $P(\lambda)$ and $R_{sp}(\lambda)$ as described in Theorems 8.1.2 and 8.3.1, respectively. Then

$$L(\lambda) := \begin{bmatrix} \widehat{A}_s(\lambda) & 0 & | & -\widehat{B}_s(\lambda) \\ 0 & A_{sp}(\lambda) & | & -B_{sp}(\lambda) \\ \hline \widehat{C}_s(\lambda) & C_{sp}(\lambda) & | & \widehat{D}_s(\lambda) \end{bmatrix}$$
(8.22)

is a strongly minimal linearization of $R(\lambda)$.

Proof. The transfer function of $L(\lambda)$ is clearly

$$\widehat{D}_s(\lambda) + \widehat{C}_s(\lambda)\widehat{A}_s(\lambda)^{-1}\widehat{B}_s(\lambda) + C_{sp}(\lambda)A_{sp}(\lambda)^{-1}B_{sp}(\lambda) = P(\lambda) + R_{sp}(\lambda) = R(\lambda).$$

The strong minimality of $L(\lambda)$ follows from the fact that the subsystems $\widehat{L}_s(\lambda)$ and $L_{sp}(\lambda)$ are strongly minimal and have no common poles.

For the construction of self-conjugate strongly minimal linearizations of structured rational matrices we proceed in the same way and obtain the following result.

Theorem 8.5.2. Let $R(\lambda) \in \mathbb{C}(\lambda)^{m \times m}$ be an arbitrary rational matrix, i.e., regular or singular, which has one of the following structures: Hermitian, skew-Hermitian, para-Hermitian or para-skew-Hermitian. Let $R(\lambda) = P(\lambda) + R_{sp}(\lambda)$ where $P(\lambda)$ is the polynomial part of $R(\lambda)$ and $R_{sp}(\lambda)$ is the strictly proper part of $R(\lambda)$. Let

$$\widehat{L}_{s}(\lambda) := \begin{bmatrix} \widehat{A}_{s}(\lambda) & -\widehat{B}_{s}(\lambda) \\ \hline \widehat{C}_{s}(\lambda) & \widehat{D}_{s}(\lambda) \end{bmatrix}, \text{ and } L_{sp}(\lambda) := \begin{bmatrix} A_{sp}(\lambda) & -B_{sp}(\lambda) \\ \hline C_{sp}(\lambda) & 0 \end{bmatrix}, \quad (8.23)$$

be strongly minimal linear polynomial system matrices of $P(\lambda)$ and $R_{sp}(\lambda)$ as described in Theorems 8.2.2 and 8.4.2, respectively, according to the corresponding structure of $R(\lambda)$. Then

$$L(\lambda) := \begin{bmatrix} \widehat{A}_s(\lambda) & 0 & -\widehat{B}_s(\lambda) \\ 0 & A_{sp}(\lambda) & -B_{sp}(\lambda) \\ \hline \widehat{C}_s(\lambda) & C_{sp}(\lambda) & \widehat{D}_s(\lambda) \end{bmatrix}$$
(8.24)

is a strongly minimal linearization of $R(\lambda)$ with the same self-conjugate structure as $R(\lambda)$.

Proof. $L(\lambda)$ is a strongly minimal linearization of $R(\lambda)$ by using the same proof as that of Theorem 8.5.1. That the self-conjugate structure of $R(\lambda)$ and $L(\lambda)$ are the same, follows from the fact that the self-conjugate structures of $\hat{L}_s(\lambda)$ and $L_{sp}(\lambda)$ coincide with that of $R(\lambda)$.

8.6 Algorithmic aspects

The constructive proofs given in the earlier sections in fact lead to possible algorithms for computing strongly minimal linearizations of rational matrices, provided the Laurent expansion (8.1) is given up to the term R_{-2k} . The two decompositions that are required for the construction of the linearizations are the rank factorizations of the matrices T in (8.13) and H in (8.17). Moreover, because of the symmetry in these problems, only the right transformation has to be computed in the structured cases.

It is worth pointing out also that both factorizations only require to construct unitary transformations that "compress" the rows and columns of a given matrix, which can be obtained by a QR factorization. Efficient algorithms that exploit the special block-Hankel and or block-Toeplitz structure can be found in the literature [49, 88]. We imposed no conditions of the normal rank of the rational matrix, although the self-conjugate structure imposes that they are square. There can therefore be a left and a right null space and their structural indices will be equal because of the self-conjugate nature of the transfer function and the system matrix.

Since the linear polynomial system matrices derived here are strongly minimal and structured, one can use the properties of such pencils to compute efficiently the left and right eigenvectors in the regular case, and the left and right null space structure in the singular case. 172CHAPTER 8. CONSTRUCTING STRONGLY MINIMAL LINEARIZATIONS

Chapter 9

Structural backward stability in rational eigenvalue problems solved via block Kronecker linearizations

In this chapter we study the backward stability of running a backward stable eigenstructure solver on a pencil $S(\lambda)$ that is a strong linearization of a rational matrix $R(\lambda)$ expressed in the form $R(\lambda) = D(\lambda) + C(\lambda I_{\ell} - A)^{-1}B$, where $D(\lambda)$ is a polynomial matrix and $C(\lambda I_{\ell} - A)^{-1}B$ is a minimal state-space realization. We consider the family of block Kronecker linearizations of $R(\lambda)$, which have the following structure

$$S(\lambda) := \begin{bmatrix} M(\lambda) & \widehat{K}_2^T C & K_2^T(\lambda) \\ B\widehat{K}_1 & A - \lambda I_\ell & 0 \\ K_1(\lambda) & 0 & 0 \end{bmatrix},$$

where the blocks have some specific structures. Backward stable eigenstructure solvers, such as the QZ or the staircase algorithms, applied to $S(\lambda)$ will compute the exact eigenstructure of a perturbed pencil

$$\widehat{S}(\lambda) := S(\lambda) + \Delta_S(\lambda)$$

and the special structure of $S(\lambda)$ will be lost, including the zero blocks below the anti-diagonal. In order to link this perturbed pencil with a nearby rational matrix, we construct in this chapter a strictly equivalent pencil

$$\widetilde{S}(\lambda) = (I - X)\widehat{S}(\lambda)(I - Y)$$

that restores the original structure, and hence is a block Kronecker linearization of a perturbed rational matrix $\widetilde{R}(\lambda) = \widetilde{D}(\lambda) + \widetilde{C}(\lambda I_{\ell} - \widetilde{A})^{-1}\widetilde{B}$, where $\widetilde{D}(\lambda)$ is a polynomial matrix with the same degree as $D(\lambda)$. Moreover, we bound appropriate norms of $\widetilde{D}(\lambda) - D(\lambda)$, $\widetilde{C} - C$, $\widetilde{A} - A$ and $\widetilde{B} - B$ in terms of an appropriate norm of $\Delta_S(\lambda)$. These bounds may be, in general, inadmissibly large, but we also introduce a scaling that allows us to make them satisfactorily tiny, by making the matrices appearing in both $S(\lambda)$ and $R(\lambda)$ have norms bounded by 1. Thus, for this scaled representation, we prove that the staircase and the QZ algorithms compute the exact eigenstructure of a rational matrix $\widetilde{R}(\lambda)$ that can be expressed in exactly the same form as $R(\lambda)$ with the parameters defining the representation very near to those of $R(\lambda)$. This shows that this approach is backward stable in a structured sense. All the results in this chapter appear in [33].

9.1 Some preliminaries

A wide family of strong linearizations called strong block minimal bases linearizations is proposed in [6, Theorem 5.11] for any $m \times n$ rational matrix $R(\lambda)$, inspired by previous results for polynomial matrices in [26] (recall Theorem 2.5.5). These linearizations are based on the splitting of $R(\lambda)$ into its strictly proper part $R_p(\lambda)$ and its polynomial part $D(\lambda)$ and in the representation:

$$R(\lambda) := R_p(\lambda) + D(\lambda) = C(\lambda I_\ell - A)^{-1}B + \sum_{i=0}^d D_i \lambda^i, \qquad (9.1)$$

where $C(\lambda I_{\ell} - A)^{-1}B$ is a minimal state-space realization of the strictly proper part $R_p(\lambda)$, represented in what follows by the triple $\{A, B, C\}$, and d > 1 is the degree of the polynomial part. Then $R(\lambda)$ is represented by the quadruple $\{\lambda I_{\ell} - A, B, C, D(\lambda)\}$. Since in this chapter we are analyzing perturbations related to backward errors of eigenvalue solvers of pencils with real or complex matrix coefficients, we restrict \mathbb{F} to be the real field \mathbb{R} or the complex field \mathbb{C} .

A particular case of the strong block minimal bases linearizations in [6, Theorem 5.11] of any $m \times n$ rational matrix $R(\lambda)$ represented as in (9.1) are (modulo block permutations) the pencils of the form

$$S(\lambda) := \begin{bmatrix} M(\lambda) & \widehat{K}_2^T C & K_2^T(\lambda) \\ B\widehat{K}_1 & A - \lambda I_\ell & 0 \\ K_1(\lambda) & 0 & 0 \end{bmatrix},$$
(9.2)

with

$$K_1(\lambda) := L_{\epsilon}(\lambda) \otimes I_n, \quad \widehat{K}_1 := \mathbf{e}_{\epsilon+1}^T \otimes I_n, \quad K_2(\lambda) := L_{\eta}(\lambda) \otimes I_m, \quad \widehat{K}_2 := \mathbf{e}_{\eta+1}^T \otimes I_m,$$

and where \otimes denotes the Kronecker product, $\mathbf{e}_k = [0 \cdots 0 \, 1]^T$ is the standard kth unit vector of dimension k and $L_k(\lambda)$ is the classical Kronecker block of dimension $k \times (k+1)$

$$L_k(\lambda) := \begin{bmatrix} 1 & -\lambda & & \\ & 1 & -\lambda & \\ & & \ddots & \ddots & \\ & & & 1 & -\lambda \end{bmatrix}.$$

Moreover, the block $M(\lambda)$ in (9.2) is related to the polynomial part $D(\lambda)$ in (9.1) by the "dual basis" vector $\Lambda_k(\lambda)$ of powers of λ ,

$$\Lambda_k^T(\lambda) := \left[\begin{array}{ccc} \lambda^k & \cdots & \lambda^2 & \lambda & 1 \end{array} \right],$$

which satisfies $L_k(\lambda)\Lambda_k(\lambda) = 0$ and also

$$D(\lambda) = (\Lambda_{\eta}(\lambda) \otimes I_m)^T M(\lambda) (\Lambda_{\epsilon}(\lambda) \otimes I_n).$$

Thus, $d = \epsilon + \eta + 1$ (see [26, eq. (4.5)]). The strong linearizations (9.2) are inspired by the so-called "block Kronecker linearizations" that were introduced in [26, Section 4] for an arbitrary $m \times n$ polynomial matrix $D(\lambda)$. Therefore, we use the same name in the rational setting. The representation of $R(\lambda)$ in (9.1) and the block Kronecker linearizations $S(\lambda)$ of $R(\lambda)$ (9.2) are the two fundamental ingredients of this chapter.

As explained in [6, Section 3.1], the finite eigenvalues, together with their partial multiplicities, of $S(\lambda)$ (resp. $A - \lambda I_{\ell}$) coincide with the finite zeros (resp. poles) of $R(\lambda)$, together with their partial multiplicities. Moreover, the eigenvalue structure at infinity of $S(\lambda)$ allows us to obtain via a simple shift rule the pole-zero structure at infinity of $R(\lambda)$.¹ In addition, as proved in [8, Section 6], the right (resp. left) minimal indices of $S(\lambda)$ are those of $R(\lambda)$ plus ϵ (resp. η). Thus, $S(\lambda)$ comprises the complete eigenstructure of $R(\lambda)$. Observe that the application to $S(\lambda)$ of the QZ algorithm [71], in the regular case, or of the staircase algorithm [83], in the singular case, gives the zeros and the minimal indices, in the singular case, of $R(\lambda)$, but not the poles, which are in $A - \lambda I_{\ell}$.

It is worth mentioning that although the families of block Kronecker linearizations of polynomial [26] and rational [6] matrices are very recent, some particular examples of strong linearizations in these families appeared much earlier in the literature. For instance, it was shown in [87] that a valid "realization" for the polynomial part $D(\lambda)$ in (9.1) is given by the following minimal polynomial system matrix

$$S_D(\lambda) := \begin{bmatrix} I_n & -\lambda I_n & & & \\ & I_n & \ddots & & \\ & & \ddots & -\lambda I_n & \\ & & & I_n & -\lambda I_n \\ \hline & & & & \lambda D_2 & | \lambda D_1 + D_0 \end{bmatrix} := \begin{bmatrix} T(\lambda) & | -U(\lambda) \\ \hline V(\lambda) & | W(\lambda) \end{bmatrix},$$

¹More precisely, according to [6, p. 1683], if r is the normal rank of $R(\lambda)$ and $e_1 \leq \cdots \leq e_r$ are the r largest partial multiplicities at infinity of $S(\lambda)$, then $e_1 - d \leq \cdots \leq e_r - d$ are the structural indices at infinity of $R(\lambda)$.

which means that $D(\lambda) = W(\lambda) + V(\lambda)T(\lambda)^{-1}U(\lambda)$. It is easy to see that after moving the bottom block row of $S_D(\lambda)$ to the top position, a block Kronecker linearization of $D(\lambda)$ is obtained with $K_2(\lambda)$ empty [26, Section 4]. Combining the minimal state-space realization $C(\lambda I_{\ell} - A)^{-1}B$ and the polynomial system matrix $S_D(\lambda)$ yields the following minimal polynomial system matrix for the rational matrix $R(\lambda)$ in (9.1):

$$S_R(\lambda) := \begin{bmatrix} A - \lambda I_\ell & & & B \\ I_n & -\lambda I_n & & & \\ & I_n & \ddots & & \\ & & I_n & \ddots & \\ & & & \ddots & -\lambda I_n \\ & & & I_n & -\lambda I_n \\ \hline C & & \lambda D_d & \dots & \dots & \lambda D_2 & \lambda D_1 + D_0 \end{bmatrix} := \begin{bmatrix} T_R(\lambda) & -U_R(\lambda) \\ V_R(\lambda) & W_R(\lambda) \end{bmatrix},$$

i.e., $R(\lambda) = W_R(\lambda) + V_R(\lambda)T_R(\lambda)^{-1}U_R(\lambda)$. A pencil with a structure similar to $S_R(\lambda)$ can also be found in [79]. It is easy to see that, modulo some block permutations, $S_R(\lambda)$ is a particular case of the block Kronecker linearizations appearing in (9.2) for $R(\lambda)$, with $K_2(\lambda)$ empty and $\hat{K}_2 = I_m$.

It was shown in [26] that perturbations of the block Kronecker linearizations of a polynomial matrix $D(\lambda)$ can be mapped to perturbations of the coefficients of $D(\lambda)$ without significant growth of the relative norms of the perturbations under mild assumptions that require to scale $D(\lambda)$ to have norm equal to 1 and to use linearizations with the norm of $M(\lambda)$ of the same order as the norm of $D(\lambda)$ (see [26, Corollary 5.24]). As a corollary of this perturbation result, we obtain that under such assumptions the computation of the eigenvalues and minimal indices of a polynomial matrix by applying the QZ or the staircase algorithm to one of its block Kronecker linearizations is a backward stable method from the point of view of the polynomial matrix. In this chapter we show that this can be extended to rational matrices as well, considering as coefficients of the rational matrix those in the quadruple $\{\lambda I_{\ell} - A, B, C, D(\lambda)\}$. However, we emphasize that the perturbation analysis for block Kronecker linearizations of rational matrices is considerably more complicated than the one in [26] and, therefore, we limit ourselves to perform a first order analysis. We also remark that the scaling needed to get satisfactory perturbation bounds is more delicate than the one in [26]. As far as we know, this is the first structural backward error analysis of this type performed in the literature for linearizations of rational matrices.

We assume throughout the chapter that $\ell > 0$ since, otherwise, $R(\lambda)$ in (9.1) is a polynomial matrix and this case was studied in [26]. Except in Section 9.6, we also assume that at least one of the parameters ϵ and η in (9.2) is larger than zero since, otherwise, none of the blocks $K_1(\lambda)$ and $K_2(\lambda)$ appears and block Kronecker linearizations collapse to much simpler pencils. Note that $\max(\eta, \epsilon) > 0$ implies that the degree d of the polynomial part $D(\lambda)$ of $R(\lambda)$ is larger than 1. The simple case $d \leq 1$ is studied in Section 9.6.

9.1.1 Norms

In order to measure perturbations, we need to introduce appropriate norms for pencils, polynomial matrices and rational matrices expressed as in (9.1). For any pair of matrices X and Y of arbitrary dimensions (that might be different), we will use the following norms

$$\|(X,Y)\|_F := \left(\|X\|_F^2 + \|Y\|_F^2\right)^{\frac{1}{2}} = \|\left[\operatorname{vec}(X)^T, \operatorname{vec}(Y)^T\right]\|_2, \\ \|(X,Y)\|_2 := \left(\|X\|_2^2 + \|Y\|_2^2\right)^{\frac{1}{2}},$$

where $||X||_F$ and $||X||_2$ are, respectively, the Frobenius and spectral matrix norms and $\operatorname{vec}(X)$ is the operator that stacks the columns of a matrix into one column vector [50]. For a pencil $S(\lambda) := A - \lambda B$ we define the corresponding norms via the two matrix coefficients :

$$||S(\lambda)||_F := ||(A, B)||_F, ||S(\lambda)||_2 := ||(A, B)||_2.$$

More generally, for a polynomial matrix $D(\lambda) := \sum_{i=0}^{d} D_i \lambda^i$, we will use the norm

$$||D(\lambda)||_F := \sqrt{\sum_{i=0}^d ||D_i||_F^2},$$

and for a list of polynomial matrices $(D_1(\lambda), \ldots, D_p(\lambda))$, the norm

$$||(D_1(\lambda), \dots, D_p(\lambda))||_F := \sqrt{\sum_{i=1}^p ||D_i(\lambda)||_F^2}$$

Finally, for a rational matrix $R(\lambda)$, represented by a quadruple $\{\lambda I_{\ell} - A, B, C, D(\lambda)\}$, as in (9.1), we use the "norm"

$$||R(\lambda)||_F := ||(\lambda I_\ell - A, B, C, D(\lambda))||_F = \sqrt{\ell + ||A||_F^2 + ||B||_F^2 + ||C||_F^2 + \sum_{i=0}^d ||D_i||_F^2}.$$

That is, the "norm" of a rational matrix $R(\lambda)$ is defined as the norm of an associated polynomial system matrix $P(\lambda)$, in this case,

$$||R(\lambda)||_F := ||P(\lambda)||_F \quad \text{where} \quad P(\lambda) := \begin{bmatrix} \lambda I_\ell - A & -B \\ C & D(\lambda) \end{bmatrix}.$$
(9.3)

We remark that $||R(\lambda)||_F$ is not rigorously a "norm" for $R(\lambda)$ because, for instance, $R(\lambda)$ is zero if B = 0 and $D(\lambda) = 0$, but $||R(\lambda)||_F$ is not. Despite this fact, and with a clear abuse of nomenclature, we will use the terminology "norm of a rational matrix" in the sense explained above.

We describe in Section 9.2 the basic systems of matrix equations we will use in this chapter, and, in Section 9.3, some bounds for the singular values of certain matrices related to these systems of matrix equations.

9.2 Generalized Sylvester equations

In order to restore the structure of perturbed block Kronecker linearizations of rational matrices, we will need to guarantee that some matrix equations have solutions and to bound the norm of their minimal norm solution. The matrix equations that we will encounter are particular cases of the generalized Sylvester equation for $m_i \times n_i$ pencils of matrices $A_i - \lambda B_i$, i = 1, 2, which is the following equation in the unknowns X and Y:

$$X(A_1 - \lambda B_1) + (A_2 - \lambda B_2)Y = \Delta^a - \lambda \Delta^b.$$
(9.4)

It is easily seen to be equivalent to a linear system of equations, when rewriting it as

$$\begin{aligned} XA_1 + A_2Y &= \Delta^a, \\ XB_1 + B_2Y &= \Delta^b, \end{aligned}$$

or, when using Kronecker products and the $vec(\cdot)$ notation, as

$$\begin{bmatrix} A_1^T \otimes I_{m_2} & I_{n_1} \otimes A_2 \\ \hline B_1^T \otimes I_{m_2} & I_{n_1} \otimes B_2 \end{bmatrix} \begin{bmatrix} \operatorname{vec}(X) \\ \hline \operatorname{vec}(Y) \end{bmatrix} = \begin{bmatrix} \operatorname{vec}(\Delta^a) \\ \hline \operatorname{vec}(\Delta^b) \end{bmatrix}.$$
(9.5)

The dimension of the unknowns X and Y are $m_2 \times m_1$ and $n_2 \times n_1$, respectively, and those of the right hand sides Δ^a and Δ^b are each $m_2 \times n_1$. These equations will be used in this chapter in two contexts, which we briefly recall here.

9.2.0.1 Block elimination

Let $A_i - \lambda B_i$ be two $m_i \times n_i$ pencils, i = 1, 2, that have respectively full column normal rank n_1 and full row normal rank m_2 . Then the problem of block antidiagonalizing the pencil $\begin{bmatrix} 0 & A_1 - \lambda B_1 \\ A_2 - \lambda B_2 & \Delta^a - \lambda \Delta^b \end{bmatrix}$, that is, finding X and Y such that

$$\begin{bmatrix} I_{m_1} & 0\\ -X & I_{m_2} \end{bmatrix} \begin{bmatrix} 0 & A_1 - \lambda B_1\\ A_2 - \lambda B_2 & \Delta^a - \lambda \Delta^b \end{bmatrix} \begin{bmatrix} I_{n_2} & -Y\\ 0 & I_{n_1} \end{bmatrix} = \begin{bmatrix} 0 & A_1 - \lambda B_1\\ A_2 - \lambda B_2 & 0 \end{bmatrix},$$
(9.6)

amounts to finding a solution for the generalized Sylvester equation (9.4). It is known that there exists a solution $(X, Y) \in \mathbb{F}^{m_2 \times m_1} \times \mathbb{F}^{n_2 \times n_1}$ for a *particular* right hand side $(\Delta^a, \Delta^b) \in \mathbb{F}^{m_2 \times n_1} \times \mathbb{F}^{m_2 \times n_1}$ if and only if the pencils

$$\begin{bmatrix} 0 & A_1 - \lambda B_1 \\ A_2 - \lambda B_2 & \Delta^a - \lambda \Delta^b \end{bmatrix} \text{ and } \begin{bmatrix} 0 & A_1 - \lambda B_1 \\ A_2 - \lambda B_2 & 0 \end{bmatrix}$$

are strictly equivalent (i.e. have the same Kronecker structure) [23]. But in order to have a solution for any right hand side $\Delta^a - \lambda \Delta^b$ one requires the stronger condition that the pencils $A_1 - \lambda B_1$ and $A_2 - \lambda B_2$ have no common generalized eigenvalues (see [86]). We recall here the result proven in [86] that is relevant for our work. **Theorem 9.2.1.** [86] Let the pencils $A_i - \lambda B_i$ of dimensions $m_i \times n_i$, i = 1, 2, be respectively of full column normal rank $n_1 \leq m_1$ and of full row normal rank $m_2 \leq n_2$, and let these two pencils have no common generalized eigenvalues. Then there always exists a solution (X, Y) to the system of equations (9.6), for any perturbation $\Delta^a - \lambda \Delta^b$. Moreover, the generalized eigenvalues of the pencil (9.6) are the union of the generalized eigenvalues of the pencils $A_i - \lambda B_i$, i = 1, 2.

The system is underdetermined if either of the two inequalities $m_1 \ge n_1$ and $n_2 \ge m_2$, is strict. Under the hypotheses of Theorem 9.2.1, the system (9.5) must be compatible for any right hand side, and hence the Kronecker product matrix in the left hand side of (9.5) must have full row rank $2m_2n_1$. A bound for the minimum Frobenius-norm solution (X, Y) is then obtained in terms of the smallest singular value $\sigma_{2m_2n_1}$ of the matrix in (9.5):

$$\|(X,Y)\|_{F} \leq \frac{\|(\Delta^{a},\Delta^{b})\|_{F}}{\sigma_{2m_{2}n_{1}}\left(\left[\begin{array}{c|c}A_{1}^{T}\otimes I_{m_{2}} & I_{n_{1}}\otimes A_{2}\\\hline B_{1}^{T}\otimes I_{m_{2}} & I_{n_{1}}\otimes B_{2}\end{array}\right]\right)}.$$
(9.7)

9.2.0.2 Equivalent pencils

The second problem in this chapter where a generalized Sylvester equation as in (9.4) arises is that of strictly equivalent pencils (see e.g. [39]). Let the pencils $A_i - \lambda B_i$, i = 1, 2, be both of dimension $m \times n$, then they are strictly equivalent if and only if there exist invertible matrices S and T such that $S(A_1 - \lambda B_1) = (A_2 - \lambda B_2)T$. Such pencils must then have the same Kronecker canonical form [39]. We are interested in finding the solution where S and T are as close as possible to the identity matrix. This can be achieved by writing the transformation matrices as

$$S = I + X, \quad T = I - Y$$

and then minimizing the Frobenius norm of the pair (X, Y). The corresponding equations are then

$$(I + X)(A_1 - \lambda B_1) = (A_2 - \lambda B_2)(I - Y)$$

or, when putting $\Delta^a - \lambda \Delta^b := (A_2 - \lambda B_2) - (A_1 - \lambda B_1)$, we finally obtain

$$X(A_1 - \lambda B_1) + (A_2 - \lambda B_2)Y = \Delta^a - \lambda \Delta^b, \qquad (9.8)$$

which is again solved by using (9.5). We will use this to "restore" a slightly perturbed pencil $(A_2 - \lambda B_2) := (A_1 - \lambda B_1) + (\Delta^a - \lambda \Delta^b)$ to its original form $(A_1 - \lambda B_1)$ using a strict equivalence transformation

$$(I+X)^{-1}(A_2 - \lambda B_2)(I-Y) = A_1 - \lambda B_1$$
(9.9)

that is very close to the identity, when we are sure that both pencils have the same Kronecker canonical form. The bounds for the norm of X and Y are in fact given by (9.7) for which we derive exact expressions in the next section. Notice that we can not apply Theorem 9.2.1 to prove existence of a solution for equation (9.8), since in this case both pencils must have the same generalized eigenvalues and the same normal rank. A sufficient condition for the consistency of (9.8) is that $A_1 - \lambda B_1$ and $A_2 - \lambda B_2$ have the same Kronecker canonical form.

The condition that the Kronecker canonical form of a pencil does not change under arbitrary sufficiently small perturbations only holds for very special pencils. In particular, it holds for the Kronecker product of Kronecker blocks times identity matrices, i.e., for $L_k(\lambda) \otimes I_r$. This is a consequence of the results in [89], because $L_k(\lambda) \otimes I_r$ has full-Sylvester-rank by [89, Theorem 4.3(a)] and, then, [89, Theorem 6.6] guarantees that $L_k(\lambda) \otimes I_r + (\Delta^a - \lambda \Delta^b)$ has the same Kronecker canonical form as $L_k(\lambda) \otimes I_r$ for all the perturbations (Δ^a, Δ^b) whose norms are smaller than the bounds in [89, Theorem 6.6]. Since we will solve (9.8)-(9.9) only in the case $A_1 - \lambda B_1 = L_k(\lambda) \otimes I_r$, these results prove that (9.8) has a solution for all sufficiently small perturbations (Δ^a, Δ^b) in the cases of interest in this chapter.

9.3 Singular value bounds

In the analysis of Section 9.4, we will need upper bounds for the minimum norm solutions of the generalized Sylvester equation (9.4) for pairs of pencils $(A_i - \lambda B_i)$, i = 1, 2, which all involve Kronecker blocks $L_k(\lambda) := E_k - \lambda F_k$, where the $k \times (k + 1)$ matrices E_k and F_k are given by

$$E_k := \begin{bmatrix} 1 & 0 & & & \\ & 1 & 0 & & \\ & & \ddots & \ddots & \\ & & & 1 & 0 \end{bmatrix} \quad \text{and} \quad F_k := \begin{bmatrix} 0 & 1 & & & \\ & 0 & 1 & & \\ & & \ddots & \ddots & \\ & & & 0 & 1 \end{bmatrix}.$$

To find such upper bounds is equivalent to find lower bounds for the singular values in the denominator of the right hand side of (9.7). We consider the generalized Sylvester equations for the following list of pencil pairs with their smallest singular value of the corresponding linear maps:

1.
$$A_1 - \lambda B_1 = A - \lambda I_\ell$$
 and $A_2 - \lambda B_2 = L_{\varepsilon}(\lambda) \otimes I_n$:

$$\omega_1 := \sigma_{2\ell\varepsilon n} \left[\frac{A^T \otimes I_{\varepsilon n} \mid I_\ell \otimes E_{\varepsilon} \otimes I_n}{I_\ell \otimes I_{\varepsilon n} \mid I_\ell \otimes F_{\varepsilon} \otimes I_n} \right].$$
(9.10)

2.
$$A_1 - \lambda B_1 = L_{\eta}^T(\lambda) \otimes I_m$$
 and $A_2 - \lambda B_2 = A - \lambda I_{\ell}$:

$$\omega_2 := \sigma_{2\eta m\ell} \left[\frac{E_\eta \otimes I_{m\ell} \mid I_{\eta m} \otimes A}{F_\eta \otimes I_{m\ell} \mid I_{\eta m} \otimes I_\ell} \right].$$
(9.11)

3.
$$A_1 - \lambda B_1 = L_{\eta}^T(\lambda) \otimes I_m$$
 and $A_2 - \lambda B_2 = L_{\varepsilon}(\lambda) \otimes I_n$:

$$\omega_3 := \sigma_{2\eta m \varepsilon n} \left[\frac{E_{\eta} \otimes I_{m \varepsilon n} \mid I_{\eta m} \otimes E_{\varepsilon} \otimes I_n}{F_{\eta} \otimes I_{m \varepsilon n} \mid I_{\eta m} \otimes F_{\varepsilon} \otimes I_n} \right].$$
(9.12)

4.
$$A_1 - \lambda B_1 = L_k(\lambda) \otimes I_r$$
 and $A_2 - \lambda B_2 = L_k(\lambda) \otimes I_r$:

$$\omega_4 := \sigma_{2(k+1)rkr} \left[\begin{array}{c|c} E_k^T \otimes I_{rkr} & I_{(k+1)r} \otimes E_k \otimes I_r \\ \hline F_k^T \otimes I_{rkr} & I_{(k+1)r} \otimes F_k \otimes I_r \end{array} \right].$$
(9.13)

In Lemma 9.3.1 we analyze the first problem and give a lower bound for ω_1 .

Lemma 9.3.1. Let ω_1 be the singular value in (9.10). Then

$$\omega_1 \ge \frac{1}{1 + 2\varepsilon \max(1, \|A\|_2^{\varepsilon})}.$$
(9.14)

Proof. It follows from the properties of singular values of Kronecker products that ω_1 is also equal to

$$\omega_1 = \sigma_{2\ell\varepsilon} \left[\begin{array}{c|c} A^T \otimes I_\varepsilon & I_\ell \otimes E_\varepsilon \\ \hline I_\ell \otimes I_\varepsilon & I_\ell \otimes F_\varepsilon \end{array} \right]$$

and using perfect shuffle permutations we also get

$$\omega_1 = \sigma_{2\varepsilon\ell} \left[\begin{array}{c|c} I_{\varepsilon} \otimes A^T & E_{\varepsilon} \otimes I_{\ell} \\ \hline I_{\varepsilon} \otimes I_{\ell} & F_{\varepsilon} \otimes I_{\ell} \end{array} \right].$$

The smallest singular value $\sigma_{2\varepsilon\ell}$ is larger than the smallest singular value of any $2\varepsilon\ell \times 2\varepsilon\ell$ submatrix. Let us take for this the submatrix obtained by dropping the last block column :

$$M = \begin{bmatrix} I_{\varepsilon} \otimes A^T & I_{\varepsilon} \otimes I_{\ell} \\ \hline I_{\varepsilon} \otimes I_{\ell} & J_{\varepsilon} \otimes I_{\ell} \end{bmatrix}, \quad \text{where} \quad J_{\varepsilon} := \begin{bmatrix} 0 & 1 & & \\ & 0 & \ddots & \\ & & \ddots & 1 \\ & & & 0 \end{bmatrix} \in \mathbb{F}^{\varepsilon \times \varepsilon}.$$

We can factorize this matrix as

$$M = \begin{bmatrix} I_{\varepsilon} \otimes A^T & I_{\varepsilon} \otimes I_{\ell} \\ \hline I_{\varepsilon} \otimes I_{\ell} & 0 \end{bmatrix} \begin{bmatrix} I_{\varepsilon\ell} & 0 \\ \hline 0 & I_{\varepsilon\ell} - J_{\varepsilon} \otimes A^T \end{bmatrix} \begin{bmatrix} I_{\varepsilon} \otimes I_{\ell} & J_{\varepsilon} \otimes I_{\ell} \\ \hline 0 & I_{\varepsilon} \otimes I_{\ell} \end{bmatrix}.$$

Therefore its inverse equals

$$M^{-1} = \begin{bmatrix} I_{\varepsilon} \otimes I_{\ell} & -J_{\varepsilon} \otimes I_{\ell} \\ 0 & I_{\varepsilon} \otimes I_{\ell} \end{bmatrix} \begin{bmatrix} I_{\varepsilon \ell} & 0 \\ 0 & (I_{\varepsilon \ell} - J_{\varepsilon} \otimes A^{T})^{-1} \end{bmatrix} \begin{bmatrix} 0 & I_{\varepsilon} \otimes I_{\ell} \\ I_{\varepsilon} \otimes I_{\ell} & -I_{\varepsilon} \otimes A^{T} \end{bmatrix}$$

$$= \left[\frac{I_{\varepsilon\ell}}{0}\right] \left[\begin{array}{c} 0 \mid I_{\varepsilon\ell} \end{array} \right] + \left[\frac{-J_{\varepsilon} \otimes I_{\ell}}{I_{\varepsilon\ell}}\right] (I_{\varepsilon\ell} - J_{\varepsilon} \otimes A^T)^{-1} \left[\begin{array}{c} I_{\varepsilon\ell} \mid -I_{\varepsilon} \otimes A^T \end{array} \right].$$

It then follows that

$$\|M^{-1}\|_{2} \leq 1 + \sqrt{2}\sqrt{1 + \|A\|_{2}^{2}} \left[1 + \|A\|_{2} + \|A\|_{2}^{2} + \ldots + \|A\|_{2}^{\varepsilon-1}\right],$$

since

$$(I_{\varepsilon\ell} - J_{\varepsilon} \otimes A^T)^{-1} = \sum_{i=0}^{\varepsilon-1} J_{\varepsilon}^i \otimes A^{i^T}$$

In particular, for $||A||_2 \leq 1$ we obtain the bound $||M^{-1}||_2 \leq 1+2\varepsilon$, while for $||A||_2 > 1$ we obtain the bound $||M^{-1}||_2 \leq 1+2\varepsilon ||A||_2^{\varepsilon}$. This finally yields the inequality

$$\omega_1 \ge \frac{1}{1 + 2\varepsilon \max(1, \|A\|_2^{\varepsilon})}.$$

The second generalized Sylvester equation is essentially the transposed of the first equation and the analysis is therefore completely analogous. This immediately yields Lemma 9.3.2.

Lemma 9.3.2. Let ω_2 be the singular value in (9.11). Then

$$\omega_2 \ge \frac{1}{1 + 2\eta \max(1, \|A\|_2^{\eta})}.$$
(9.15)

The third generalized Sylvester equation was analyzed in [26] and its associated smallest singular value is exactly equal to $\omega_3 = 2\sin(\pi/(4\min(\varepsilon,\eta)+2))$ if $\varepsilon \neq \eta$, and to $2\sin(\pi/4\eta)$ if $\varepsilon = \eta$. Notice that we can assume $\min(\varepsilon,\eta) \geq 1$ since otherwise the equation is void. For $\varepsilon \neq \eta$ we then obtain $\omega_3 \geq \frac{3}{2\min(\varepsilon,\eta)+1}$ since $\sin x \geq 3x/\pi$ for $0 \leq x \leq \pi/6$, and for $\varepsilon = \eta$ we then obtain $\omega_3 \geq \frac{\sqrt{2}}{\eta}$ since $\sin x \geq 2\sqrt{2}x/\pi$ for $0 \leq x \leq \pi/4$. We have also that $2\eta = \varepsilon + \eta$ if $\varepsilon = \eta$ and $2\min(\varepsilon,\eta) + 1 \leq \varepsilon + \eta$ if $\varepsilon \neq \eta$, which finally yields the lower bound in Lemma 9.3.3 for ω_3 .

Lemma 9.3.3. Let ω_3 be the singular value in (9.12). Then

$$\omega_3 \ge \frac{2\sqrt{2}}{(\varepsilon + \eta)}.\tag{9.16}$$

In Lemma 9.3.4, we give a lower bound for the smallest singular value ω_4 corresponding to the fourth generalized Sylvester equation.

Lemma 9.3.4. Let ω_4 be the singular value in (9.13). Then

$$\omega_4 \ge \frac{3}{4k-1}.\tag{9.17}$$

Proof. We prove first that $\omega_4 = 2\sin(\pi/(8k-2))$. This is obtained as follows. We can again use the properties of Kronecker products to prove that

$$\omega_4 = \sigma_{2k(k+1)} \left[\begin{array}{c|c} E_k^T \otimes I_k & I_{(k+1)} \otimes E_k \\ \hline F_k^T \otimes I_k & I_{(k+1)} \otimes F_k \end{array} \right]$$

This matrix can be transformed by row and column permutations to the direct sum of smaller matrices :

$$M_1 \oplus M_1 \oplus M_3 \oplus M_3 \oplus \cdots \oplus M_{2k-1} \oplus M_{2k-1} \oplus N_{2k},$$

see Appendix C, where the blocks

$$M_{k} := \begin{bmatrix} 1 & 1 & & \\ & 1 & \ddots & \\ & & \ddots & 1 \\ & & & & 1 \end{bmatrix} \in \mathbb{F}^{k \times k}, \quad N_{k} := \begin{bmatrix} 1 & 1 & & & \\ & 1 & \ddots & & \\ & & \ddots & 1 & \\ & & & & 1 & 1 \end{bmatrix} \in \mathbb{F}^{k \times (k+1)} \quad (9.18)$$

have as smallest singular values $2 \sin \frac{\pi}{4k+2}$ and $2 \sin \frac{\pi}{2k+2}$, respectively (see [26, Proof of Proposition B.4]). The smallest singular value therefore corresponds to M_{2k-1} and equals $\omega_4 = 2 \sin(\pi/(8k-2))$. For $k \ge 1$, we use again that $\sin x \ge 3x/\pi$ for $0 \le x \le \pi/6$, to obtain the bound $\omega_4 \ge \frac{3}{4k-1}$.

9.4 Restoring the structure of the linearization after perturbations

We now consider perturbations of the following block Kronecker linearization introduced in (9.2)

$$S(\lambda) := \begin{bmatrix} S_{11}(\lambda) & S_{12}(\lambda) & S_{13}(\lambda) \\ S_{21}(\lambda) & S_{22}(\lambda) & 0 \\ S_{31}(\lambda) & 0 & 0 \end{bmatrix} := \begin{bmatrix} M(\lambda) & \hat{K}_2^T C & K_2^T(\lambda) \\ B\hat{K}_1 & A - \lambda I_\ell & 0 \\ K_1(\lambda) & 0 & 0 \end{bmatrix}, \quad (9.19)$$

where $S_{13}(\lambda)$ is $(\eta + 1)m \times \eta m$ and has full column rank ηm , $S_{22}(\lambda)$ is $\ell \times \ell$ and is a regular pencil, $S_{31}(\lambda)$ is $\varepsilon n \times (\varepsilon + 1)n$ and has full row rank εn , and where no two of these three pencils have common generalized eigenvalues. As explained in the introduction, if the state-space triple $\{A, B, C\}$ is minimal, then $S(\lambda)$ is a strong linearization of the $m \times n$ rational matrix

$$R(\lambda) = C(\lambda I_{\ell} - A)^{-1}B + (\Lambda_{\eta}(\lambda) \otimes I_m)^T M(\lambda)(\Lambda_{\epsilon}(\lambda) \otimes I_n).$$
(9.20)

Except in Section 9.6, we assume in this section that $\max(\eta, \epsilon) > 0$. This means that the degree $d = \epsilon + \eta + 1$ of the polynomial part $D(\lambda) = (\Lambda_{\eta}(\lambda) \otimes I_m)^T M(\lambda) (\Lambda_{\epsilon}(\lambda) \otimes I_n)$

of $R(\lambda)$ is greater than 1 and that at least one of the blocks $K_1(\lambda)$ or $K_2(\lambda)$ is not an empty matrix. The degenerate case in which $\varepsilon = 0$ and $\eta = 0$ will be studied in Section 9.6.

Since $S(\lambda)$ is a strong linearization of $R(\lambda)$, $S(\lambda)$ has the exact eigenstructure of the finite zeros of $R(\lambda)$, and its infinite zero structure as well as its left and right null-space structure can be correctly retrieved from the pencil via simple constant shifts, as explained in the introduction. In order to compute this eigenstructure, we make use of the staircase algorithm [83], followed by the QZ algorithm [71], on $S(\lambda)$. The backward stability of these two algorithms guarantees in fact that we computed the exact eigenstructure of a slightly perturbed pencil

$$\widehat{S}(\lambda) := S(\lambda) + \Delta_S(\lambda), \quad \Delta_S(\lambda) := \begin{bmatrix} \Delta_{11}(\lambda) & \Delta_{12}(\lambda) & \Delta_{13}(\lambda) \\ \Delta_{21}(\lambda) & \Delta_{22}(\lambda) & \Delta_{23}(\lambda) \\ \Delta_{31}(\lambda) & \Delta_{32}(\lambda) & \Delta_{33}(\lambda) \end{bmatrix}, \quad (9.21)$$

where the pencil $\Delta_S(\lambda)$ has a norm which is much smaller than the norm of $S(\lambda)$. More precisely, $\|\Delta_S(\lambda)\|_F = O(\epsilon_M) \|S(\lambda)\|_F$, where ϵ_M is the machine precision of the computer. But even for very small perturbations, the structure of the pencil $S(\lambda)$ is lost, and therefore also the connection between $\widehat{S}(\lambda)$ and some rational matrix $\widehat{R}(\lambda)$ is lost. In this section, we will show that this structure can be restored, without affecting the computed eigenstructure. For this, one needs only to find a strict equivalence transformation that is close to the identity and restores the structure of $\widehat{S}(\lambda)$ to a new pencil $\widetilde{S}(\lambda)$ that is a block Kronecker linearization, with the same parameters ϵ and η as $S(\lambda)$, of a rational matrix $\widetilde{R}(\lambda)$:

$$\widetilde{S}(\lambda) := (I - X)(S(\lambda) + \Delta_S(\lambda))(I - Y) = \begin{bmatrix} \widetilde{M}(\lambda) & \widehat{K}_2^T \widetilde{C} & K_2^T(\lambda) \\ \widetilde{B} \widehat{K}_1 & \widetilde{A} - \lambda I_\ell & 0 \\ K_1(\lambda) & 0 & 0 \end{bmatrix}.$$
 (9.22)

We will see that if $\|\Delta_S(\lambda)\|_F$ is sufficiently small, then the perturbed system triple $\{\widetilde{A}, \widetilde{B}, \widetilde{C}\}$ is very close to the unperturbed minimal one $\{A, B, C\}$ and, so, $\{\widetilde{A}, \widetilde{B}, \widetilde{C}\}$ is still minimal, since minimality is a generic property equivalent to the controllability matrix having full row rank and the observability matrix having full column rank [55, Chapter 6]. Observe that according to [6], or the discussion in the introduction, $\widetilde{S}(\lambda)$ is a strong linearization of the $m \times n$ rational matrix

$$\widetilde{R}(\lambda) := \widetilde{C}(\lambda I_{\ell} - \widetilde{A})^{-1}\widetilde{B} + (\Lambda_{\eta}(\lambda) \otimes I_m)^T \widetilde{M}(\lambda)(\Lambda_{\epsilon}(\lambda) \otimes I_n)$$

=: $\widetilde{C}(\lambda I_{\ell} - \widetilde{A})^{-1}\widetilde{B} + \widetilde{D}(\lambda).$ (9.23)

Since the eigenstructures of the pencils $\widehat{S}(\lambda)$ and $\widetilde{S}(\lambda)$ are identical, the results in this section prove that the computed finite eigenvalues of $S(\lambda)$ and their partial multiplicities are the exact finite zeros and their partial multiplicities of $\widetilde{R}(\lambda)$, the computed right (resp. left) minimal indices of $S(\lambda)$ minus ϵ (resp. η) are the exact right (resp. left) minimal indices of $\widetilde{R}(\lambda)$, and, if a number ν_r of right minimal indices of $S(\lambda)$ have been computed, then the computed $n - \nu_r$ largest partial multiplicities at infinity of $S(\lambda)$ minus d are the exact structural indices at infinity of $\widetilde{R}(\lambda)$. This is a very strong backward error result for the computation of the eigenstructure of $R(\lambda)$ in the case we are able to prove that $\|\widetilde{A} - A\|_F, \|\widetilde{B} - B\|_F, \|\widetilde{C} - C\|_F$ and $\|\widetilde{D}(\lambda) - D(\lambda)\|_F$ are very small.

The restoration of the structure in $\widehat{S}(\lambda)$ will be done in three steps, each of them involving a strict equivalence transformation close to the identity:

- Step 1: We restore the block anti-triangular structure of the perturbed pencil $\widehat{S}(\lambda)$, i.e., the blocks (2,3), (3,2) and (3,3) are transformed to become 0.
- Step 2: We take care of the anti-diagonal blocks (1,3), (2,2) and (3,1), by restoring their 0 and I block matrices.
- Step 3: We restore the special structure of the blocks (1,2) and (2,1).

At each step k, for k = 1, 2, 3, we obtain a pencil

$$\widehat{S}_k(\lambda) := (I - X_k)\widehat{S}_{k-1}(\lambda)(I - Y_k) := \widehat{S}_{k-1}(\lambda) + \Delta_k(\lambda), \qquad (9.24)$$

where $\widehat{S}_0(\lambda) := \widehat{S}(\lambda)$ and $\Delta_0(\lambda) := \Delta_S(\lambda)$:

$$S(\lambda) \xrightarrow{+\Delta_0(\lambda)} \widehat{S}(\lambda) = \widehat{S}_0(\lambda) \xrightarrow{+\Delta_1(\lambda)} \widehat{S}_1(\lambda) \xrightarrow{+\Delta_2(\lambda)} \widehat{S}_2(\lambda) \xrightarrow{+\Delta_3(\lambda)} \widehat{S}_3(\lambda) = \widetilde{S}(\lambda).$$

We will compute bounds for $||(X_k, Y_k)||_F$ as a function of $||\widehat{S}_{k-1}(\lambda)||_F$, where the Frobenius norms are computed as defined in the introduction. Moreover, we define the cumulative errors:

$$\Delta_k^{old}(\lambda) := \sum_{i=0}^{k-1} \Delta_i(\lambda), \text{ and}$$

$$\Delta_k^{new}(\lambda) := \Delta_k^{old}(\lambda) + \Delta_k(\lambda) = \sum_{i=0}^k \Delta_i(\lambda),$$
(9.25)

and we will also compute bounds for the Frobenius norm of these error pencils. In our analysis, we will assume that $\delta := \frac{\|\Delta_S(\lambda)\|_F}{\|S(\lambda)\|_F}$ is very small, since in practice is of the order of the machine precision ϵ_M , and we will neglect, when appropriate, terms of order larger than 1 in δ to simplify our bounds. Moreover, we will assume that δ is sufficiently small for guaranteeing that all the steps in the analysis can be performed, for instance, for guaranteeing that some perturbed matrices are invertible, etc. In particular, we have Lemma 9.4.1 for computing bounds of the growth of the cumulative errors $\Delta_k^{new}(\lambda)$. **Lemma 9.4.1.** At each step k of our method, the perturbation $\Delta_k^{new}(\lambda)$ can be bounded by

$$\|\Delta_{k}^{new}(\lambda)\|_{F} \leq \sqrt{2} \|\widehat{S}_{k-1}(\lambda)\|_{2} \|(X_{k}, Y_{k})\|_{F} + \|\Delta_{k}^{old}(\lambda)\|_{F} + \mathcal{O}(\delta^{2}),$$

assuming that $||(X_k, Y_k)||_F$ is of the order of $||\Delta_S(\lambda)||_F$.

Proof. At step k, we have $\widehat{S}_k(\lambda) = (I - X_k)\widehat{S}_{k-1}(\lambda)(I - Y_k)$. Therefore

$$\Delta_k^{new}(\lambda) = \Delta_k^{old}(\lambda) - X_k \widehat{S}_{k-1}(\lambda) - \widehat{S}_{k-1}(\lambda)Y_k + X_k \widehat{S}_{k-1}(\lambda)Y_k.$$

It then follows that the increment (up to $\mathcal{O}(\delta^2)$ terms) is given by

$$-X_k S_a - S_a Y_k + \lambda (X_k S_b + S_b Y_k) + \mathcal{O}(\delta^2),$$

where $S_a - \lambda S_b := \widehat{S}_{k-1}(\lambda)$. We then use the inequalities

$$||X_kS_a + S_aY_k||_F^2 \le 2||S_a||_2^2||(X_k, Y_k)||_F^2, \quad ||X_kS_b + S_bY_k||_F^2 \le 2||S_b||_2^2||(X_k, Y_k)||_F^2$$

and the definition for $\|\widehat{S}_{k-1}(\lambda)\|_2$, to finally get the required bound.

9.4.1 Step 1: Restoring the block anti-triangular structure

For step 1, that is, restoring the block anti-triangular structure of $S(\lambda)$ in the perturbed matrix pencil (9.21), we apply a strict equivalence transformation of the type :

$$\begin{bmatrix} I_{(\eta+1)m} & 0 & 0\\ -X_{21} & I_{\ell} & 0\\ -X_{31} & -X_{32} & I_{\varepsilon n} \end{bmatrix} \widehat{S}(\lambda) \begin{bmatrix} I_{(\varepsilon+1)n} & -Y_{12} & -Y_{13}\\ 0 & I_{\ell} & -Y_{23}\\ 0 & 0 & I_{\eta m} \end{bmatrix}$$
(9.26)

in order to eliminate the perturbations $\Delta_{23}(\lambda)$, $\Delta_{32}(\lambda)$ and $\Delta_{33}(\lambda)$ of the error matrix pencil $\Delta_0(\lambda)$. The notation $\widehat{S}_{ij}^a - \lambda \widehat{S}_{ij}^b := \widehat{S}_{ij} := \widehat{S}_{ij}(\lambda)$ will be used in this section to refer to sub-blocks of $\widehat{S}_0(\lambda)$. Let us write down the equations that we get by setting the blocks (2,3), (3,2) and (3,3) of the matrix in (9.26) equal to zero :

$$\begin{split} \Delta_{23}(\lambda) &:= \Delta_{23}^{a} - \lambda \Delta_{23}^{b} = X_{21} \widehat{S}_{13} + \widehat{S}_{21} Y_{13} + \widehat{S}_{22} Y_{23} - X_{21} \widehat{S}_{11} Y_{13} - X_{21} \widehat{S}_{12} Y_{23}, \\ \Delta_{32}(\lambda) &:= \Delta_{32}^{a} - \lambda \Delta_{32}^{b} = \widehat{S}_{31} Y_{12} + X_{31} \widehat{S}_{12} + X_{32} \widehat{S}_{22} - X_{31} \widehat{S}_{11} Y_{12} - X_{32} \widehat{S}_{21} Y_{12}, \\ \Delta_{33}(\lambda) &:= \Delta_{33}^{a} - \lambda \Delta_{33}^{b} = X_{31} \widehat{S}_{13} + \widehat{S}_{31} Y_{13} + X_{32} \Delta_{23} + \Delta_{32} Y_{23} \\ - X_{31} \widehat{S}_{11} Y_{13} - X_{32} \widehat{S}_{21} Y_{13} - X_{31} \widehat{S}_{12} Y_{23} - X_{32} \widehat{S}_{22} Y_{23}. \end{split}$$

$$(9.27)$$

This is a system of nonlinear matrix equations for the six matrix unknowns X_{21} , X_{31} , X_{32} , Y_{12} , Y_{13} and Y_{23} . We will show that it is consistent and that it has a solution for which the norms of the unknowns are of the order of $\|\Delta_0(\lambda)\|_F$, which

implies that there are many terms in the above three equations that are of second order.

Using Kronecker product and the vec(\cdot) notation, the system of matrix equations (9.27) can be rewritten as :

$$\underbrace{\begin{bmatrix} \operatorname{vec}(\Delta_{23}^{a}) \\ \operatorname{vec}(\Delta_{23}^{b}) \\ \operatorname{vec}(\Delta_{32}^{a}) \\ \operatorname{vec}(\Delta_{32}^{b}) \\ \operatorname{vec}(\Delta_{33}^{a}) \\ \operatorname{vec}(\Delta_{33}^{b}) \\ \operatorname{vec}(\Delta_{33}^{b}) \\ \end{array}}_{:=c} = (T + \Delta T) \underbrace{\begin{bmatrix} \operatorname{vec}(X_{21}) \\ \operatorname{vec}(Y_{23}) \\ \operatorname{vec}(X_{32}) \\ \operatorname{vec}(X_{32}) \\ \operatorname{vec}(Y_{12}) \\ \operatorname{vec}(X_{31}) \\ \operatorname{vec}(Y_{13}) \\ \end{array}}_{:=x} - \underbrace{\begin{bmatrix} \operatorname{vec}(Z_{1}) \\ \operatorname{vec}(Z_{2}) \\ \operatorname{vec}(Z_{3}) \\ \operatorname{vec}(Z_{3}) \\ \operatorname{vec}(Z_{4}) \\ \operatorname{vec}(Z_{5}) \\ \operatorname{vec}(Z_{6}) \\ \end{array}}_{:=z}, \tag{9.28}$$

where

$$Z_{1} := X_{21}\widehat{S}_{11}^{a}Y_{13} + X_{21}\widehat{S}_{12}^{a}Y_{23}, \quad Z_{2} := X_{21}\widehat{S}_{11}^{b}Y_{13} + X_{21}\widehat{S}_{12}^{b}Y_{23},$$

$$Z_{3} := X_{31}\widehat{S}_{11}^{a}Y_{12} + X_{32}\widehat{S}_{21}^{a}Y_{12}, \quad Z_{4} := X_{31}\widehat{S}_{11}^{b}Y_{12} + X_{32}\widehat{S}_{21}^{b}Y_{12},$$

$$Z_{5} := X_{31}\widehat{S}_{11}^{a}Y_{13} + X_{32}\widehat{S}_{21}^{a}Y_{13} + X_{31}\widehat{S}_{12}^{a}Y_{23} + X_{32}\widehat{S}_{22}^{a}Y_{23},$$

$$Z_{6} := X_{31}\widehat{S}_{11}^{b}Y_{13} + X_{32}\widehat{S}_{21}^{b}Y_{13} + X_{31}\widehat{S}_{12}^{b}Y_{23} + X_{32}\widehat{S}_{22}^{b}Y_{23},$$

$$\Delta T = \begin{bmatrix} \Delta_{13}^{a} \stackrel{T}{\longrightarrow} SI_{\ell} & I_{\eta m} \otimes \Delta_{22}^{a} & 0 & 0 & 0 & I_{\eta m} \otimes \Delta_{21}^{a} \\ \Delta_{13}^{b} \stackrel{T}{\longrightarrow} I_{\ell} & I_{\eta m} \otimes \Delta_{22}^{b} & 0 & 0 & 0 & I_{\eta m} \otimes \Delta_{21}^{b} \\ 0 & 0 & \Delta_{22}^{a} \stackrel{T}{\longrightarrow} I_{\varepsilon n} & I_{\ell} \otimes \Delta_{31}^{a} & \Delta_{12}^{a} \stackrel{T}{\longrightarrow} I_{\varepsilon n} & 0 \\ 0 & 0 & \Delta_{22}^{b} \stackrel{T}{\longrightarrow} SI_{\varepsilon n} & I_{\ell} \otimes \Delta_{31}^{b} & \Delta_{12}^{b} \stackrel{T}{\longrightarrow} I_{\varepsilon n} & 0 \\ 0 & I_{\eta m} \otimes \Delta_{32}^{a} & \Delta_{23}^{a} \stackrel{T}{\longrightarrow} I_{\varepsilon n} & 0 & \Delta_{13}^{a} \stackrel{T}{\longrightarrow} I_{\varepsilon n} & I_{\eta m} \otimes \Delta_{31}^{a} \\ 0 & I_{\eta m} \otimes \Delta_{32}^{b} & \Delta_{23}^{b} \stackrel{T}{\longrightarrow} I_{\varepsilon n} & 0 & \Delta_{13}^{b} \stackrel{T}{\longrightarrow} I_{\varepsilon n} & I_{\eta m} \otimes \Delta_{31}^{a} \end{bmatrix},$$

and

$$T = \begin{bmatrix} T_{11} & T_{12} & 0 & 0 & 0 & T_{16} \\ T_{21} & T_{22} & 0 & 0 & 0 & 0 \\ & & T_{33} & T_{34} & T_{35} & 0 \\ & & & T_{43} & T_{44} & 0 & 0 \\ & & & & & T_{55} & T_{56} \\ & & & & & T_{65} & T_{66} \end{bmatrix},$$

with

$$\begin{bmatrix} T_{11} & T_{12} \\ T_{21} & T_{22} \end{bmatrix} := \begin{bmatrix} E_{\eta} \otimes I_{m\ell} & I_{\eta m} \otimes A \\ F_{\eta} \otimes I_{m\ell} & I_{\eta m} \otimes I_{\ell} \end{bmatrix}, \begin{bmatrix} T_{33} & T_{34} \\ T_{43} & T_{44} \end{bmatrix} := \begin{bmatrix} A^T \otimes I_{\varepsilon n} & I_{\ell} \otimes E_{\varepsilon} \otimes I_n \\ I_{\ell} \otimes I_{\varepsilon n} & I_{\ell} \otimes F_{\varepsilon} \otimes I_n \end{bmatrix}, \\ \begin{bmatrix} T_{55} & T_{56} \\ T_{65} & T_{66} \end{bmatrix} := \begin{bmatrix} E_{\eta} \otimes I_{m\varepsilon n} & I_{\eta m} \otimes E_{\varepsilon} \otimes I_n \\ F_{\eta} \otimes I_{m\varepsilon n} & I_{\eta m} \otimes F_{\varepsilon} \otimes I_n \end{bmatrix}, \quad \begin{cases} T_{16} := I_{\eta m} \otimes \mathbf{e}_{\varepsilon+1}^T \otimes B \\ T_{35} := \mathbf{e}_{\eta+1}^T \otimes C^T \otimes I_{\varepsilon n} \end{bmatrix}.$$

We emphasize that the matrices in the two lines above are precisely those appearing in equations (9.11), (9.10) and (9.12), respectively.

The smallest singular value of T and the 2-norm of ΔT will be needed in the analysis of the bound for the structured backward errors. More precisely for proving that (9.27) is consistent and bounding the norm of one of its solutions. A lower bound for $\sigma_{\min}(T)$ and an upper bound for $\|\Delta T\|_2$ are given in Lemma 9.4.2 and Lemma 9.4.3, respectively.

Lemma 9.4.2. Let T be the matrix in (9.28). Let $\alpha := 1 + 2\varepsilon \max(1, \|A\|_2^{\varepsilon}), \beta := 1 + 2\eta \max(1, \|A\|_2^{\eta}), \gamma := \frac{\varepsilon + \eta}{2\sqrt{2}}$ and $s := \max(\alpha, \beta, \gamma) + \gamma(\beta \|B\|_2 + \alpha \|C\|_2)$ then

$$\sigma_{\min}(T) \ge \frac{1}{s}.$$

Proof. If we partition the matrix T as a block triangular matrix

$$T = \begin{bmatrix} T_1 & 0 & T_B \\ & T_2 & T_C \\ & & T_3 \end{bmatrix},$$

then the diagonal blocks have full row ranks because their smallest singular values are strictly larger than zero according to Lemmas 9.3.2, 9.3.1 and 9.3.3, respectively. Therefore, they are right invertible, with Moore–Penrose pseudoinverses T_i^r satisfying $T_iT_i^r = I$, for i = 1, 2, 3. Moreover, $||T_1^r||_2 = \omega_2^{-1}$, $||T_2^r||_2 = \omega_1^{-1}$ and $||T_3^r||_2 = \omega_3^{-1}$, with ω_1, ω_2 and ω_3 as in (9.10), (9.11) and (9.12). A right inverse T^r for T is given by

$$T^{r} = \begin{bmatrix} T_{1}^{r} & 0 & -T_{1}^{r}T_{B}T_{3}^{r} \\ T_{2}^{r} & -T_{2}^{r}T_{C}T_{3}^{r} \\ T_{3}^{r} \end{bmatrix}$$

since $TT^r = I$. It then follows that the smallest singular value of T is lower bounded by $||T^r||_2^{-1}$. This right inverse can be written as the sum of three matrices (one of them being diag (T_1^r, T_2^r, T_3^r)), and the 2-norm of each of them can be upper bounded using the results of Section 9.3 and the fact that $||T_B||_2 = ||B||_2$ and $||T_C||_2 = ||C||_2$. We then obtain the bound :

$$\sigma_{\min}(T) \ge 1/\left[\max(\omega_1^{-1}, \omega_2^{-1}, \omega_3^{-1}) + \omega_3^{-1}(\omega_2^{-1} \|B\|_2 + \omega_1^{-1} \|C\|_2)\right] \\\ge 1/\left[\max(\alpha, \beta, \gamma) + \gamma(\beta \|B\|_2 + \alpha \|C\|_2)\right],$$

by taking into account inequalities (9.14), (9.15), (9.16).

Lemma 9.4.3. Let ΔT be the matrix in (9.28) and let $\Delta_S(\lambda)$ be the pencil in (9.21). Then

$$\|\Delta T\|_2 \le \sqrt{3} \|\Delta_S(\lambda)\|_2.$$

Proof. We consider a permutation matrix P such that

$$\Delta T = \begin{bmatrix} \Delta_{13}^{a} \stackrel{I}{\otimes} I_{\ell} & 0 & 0 & I_{\eta m} \otimes \Delta_{21}^{a} & I_{\eta m} \otimes \Delta_{21}^{a} \\ \Delta_{13}^{b} \stackrel{T}{\otimes} I_{\ell} & 0 & 0 & I_{\eta m} \otimes \Delta_{22}^{b} & I_{\eta m} \otimes \Delta_{21}^{b} \\ 0 & I_{\ell} \otimes \Delta_{31}^{a} & \Delta_{22}^{a} \stackrel{T}{\otimes} I_{\varepsilon n} & \Delta_{12}^{a} \stackrel{T}{\otimes} I_{\varepsilon n} & 0 & 0 \\ 0 & I_{\ell} \otimes \Delta_{31}^{b} & \Delta_{22}^{b} \stackrel{T}{\otimes} I_{\varepsilon n} & \Delta_{12}^{b} \stackrel{T}{\otimes} I_{\varepsilon n} & 0 & 0 \\ 0 & 0 & \Delta_{23}^{a} \stackrel{T}{\otimes} I_{\varepsilon n} & \Delta_{13}^{a} \stackrel{T}{\otimes} I_{\varepsilon n} & I_{\eta m} \otimes \Delta_{32}^{a} & I_{\eta m} \otimes \Delta_{31}^{a} \\ 0 & 0 & \Delta_{23}^{b} \stackrel{T}{\otimes} I_{\varepsilon n} & \Delta_{13}^{a} \stackrel{T}{\otimes} I_{\varepsilon n} & I_{\eta m} \otimes \Delta_{32}^{a} & I_{\eta m} \otimes \Delta_{31}^{a} \\ 0 & 0 & \Delta_{23}^{b} \stackrel{T}{\otimes} I_{\varepsilon n} & \Delta_{13}^{b} \stackrel{T}{\otimes} I_{\varepsilon n} & I_{\eta m} \otimes \Delta_{32}^{b} & I_{\eta m} \otimes \Delta_{31}^{a} \\ \vdots = [T_{1}|T_{2}|T_{3}] P. \end{bmatrix} P$$

Using properties of norms and Kronecker products (see [54, Chapter 4]) we have that $||T_i||_2 \leq ||\Delta_S(\lambda)||_2$ for i = 1, 2, 3. Finally, by [52, Lemma 3.5],

$$|\Delta T||_{2} \le \sqrt{3} \max\{||T_{1}||_{2}, ||T_{2}||_{2}, ||T_{3}||_{2}\} \le \sqrt{3} ||\Delta_{S}(\lambda)||_{2}.$$

In order to prove that the system of nonlinear matrix equations (9.27) is consistent, first, we remove quadratic terms in X_{ij} and Y_{ij} of these equations and we get the following system of linear equations :

$$\begin{split} \Delta_{23}(\lambda) &= X_{21}\widehat{S}_{13} + \widehat{S}_{21}Y_{13} + \widehat{S}_{22}Y_{23}, \\ \Delta_{32}(\lambda) &= \widehat{S}_{31}Y_{12} + X_{31}\widehat{S}_{12} + X_{32}\widehat{S}_{22}, \\ \Delta_{33}(\lambda) &= X_{31}\widehat{S}_{13} + \widehat{S}_{31}Y_{13} + X_{32}\Delta_{23} + \Delta_{32}Y_{23}. \end{split}$$

This linear system of matrix equations can be rewritten as the underdetermined linear system :

$$(T + \Delta T)x = c, \tag{9.29}$$

with the same notation as in (9.28). Next we prove that (9.29) is consistent for any right hand side if ΔT is sufficiently small. From the minimum norm solution of (9.29), we obtain in Theorem 9.4.6 that there exists a solution for the quadratic system (9.28) under certain conditions and bound its norm.

Lemma 9.4.4. Let $(T + \Delta T)x = c$ be the underdetermined linear system in (9.29), and let us assume that $\sigma_{\min}(T) > \|\Delta T\|_2$. Then $(T + \Delta T)x = c$ is consistent and its minimum norm solution $(X^0, Y^0) := (X^0_{21}, X^0_{31}, X^0_{32}, Y^0_{12}, Y^0_{13}, Y^0_{23})$ satisfies

$$\|(X^0, Y^0)\|_F \le \frac{1}{\sigma} \|(\Delta_{23}(\lambda), \Delta_{32}(\lambda), \Delta_{33}(\lambda))\|_F,$$

where $\sigma := \sigma_{\min}(T) - \|\Delta T\|_2$.

Proof. Analogous proof as for [26, Lemma 5.6].

189

The notation $\sigma := \sigma_{\min}(T) - \|\Delta T\|_2$ has been chosen to remind that σ is a lower bound for the smallest singular value of $T + \Delta T$, since $\sigma_{\min}(T + \Delta T) \geq \sigma_{\min}(T) - \|\Delta T\|_2$ by Weyl's perturbation theorem for singular values [54, Theorem 3.3.16]. Lemma 9.4.5 gives a sufficient condition on $\|\Delta_S(\lambda)\|_2$ that guarantees $\sigma > 0$ and, hence, that allows us to apply Lemma 9.4.4.

Lemma 9.4.5. Consider the real number *s* defined as in Lemma 9.4.2. Let *T* and ΔT be the matrices in (9.29), and let $\Delta_S(\lambda)$ be the pencil in (9.21). If $\|\Delta_S(\lambda)\|_2 < \frac{1}{2s}$ then

$$\sigma = \sigma_{\min}(T) - \|\Delta T\|_2 > \frac{2 - \sqrt{3}}{2s} > 0.$$

Proof. If $\|\Delta_S(\lambda)\|_2 < \frac{1}{2s}$ we have, by Lemmas 9.4.2 and 9.4.3, that $\sigma_{\min}(T) - \|\Delta T\|_2 \geq \frac{1}{s} - \sqrt{3} \|\Delta_S(\lambda)\|_2 > \frac{2-\sqrt{3}}{2s} > 0.$

Theorem 9.4.6 establishes conditions in order the system of matrix equations (9.27) to have a solution as we announced. Moreover, it gives an upper bound for the Frobenius norm of this solution. We remark that Theorem 9.4.6 is similar to [26, Theorem 5.8], though the involved systems of matrix equations are very different from each other. Therefore, some details in the proof of Theorem 9.4.6 are omitted since can be found in [26].

Theorem 9.4.6. There exists a solution $(X, Y) := (X_{21}, X_{31}, X_{32}, Y_{12}, Y_{13}, Y_{23})$ of the quadratic system of equations (9.28) satisfying

$$\|(X,Y)\|_F \le 2\frac{\theta}{\sigma},$$

whenever

$$\sigma > 0 \quad and \quad \frac{\theta\omega}{\sigma^2} < \frac{1}{4},$$
(9.30)

where $\omega := \|(M(\lambda), A - \lambda I_{\ell}, B, C)\|_F + \|\Delta_S(\lambda)\|_F$, $\theta := \|(\Delta_{23}(\lambda), \Delta_{32}(\lambda), \Delta_{33}(\lambda))\|_F$, and $\sigma = \sigma_{\min}(T) - \|\Delta T\|_2$.

Proof. Since $\sigma > 0$, we can apply Lemma 9.4.4 and consider (X^0, Y^0) the minimum norm solution of (9.29). Let

$$x_0 := \left[\operatorname{vec}(X_{21}^0)^T \operatorname{vec}(Y_{23}^0)^T \operatorname{vec}(X_{32}^0)^T \operatorname{vec}(Y_{12}^0)^T \operatorname{vec}(X_{31}^0)^T \operatorname{vec}(Y_{13}^0)^T \right]^T$$

Let us define the sequence $\{(X^i, Y^i) := (X_{21}^i, X_{31}^i, X_{32}^i, Y_{12}^i, Y_{13}^i, Y_{23}^i)\}_{i=0}^{\infty}$ such that, for each i > 0, (X^i, Y^i) is the minimum norm solution of the linear system

$$(T + \Delta T) \begin{bmatrix} \operatorname{vec}(X_{21}^{i}) \\ \operatorname{vec}(Y_{23}^{i}) \\ \operatorname{vec}(X_{32}^{i}) \\ \operatorname{vec}(Y_{12}^{i}) \\ \operatorname{vec}(X_{31}^{i}) \\ \operatorname{vec}(Y_{13}^{i}) \end{bmatrix} = c + \begin{bmatrix} \operatorname{vec}(Z_{1}^{i-1}) \\ \operatorname{vec}(Z_{2}^{i-1}) \\ \operatorname{vec}(Z_{3}^{i-1}) \\ \operatorname{vec}(Z_{4}^{i-1}) \\ \operatorname{vec}(Z_{5}^{i-1}) \\ \operatorname{vec}(Z_{6}^{i-1}) \end{bmatrix},$$
(9.31)

where

$$\begin{split} Z_1^{i-1} &:= X_{21}^{i-1} \widehat{S}_{11}^a Y_{13}^{i-1} + X_{21}^{i-1} \widehat{S}_{12}^a Y_{23}^{i-1}, \quad Z_2^{i-1} := X_{21}^{i-1} \widehat{S}_{11}^b Y_{13}^{i-1} + X_{21}^{i-1} \widehat{S}_{12}^b Y_{23}^{i-1}, \\ Z_3^{i-1} &:= X_{31}^{i-1} \widehat{S}_{11}^a Y_{12}^{i-1} + X_{32}^{i-1} \widehat{S}_{21}^a Y_{12}^{i-1}, \quad Z_4^{i-1} := X_{31}^{i-1} \widehat{S}_{11}^b Y_{12}^{i-1} + X_{32}^{i-1} \widehat{S}_{21}^b Y_{12}^{i-1}, \\ Z_5^{i-1} &:= X_{31}^{i-1} \widehat{S}_{11}^a Y_{13}^{i-1} + X_{32}^{i-1} \widehat{S}_{21}^a Y_{13}^{i-1} + X_{31}^{i-1} \widehat{S}_{12}^a Y_{23}^{i-1} + X_{32}^{i-1} \widehat{S}_{22}^a Y_{23}^{i-1}, \quad \text{and} \\ Z_6^{i-1} &:= X_{31}^{i-1} \widehat{S}_{11}^b Y_{13}^{i-1} + X_{32}^{i-1} \widehat{S}_{21}^b Y_{13}^{i-1} + X_{31}^{i-1} \widehat{S}_{12}^b Y_{23}^{i-1} + X_{32}^{i-1} \widehat{S}_{22}^b Y_{23}^{i-1}. \end{split}$$

Note that the minimum norm solution of (9.31) is obtained by multiplying the right hand side of (9.31) by the Moore-Penrose pseudoinverse of $T + \Delta T$, denoted by $(T + \Delta T)^{\dagger}$, and that $x_0 = (T + \Delta T)^{\dagger}c$.

Now we assume that $\frac{\theta_{\omega}}{\sigma^2} < \frac{1}{4}$ holds. Then we can prove that the sequence $\{(X^i, Y^i)\}_{i=0}^{\infty}$ converges to a solution (X, Y) of the quadratic system of equations (9.28) analogously as it is done in [26, Theorem 5.8]. For that, we have to take into account that, if $\|(X^{i-1}, Y^{i-1})\|_F \leq \rho_{i-1}$, then

$$\begin{aligned} \|(X^{i},Y^{i})\|_{F} \\ &\leq \|(X^{0},Y^{0})\|_{F} + \|(T+\Delta T)^{\dagger}\|_{2} \left\| \begin{bmatrix} X_{21}^{i-1} & 0\\ X_{31}^{i-1} & X_{32}^{i-1} \end{bmatrix} \begin{bmatrix} \widehat{S}_{11} & \widehat{S}_{12}\\ \widehat{S}_{21} & \widehat{S}_{22} \end{bmatrix} \begin{bmatrix} Y_{12}^{i-1} & Y_{13}^{i-1}\\ 0 & Y_{23}^{i-1} \end{bmatrix} \right\|_{F} \\ &\leq \rho_{0} + \sigma^{-1}\rho_{i-1}^{2}\omega := \rho_{i}, \end{aligned}$$

where $||(X^0, Y^0)||_F \leq \theta \sigma^{-1} := \rho_0$. Therefore, we can define the same fixed point iteration as in the proof of [26, Theorem 5.8] and we obtain that the sequence is bounded, i.e., $||(X^i, Y^i)||_F \leq \rho$, with $\rho < 2\sigma^{-1}\theta$, for all $i \geq 0$. In addition, if we define the sequence $\{C_i := (X^{i+1}, Y^{i+1}) - (X^i, Y^i)\}_{i=0}^{\infty}$ then

$$\begin{split} \|C_i\|_F \leq \|(T+\Delta T)^{\dagger}\|_2 \left(\left\| \begin{bmatrix} X_{21}^i & 0\\ X_{31}^i & X_{32}^i \end{bmatrix} \begin{bmatrix} \widehat{S}_{11} & \widehat{S}_{12}\\ \widehat{S}_{21} & \widehat{S}_{22} \end{bmatrix} \begin{bmatrix} Y_{12}^i & Y_{13}^i\\ 0 & Y_{23}^i \end{bmatrix} \right. \\ \left. - \begin{bmatrix} X_{21}^{i-1} & 0\\ X_{31}^{i-1} & X_{32}^{i-1} \end{bmatrix} \begin{bmatrix} \widehat{S}_{11} & \widehat{S}_{12}\\ \widehat{S}_{21} & \widehat{S}_{22} \end{bmatrix} \begin{bmatrix} Y_{12}^{i-1} & Y_{13}^{i-1}\\ 0 & Y_{23}^{i-1} \end{bmatrix} \right\|_F \right) \\ \leq \|(T+\Delta T)^{\dagger}\|_2 \left(\left\| \begin{bmatrix} X_{21}^i - X_{21}^{i-1} & 0\\ X_{31}^i - X_{31}^{i-1} & X_{32}^i - X_{32}^{i-1} \end{bmatrix} \begin{bmatrix} \widehat{S}_{11} & \widehat{S}_{12}\\ \widehat{S}_{21} & \widehat{S}_{22} \end{bmatrix} \begin{bmatrix} Y_{12}^i & Y_{13}^i\\ 0 & Y_{23}^{i-1} \end{bmatrix} \right\|_F \right) \\ + \left\| \begin{bmatrix} X_{21}^{i-1} & 0\\ X_{31}^{i-1} & X_{32}^{i-1} \end{bmatrix} \begin{bmatrix} \widehat{S}_{11} & \widehat{S}_{12}\\ \widehat{S}_{21} & \widehat{S}_{22} \end{bmatrix} \begin{bmatrix} Y_{12}^i - Y_{12}^{i-1} & Y_{13}^i - Y_{13}^{i-1}\\ 0 & Y_{23}^i - Y_{23}^{i-1} \end{bmatrix} \right\|_F \right) \\ \leq 2\sigma^{-1}\rho\omega\|C_{i-1}\|_F. \end{split}$$

The above inequality implies that $\{(X^i, Y^i)\}_{i=0}^{\infty}$ is a Cauchy sequence, since $2\sigma^{-1}\rho\omega < 1$. Thus, taking limits in both sides of (9.31), we see that $\{(X^i, Y^i)\}_{i=0}^{\infty}$ converges to a solution (X, Y) of the system of equations in (9.28) with $||(X, Y)||_F \leq \rho$.

Theorem 9.4.6, together with Lemma 9.4.5, allow us to prove in Theorem 9.4.7 that there exists a solution (X, Y) of (9.27) which is of the order of the perturbation $\Delta_S(\lambda)$ whenever $\|\Delta_S(\lambda)\|_F$ is properly upper bounded.

Theorem 9.4.7. Consider the real number s defined as in Lemma 9.4.2. Let $S(\lambda)$ be a block Kronecker linearization as in (9.19), and let $\Delta_S(\lambda)$ be a perturbation of $S(\lambda)$ as in (9.21) such that

$$\|\Delta_S(\lambda)\|_F < \left(\frac{2-\sqrt{3}}{4s}\right)^2 \frac{1}{1+\|(M(\lambda), A-\lambda I_\ell, B, C)\|_F}.$$
(9.32)

Then there exists a solution $(X, Y) := (X_{21}, X_{31}, X_{32}, Y_{12}, Y_{13}, Y_{23})$ of the quadratic system of matrix equations in (9.27) that satisfies

$$\|(X,Y)\|_{F} \le \frac{4s \|\Delta_{S}(\lambda)\|_{F}}{2 - \sqrt{3}}.$$
(9.33)

Proof. We have

$$\|\Delta_S(\lambda)\|_F < \left(\frac{2-\sqrt{3}}{4s}\right)^2 \frac{1}{1+\|(M(\lambda), A-\lambda I_\ell, B, C)\|_F} \le \frac{1}{2s}$$

since $s \ge 1$. Then, by Lemma 9.4.5, $\sigma = \sigma_{\min}(T) - \|\Delta T\|_2 > \frac{2-\sqrt{3}}{2s} > 0$. In addition, using the same notation as in Theorem 9.4.6,

$$\frac{\theta\omega}{\sigma^2} \leq \frac{\|\Delta_S(\lambda)\|_F(\|(M(\lambda), A - \lambda I_\ell, B, C)\|_F + \|\Delta_S(\lambda)\|_F)}{\left(\frac{2-\sqrt{3}}{2s}\right)^2} < \frac{1}{4},$$

by (9.32). Therefore, conditions in (9.30) hold and, by Theorem 9.4.6, there exists a solution (X, Y) of the system in (9.27) satisfying

$$\|(X,Y)\|_F \le 2\frac{\theta}{\sigma} \le \frac{4s\|\Delta_S(\lambda)\|_F}{2-\sqrt{3}}.$$

After restoring the block anti-triangular structure of $S(\lambda)$, we get the perturbation error $\Delta_1^{new}(\lambda)$ defined in (9.25). The following first order bound for the norm of $\Delta_1^{new}(\lambda)$ in Corollary 9.4.8 follows from Lemma 9.4.1 and Theorem 9.4.7.

Corollary 9.4.8. Let us define the scalar $f_1 := \frac{4\sqrt{2}s}{2-\sqrt{3}}$. Then

$$\begin{aligned} \|\Delta_{1}^{new}(\lambda)\|_{F} &\leq [1 + f_{1} \|\widehat{S}_{0}(\lambda)\|_{2}] \|\Delta_{S}(\lambda)\|_{F} + \mathcal{O}(\delta^{2}) \\ &\leq [1 + f_{1} \|S(\lambda)\|_{2}] \|\Delta_{S}(\lambda)\|_{F} + \mathcal{O}(\delta^{2}). \end{aligned}$$

9.4.2 Step 2: Restoring the Kronecker blocks $K_1(\lambda)$, $K_2(\lambda)$ and the identity I_{ℓ}

At this stage we have obtained a pencil $\widehat{S}_1(\lambda) = S(\lambda) + \Delta_1^{new}(\lambda)$ of the type

$$\widehat{S}_{1}(\lambda) := \begin{bmatrix} \widehat{M}(\lambda) & \widehat{C}(\lambda) & \widehat{K}_{2}^{T}(\lambda) \\ \widehat{B}(\lambda) & \widehat{A} - \lambda \widehat{I}_{\ell} & 0 \\ \widehat{K}_{1}(\lambda) & 0 & 0 \end{bmatrix}, \qquad (9.34)$$

where the zero blocks below the anti-diagonal are exact and $\widehat{S}_1(\lambda)$ is strictly equivalent to $\widehat{S}(\lambda)$. In this subsection, we will use $\Delta_{ij}^a - \lambda \Delta_{ij}^b$ to denote the corresponding blocks of the updated perturbation matrix $\Delta_1^{new}(\lambda)$. We assume that the norm of the perturbation $\Delta_1^{new}(\lambda)$ is small enough for $\widehat{K}_1(\lambda)$ and $\widehat{K}_2(\lambda)$ to be also minimal bases with row degrees all equal to 1 and the row degrees of their dual minimal bases all equal to ϵ and η , respectively [26, Corollary 5.15]. Thus, $\widehat{K}_1(\lambda)$ and $\widehat{K}_2(\lambda)$ have the same Kronecker canonical forms as $K_1(\lambda)$ and $K_2(\lambda)$, respectively, and are strictly equivalent to them. We will then perform step 2, that is, an updating block-diagonal strict equivalent transformation of the type

$$\begin{bmatrix} I_{(\eta+1)m} - X_{11} & 0 & 0\\ 0 & I_{\ell} - X_{22} & 0\\ 0 & 0 & I_{\varepsilon n} - X_{33} \end{bmatrix} \widehat{S}_{1}(\lambda) \begin{bmatrix} I_{(\epsilon+1)n} - Y_{11} & 0 & 0\\ 0 & I_{\ell} - Y_{22} & 0\\ 0 & 0 & I_{\eta m} - Y_{33} \end{bmatrix}$$
(9.35)

such that

$$(I - X_{33})\widehat{K}_1(\lambda)(I - Y_{11}) = K_1(\lambda), \quad (I - X_{11})\widehat{K}_2^T(\lambda)(I - Y_{33}) = K_2^T(\lambda),$$

and

$$(I - X_{22})\widehat{I}_{\ell}(I - Y_{22}) = I_{\ell}.$$

In the last three equations the sizes of some identity matrices are not specified for simplicity. Clearly, these three problems are independent from each other and can be treated separately.

Let us first look at the equation restoring $K_1(\lambda)$. As pointed out in Section 9.2, this can be reduced to the solution of a Sylvester equation. Let

$$\widehat{K}_1(\lambda) = K_1(\lambda) + \Delta_{K_1}(\lambda) := L_{\varepsilon}(\lambda) \otimes I_n + \Delta_{K_1}(\lambda) := (E_{\varepsilon} - \lambda F_{\varepsilon}) \otimes I_n + (\Delta_{31}^a - \lambda \Delta_{31}^b).$$

Then, making the change of variables $Y_{11} := Y$ and $X_{33} := X(I + X)^{-1}$, it suffices to solve

$$(K_1(\lambda) + \Delta_{K_1}(\lambda))Y + XK_1(\lambda) = \Delta_{K_1}(\lambda),$$

or, equivalently,

$$\begin{bmatrix} E_{\varepsilon}^T \otimes I_{n\varepsilon n} & I_{(\varepsilon+1)n} \otimes (E_{\varepsilon} \otimes I_n + \Delta_{31}^a) \\ F_{\varepsilon}^T \otimes I_{n\varepsilon n} & I_{(\varepsilon+1)n} \otimes (F_{\varepsilon} \otimes I_n + \Delta_{31}^b) \end{bmatrix} \begin{bmatrix} \operatorname{vec}(X) \\ \operatorname{vec}(Y) \end{bmatrix} = \begin{bmatrix} \operatorname{vec}(\Delta_{31}^a) \\ \operatorname{vec}(\Delta_{31}^b) \end{bmatrix}.$$
(9.36)

By Lemma 9.3.4, the smallest singular value of the unperturbed problem satisfies

$$\sigma_{2\varepsilon n(\varepsilon+1)n} \left[\begin{array}{c|c} E_{\varepsilon}^T \otimes I_{n\varepsilon n} & I_{(\varepsilon+1)n} \otimes E_{\varepsilon} \otimes I_n \\ \hline F_{\varepsilon}^T \otimes I_{n\varepsilon n} & I_{(\varepsilon+1)n} \otimes F_{\varepsilon} \otimes I_n \end{array} \right] \ge \frac{3}{4\varepsilon - 1}$$

Then, by using Weyl's perturbation theorem for singular values [54, Theorem 3.3.16], one obtains the following bound for the minimum norm solution of (9.36)

$$\|(X,Y)\|_{F} \leq \left[\frac{3}{4\varepsilon - 1} - \|\Delta_{31}^{a}\|_{2} - \|\Delta_{31}^{b}\|_{2}\right]^{-1} \|(\Delta_{31}^{a}, \Delta_{31}^{b})\|_{F}$$

assuming that the perturbation is small enough for satisfying $\frac{3}{4\varepsilon-1} - \|\Delta_{31}^a\|_2 - \|\Delta_{31}^b\|_2 > 0$. In addition,

$$||(X_{33}, Y_{11})||_F \le ||(X, Y)||_F / (1 - ||(X, Y)||_F).$$

Since $\|\Delta_{31}^a\|_2$ and $\|\Delta_{31}^b\|_2$ are of the order of δ , finally yields

$$\|(X_{33}, Y_{11})\|_F \le \frac{4\varepsilon - 1}{3} \|(\Delta_{31}^a, \Delta_{31}^b)\|_F + \mathcal{O}(\delta^2),$$
(9.37)

by neglecting quantities of the order of $\mathcal{O}(\delta^2)$.

The problem for restoring $K_2(\lambda)$ is clearly dual to the problem of $K_1(\lambda)$ and will therefore yield the bound

$$\|(X_{11}, Y_{33})\|_F \le \frac{4\eta - 1}{3} \|(\Delta_{13}^a, \Delta_{13}^b)\|_F + \mathcal{O}(\delta^2).$$
(9.38)

The problem of restoring I_{ℓ} amounts to solving $(I_{\ell} - X_{22})(I_{\ell} + \Delta_{22}^b)(I_{\ell} - Y_{22}) = I_{\ell}$, with $\widehat{I}_{\ell} = I_{\ell} + \Delta_{22}^b$. There are many possible solutions. A very simple one is to take $Y_{22} = 0$ and $I_{\ell} - X_{22} = (I_{\ell} + \Delta_{22}^b)^{-1}$, assuming Δ_{22}^b is small enough for the inverse to exist. This means that $X_{22} = \Delta_{22}^b + \mathcal{O}(\|\Delta_{22}^b\|_F^2)$ and

$$\|(X_{22}, Y_{22})\|_F = \|\Delta_{22}^b\|_F + \mathcal{O}(\delta^2).$$
(9.39)

We summarize this discussion in the following Theorem.

Theorem 9.4.9. Let the pencil $\widehat{S}_1(\lambda)$ have the block anti-triangular form given in (9.34). If $\max(\varepsilon, \eta) > 0$, then the updating strict equivalence transformation $(I - X)\widehat{S}_1(\lambda)(I - Y)$ detailed in (9.35) exists and can be bounded by

$$\|(X,Y)\|_F \le \frac{4\max(\varepsilon,\eta)-1}{3} \|\Delta_1^{new}(\lambda)\|_F + \mathcal{O}(\delta^2).$$

Proof. The bound for $||(X,Y)||_F$ follows directly from the identity

$$||(X,Y)||_F^2 = ||(X_{11},Y_{33})||_F^2 + ||(X_{22},Y_{22})||_F^2 + ||(X_{33},Y_{11})||_F^2,$$

from the inequality

$$\|(\Delta_{13}^{a}, \Delta_{13}^{b})\|_{F}^{2} + \|\Delta_{22}^{b}\|_{F}^{2} + \|(\Delta_{31}^{a}, \Delta_{31}^{b})\|_{F}^{2} \le \|\Delta_{1}^{new}(\lambda)\|_{F}^{2}$$

and from the individual inequalities (9.37), (9.38) and (9.39).

The following first order bound in Corollary 9.4.10 for the norm of the perturbation error $\Delta_2^{new}(\lambda)$ follows from Lemma 9.4.1, Theorem 9.4.9 and Corollary 9.4.8.

Corollary 9.4.10. Let us define the scalar $f_2 := \frac{\sqrt{2}(4 \max(\varepsilon, \eta) - 1)}{3}$. Then $\|\Delta_2^{new}(\lambda)\|_F < [1 + f_2 \|\widehat{S}_1(\lambda)\|_2] \|\Delta_1^{new}(\lambda)\|_F + \mathcal{O}(\delta^2)$

$$\leq [1 + f_2 \| S(\lambda) \|_2] \| \Delta_1^{new}(\lambda) \|_F + \mathcal{O}(\delta^2).$$

9.4.3 Step 3: Restoring the constant B and C matrices

From steps 1 and 2, described in the previous subsections, we have obtained a pencil $\widehat{S}_2(\lambda) = S(\lambda) + \Delta_2^{new}(\lambda)$ of the type

$$\widehat{S}_{2}(\lambda) := \begin{bmatrix} \widehat{M}(\lambda) & \widehat{C}(\lambda) & K_{2}^{T}(\lambda) \\ \widehat{B}(\lambda) & \widehat{A} - \lambda I_{\ell} & 0 \\ K_{1}(\lambda) & 0 & 0 \end{bmatrix}$$
(9.40)

strictly equivalent to $\widehat{S}(\lambda)$. We emphasize that the blocks $\widehat{M}(\lambda)$, $\widehat{B}(\lambda)$, $\widehat{C}(\lambda)$ and the matrix \widehat{A} are obviously different in (9.40) and in (9.34). We use the same symbols for avoiding a cumbersome notation. In this subsection, we will use $\Delta_{ij}(\lambda) = \Delta_{ij}^a - \lambda \Delta_{ij}^b$ to denote the corresponding blocks of the updated perturbation matrix $\Delta_2^{new}(\lambda)$. In this third step, we will restore the pencil $\widehat{S}_2(\lambda)$ to one where the blocks

$$\widehat{B}(\lambda) = B\widehat{K}_1 + \Delta_{21}(\lambda), \text{ and } \widehat{C}(\lambda) = \widehat{K}_2^T C + \Delta_{12}(\lambda)$$

are transformed to $\widetilde{B}\widehat{K}_1$ and $\widehat{K}_2^T\widetilde{C}$, respectively. We recall that

$$K_1(\lambda) = L_{\epsilon}(\lambda) \otimes I_n, \quad \widehat{K}_1 = \mathbf{e}_{\epsilon+1}^T \otimes I_n, \quad K_2(\lambda) = L_{\eta}(\lambda) \otimes I_m, \quad \widehat{K}_2 = \mathbf{e}_{\eta+1}^T \otimes I_m,$$

where \mathbf{e}_k is the standard kth unit vector of dimension k and $L_k(\lambda)$ is the classical Kronecker block of dimension $k \times (k+1)$, as introduced below (9.2). We will construct for this a strict equivalence transformation of the type

$$\begin{bmatrix} I_{m(\eta+1)} & -X_{12} & 0\\ & I_{\ell} & -X_{23}\\ & & I_{n\varepsilon} \end{bmatrix} \widehat{S}_{2}(\lambda) \begin{bmatrix} I_{n(\varepsilon+1)} & \\ -Y_{21} & I_{\ell} \\ 0 & -Y_{32} & I_{m\eta} \end{bmatrix}$$

$$= \begin{bmatrix} \widetilde{M}(\lambda) & \widehat{K}_{2}^{T}\widetilde{C} & K_{2}^{T}(\lambda)\\ \widetilde{B}\widehat{K}_{1} & \widehat{A} - \lambda I_{\ell} & 0\\ K_{1}(\lambda) & 0 & 0 \end{bmatrix}$$
(9.41)

The problems for $\widehat{B}(\lambda)$ and $\widehat{C}(\lambda)$ can again be treated separately. Let us first focus on the subsystem

$$\begin{bmatrix} I_{m(\eta+1)} & -X_{12} \\ & I_{\ell} \end{bmatrix} \begin{bmatrix} \widehat{C}(\lambda) & L_{\eta}^{T}(\lambda) \otimes I_{m} \\ \widehat{A} - \lambda I_{\ell} & 0 \end{bmatrix} \begin{bmatrix} I_{\ell} \\ -Y_{32} & I_{m\eta} \end{bmatrix}$$
$$= \begin{bmatrix} \mathbf{e}_{\eta+1} \otimes \widetilde{C} & L_{\eta}^{T}(\lambda) \otimes I_{m} \\ \widehat{A} - \lambda I_{\ell} & 0 \end{bmatrix}.$$

If we partition the matrices X_{12} , Y_{32} and $\widehat{C}(\lambda)$ as follows :

$$X_{12} := \begin{bmatrix} E_1 \\ E_2 \\ \vdots \\ E_\eta \\ E_{\eta+1} \end{bmatrix}, \quad Y_{32} := \begin{bmatrix} F_1 \\ F_2 \\ \vdots \\ F_\eta \end{bmatrix}, \quad \widehat{C}(\lambda) := \begin{bmatrix} C_{01} \\ C_{02} \\ \vdots \\ C_{0\eta} \\ C_{0(\eta+1)} \end{bmatrix} - \begin{bmatrix} C_{11} \\ C_{12} \\ \vdots \\ C_{1\eta} \\ C_{1(\eta+1)} \end{bmatrix} \lambda,$$

where all blocks have dimension $m \times \ell$, then we need to solve the following system of equations

$$\begin{bmatrix} E_1 & F_1 & E_2 & \dots & F_\eta & E_{\eta+1} \end{bmatrix} (I_{(2\eta+1)\ell} + N)$$

=
$$\begin{bmatrix} C_{11} & C_{01} & C_{12} & \dots & C_{0\eta} & C_{1(\eta+1)} \end{bmatrix},$$

where

$$I_{(2\eta+1)\ell} + N := \begin{bmatrix} I_{\ell} & A & & & \\ & I_{\ell} & I_{\ell} & & \\ & & I_{\ell} & \widehat{A} & & \\ & & & \ddots & \ddots & \\ & & & & & I_{\ell} & I_{\ell} \\ & & & & & & & I_{\ell} \end{bmatrix},$$

and $\widetilde{C} := C_{0(\eta+1)} - E_{\eta+1}\widehat{A}$. Clearly

$$\| \begin{bmatrix} E_1 & F_1 & E_2 & \dots & F_\eta & E_{\eta+1} \end{bmatrix} \|_F = \| (X_{12}, Y_{32}) \|_F,$$
$$\| \begin{bmatrix} C_{11} & C_{01} & C_{12} & \dots & C_{0\eta} & C_{1(\eta+1)} \end{bmatrix} \|_F \le \| \Delta_{12}(\lambda) \|_F,$$

and, since the matrix N is nilpotent with $N^{2\eta+1} = 0$,

$$(I_{(2\eta+1)\ell} + N)^{-1} = \sum_{i=0}^{2\eta} (-N)^i.$$

In addition, N has even powers N^{2i} of 2-norm $\|\widehat{A}^i\|_2 \leq \|\widehat{A}\|_2^i$, whereas the odd powers N^{2i-1} have 2-norm $\max(\|\widehat{A}^{i-1}\|_2, \|\widehat{A}^i\|_2) \leq \max(\|\widehat{A}\|_2^{i-1}, \|\widehat{A}\|_2^i)$. Since both of them can be bounded by $\max(1, \|\widehat{A}\|_2^i)$, it then follows that

$$\| (X_{12}, Y_{32}) \|_{F} \leq \| \Delta_{12}(\lambda) \|_{F} (1 + 2 \max(1, \|\widehat{A}\|_{2}) + \dots + 2 \max(1, \|\widehat{A}\|_{2}^{\eta}))$$

$$\leq [1 + 2\eta \max(1, \|\widehat{A}\|_{2}^{\eta})] \| \Delta_{12}(\lambda) \|_{F}.$$

$$(9.42)$$

The discussion for the $\widehat{B}(\lambda)$ block is clearly analogous and will yield the bound

$$\|(X_{23}, Y_{21})\|_F \le [1 + 2\varepsilon \max(1, \|\widehat{A}\|_2^{\varepsilon})] \|\Delta_{21}(\lambda)\|_F.$$
(9.43)

We can thus summarize this discussion in the following Theorem.

Theorem 9.4.11. Let the pencil $\hat{S}_2(\lambda)$ have the anti-triangular form given in (9.40). Then the updating strict equivalence transformation $(I - X)\hat{S}_2(\lambda)(I - Y)$ detailed in (9.41) exists and can be bounded by

$$\|(X,Y)\|_F \le [1+2\max(\eta,\varepsilon)\max(1,\|\widehat{A}\|_2^{\max(\eta,\varepsilon)})] \|\Delta_2^{new}(\lambda)\|_F.$$

Proof. The bound for $||(X,Y)||_F$ follows directly from the identity

$$||(X,Y)||_F^2 = ||(X_{12},Y_{32})||_F^2 + ||(X_{23},Y_{21})||_F^2,$$

from the inequality $\|\Delta_{12}(\lambda)\|_F^2 + \|\Delta_{21}(\lambda)\|_F^2 \le \|\Delta_2^{new}(\lambda)\|_F^2$ and from the individual inequalities (9.42) and (9.43).

The following first order bound in Corollary 9.4.12 for the norm of the perturbation error $\Delta_3^{new}(\lambda)$ follows from Lemma 9.4.1, Theorem 9.4.11 and Corollaries 9.4.8 and 9.4.10.

Corollary 9.4.12. Let us define the scalar $f_3 := \sqrt{2} \left[1 + 2 \max(\eta, \varepsilon) \max(1, \|\widehat{A}\|_2^{\max(\eta, \varepsilon)})\right]$. Then

$$\begin{aligned} \|\Delta_{3}^{new}(\lambda)\|_{F} &\leq [1 + f_{3} \|\widehat{S}_{2}(\lambda)\|_{2}] \|\Delta_{2}^{new}(\lambda)\|_{F} + \mathcal{O}(\delta^{2}) \\ &\leq [1 + f_{3} \|S(\lambda)\|_{2}] \|\Delta_{2}^{new}(\lambda)\|_{F} + \mathcal{O}(\delta^{2}). \end{aligned}$$

9.4.4 Putting it all together

In this subsection, we combine the obtained results regarding the strict equivalence transformation that restores in $\widehat{S}(\lambda)$ of (9.21) the special structure of the unperturbed block Kronecker linearization $S(\lambda)$ defined in (9.2), in such a way that the eigenstructure of $\widehat{S}(\lambda)$ can be linked to that of a particular rational matrix $\widetilde{R}(\lambda)$ as in (9.23). The final goal is to bound the norms of the differences between the quadruples $\{\lambda I_{\ell} - A, B, C, D(\lambda)\}$ and $\{\lambda I_{\ell} - \widetilde{A}, \widetilde{B}, \widetilde{C}, \widetilde{D}(\lambda)\}$ that are used for representing the unperturbed rational matrix $R(\lambda)$ and the perturbed one $\widetilde{R}(\lambda)$, respectively.

Recall that we were given the pencil $S(\lambda)$ of which we want to compute the eigenstructure, since it gives the one of the rational matrix $R(\lambda)$ in (9.20). Instead, our backward stable algorithm applied to $S(\lambda)$ computes the exact eigenstructure of a slightly perturbed pencil $\widehat{S}(\lambda)$ with additive error $\Delta_S(\lambda)$ which is induced by the eigenstructure algorithm and is bounded as :

$$\|\Delta_S(\lambda)\|_F \le c(\ell, m\eta, n\varepsilon) \cdot \epsilon_M \cdot \|S(\lambda)\|_F,$$

where ϵ_M is the machine precision of the used computer, and $c(\ell, m\eta, n\varepsilon)$ is a moderate function depending only on the size of the matrix pencil. We then constructed in three steps a new modified block Kronecker linearization

$$\widetilde{S}(\lambda) := (I - X)\widehat{S}(\lambda)(I - Y) := (I - X_3)(I - X_2)(I - X_1)\widehat{S}(\lambda)(I - Y_1)(I - Y_2)(I - Y_3)$$
(9.44)

as in (9.22), strictly equivalent to $\widehat{S}(\lambda)$, where both $||X||_F$ and $||Y||_F$ are also of the order of the machine precision times some factors and such that the corresponding rational matrix $\widetilde{R}(\lambda)$ (9.23) has a similar representation as $R(\lambda)$. Since $\widehat{S}(\lambda)$ and $\widetilde{S}(\lambda)$ are strictly equivalent pencils, they have *exactly* the same eigenstructure, which implies that we have computed the exact eigenstructure of the nearby rational matrix $\widetilde{R}(\lambda)$.

For convenience, the blocks of $\tilde{S}(\lambda)$ will be expressed in the sequel as $\tilde{M}(\lambda) := M(\lambda) + \Delta M(\lambda)$, $\tilde{A} := A + \Delta A$, $\tilde{B} := B + \Delta B$ and $\tilde{C} := C + \Delta C$. In the previous subsections, we rewrote $\tilde{S}(\lambda)$ as an additive perturbation

$$\widetilde{S}(\lambda) = S(\lambda) + \Delta_3^{new}(\lambda)$$

and derived a first order bound for the norm of the error pencil $\Delta_3^{new}(\lambda)$ in Corollaries 9.4.8, 9.4.10 and 9.4.12 :

$$\|\Delta_3^{new}(\lambda)\|_F \le (1+f_1\|S(\lambda)\|_2)(1+f_2\|S(\lambda)\|_2)(1+f_3\|S(\lambda)\|_2)\|\Delta_S(\lambda)\|_F + \mathcal{O}(\delta^2).$$
(9.45)

This implies, in particular, that if $\|\Delta_S(\lambda)\|_F$ is sufficiently small, then the norms of the perturbations ΔA , ΔB and ΔC are sufficiently small to guarantee that $\widetilde{C}(\lambda I_\ell - \widetilde{A})^{-1}\widetilde{B}$ is a minimal state-space realization, as announced. Then, according to [6], $\widetilde{S}(\lambda)$ is indeed a strong linearization of the rational matrix $\widetilde{R}(\lambda)$ in (9.23). Moreover, (9.45) also implies that if $\|\Delta_S(\lambda)\|_F$ is sufficiently small, then $\widetilde{D}(\lambda) := \sum_{i=0}^d (D_i + \Delta D_i)\lambda^i$ in (9.23) is a polynomial matrix with the same degree $d = \eta + \varepsilon + 1$ as the polynomial part $D(\lambda)$ of $R(\lambda)$ (recall that we are assuming that d is the degree of $D(\lambda)$ or, equivalently, that $D_d \neq 0$).

Notice that $R(\lambda)$ in (9.23) is the transfer function of the following perturbed polynomial system matrix

$$P(\lambda) + \Delta P(\lambda) := \begin{bmatrix} \lambda I_{\ell} - A & -B \\ C & D(\lambda) \end{bmatrix} + \begin{bmatrix} -\Delta A & -\Delta B \\ \Delta C & \sum_{i=0}^{d} \Delta D_{i} \lambda^{i} \end{bmatrix}, \quad (9.46)$$

where $P(\lambda)$ is a polynomial system matrix of the original rational matrix $R(\lambda)$. Recall that $||R(\lambda)||_F$ is defined in (9.3) as $||P(\lambda)||_F$. This motivates us to define the norm of the perturbation of $R(\lambda)$ as

$$\|\Delta R(\lambda)\|_F := \|\Delta P(\lambda)\|_F = \sqrt{\|\Delta A\|_F^2 + \|\Delta B\|_F^2 + \|\Delta C\|_F^2 + \sum_{i=0}^d \|\Delta D_i\|_F^2}.$$

After this discussion, we present our main perturbation results in Theorems 9.4.13 and 9.4.15. The first one focuses on block Kronecker linearizations and the second one on the corresponding rational matrices.

Theorem 9.4.13. Let $R(\lambda)$ be the $m \times n$ rational matrix in (9.1) and let $S(\lambda)$ be a block Kronecker linearization of $R(\lambda)$ as in (9.2). Let us define $\alpha := 1 + 2\varepsilon \max(1, ||A||_2^{\varepsilon}), \beta := 1 + 2\eta \max(1, ||A||_2^{\eta}), \gamma := \frac{\varepsilon + \eta}{2\sqrt{2}}$ and $s := \max(\alpha, \beta, \gamma) + \gamma(\beta ||B||_2 + \alpha ||C||_2)$. Assume that $\max(\varepsilon, \eta) > 0$ and consider the functions dependent on the initial data

$$f_{1} := f_{1}(\varepsilon, \eta, ||A||_{2}, ||B||_{2}, ||C||_{2}) := \frac{4\sqrt{2s}}{2-\sqrt{3}},$$

$$f_{2} := f_{2}(\varepsilon, \eta) := \frac{\sqrt{2} (4 \max(\varepsilon, \eta) - 1)}{3},$$

$$f_{3} := f_{3}(\varepsilon, \eta, ||A||_{2}) := \sqrt{2} [1 + 2 \max(\eta, \varepsilon) \max(1, ||A||_{2}^{\max(\eta, \varepsilon)})].$$

Let $\widehat{S}(\lambda) := S(\lambda) + \Delta_S(\lambda)$ be a perturbed pencil as in (9.21). If $\|\Delta_S(\lambda)\|_F$ is sufficiently small, then $\widehat{S}(\lambda)$ is strictly equivalent to a block Kronecker linearization $\widetilde{S}(\lambda)$ as in (9.22) with the same parameters ε and η as $S(\lambda)$, i.e., the transformation (9.44) exists. Moreover, $\widetilde{S}(\lambda) = S(\lambda) + \Delta_3^{new}(\lambda)$ with

$$\|\Delta_{3}^{new}(\lambda)\|_{F} \leq (1 + f_{1}\|S(\lambda)\|_{2})(1 + f_{2}\|S(\lambda)\|_{2})(1 + f_{3}\|S(\lambda)\|_{2})\|\Delta_{S}(\lambda)\|_{F} + \mathcal{O}(\delta^{2}),$$
(9.47)

where $\delta := \frac{\|\Delta_S(\lambda)\|_F}{\|S(\lambda)\|_F}$.

Proof. This follows directly from (9.45), except that we have replaced the 2-norm of \widehat{A} in f_3 in Corollary 9.4.12 by that of A, because the difference can be absorbed in the $\mathcal{O}(\delta^2)$ term.

Theorem 9.4.13 does not provide directly bounds on the norms of the differences between the quadruples representing the rational matrices $R(\lambda)$ and $\widetilde{R}(\lambda)$ corresponding to the block Kronecker linearizations $S(\lambda)$ and $\widetilde{S}(\lambda)$. The reason is that the polynomial parts $D(\lambda) = (\Lambda_{\eta}(\lambda) \otimes I_m)^T M(\lambda) (\Lambda_{\epsilon}(\lambda) \otimes I_n)$ and $\widetilde{D}(\lambda) =$ $(\Lambda_{\eta}(\lambda) \otimes I_m)^T \widetilde{M}(\lambda) (\Lambda_{\epsilon}(\lambda) \otimes I_n)$ of $R(\lambda)$ and $\widetilde{R}(\lambda)$ are not directly visible in $S(\lambda)$ and $\widetilde{S}(\lambda)$. For this reason, we will need Lemma 9.4.14, that follows from [26, Lemma 2.15, Theorem 4.4 and Lemma 5.23(b)].

Lemma 9.4.14. Let $M(\lambda)$ be a $m(\eta + 1) \times n(\varepsilon + 1)$ pencil and let $\Lambda_k(\lambda) := \begin{bmatrix} \lambda^k & \cdots & \lambda & 1 \end{bmatrix}^T$. If we define the polynomial matrix $Q(\lambda)$ as

$$Q(\lambda) := (\Lambda_{\eta}(\lambda) \otimes I_m)^T M(\lambda) (\Lambda_{\varepsilon}(\lambda) \otimes I_n), \qquad (9.48)$$

then we can bound its norm as follows

$$\|Q(\lambda)\|_F \le \sqrt{2\min(\varepsilon+1,\eta+1)} \,\|M(\lambda)\|_F.$$

Moreover, for every polynomial matrix $Q(\lambda)$ of degree at molerstewart1973t $d = \varepsilon + \eta + 1$, there exist infinitely many pencils $M(\lambda)$ satisfying (9.48). For each of these pencils $||M(\lambda)||_F \geq ||Q(\lambda)||_F / \sqrt{2d}$ and there exist pencils such that $||Q(\lambda)||_F = ||M(\lambda)||_F$.

As commented in [26], Fiedler and proper generalized Fiedler pencils (modulo permutations) of a polynomial matrix $Q(\lambda)$ satisfy $||Q(\lambda)||_F = ||M(\lambda)||_F$ in Lemma 9.4.14. On the other hand, it might be worth to remind that there exist pencils $M(\lambda)$ satisfying (9.48) with norm arbitrarily larger than the norm of $Q(\lambda)$.

We are finally in the position of proving the main perturbation result of this chapter.

Theorem 9.4.15. Let $R(\lambda) = C(\lambda I_{\ell} - A)^{-1}B + \sum_{i=0}^{d} D_{i}\lambda^{i}$ be an $m \times n$ rational matrix, where $C(\lambda I_{\ell} - A)^{-1}B$ is a minimal state-space realization of the strictly proper part of $R(\lambda)$, let $S(\lambda)$ be a block Kronecker linearization of $R(\lambda)$ as in (9.2) with $\max(\varepsilon, \eta) > 0$, and let f_{1}, f_{2}, f_{3} be the functions defined in Theorem 9.4.13. Let $\widehat{S}(\lambda) := S(\lambda) + \Delta_{S}(\lambda)$ be a perturbed pencil as in (9.21). If $\|\Delta_{S}(\lambda)\|_{F}$ is sufficiently small, then $\widehat{S}(\lambda)$ is strictly equivalent to a block Kronecker linearization $\widetilde{S}(\lambda)$ as in (9.22), with the same parameters ε and η as $S(\lambda)$, of a rational matrix

$$\widetilde{R}(\lambda) = \widetilde{C}(\lambda I_{\ell} - \widetilde{A})^{-1}\widetilde{B} + \sum_{i=0}^{d} \widetilde{D}_{i}\lambda^{i},$$

where $\widetilde{C}(\lambda I_{\ell} - \widetilde{A})^{-1}\widetilde{B}$ is a minimal state-space realization of the strictly proper part of $\widetilde{R}(\lambda)$. Moreover, if $\widetilde{A} := A + \Delta A$, $\widetilde{B} := B + \Delta B$, $\widetilde{C} := C + \Delta C$ and $\widetilde{D}_i := D_i + \Delta D_i$, $i = 0, 1, \ldots, d$, then

$$\frac{\sqrt{\|\Delta A\|_F^2 + \|\Delta B\|_F^2 + \|\Delta C\|_F^2 + \sum_{i=0}^d \|\Delta D_i\|_F^2}}{\|R(\lambda)\|_F} \le K_{S,R} \frac{\|\Delta_S(\lambda)\|_F}{\|S(\lambda)\|_F} + \mathcal{O}(\delta^2), \quad (9.49)$$

where

$$K_{S,R} := \sqrt{2\min(\varepsilon + 1, \eta + 1)} (1 + f_1 \| S(\lambda) \|_2) (1 + f_2 \| S(\lambda) \|_2) (1 + f_3 \| S(\lambda) \|_2) \frac{\| S(\lambda) \|_F}{\| R(\lambda) \|_F}$$

and $\delta = \frac{\| \Delta_S(\lambda) \|_F}{\| S(\lambda) \|_F}.$

Proof. Since $\widetilde{S}(\lambda)$ and $S(\lambda)$ have the same structure according to Theorem 9.4.13,

$$\Delta_3^{new}(\lambda) = \widetilde{S}(\lambda) - S(\lambda) = \begin{bmatrix} \widetilde{M}(\lambda) - M(\lambda) & \widehat{K}_2^T(\widetilde{C} - C) & 0\\ (\widetilde{B} - B)\widehat{K}_1 & \widetilde{A} - A & 0\\ 0 & 0 & 0 \end{bmatrix}$$

and $\|\Delta_3^{new}(\lambda)\|_F = \sqrt{\|\Delta A\|_F^2 + \|\Delta B\|_F^2 + \|\Delta C\|_F^2 + \|\widetilde{M}(\lambda) - M(\lambda)\|_F^2}$. Next, we combine this expression of $\|\Delta_3^{new}(\lambda)\|_F$ with $\sum_{i=0}^d D_i \lambda^i = (\Lambda_\eta(\lambda) \otimes I_m)^T M(\lambda) (\Lambda_\epsilon(\lambda) \otimes I_n)$, $\sum_{i=0}^d \widetilde{D}_i \lambda^i = (\Lambda_\eta(\lambda) \otimes I_m)^T \widetilde{M}(\lambda) (\Lambda_\epsilon(\lambda) \otimes I_n)$ and Lemma 9.4.14, and we get

$$\sqrt{\|\Delta A\|_F^2 + \|\Delta B\|_F^2 + \|\Delta C\|_F^2 + \sum_{i=0}^d \|\Delta D_i\|_F^2} \le \sqrt{2\min(\varepsilon + 1, \eta + 1)} \|\Delta_3^{new}(\lambda)\|_F.$$

The rest of the proof follows from (9.47).

The strength of the new structured backward error analysis that we present in this chapter for the computation of the eigenstructure of a rational matrix $R(\lambda)$ by applying a backward stable generalized eigenvalue algorithm to a block Kronecker linearization $S(\lambda)$ of $R(\lambda)$ is that we can interpret the computed eigenstructure as the exact eigenstructure for a slightly perturbed rational matrix $\tilde{R}(\lambda)$ corresponding to the nearby quadruple $\{\lambda I_{\ell} - \tilde{A}, \tilde{B}, \tilde{C}, \tilde{D}(\lambda)\}$, and that we have a bound on the error because we have a specific coordinate system in which we can describe both the original rational matrix $R(\lambda)$ and its perturbed version $\tilde{R}(\lambda)$, namely by the quadruples $\{\lambda I_{\ell} - A, B, C, D(\lambda)\}$ and $\{\lambda I_{\ell} - \tilde{A}, \tilde{B}, \tilde{C}, \tilde{D}(\lambda)\}$. It still remains to analyze under which conditions this bound is satisfactory. This is the purpose of the next section.

9.5 Sufficient conditions for structural backward stability

The goal of this section is to establish sufficient conditions on $R(\lambda)$ and $S(\lambda)$ that guarantee that $K_{S,R}$ in (9.49) is moderate and, thus, that guarantee structural backward stability. We advance that these conditions are the following

$$\max(\|A\|_F, \|B\|_F, \|C\|_F, \|D(\lambda)\|_F) \le 1 \text{ and } \|M(\lambda)\|_F \approx \|D(\lambda)\|_F, \quad (9.50)$$

where the notation introduced in the previous section is used. Observe that the first condition is a condition on $R(\lambda)$ while the second one is on $S(\lambda)$. According to Lemma 9.4.14, the second condition can be satisfy simply by choosing an adequate block Kronecker linearization $S(\lambda)$. In addition, we will see that the conditions (9.50) are essentially necessary for $K_{S,R}$ to be moderate, though this does not mean that they are necessary for structural backward stability since (9.49) is an upper bound. For the sake of clarity, the discussion in this section focuses on identifying the key ingredients for structural backward stability instead of on providing precise bounds. There exist, obviously, rational matrices which do not satisfy the first condition in (9.50). We will discuss in Section 9.7 how to proceed in such cases.

In the first place observe that each of the essential four factors of $K_{S,R}$, that is, $(1 + f_1 || S(\lambda) ||_2)$, $(1 + f_2 || S(\lambda) ||_2)$, $(1 + f_3 || S(\lambda) ||_2)$ and $\frac{|| S(\lambda) ||_F}{|| R(\lambda) ||_F}$, is larger than 1. This is obvious for the first three factors. For the fourth factor, it follows from the equalities

$$||S(\lambda)||_F^2 = ||A||_F^2 + ||B||_F^2 + ||C||_F^2 + ||M(\lambda)||_F^2 + \ell + 2(m\eta + n\varepsilon) \quad \text{and} ||R(\lambda)||_F^2 = ||A||_F^2 + ||B||_F^2 + ||C||_F^2 + ||D(\lambda)||_F^2 + \ell.$$
(9.51)

To find upper bounds for the three factors $(1 + f_1 || S(\lambda) ||_2)$, $(1 + f_2 || S(\lambda) ||_2)$, $(1 + f_3 || S(\lambda) ||_2)$ of $K_{S,R}$ requires to upper bound each f_i and $|| S(\lambda) ||_2$. For this purpose, we consider Lemmas 9.5.1 and 9.5.2. Lemma 9.5.1 provides a bound on the function f_1 that allows us to identify its molerstewart1973t relevant dependencies. Moreover, Lemma 9.5.1 emphasizes the key role of $t := \max(\eta, \varepsilon)$ in our perturbation analysis. Lemma 9.5.2 bounds $||S(\lambda)||_2$.

Lemma 9.5.1. Let us define $M_a := \max(1, ||A||_2)$, $M_b := \max(||B||_2, ||C||_2)$ and $t := \max(\eta, \varepsilon) > 0$ and consider the functions f_1 , f_2 and f_3 in Theorem 9.4.13. Then

$$1 \le f_1 \le 22(1+2tM_a^t)(1+\sqrt{2}tM_b), \ 1 \le f_2 = \frac{\sqrt{2}}{3}(4t-1), \ 1 \le f_3 = \sqrt{2}(1+2tM_a^t).$$

Proof. It follows by taking into account the inequalities $\gamma \leq \frac{t}{\sqrt{2}}$ and $s \leq (1 + 2tM_a^t)(1 + \sqrt{2}tM_b)$.

Lemma 9.5.2. Let $S(\lambda)$ be the block Kronecker linearization (9.2). Then

$$\max(1, \|A\|_2, \|B\|_2, \|C\|_2, \|M(\lambda)\|_2) \le \|S(\lambda)\|_2$$

and

$$\|S(\lambda)\|_{2} \leq \sqrt{2} + \| \begin{bmatrix} M(\lambda) & \widehat{K}_{2}^{T}C \\ B\widehat{K}_{1} & A \end{bmatrix} \|_{2} \leq \sqrt{2} + \sqrt{\|A\|_{F}^{2} + \|B\|_{F}^{2} + \|C\|_{F}^{2} + \|M(\lambda)\|_{F}^{2}}.$$

Proof. The first inequality follows from the definition of the 2-norm of a pencil given in the introduction and the fact that the 2-norm of a matrix is larger than or equal to the 2-norm of any of its submatrices. The second inequality follows from applying the triangular inequality to

$$S(\lambda) = \begin{bmatrix} M(\lambda) & \widehat{K}_2^T C & 0 \\ B\widehat{K}_1 & A & 0 \\ 0 & 0 & 0 \end{bmatrix} + \begin{bmatrix} 0 & 0 & K_2^T(\lambda) \\ 0 & -\lambda I_\ell & 0 \\ K_1(\lambda) & 0 & 0 \end{bmatrix}.$$

Note that the 2-norm of a pencil as defined in the introduction is indeed a norm and, so, the triangular inequality can be applied.

We remark that Lemmas 9.5.1 and 9.5.2 imply that the conditions (9.50) are essentially necessary for $K_{S,R}$ to be moderate. This can be seen as follows. First, from Lemma 9.4.14, we have $||M(\lambda)||_F \geq ||D(\lambda)||_F/\sqrt{2(\varepsilon + \eta + 1)}$. Thus, $\max(||A||_F, ||B||_F, ||C||_F, ||D(\lambda)||_F) \gg 1$ implies $||S(\lambda)||_2 \gg 1$, which in turns implies $K_{S,R} \gg 1$, since $f_i \geq 1$ for i = 1, 2, 3. Moreover, if $||M(\lambda)||_F \gg ||D(\lambda)||_F$, then $\frac{||S(\lambda)||_F}{||R(\lambda)||_F} \gg 1$ may happen, according to (9.51), and $K_{S,R} \gg 1$ in that situation. We emphasize that the condition $||M(\lambda)||_F \approx ||D(\lambda)||_F$ was also used in the analysis in [26, Corollary 5.24].

Next, we prove the announced result that conditions (9.50) are sufficient for $K_{S,R}$ to be moderate and, thus, for structural backward stability.
Corollary 9.5.3. Under the hypotheses and with the notation of Theorem 9.4.15, assume, in addition, that (9.50) holds and let $t := \max(\eta, \varepsilon) > 0$. Then,

$$K_{S,R} \leq g t^q \sqrt{m+n}$$
,

where q = 5, if $\eta > 0$ and $\varepsilon > 0$, q = 9/2, if $\eta = 0$ or $\varepsilon = 0$, and g is a moderate number (a constant that does not depend on $\eta, \varepsilon, m, n, \ell$). Moreover

$$\frac{\sqrt{\|\Delta A\|_F^2 + \|\Delta B\|_F^2 + \|\Delta C\|_F^2 + \sum_{i=0}^d \|\Delta D_i\|_F^2}}{\|R(\lambda)\|_F} \le g t^q \sqrt{m+n} \frac{\|\Delta_S(\lambda)\|_F}{\|S(\lambda)\|_F} + \mathcal{O}(\delta^2).$$

Proof. Note that (9.50) and Lemmas 9.5.1 and 9.5.2 imply $||S(\lambda)||_2 \leq 2 + \sqrt{2}$, $f_1 \leq g_1 t^2$, $f_2 \leq g_2 t$, and $f_3 \leq g_3 t$, with g_1, g_2, g_3 moderate numbers. Moreover, from (9.51), (9.50) and $||R(\lambda)||_F \geq 1$, we get that $||S(\lambda)||_F^2 \approx ||R(\lambda)||_F^2 + 2(m\eta + n\varepsilon)$ and

$$|S(\lambda)||_F^2 \le (1 + 2(m\eta + n\varepsilon)) ||R(\lambda)||_F^2 \le 3(m+n) t ||R(\lambda)||_F^2$$

It only remains to analyze the factor $\sqrt{2\min(\varepsilon+1,\eta+1)}$ of $K_{S,R}$, which is less than or equal to $\sqrt{2(t+1)}$, if $\eta > 0$ and $\varepsilon > 0$, or equal to $\sqrt{2}$, if $\eta = 0$ or $\varepsilon = 0$. Combining all these bounds with the fact that $t \ge 1$, the result follows as a corollary of Theorem 9.4.15.

Remark 9.5.4. Observe that (9.50) allow $\max(||A||_F, ||B||_F, ||C||_F, ||D(\lambda)||_F) \ll 1$. However, since the rational matrix $R(\lambda)$ in (9.1) can be multiplied by a nonzero number without affecting at all its eigenstructure, it is natural and convenient to use as sufficient conditions

$$\max(\|A\|_F, \|B\|_F, \|C\|_F, \|D(\lambda)\|_F) = 1 \text{ and } \|M(\lambda)\|_F \approx \|D(\lambda)\|_F.$$
(9.52)

Such conditions would have appeared as sufficient in the analysis if we had defined the norm of $R(\lambda)$ as

$$|||R(\lambda)|||_{F} := \sqrt{||A||_{F}^{2} + ||B||_{F}^{2} + ||C||_{F}^{2} + \sum_{i=0}^{d} ||D_{i}||_{F}^{2}}, \qquad (9.53)$$

instead as in (9.3) (observe that we have removed the ℓ summand), depending only on the free parameters of the representation of $R(\lambda)$ in (9.1). We have chosen to use (9.3) because, first, it identifies the informal "norm" of $R(\lambda)$ with the formal norm of the polynomial system matrix $P(\lambda)$ and, second, it corresponds to the particular case $E = I_{\ell}$ of the more general representation $R(\lambda) = C(\lambda E - A)^{-1}B + D(\lambda)$, with E nonsingular, when taking as norm the one of the corresponding polynomial system matrix. Under the conditions (9.52), it is essentially equivalent to use (9.3) or (9.53) as "norm" of $R(\lambda)$. The use of representations $R(\lambda) = C(\lambda E - A)^{-1}B + D(\lambda)$ for rational matrices is of interest in certain applications and the block Kronecker linearizations in this case are obtained just by replacing $A - \lambda I_{\ell}$ by $A - \lambda E$ in (9.2). We will consider the analysis of this general case in the future.

9.6 Restoring the structure when the polynomial part of the rational matrix is linear

In this section, we consider the particular case of having a rational matrix with linear polynomial part. That is, the case of having a rational matrix that can be written in the form

$$R(\lambda) = C(\lambda I_{\ell} - A)^{-1}B + M(\lambda),$$

where $C(\lambda I_{\ell} - A)^{-1}B$ is a minimal state-space realization and $M(\lambda)$ is a matrix pencil. Then $R(\lambda)$ can be strongly linearized using the following linear polynomial system matrix

$$S(\lambda) := \begin{bmatrix} M(\lambda) & C \\ B & A - \lambda I_{\ell} \end{bmatrix}.$$
(9.54)

Notice that, in this case, the linearization does not have the block anti-triangular structure as the block Kronecker linearization in (9.2) since $K_1(\lambda)$ and $K_2(\lambda)$ are empty matrices. The strong linearization (9.54) can be seen as the limit case of (9.2) when $\varepsilon = \eta = 0$.

If we compute the eigenstructure of $S(\lambda)$, the backward stability of the staircase algorithm [83] and the QZ algorithm [71] guarantees that we computed the exact eigenstructure of a slightly perturbed pencil

$$\widehat{S}(\lambda) := S(\lambda) + \Delta_S(\lambda), \quad \Delta_S(\lambda) := \begin{bmatrix} \Delta_{11}(\lambda) & \Delta_{12}(\lambda) \\ \Delta_{21}(\lambda) & \Delta_{22}(\lambda) \end{bmatrix}.$$
(9.55)

The structure of (9.54) is lost in (9.55) since the off-diagonal blocks of $\widehat{S}(\lambda)$ are not constant matrices and the identity block I_{ℓ} is not preserved by the perturbation.

Notice that restoring in $\widehat{S}(\lambda)$ the original structure of $S(\lambda)$ is much simpler than in previous sections, as we do not have to restore any anti-triangular zero block nor the minimal bases $K_1(\lambda)$ and $K_2(\lambda)$ in (9.19). We only have to take care of restoring the identity matrix I_{ℓ} and the constant matrices B and C to obtain in two steps a new strictly equivalent linear polynomial system matrix

$$\widetilde{S}(\lambda) := (I - X)\widehat{S}(\lambda)(I - Y) := (I - X_2)(I - X_1)\widehat{S}(\lambda)(I - Y_1)(I - Y_2)$$
(9.56)

of the form

$$\widetilde{S}(\lambda) := \begin{bmatrix} \widetilde{M}(\lambda) & \widetilde{C} \\ \widetilde{B} & \widetilde{A} - \lambda I_{\ell} \end{bmatrix}, \qquad (9.57)$$

where $\widetilde{M}(\lambda) := M(\lambda) + \Delta M(\lambda)$, $\widetilde{A} := A + \Delta A$, $\widetilde{B} := B + \Delta B$ and $\widetilde{C} := C + \Delta C$. For that, we consider the discussion in Subsection 9.4.2, for restoring I_{ℓ} ; and a simplified version of the discussion in Subsection 9.4.3, for restoring the constant matrices Band C. In particular, from the bound in (9.39) and a counterpart of Theorem 9.4.11 we get the following result. **Theorem 9.6.1.** Let $S(\lambda)$ be a minimal linear system matrix as in (9.54). The transformation (X, Y) in (9.56) exists and we can bound the corresponding perturbation $\widetilde{S}(\lambda) - S(\lambda)$ as follows :

$$\|\widetilde{S}(\lambda) - S(\lambda)\|_F \le (1 + \sqrt{2} \|S(\lambda)\|_2)^2 \|\Delta_S(\lambda)\|_F + \mathcal{O}(\delta^2).$$
(9.58)

In addition, if $\|\widetilde{S}(\lambda) - S(\lambda)\|_F$ is sufficiently small, then the perturbed pencil $\widetilde{S}(\lambda)$ is a minimal linear system matrix of the rational matrix $\widetilde{R}(\lambda) = \widetilde{C}(\lambda I_\ell - \widetilde{A})^{-1}\widetilde{B} + \widetilde{M}(\lambda)$ and

$$\frac{\sqrt{\|\Delta A\|_F^2 + \|\Delta B\|_F^2 + \|\Delta C\|_F^2 + \|\Delta M(\lambda)\|_F^2}}{\|R(\lambda)\|_F} \le (1 + \sqrt{2} \|S(\lambda)\|_2)^2 \frac{\|\Delta_S(\lambda)\|_F}{\|S(\lambda)\|_F} + \mathcal{O}(\delta^2),$$

where $\delta = \|\Delta_S(\lambda)\|_F / \|S(\lambda)\|_F$.

The simplicity of the bound in Theorem 9.6.1 is also a consequence of $||S(\lambda)||_F = ||R(\lambda)||_F$.

9.7 Scaling for obtaining structural backward stability

Once a block Kronecker linearization $S(\lambda)$ in (9.2) of $R(\lambda)$ in (9.1) satisfying

$$||M(\lambda)||_F \approx ||D(\lambda)||_F$$

is chosen and the staircase or the QZ algorithm is applied to $S(\lambda)$, structural backward stability is guaranteed for the computed eigenstructure if the first condition in (9.50) holds. However, there exist rational matrices which do not satisfy

$$\max(\|A\|_F, \|B\|_F, \|C\|_F, \|D(\lambda)\|_F) \le 1$$

and, therefore, the computation of their eigenstructure via a block Kronecker linearization might not be structurally backward stable. In this section, we study how to proceed in these cases.

Multiplication by a constant. Observe that the eigenstructure of the rational matrix $R(\lambda)$ does not change at all if it is multiplied by a positive real constant d_R . Choosing appropriately d_R , we get easily a rational matrix such that $\max(\|B\|_F, \|C\|_F, \|D(\lambda)\|_F) \leq 1$. Even more, if d_R is an integer power of 2, this multiplication can be performed without introducing any rounding error.

Diagonal scaling. The above explanation indicates that the crucial point is how to deal with rational matrices with $||A||_F > 1$. For this, note that when representing a rational matrix $R(\lambda)$ by a realization quadruple $\{\lambda I_{\ell} - A, B, C, D(\lambda)\}$, where $D(\lambda)$ is polynomial,

$$R(\lambda) := C(\lambda I_{\ell} - A)^{-1}B + \sum_{i=0}^{d} D_{i}\lambda^{i},$$

one can change the coordinate system of the state-space realization $\{A, B, C\}$ of the strictly proper part of $R(\lambda)$ by a diagonal similarity scaling

$$T := \operatorname{diag}(d_1, \ldots, d_\ell) \quad \text{with} \quad d_i > 0,$$

without changing $R(\lambda)$ since

$$C(\lambda I_{\ell} - A)^{-1}B = CT(\lambda I_{\ell} - T^{-1}AT)^{-1}T^{-1}B.$$

Thus, before multiplying $R(\lambda)$ by d_R , we can choose T to balance A, i.e., to minimize its Frobenius norm under all diagonal similarities by making the 2-norms of the rows and columns of $T^{-1}AT$ become equal [74]. Moreover, at the same time, the Frobenius norms of $T^{-1}B$ and CT can be made equal by considering a positive scalar factor multiplying T. Observe, in addition, that if the entries of T are integer powers of 2, this process does not introduce rounding errors, though, in this case, the norm of $T^{-1}AT$ is only approximately minimized. However, the effects of T are limited since $||T^{-1}AT||_F \ge \sqrt{|\lambda_1|^2 + \cdots + |\lambda_\ell|^2}$, where $\lambda_1, \ldots, \lambda_\ell$ are the eigenvalues of A, for any invertible T, i.e., diagonal or not. Therefore, other approaches are needed for dealing with all instances of matrices A with large norms. It is important to emphasize at this point that the influence of a large norm matrix A on the bound (9.49) is huge, because it contributes to $||S(\lambda)||_2$, but also the factor $||A||_2^{\max(\eta,\varepsilon)}$ is present in both f_1 and f_3 .

Change of variable. The final solution comes from changing the variable λ to

$$\widehat{\lambda} := d_{\lambda} \lambda$$

and from combining this with the multiplication by the constant d_R and the diagonal scaling T discussed above. Note that the change of variable transforms the zeros and the poles of $R(\lambda)$ in a very simple way, preserving their partial multiplicities, and that does not change at all its minimal indices [65, 82].

The combination of all these scalings yields a new transfer function

$$\widehat{R}(\widehat{\lambda}) := \widehat{D}(\widehat{\lambda}) + \widehat{C}(\widehat{\lambda}I_{\ell} - \widehat{A})^{-1}\widehat{B} := d_R R(\widehat{\lambda}/d_{\lambda})$$
(9.59)

where

$$\widehat{A} := d_{\lambda} T^{-1} A T, \quad \widehat{B} := \sqrt{d_{\lambda} d_R} T^{-1} B, \quad \widehat{C} := \sqrt{d_{\lambda} d_R} C T \tag{9.60}$$

and

$$\widehat{D}_i := d_R d_\lambda^{-i} D_i, \quad \text{for all} \quad i = 0, 1, \dots, d.$$
(9.61)

Then, we can choose $d_{\lambda} := \min(1, ||T^{-1}AT||_F^{-1})$, such that \widehat{A} has norm smaller than or equal to 1. Note that the preliminary balancing will make this step milder, in the sense that d_{λ} will be closer to 1. Finally, based on (9.59), we summarize the following scaling procedure for obtaining a rational matrix $\widehat{R}(\widehat{\lambda})$ with

$$\max(\|\widehat{A}\|_F, \|\widehat{B}\|_F, \|\widehat{C}\|_F, \|\widehat{D}(\widehat{\lambda})\|_F) = 1$$

from the data $\{A, B, C, D_0, D_1, \dots, D_d\}$:

- **Step 1.** Compute $T = \text{diag } d_1, \ldots, d_\ell$ to balance A and to make equal the norms of $T^{-1}B$ and CT.
- **Step 2.** Choose $d_{\lambda} := \min(1, \|T^{-1}AT\|_F^{-1}).$

Step 3. Choose

$$d_R = \frac{1}{\max(\|\sqrt{d_\lambda} T^{-1}B\|_F^2, \|\sqrt{d_\lambda} CT\|_F^2, \sqrt{\sum_{i=0}^d \|d_\lambda^{-i}D_i\|_F^2})}.$$

Step 4. Compute $\{\widehat{A}, \widehat{B}, \widehat{C}, \widehat{D}_0, \widehat{D}_1, \dots, \widehat{D}_d\}$ as in (9.60)-(9.61).

This process can be easily arranged to use scale factors that are all integer powers of two and, thus, can be implemented without any rounding error. Moreover, this scaling can be applied directly to the pencil $S(\lambda)$. More precisely, the pencil

$$\widehat{S}(\widehat{\lambda}) := D_{\ell} S(\widehat{\lambda}/d_{\lambda}) D_r,$$

where the left and right diagonal scalings D_{ℓ} and D_r are given by

$$D_{\ell} := \operatorname{diag} d_{R}^{\frac{1}{2}} d_{\lambda}^{-\eta} I_{m}, \dots, d_{R}^{\frac{1}{2}} d_{\lambda}^{0} I_{m}, d_{\lambda}^{\frac{1}{2}} d_{1}^{-1}, \dots, d_{\lambda}^{\frac{1}{2}} d_{\ell}^{-1}, d_{R}^{-\frac{1}{2}} d_{\lambda}^{\epsilon} I_{n}, \dots, d_{R}^{-\frac{1}{2}} d_{\lambda}^{1} I_{n},$$
$$D_{r} := \operatorname{diag} d_{R}^{\frac{1}{2}} d_{\lambda}^{-\epsilon} I_{n}, \dots, d_{R}^{\frac{1}{2}} d_{\lambda}^{0} I_{n}, d_{\lambda}^{\frac{1}{2}} d_{1}, \dots, d_{\lambda}^{\frac{1}{2}} d_{\ell}, d_{R}^{-\frac{1}{2}} d_{\lambda}^{\eta} I_{m}, \dots, d_{R}^{-\frac{1}{2}} d_{\lambda}^{1} I_{m},$$

is a block Kronecker linearization of the rational matrix $\widehat{R}(\widehat{\lambda})$ in (9.59).

9.8 Numerical experiments

In this section, we describe three experiments illustrating that the potential sources of structural backward instability revealed by the bound (9.49) are indeed observed in practice. More precisely, the experiments will illustrate that if a rational matrix $R(\lambda)$ as in (9.1) does not satisfy the first condition in (9.50), then the computation of the eigenstructure of $R(\lambda)$ by applying the QZ algorithm to a block Kronecker linearization $S(\lambda)$ of $R(\lambda)$ that satisfies $||M(\lambda)||_F = ||D(\lambda)||_F$ is not structurally backward stable. Moreover, the experiments also illustrate that the scaling described in Section 9.7 is effective and leads to structured backward stability for the scaled rational matrices and linearizations.

A difficulty for performing fully reliable numerical experiments in this setting is that to estimate the actual global backward error for the *whole* computed eigenstructure, i.e., the left-hand side of (9.49), is a challenging optimization problem for which we do not know yet a solution. Therefore, we will limit ourselves to computing a lower bound for the backward error based on the "local" backwards errors of each computed zero of the rational matrix, as we explain below. This lower bound might severely underestimate the actual global backward error. Thus, we cannot check from our experiments the sharpness of the bound (9.49), which, on the other hand, was deduced through many potentially overestimating inequalities with the main goal of getting a bound as clear as possible instead of optimizing its sharpness.

For simplicity, we will restrict our numerical experiments to square and regular rational matrices $R(\lambda)$ with a corresponding quadruple $\{A, B, C, D(\lambda)\}$ of moderate dimensions and degree of its polynomial part: $m = n = 2, \ell = 5, d = 3$. The block Kronecker pencil we choose for our computations is

$$S(\lambda) := \begin{bmatrix} \lambda D_3 + D_2 & 0 & 0 & I_2 \\ 0 & \lambda D_1 + D_0 & C & -\lambda I_2 \\ 0 & B & A - \lambda I_\ell & 0 \\ I_2 & -\lambda I_2 & 0 & 0 \end{bmatrix},$$

which has η and ε equal to 1, size 11×11 and satisfies $||M(\lambda)||_F = ||D(\lambda)||_F$. We also will look at the polynomial system matrix

$$P(\lambda) := \begin{bmatrix} A - \lambda I_{\ell} & B \\ C & D(\lambda) \end{bmatrix}, \quad D(\lambda) := D_0 + \lambda D_1 + \lambda^2 D_2 + \lambda^3 D_3$$

of $R(\lambda)$ because it allows us to estimate the backward errors of our algorithm as follows. We look for a rational matrix $\widetilde{R}(\lambda)$ corresponding to a quadruple $\{A + \Delta A, B + \Delta B, C + \Delta C, (D + \Delta D)(\lambda)\}$ such that all its finite zeros are exactly all the computed finite eigenvalues obtained by applying the QZ algorithm to $S(\lambda)$ and such that $\|(\Delta A, \Delta B, \Delta C, (\Delta D)(\lambda))\|_F$ is as small as possible. As a consequence of the classical results of Rosenbrock [78], this is equivalent to find a perturbed polynomial system matrix $P(\lambda) + \Delta P(\lambda)$ of $\widetilde{R}(\lambda)$, whose finite zeros are the computed eigenvalues λ_i and such that $\|(\Delta A, \Delta B, \Delta C, (\Delta D)(\lambda))\|_F$ is as small as possible. Therefore, $\{\Delta A, \Delta B, \Delta C, \Delta D_0, \Delta D_1, \Delta D_2, \Delta D_3\}$ must have the property that simultaneously, at each computed eigenvalue λ_i , the matrix

$$P(\lambda_i) + \Delta P(\lambda_i) = P(\lambda_i) + \begin{bmatrix} \Delta A & \Delta B & | & 0 & 0 & | & 0 & 0 & | & 0 & 0 \\ \Delta C & \Delta D_0 & | & 0 & \Delta D_1 & | & 0 & \Delta D_2 & | & 0 & \Delta D_3 \end{bmatrix} \begin{bmatrix} I_{\ell+m} \\ \lambda_i I_{\ell+m} \\ \lambda_i^2 I_{\ell+m} \\ \lambda_i^3 I_{\ell+m} \end{bmatrix}$$

must be singular. To find the smallest possible Frobenius norm of all possible $\{\Delta A, \Delta B, \Delta C, \Delta D_0, \Delta D_1, \Delta D_2, \Delta D_3\}$ that satisfy this property for all computed λ_i is not obvious, however to solve this problem for only one computed λ_i is easy. For this purpose, let $\Delta^{(i)}$ be the minimum Frobenius norm matrix that makes $P(\lambda_i) + \Delta^{(i)}$ singular. Note that $\Delta^{(i)}$ can be computed through the singular value decomposition of $P(\lambda_i)$ and that, generically, it is a rank one matrix with Frobenius norm equal to

 $\sigma_{\min} P(\lambda_i)$. Then, the linear system

$$\Delta^{(i)} := \begin{bmatrix} \Delta_{11}^{(i)} & \Delta_{12}^{(i)} \\ \Delta_{21}^{(i)} & \Delta_{22}^{(i)} \end{bmatrix} = \begin{bmatrix} \Delta A & \Delta B & | & 0 & 0 & | & 0 & 0 & | & 0 & 0 \\ \Delta C & \Delta D_0 & | & 0 & \Delta D_1 & | & 0 & \Delta D_2 & | & 0 & \Delta D_3 \end{bmatrix} \begin{bmatrix} I_{\ell+m} \\ \lambda_i I_{\ell+m} \\ \lambda_i^2 I_{\ell+m} \\ \lambda_i^3 I_{\ell+m} \end{bmatrix}$$

for the unknowns $\{\Delta A, \Delta B, \Delta C, \Delta D_0, \Delta D_1, \Delta D_2, \Delta D_3\}$ is consistent and its minimum Frobenius norm solution is given by

where $g(\lambda_i) := (1 + |\lambda_i|^2 + |\lambda_i|^4 + |\lambda_i|^6)$, and the Frobenius norm of this 7-tuple of matrices is given by

$$r(P,\lambda_i) := \| \left[\begin{array}{cc} \Delta_{11}^{(i)} & \Delta_{12}^{(i)} \\ \Delta_{21}^{(i)} & \Delta_{22}^{(i)} / \sqrt{g(\lambda_i)} \end{array} \right] \|_F.$$

This leads us to use in our experiments

$$r(P) := \max_{i} r(P, \lambda_i) \tag{9.62}$$

as an estimate for the structured absolute backward error induced by our algorithm, i.e., as an estimate for the numerator of the left-hand side of (9.49). We emphasize that this is a lower bound for the actual global structured backward error, since it corresponds to a rational matrix that has only one of the computed eigenvalues as a finite zero.

Experiment 1. In the first experiment, we investigate the behavior of the structured backward error for rational matrices with matrices A of increasing (large) norms, and with the rest of the matrices in the quadruple $\{A, B, C, D(\lambda)\}$ having norms of order 1. The reason why we pay first particular attention to the norm of Ais because according to the bound (9.49) the influence of A should be huge because it contributes to $||S(\lambda)||_2$ and also to f_1 and f_3 . For this purpose, we generated with the Matlab function randn, 7 batchs of samples of 50 random matrix-tuples $\{A, B, C, D_0, D_1, D_2, D_3\}$, and in each batch indexed with i, we multiplied the matrix A by 10^i , with i going from 1 till 7, in each of the 50 runs of each batch. In each batch, we computed the average of the absolute backward error estimators (9.62) for both the original matrix-tuples and the scaled ones after applying the procedure in Section 9.7. In Figure 9.1, we plot the results of these computations: the horizontal axis represents the index i defining each batch and the vertical axis the logarithm of the average absolute backward errors. Ideally, the absolute backward error should be of order $\epsilon_M ||R(\lambda)||_F$, where ϵ_M is the machine precision, and, so, we also plot this magnitude for the unscaled original data taking in each batch the average of all $||R(\lambda)||_F$ (for the scaled data, this magnitude is always of order ϵ_M and is not plotted).

We observe that the absolute backward errors for the unscaled problem grow very strongly with the index *i*, i.e., with the norm of *A*, and that computing the zeros of a rational matrix by applying the QZ algorithm to the block Kronecker linearization $S(\lambda)$ is highly structurally backward unstable for large norms of *A*, as predicted by the bound (9.49). In contrast, when applying the scaling procedure described in Section 9.7, this growth is absent and we get perfect structural backward stability for the scaled rational matrix, as predicted by (9.49).



Figure 9.1: Experiment 1: behavior of absolute structured backward errors for increasing values of the norm of A.

Experiment 2. In the second experiment, we investigate the behavior of the structured backward error for rational matrices with matrices A of norms of order 1, and with the rest of the matrices in the quadruple $\{A, B, C, D(\lambda)\}$ having increasing (large) norms. The situation in this experiment is opposite to the one in the first experiment. The matrices are generated following the same pattern of the first experiment except by the fact that once the matrices $\{A, B, C, D_0, D_1, D_2, D_3\}$ are generated with randn, B is multiplied by $10^{i/2}$, C by $10^{i/3}$, D_1 by 10^i , D_2 by $10^{i/2}$ and D_3 by $10^{i/3}$, for $i = 1, \ldots, 7$. The results are plotted in Figure 9.2 and the conclusions are the same as in the first experiment and are in agreement with our analysis. However, note that the growth of the absolute backward errors of the original unscaled data is much smaller than in the first experiment. This effect is qualitatively expected from the bound (9.49), since f_3 does not depend on the norms of B, C and $D(\lambda)$, but the observed very large quantitative difference is not fully

explained by (9.49). Possible reasons of this are that, as we have emphasized before, our backward error estimator is a lower bound that may underestimate severely the actual global backward error and/or that the bound in (9.49) overestimates the actual error.



Figure 9.2: Experiment 2: behavior of absolute structured backward errors for increasing values of the norms of B, C and $D(\lambda)$.

Experiment 3. The last experiment we present combines the scalings used in the first and second experiments. That is, once the matrices $\{A, B, C, D_0, D_1, D_2, D_3\}$ are generated with randn, A is multiplied by the factor used in Experiment 1 and B, C, D_1, D_2 , and D_3 are multiplied by the factors used in Experiment 2. Taking into account that the function f_1 appearing in the bound (9.49) includes a product of the norm of A times the norm of B and B and

The main conclusion of this section is that our main *a priori* structured backward error bound (9.49) identifies correctly the sources of instability of computing the eigenstructure of a rational matrix by applying the QZ algorithm to its block Kronecker linearizations and that the scaling proposed in Section 9.7 leads to structural backward stability.



Figure 9.3: Experiment 3: behavior of absolute structured backward errors for increasing values of the norms of A, B, C and $D(\lambda)$.

Chapter 10

Conclusions, publications and open problems

In this chapter, we give some concluding remarks of the results presented in this PhD thesis and summarize the main original contributions. We also list all the papers (published or in development) that include the results obtained in this dissertation, and the conferences where many of them have been presented. Finally, we suggest a collection of related open problems and recommendations for future research directions.

10.1 Conclusions and original results

In this section, we highlight the main original results in this dissertation together with some conclusions of each chapter. In general, it is worth mentioning that as a consequence of the results of this thesis and, in particular, as a consequence of the new definition of local linearization of rational matrices in Chapter 4, most of the linearizations for rational matrices in the literature have been unified in a general framework. In particular, the theory of local linearizations of rational matrices captures all the pencils that have been used (as far as we know) in the literature for solving REPs arising from approximating NLEPs; some of them are analysed in Chapter 5. In addition, the definition of local linearizations includes the definition of linearizations and strong linearizations of arbitrary rational matrices presented in [6], just by considering as set the whole underlying field and including infinity in the strong case. But the conditions for a linearization to be strong in Chapter 4 and those in [6] are different. More precisely, given a minimal linear polynomial system matrix

$$\mathcal{L}(\lambda) = \begin{bmatrix} A_1 \lambda + A_0 & B_1 \lambda + B_0 \\ -(C_1 \lambda + C_0) & D_1 \lambda + D_0 \end{bmatrix} \in \mathbb{F}[\lambda]^{(n+q) \times (n+r)},$$

for the linearization to be strong in [6] it is required A_1 to be invertible (recall Proposition 3.1.1). However, we do not require A_1 to be invertible but

$$\operatorname{rank} \begin{bmatrix} A_1 \\ C_1 \end{bmatrix} = \operatorname{rank} \begin{bmatrix} A_1 & B_1 \end{bmatrix} = n,$$

which is the condition for minimality at infinity introduced in Chapter 4. A particular case of local linearizations in the strong sense are the strongly minimal linearizations in Chapter 7, for which the transfer function matrix is exactly the desired rational matrix. Strongly minimal linearizations include the generalized state-space systems originally used in [92] for linearizing rational matrices. Thus, the new local theory of linearizations for rational matrices is a flexible tool that generalizes and includes most of the previous results available in the literature in this area. Moreover, local linearizations allow us to introduce in Chapters 6 and 8 linearizations of rational matrices that can not be constructed with the previous definitions of linearization. All these results are in part possible due to the new treatment of minimality of polynomial system matrices at infinity.

We first started in **Chapter 3** by considering the definition of strong linearization for rational matrices given in [6], and we proved the key Lemma 3.1.2, which allows us to construct infinitely many strong linearizations for any rational matrix $R(\lambda) \in \mathbb{F}(\lambda)^{p \times m}$ from any given strong linearization of $R(\lambda)$. Then Lemma 3.1.2 was applied to construct linearizations of square rational matrices $R(\lambda) \in \mathbb{F}(\lambda)^{m \times m}$, written as the sum of their polynomial and strictly proper parts. Namely,

$$R(\lambda) := D(\lambda) + R_{sp}(\lambda).$$

More precisely, Lemma 3.1.2 has been combined with some of the strong linearizations constructed in [6, Theorem 5.11] for rational matrices and with strong linearizations of its polynomial part $D(\lambda)$ to create new families of strong linearizations of square rational matrices. In particular, we considered the families of strong linearizations of polynomial matrices presented in [36] to linearize the polynomial parts. These are, the ansatz spaces $\mathbb{M}_1(D)$ and $\mathbb{M}_2(D)$, where the corresponding polynomial matrix is expressed in terms of any polynomial basis satisfying a three term recurrence relation. The main results for the construction of the strong linearizations of rational matrices from $\mathbb{M}_1(D)$ and $\mathbb{M}_2(D)$ are Theorems 3.2.9 and 3.3.4, respectively. Such linearizations were called M_1 - and M_2 -strong linearizations, respectively. The recovery of the eigenvectors of the rational matrix from those of the linearizations in these families has been studied and stated in different results. For \mathbb{M}_1 -strong linearizations, Theorems 3.4.4 and 3.4.5 state the recovery of right and left eigenvectors associated with finite eigenvalues, respectively, and Theorem 3.4.7 the recovery of right and left eigenvectors associated with infinity. For \mathbb{M}_2 -strong linearizations, Theorems 3.4.9 and 3.4.10 state the recovery of right and left eigenvectors associated with finite eigenvalues, respectively, and Theorem

3.4.12 the recovery of right and left eigenvectors associated with infinity. Moreover, the preservation of symmetric and Hermitian structures of the rational matrix in the linearizations have been studied in Theorem 3.6.9 and Corollary 3.6.11, for the symmetric case, and in Theorem 3.7.5 and Corollary 3.7.6, for Hermitian structures. Finally, a result on how to construct strong linearizations when the polynomial part is expressed in other polynomial bases is given in Theorem 3.8.1.

We would like to emphasize that the techniques developed in this chapter could be applied to construct other families of strong linearizations of rational matrices $R(\lambda)$ expressed as the sum of its polynomial part $D(\lambda)$ and its strictly proper part $R_{sp}(\lambda)$, given a strong linearization of $D(\lambda)$ in any of the families of strong linearizations of polynomial matrices developed in the last years and a minimal state-space realization of $R_{sp}(\lambda)$. However, although any rational matrix can be expressed as the sum of its polynomial and strictly proper parts, this expression may not be easily available from the applications and/or may not be the best representation in a particular problem. Then linearizations for more general representations are presented in Chapter 6. But in Chapter 6 we used the definition of linearization for rational matrices introduced in Chapter 4, since with other definitions of linearization for rational matrices existing in the literature it is not possible to develop the results in Chapter 6.

In Chapter 4 a theory of local linearizations of rational matrices has been carefully introduced and analysed. For that, we first developed the extension of Rosenbrock's minimal polynomial system matrices to a local sense. The notions of polynomial system matrices minimal in a set and at infinity are given in Definitions 4.1.1 and 4.1.8, respectively. For minimality at infinity we used the notion of qreversal of rational matrices introduced in Definition 4.1.7. We studied the pole and zero information that one can obtain from polynomial system matrices minimal in a set and at infinity in Theorems 4.1.5 and 4.1.11, respectively. The study of the recovery of the invariant orders at infinity is given in Theorem 4.1.13. Polynomial system matrices that are minimal in the whole underlying field and also at infinity are called strongly minimal in Definition 4.1.15. Then we present the notions of linearizations of rational matrices in a set and also at infinity in Definitions 4.2.1 and 4.2.9, respectively, together with spectral characterizations in Theorems 4.2.6 and 4.2.14 and the recovery of the invariant orders at infinity in Proposition 4.2.15. When a linearization of a rational matrix satisfy the minimality conditions in the whole underlying field and also at infinity for a particular choice of the q-reversal, then we say it is a strong linearization of grade q (Definition 4.2.16). We also define and analyze the very general family of block full rank pencils and linearizations, as templates that cover many of the pencils, available in the literature, that linearize rational matrices in a target set, in the whole field and/or at infinity. In particular, block full rank pencils and linearizations in a set are introduced in Theorems 4.3.5 and 4.4.1, and block full rank pencils and linearizations at infinity in Theorems 4.3.7 and 4.4.10.

In **Chapter 5** we apply the theory of local linearizations in Chapter 4 to a number of pencils that have appeared in the influential papers [47, 60] on solving numerically NLEPs by combining rational approximations and linearizations of the resulting rational matrices. In particular, the pencils introduced in [47] have been studied in Theorems 5.1.1 and 5.1.3. In addition, in [47] trimmed linearizations are considered when the coefficients defining the NLEPs have low rank. These trimmed linearizations are studied in Theorems 5.1.5 and 5.1.7. The pencils introduced in [60] have been studied in Theorem 5.2.6 and their minimality conditions in Theorems 5.2.8 and 5.2.9. In [60] trimmed linearizations exploiting low rank structures are also considered, which we analyzed in Theorem 5.2.13. It has been emphasized throughout the chapter that the theory of local linearizations allows us to view these pencils, and to explain their properties, from rather different perspectives, by using the notions of block full rank pencils and linearizations in Chapter 4.

In Chapter 6 rational matrices $R(\lambda)$ are expressed in the general form

$$R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda),$$

where $A(\lambda)$, $B(\lambda)$, $C(\lambda)$ and $D(\lambda)$ are polynomial matrices with arbitrary degrees. From these representations we have constructed a family of linear polynomial matrices that, under some minimality conditions, are local linearizations for $R(\lambda)$ in the sense defined in Chapter 4. These linearizations are presented in Theorems 6.1.2 and 6.2.1. Moreover, we have showed in Theorem 6.4.12 how to recover the eigenvectors of $R(\lambda)$, when $R(\lambda)$ is regular, from those of any of the new linearizations. The recovery of right and left minimal bases and minimal indices of $R(\lambda)$, when $R(\lambda)$ is singular, is studied in Theorems 6.4.9 and 6.4.11, respectively. Finally, in Section 6.5 we apply the theory in this chapter to the solution of scalar rational equations.

We emphasize that in contrast to the construction of other families of lineariations for rational matrices, as those introduced in Chapter 3 or in the references [2, 7, 27, 79], the construction of the linearizations in this chapter do not require neither to write the corresponding rational matrix $R(\lambda)$ as the sum of its polynomial part and its strictly proper part nor to express the strictly proper rational part in state-space form, which can introduce errors that were not in the original problem.

In **Chapter 7** we considered realizations $\{A(\lambda), B(\lambda), C(\lambda), D(\lambda)\}$ for a given rational transfer function matrix $R(\lambda) = C(\lambda)A(\lambda)^{-1}B(\lambda) + D(\lambda)$, where the matrices $A(\lambda), B(\lambda), C(\lambda)$ and $D(\lambda)$ are pencils, and where $A(\lambda)$ is assumed to be regular. We showed that if the corresponding linear polynomial system matrix

$$L(\lambda) := \begin{bmatrix} A(\lambda) & -B(\lambda) \\ C(\lambda) & D(\lambda) \end{bmatrix}$$
(10.1)

is strongly minimal then the poles, zeros and left and right null space structures of the rational matrix $R(\lambda)$ can be recovered from the generalized eigenstructures of $A(\lambda)$ and $L(\lambda)$, and $L(\lambda)$ is named as a strongly minimal linearization of $R(\lambda)$ (Definition 7.2.4). In particular, in Proposition 7.2.11 we proved that if $L(\lambda)$ is a strongly minimal linearization of a rational matrix $R(\lambda)$ then the left and right minimal indices of $R(\lambda)$ and those of $L(\lambda)$ are the same. In Subsection 7.2.2 we studied the relation of strongly minimal linearizations with other definitions of strong linearizations for polynomial and rational matrices in the literature. In addition, by Theorems 7.2.6 and 7.2.7, we showed that the strong minimality conditions imply the strong irreducibility conditions in [91]. We stated such result in Proposition 7.2.10. We also showed in Section 7.3 how to obtain such a strongly minimal linear system matrix from a non-minimal one, by applying a reduction procedure that is based on the classical staircase algorithm developed in [83]. The main results in that section are stated in Theorems 7.3.1 and 7.3.3. These results extend those previously obtained in the literature for generalized state space systems and polynomial matrices.

In Chapter 8 we looked at strongly minimal linear polynomial system matrices for any given rational matrix $R(\lambda)$, preserving the structure whenever we have a specific type of self-conjugate structure on $R(\lambda)$. We showed that there always exist strongly minimal linear polynomial system matrices that have the same selfconjugate structure as the transfer function matrix. For that we first constructed, in Theorem 8.1.2, strongly minimal linearizations for arbitrary polynomial matrices and, in Theorem 8.2.2, for self-conjugate polynomial matrices. Then, in Theorem 8.3.1, we constructed strongly minimal linearizations for arbitrary strictly proper rational matrices and, in Theorem 8.4.2, for self-conjugate strictly proper rational matrices. Finally, in Theorems 8.5.1 and 8.5.2 we stated the results for both arbitrary and structured rational matrices. These results were known for the case of arbitrary proper transfer functions [48], but were extended in this chapter to rational matrices that are not proper. Moreover, the derivation is new and is based on arguments that are very similar for the strictly proper part and the polynomial part of the rational matrix. The proofs are also constructive and lead to efficient algorithms for their construction.

It is worth mentioning that the strong linearizations constructed in Chapter 3 preserving symmetric and Hermitian structures require the degree of the polynomial part of the rational matrix to be odd or that, when it is even, the leading matrix coefficient to be invertible. Both restrictions are inherited from the classical theory of linearizations for structured polynomial matrices and they do not appear by considering strongly minimal linearizations as we showed in this chapter. As far as we know, it is the first time that such a completely restriction free construction of structured linearizations has been achieved.

In **Chapter 9** we developed the first structured backward error analysis for an algorithm that computes the eigenstructure of a rational matrix via linearization. More precisely, the considered algorithm starts from a rational matrix expressed as in (9.1) and computes its eigenstructure by applying a backward stable generalized eigenproblem algorithm to its block Kronecker linearizations described in (9.2). The

main result is stated in Theorem 9.4.15, and in Theorem 9.6.1 for the case of rational matrices with linear polynomial part. As a consequence of this analysis, we have identified the simple sufficient conditions (9.50) for global structural backward stability. In the case of rational matrices which do not satisfy these conditions, we have developed in Section 9.7 an scaling procedure that transforms the original rational matrix in another one for which structural backward stability is guaranteed. A number of numerical experiments confirming the predictions of the backward error analysis have been performed and discussed in Section 9.8. The results in Chapter 9 showed that solving numerically REPs via block Kronecker linearizations is a backward stable process in a global sense, under certain conditions involving both the representation of the rational matrix and the choice of the linearization.

10.2 Publications

The original results in Chapter 3 are contained in the following paper:

1 – F. M. DOPICO, S. MARCAIDA, M. C. QUINTANA, Strong linearizations of rational matrices with polynomial part expressed in an orthogonal basis, Linear Algebra and its Applications, 570 (2019) 1–45.

The original results in Chapters 4 and 5 are contained in the following two papers:

2 – F. M. DOPICO, S. MARCAIDA, M. C. QUINTANA, P. VAN DOOREN, Local linearizations of rational matrices with application to rational approximations of nonlinear eigenvalue problems, Linear Algebra and its Applications, 604 (2020) 441 – 475.

3– F. M. DOPICO, S. MARCAIDA, M. C. QUINTANA, P. VAN DOOREN, *Block full rank linearizations of rational matrices*, submitted. Available as arXiv: 2011.00955v1.

The original results in Chapter 6 are contained in the following paper:

4 – J. PÉREZ, M. C. QUINTANA, *Linearizations of rational matrices from general representations*, submitted. Available as arXiv:2003.02934v1.

The original results in Chapter 7 are contained in the following paper:

5 – F. M. DOPICO, M. C. QUINTANA, P. VAN DOOREN, *Linear system matrices of rational transfer functions*, to appear in "Realization and Model Reduction of Dynamical Systems", A Festschrift to honor the 70th birthday of Thanos Antoulas", Springer-Verlag. Available as arXiv:1903.05016v2.

The original results in Chapter 8 are contained in the following paper:

6- F. M. DOPICO, M. C. QUINTANA, P. VAN DOOREN, Strongly minimal self-conjugate linearizations for polynomial and rational matrices, in preparation.

The original results in Chapter 9 are contained in the following paper:

7- F. M. DOPICO, M. C. QUINTANA, P. VAN DOOREN, *Structural backward stability in rational eigenvalue problems solved via block Kronecker linearizations*, submitted. Available as arXiv:2103.16395v1.

10.3 Contributions to conferences

The author of this dissertation has presented many of the original results in this thesis in several conferences. Among them, we emphasize one of the main international conferences in the area of Applied Mathematics: the International Congress on Industrial and Applied Mathematics (ICIAM), and one of the main international conferences in the area of Applied Linear Algebra: the Society for Industrial and Applied Mathematics (SIAM) Conference on Applied Linear Algebra. All the contributions are listed in what follows in reversed chronological order:

- (1) MAY 17 21, 2021: SIAM CONFERENCE ON APPLIED LINEAR ALGEBRA (LA21). Virtual conference, originally scheduled in New Orleans, Louisiana, USA. Type of participation: Invited talk in minisymposium "Structured Eigenvalue Problems." Title of the work: Block Full Rank Linearizations of Rational Matrices.
- (2) JULY 15-19, 2019: INTERNATIONAL CONGRESS ON INDUSTRIAL AND AP-PLIED MATHEMATICS (ICIAM 19). Universidad de Valencia, Valencia, Spain. Type of participation: Invited talk in minisymposium "Nonlinear and multiparameter eigenvalue problems". Title of the work: Local Linearizations of Rational Matrices and their Applications to Nonlinear Eigenvalue Problems.
- (3) MAY 29-30, 2019: WORKSHOP "ADVANCES IN NUMERICAL LINEAR ALGE-BRA: CELEBRATING THE CENTENARY OF THE BIRTH OF JAMES H. WILKIN-SON". University of Manchester, Manchester, United Kingdom. Type of participation: Poster. Title of the work: On the Structure of Linearizations for Rational Approximations of Nonlinear Eigenvalue Problems.
- (4) SEPTEMBER 19, 2018: AUTUMN OF EIGENVALUES SEMINAR SERIES PART II. Instituto de Ciencias Matematicas (ICMAT), Madrid, Spain. Type of participation: Invited talk. Title of the work: Symmetric Strong Linearizations of Symmetric Rational Matrices.

- (5) MAY 30 JUNE 1, 2018: ALAMA 2018 MEETING. Complejo San Juan, Sant Joan d'Alacant, Spain. Type of participation: Contributed talk. Title of the work: Strong linearizations of rational matrices with polynomial part expressed in an orthogonal basis.
- (6) MAY 4-8, 2018: SIAM CONFERENCE ON APPLIED LINEAR ALGEBRA (LA18). Hong Kong Baptist University, Hong Kong, China. Type of participation: Invited talk in minisymposium "Polynomial and Rational Matrices". Title of the work: Strong linearizations of rational matrices with polynomial part expressed in an orthogonal basis.
- (7) SEPTEMBER 11-13, 2017: WORKSHOP OF YOUNG RESEARCHERS IN MATH-EMATICS. Universidad Complutense de Madrid, Madrid, Spain. Type of participation: Contributed talk. Title of the work: Constructing new classes of Strong Linearizations of Rational Matrices.

10.4 Open problems and future research

In this section, we discuss some open problems and directions for future research related to the results developed in this dissertation. We think that, after the results presented in this thesis and others existing in the literature, the study of the theoretical properties of linearizations of rational matrices is quite developed. However, there are few numerical stability results in this regard, in the spirit of those we presented in Chapter 9 that are limited to the family of "block Kronecker linearizations" in [6]. Therefore, the problems proposed in this section will focus more on addressing numerical properties for rational matrices and their linearizations than on theoretical results. For the case of polynomial matrices, the study of the numerical properties of their linearizations is well understood. The general idea is to extend those results for rational matrices.

Problem 1: Study the numerical properties of the linearizations introduced in this dissertation. In particular, given a linearization of a REP in a set, it is important to study the global backward stability in terms of the structure of the representation of the rational matrix defining the REP when applying a numerical method to compute the eigenvalues of the corresponding linearization and it gets perturbed, as it is done for the family of "block Kronecker linearizations" in Chapter 9. A first interesting problem inside the general Problem 1 is to extend the analysis in Chapter 9 for block Kronecker linearizations of the rational matrix ($\lambda I_{\ell} - A$) is replaced by ($\lambda E - A$) with E nonsingular. This might help to avoid the scaling of the variable for getting backward stability in certain cases.

Now we briefly recall the families of linearizations for rational matrices introduced in this thesis:

- (1) The strong linearizations of rational matrices in Chapter 3, where the corresponding rational matrix $R(\lambda)$ is written in the form $R(\lambda) = D(\lambda) + C(I_n\lambda A)^{-1}B$ with polynomial part $D(\lambda)$ expressed in an orthogonal basis and $C(I_n\lambda A)^{-1}B$ being a minimal state space realization.
- (2) The family of block full rank linearizations in Chapter 4. Associated with such linearizations, rational matrices $R(\lambda)$ are written in the form $R(\lambda) = N_2(\lambda)[M(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)]N_1(\lambda)^T$ where $M(\lambda)$, $C(\lambda)$, $A(\lambda)$ and $B(\lambda)$ are pencils and $N_1(\lambda)$ and $N_2(\lambda)$ are rational bases.
- (3) The linearizations of rational matrices from general representations in Chapter 6, where rational matrices are written as $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, with $D(\lambda)$, $C(\lambda)$, $A(\lambda)$ and $B(\lambda)$ being arbitrary polynomial matrices.
- (4) The strongly minimal linearizations in Chapter 8, where rational matrices $R(\lambda)$ are written from their Laurent expansions around the point at infinity. That is, $R(\lambda) = R_d \lambda^d + \ldots + R_1 \lambda + R_0 + R_{-1} \lambda^{-1} + R_{-2} \lambda^{-2} + R_{-3} \lambda^{-3} + \ldots$

We notice that for the families of linearizations in (1), (2) and (3) the matrix coefficients defining the representations of the rational matrix are reflected on the linearizations. Then, in principle, one might use ideas similar to those in Chapter 9 in order to recover the structure of the rational matrix in the corresponding linearization when it gets perturbed. However, the coefficients of the Laurent expansions do no appear in the strongly minimal linearizations in (4). Thus one should take into account, in addition, the numerical transformations done to build the linearizations in order to determine which rational matrix is linearized by the perturbed pencil for studying backward stability. The fact that this construction is based on unitary transformations gives us some hope of solving this difficult problem.

Problem 2: Define different condition numbers of eigenvalues λ of REPs, for different representations of the rational matrix defining the REP, that is their sensitivity to perturbations of the parameters defining the representation. For the case of polynomial matrices, definitions of condition numbers in the literature take into account that the corresponding polynomial matrix can be written in terms of different polynomial bases. For the case of rational matrices, it is important to consider as well different representations of the corresponding rational matrix and different types of perturbations. After the condition numbers are defined, the next step would be to compare the condition numbers with the standard condition numbers of the generalized eigenvalue problems associated with different linearizations by analyzing their ratios. An eigenvalue λ could then be computed stably from the linearization of rational matrices and which linearizations have favorable properties with respect to conditioning.

Problem 3: Define different local backward errors for particular computed pairs of eigenvalue λ and associated eigenvector x of REPs considering different represen-

tations of the corresponding rational matrix and different types of perturbations. Obtaining explicit formulas of backward errors of an approximate eigenpair (λ, x) for rational matrices can be much more involved than for polynomial matrices. This is due to the fact that in many of the representations of rational matrices appear inverses of polynomial matrices containing the pole information. After expressions or methods for computing local backward errors are obtained, the next step would be to compare the obtained local backward errors with the standard backward errors of the generalized eigenvalue problems associated with linearizations. These results would determine which representations of rational matrices and which linearizations have favorable properties with respect to local backward stability.

Problem 4: Study the stability and accuracy of the linearization-based algorithm proposed in Section 6.5 for solving (scalar) rational equations as in (6.25).

Problem 5: For rational matrices $R(\lambda)$ expressed in the general form $R(\lambda) = D(\lambda) + C(\lambda)A(\lambda)^{-1}B(\lambda)$, the linearizations constructed in Chapter 6 are of the form

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix}$$

where $\begin{bmatrix} M_A(\lambda) \\ K_A(\lambda) \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) \\ K_D(\lambda) \end{bmatrix}$ are (degenerate) strong block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$, respectively. A related problem is to develop the construction by considering general strong block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$ as in Definition 2.5.1. That is, strong block minimal bases linearizations of $A(\lambda)$ and $D(\lambda)$ as in Definition 2.5.1. That is, strong block minimal bases linearizations of the form $\begin{bmatrix} M_A(\lambda) & K_A^2(\lambda)^T \\ K_A^1(\lambda) & 0 \end{bmatrix}$ and $\begin{bmatrix} M_D(\lambda) & K_D^2(\lambda)^T \\ K_D^1(\lambda) & 0 \end{bmatrix}$ for $A(\lambda)$ and $D(\lambda)$, respectively. In such a case, the recovery rules for minimal bases and minimal indices from the linearization would be analogous for the right and left null spaces, as well as the recovery of eigenvectors. In addition, the linearizations could also preserve structures.

Problem 6: The results in Chapter 8 are not only important for preserving structures of rational matrices but for the particular case of structured polynomial matrices. In particular, in Sections 8.1 and 8.2, we construct structured strongly minimal linearizations of structured polynomial matrices with (skew-)Hermitian and alternating structures. An open problem is to construct strongly minimal linearizations preserving other structures of polynomial matrices as (anti-)palindromic, that are also important in applications [64]. A polynomial matrix $P(\lambda)$ is palindromic if $[P(\lambda)]^* = \operatorname{rev} P(\lambda)$, and anti-palindromic if $[P(\lambda)]^* = -\operatorname{rev} P(\lambda)$.

Appendix A Proof of Theorem 6.4.9

Proof. Let us consider a right minimal basis $\{z_i(\lambda)\}_{i=1}^s$ of $\mathcal{L}(\lambda)$ with right minimal indices $\epsilon_i = \deg z_i(\lambda)$, for $i = 1, \ldots, s$. By Lemma 2.4.5, we have that the polynomial vectors $z_i(\lambda)$ must be of the form

$$z_i(\lambda) = \begin{bmatrix} y_i(\lambda) \\ x_i(\lambda) \end{bmatrix} \quad (i = 1, \dots, s)$$

for some basis $\{x_i(\lambda)\}_{i=1}^s$ of $\mathcal{N}_r(\widehat{R})$. We notice that the vectors $x_i(\lambda)$ must be polynomial vectors, otherwise the vectors $z_i(\lambda)$ would not be polynomial vectors.

We will prove that the polynomial basis $\{x_i(\lambda)\}_{i=1}^s$ is minimal by using Theorem 2.3.3. For this purpose, let us define the polynomial matrices

$$B(\lambda) := \begin{bmatrix} z_1(\lambda) & \cdots & z_s(\lambda) \end{bmatrix}$$
 and $\widehat{B}(\lambda) := \begin{bmatrix} x_1(\lambda) & \cdots & x_s(\lambda) \end{bmatrix}$.

First, let us show that $\widehat{B}(\lambda)$ has full column rank for every $\lambda_0 \in \mathbb{F}$. Consider the unimodular (and, so, invertible at every $\lambda_0 \in \mathbb{F}$) matrix $U_A(\lambda)^{-1} = \begin{bmatrix} \widehat{N}_A(\lambda)^T & N_A(\lambda)^T \end{bmatrix}$ defined in (6.7). By (6.8), we have

$$\begin{bmatrix} M_A(\lambda_0) \\ K_A(\lambda_0) \\ -M_C(\lambda_0) \\ 0 \end{bmatrix} U_A(\lambda_0)^{-1} = \begin{bmatrix} * & A(\lambda_0) \\ I_{n\rho_A} & 0 \\ * & -C(\lambda_0) \\ 0 & 0 \end{bmatrix},$$

for every $\lambda_0 \in \mathbb{F}$. The equation above, together with (6.21), implies that the (constant) matrix

$$\begin{bmatrix} M_A(\lambda_0) \\ K_A(\lambda_0) \\ -M_C(\lambda_0) \\ 0 \end{bmatrix}$$

has full column rank for every $\lambda_0 \in \mathbb{F}$. Then, from $\mathcal{L}(\lambda_0)B(\lambda_0) = 0$, we obtain

$$\begin{bmatrix} M_A(\lambda_0) & M_B(\lambda_0) \\ K_A(\lambda_0) & 0 \\ -M_C(\lambda_0) & M_D(\lambda_0) \\ 0 & K_D(\lambda_0) \end{bmatrix} \begin{bmatrix} * \\ \widehat{B}(\lambda_0) \end{bmatrix} = 0,$$

where * indicates a constant matrix that is not important for the argument. By Lemma 6.4.8, we conclude that $\widehat{B}(\lambda_0)$ has full column rank.

Let us consider the highest column degree coefficient matrices of $B(\lambda)$ and $\widehat{B}(\lambda)$, which we denote by B_{hcd} and \widehat{B}_{hcd} , respectively. Let us show, next, that the matrix \widehat{B}_{hcd} has full column rank. For this purpose, let us write

$$\mathcal{L}(\lambda) = \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ K_A(\lambda) & 0 \\ \hline -M_C(\lambda) & M_D(\lambda) \\ 0 & K_D(\lambda) \end{bmatrix} =: \begin{bmatrix} M_{1A}\lambda + M_{0A} & M_{1B}\lambda + M_{0B} \\ K_{1A}\lambda + K_{0A} & 0 \\ \hline -M_{1C}\lambda - M_{0C} & M_{1D}\lambda + M_{0D} \\ 0 & K_{1D}\lambda + K_{0D} \end{bmatrix}.$$
(A.1)

Consider the unimodular matrix $\widetilde{U}_A(\lambda)^{-1} = \left[\widetilde{N}_A(\lambda)^T \quad \operatorname{rev}_{\rho_A} N_A(\lambda)^T\right]$ defined in (6.12). By (6.14), we have

$$\begin{bmatrix} \operatorname{rev}_1 M_A(0) \\ \operatorname{rev}_1 K_A(0) \\ -\operatorname{rev}_1 M_C(0) \\ 0 \end{bmatrix} \widetilde{U}_A(0)^{-1} = \begin{bmatrix} M_{1A} \\ K_{1A} \\ -M_{1C} \\ 0 \end{bmatrix} \widetilde{U}_A(0)^{-1} = \begin{bmatrix} * & \operatorname{rev}_{\rho_A+1} A(0) \\ I_{n\rho_A} & 0 \\ * & -\operatorname{rev}_{\rho_A+1} C(0) \\ 0 & 0 \end{bmatrix}.$$

The above equation, together with (6.22), implies that the matrix

$$\begin{bmatrix} M_{1A} \\ K_{1A} \\ -M_{1C} \\ 0 \end{bmatrix}$$

has full column rank. Moreover, from Lemma 6.4.5, we obtain

$$z_i(\lambda) = \begin{bmatrix} y_i(\lambda) \\ x_i(\lambda) \end{bmatrix} = \begin{bmatrix} y_{\epsilon_i} \\ x_{\epsilon_i} \end{bmatrix} \lambda^{\epsilon_i} + \text{ lower degree terms, with } x_{\epsilon_i} \neq 0.$$

Hence, from $\mathcal{L}(\lambda)z_i(\lambda) = 0$, we get

$$\begin{bmatrix} M_{1A} \\ K_{1A} \\ -M_{1C} \\ 0 \end{bmatrix} y_{\epsilon_i} + \begin{bmatrix} M_{1B} \\ 0 \\ M_{1D} \\ K_{1D} \end{bmatrix} x_{\epsilon_i} = 0,$$

which implies

$$y_{\epsilon_i} = -\underbrace{\begin{bmatrix} M_{1A} \\ K_{1A} \\ -M_{1C} \\ 0 \end{bmatrix}}^{\mathsf{T}} \begin{bmatrix} M_{1B} \\ 0 \\ M_{1D} \\ K_{1D} \end{bmatrix}}_{=:E} x_{\epsilon_i} \quad (i = 1, \dots, s),$$

where \dagger denotes the pseudoinverse operation. Thus, we have $B_{\text{hcd}} = \begin{bmatrix} E\widehat{B}_{\text{hcd}} \\ \widehat{B}_{\text{hcd}} \end{bmatrix}$. Therefore, the matrix \widehat{B}_{hcd} must have full column rank since, otherwise, the matrix B_{hcd} would not have full column rank.

By Theorem 2.3.3, we conclude that $\{x_i(\lambda)\}_{i=1}^s$ is a right minimal basis of $\widehat{R}(\lambda)$. Moreover, since $\epsilon_i = \deg z_i(\lambda) = \deg x_i(\lambda)$, for $i = 1, \ldots, s$, the right minimal indices of $\widehat{R}(\lambda)$ are equal to the right minimal indices of $\mathcal{L}(\lambda)$. This establishes the first statement in part (a) and in part (b).

Finally, let us prove that $x_i(\lambda) = N_D(\lambda)^T u_i(\lambda)$, for some right minimal basis $\{u_i(\lambda)\}_{i=1}^s$ of $R(\lambda)$. First, from Lemma 6.4.2, we get $x_i(\lambda) = N_D(\lambda)^T u_i(\lambda)$, for $i = 1, \ldots, s$, for some basis $\{u_i(\lambda)\}_{i=1}^s$ of $\mathcal{N}_r(R)$. Since $N_D(\lambda)$ is a minimal basis, by [38, Main Theorem, part 4], the vectors $u_i(\lambda)$ must be polynomial vectors.

We will show that the polynomial basis $\{u_i(\lambda)\}_{i=1}^s$ is minimal by using Theorem 2.3.3. For this purpose, let us define

$$\widetilde{B}(\lambda) := \begin{bmatrix} u_1(\lambda) & \cdots & u_s(\lambda) \end{bmatrix}.$$

Clearly, we have $\widehat{B}(\lambda) = N_D(\lambda)^T \widetilde{B}(\lambda)$. Since $\widehat{B}(\lambda_0)$ has full column rank for every $\lambda_0 \in \mathbb{F}$, because $\{x_i(\lambda)\}_{i=1}^s$ is a right minimal basis of $\widehat{R}(\lambda)$, we obtain that $\widetilde{B}(\lambda_0)$ must also have full column rank for every $\lambda_0 \in \mathbb{F}$. Next, let us denote by \widetilde{B}_{hcd} the highest column degree coefficient matrix of \widetilde{B} . Since $N_D(\lambda)$ is a minimal basis with all its row degrees equal, the highest column degree coefficient of $N_D(\lambda)^T$ is its leading coefficient matrix, which we denote by N_{ρ_D} . Then, we have $\widehat{B}_{hcd} = N_{\rho_D} \widetilde{B}_{hcd}$. Since \widehat{B}_{hcd} has full column rank, so does \widetilde{B}_{hcd} .

By Theorem 2.3.3, we conclude that $\{u_i(\lambda)\}_{i=1}^s$ is a right minimal basis of $R(\lambda)$. Moreover, by [38, Main Theorem, part 5] and the fact that $N_D(\lambda)$ is a minimal basis with all its row degrees equal to ρ_D , we have

$$\deg x_i(\lambda) = \rho_D + \deg u_i(\lambda) \quad (i = 1, \dots, s),$$

which shows that the right minimal indices of $R(\lambda)$ are equal to $\epsilon_1 - \rho_D \leq \cdots \leq \epsilon_s - \rho_D$. This concludes the proof.

Appendix B

Proof of Theorem 6.4.11

Proof. Let us consider a left minimal basis $\{z_i(\lambda)^T\}_{i=1}^t$ of $\mathcal{L}(\lambda)$ with left minimal indices $\eta_i = \deg z_i(\lambda)$, for $i = 1, \ldots, t$. By Lemma 2.4.5, the polynomial vector $z_i(\lambda)^T$ must be of the form

$$z_i(\lambda)^T = \begin{bmatrix} y_i(\lambda)^T & x_i(\lambda)^T \end{bmatrix} \quad (i = 1, \dots, t),$$

for some basis $\{x_i(\lambda)^T\}_{i=1}^t$ of the left nullspace of $\widehat{R}(\lambda)$. We observe the vectors $x_i(\lambda)^T$ must be polynomial vectors because the vectors $z_i(\lambda)^T$ are polynomial.

We will prove that the polynomial basis $\{x_i(\lambda)^T\}_{i=1}^t$ is a minimal basis by using the characterization in Theorem 2.3.3. With this goal in mind, let us introduce the polynomial matrices

$$X(\lambda) := \begin{bmatrix} x_1(\lambda)^T \\ \vdots \\ x_t(\lambda)^T \end{bmatrix} \quad \text{and} \quad Z(\lambda) := \begin{bmatrix} z_1(\lambda)^T \\ \vdots \\ z_t(\lambda)^T \end{bmatrix} = \begin{bmatrix} y_1(\lambda)^T & x_1(\lambda)^T \\ \vdots & \vdots \\ y_t(\lambda)^T & x_t(\lambda)^T \end{bmatrix} = : \begin{bmatrix} Y(\lambda) & X(\lambda) \end{bmatrix}.$$

Let us show, first, that the polynomial matrix $X(\lambda)$ has full row rank for every $\lambda_0 \in \mathbb{F}$. For this purpose, consider the unimodular matrices $U_A(\lambda)^{-1}$ and $U_D(\lambda)^{-1}$ defined in (6.7). By (6.8), we have that the matrix

$$\begin{bmatrix} M_A(\lambda_0) & M_B(\lambda_0) \\ K_A(\lambda_0) & 0 \end{bmatrix} \begin{bmatrix} U_A(\lambda_0)^{-1} & 0 \\ 0 & U_D(\lambda_0)^{-1} \end{bmatrix} = \begin{bmatrix} * & A(\lambda_0) & * & B(\lambda_0) \\ I_{n\,\rho_A} & 0 & 0 & 0 \end{bmatrix}$$

has full row rank for every $\lambda_0 \in \mathbb{F}$ because of (6.23). Hence, $\begin{bmatrix} M_A(\lambda_0) & M_B(\lambda_0) \\ K_A(\lambda_0) & 0 \end{bmatrix}$ has full row rank for every $\lambda_0 \in \mathbb{F}$. Then, from $Z(\lambda_0)\mathcal{L}(\lambda_0) = 0$, we obtain

$$\begin{bmatrix} Y(\lambda_0) & X(\lambda_0) \end{bmatrix} \begin{bmatrix} M_A(\lambda_0) & M_B(\lambda_0) \\ K_A(\lambda_0) & 0 \\ \hline -M_C(\lambda_0) & M_D(\lambda_0) \\ 0 & K_D(\lambda) \end{bmatrix} = 0.$$

By Lemma 6.4.8, we conclude that $X(\lambda_0)$ has full row rank for every $\lambda_0 \in \mathbb{F}$.

Let Z_{hrd} and X_{hrd} be the highest row degree matrix coefficients of $Z(\lambda)$ and $X(\lambda)$, respectively. Let us show that the matrix X_{hrd} has full row rank. Consider the unimodular matrices $\widetilde{U}_A(\lambda)^{-1}$ and $\widetilde{U}_D(\lambda)^{-1}$ defined in (6.12). By (6.14), we have that

$$\begin{bmatrix} \operatorname{rev}_1 M_A(0) & \operatorname{rev}_1 M_B(0) \\ \operatorname{rev}_1 K_A(0) & 0 \end{bmatrix} \begin{bmatrix} \widetilde{U}_A(0)^{-1} & 0 \\ 0 & \widetilde{U}_D(0)^{-1} \end{bmatrix} = \begin{bmatrix} * & \operatorname{rev}_{\rho_A+1} A(0) & * & \operatorname{rev}_{\rho_D+1} B(0) \\ I_n \rho_A & 0 & 0 & 0 \end{bmatrix}$$

has full row rank for every $\lambda_0 \in \mathbb{F}$ because of (6.24). Hence, using the notation introduced in (A.1), we have that the matrix $\begin{bmatrix} \operatorname{rev}_1 M_A(0) & \operatorname{rev}_1 M_B(0) \\ \operatorname{rev}_1 K_A(0) & 0 \end{bmatrix} = \begin{bmatrix} M_{1A} & M_{1B} \\ K_{1A} & 0 \end{bmatrix}$ has full row rank. Thus, Lemma 6.4.5 implies

$$z_i(\lambda)^T = \begin{bmatrix} y_i(\lambda)^T & x_i(\lambda)^T \end{bmatrix} = \begin{bmatrix} y_{\eta_i}^T & x_{\eta_i}^T \end{bmatrix} \lambda^{\eta_i} + \text{ lower degree terms, with } x_{\eta_i} \neq 0.$$

Then, from $z_i(\lambda)^T \mathcal{L}(\lambda) = 0$, we obtain

$$\begin{bmatrix} y_{\eta_i}^T & x_{\eta_i}^T \end{bmatrix} \begin{bmatrix} M_{1A} & M_{1B} \\ K_{1A} & 0 \\ -M_{1C} & M_{1D} \\ 0 & K_{1D} \end{bmatrix} = 0.$$

Since the matrix $\begin{bmatrix} M_{1A} & M_{1B} \\ K_{1A} & 0 \end{bmatrix}$ has full row rank, we have

$$y_{\eta_i}^T = -x_{\eta_i}^T \underbrace{ \begin{bmatrix} -M_{1C} & M_{1D} \\ 0 & K_{1D} \end{bmatrix} \begin{bmatrix} M_{1A} & M_{1B} \\ K_{1A} & 0 \end{bmatrix}^{\dagger}}_{=:F},$$

where \dagger indicates the pseudoinverse operation. Therefore, $Z_{hrd} = \begin{bmatrix} X_{hrd}F & X_{hrd} \end{bmatrix}$. Conclusively, the matrix X_{hrd} has full row rank because Z_{hrd} has full row rank.

From Theorem 2.3.3, we get that $\{x_i(\lambda)^T\}_{i=1}^t$ is a left minimal basis of $\widehat{R}(\lambda)$. Moreover, by Lemma 6.4.5, we have $\eta_i = \deg z_i(\lambda) = \deg x_i(\lambda)$, for $i = 1, \ldots, t$. Therefore, $\mathcal{L}(\lambda)$ and $\widehat{R}(\lambda)$ have the same left minimal indices. This establishes the first statement in part (a) and in part (b).

By Lemma 6.4.2, the vector $x_i(\lambda)^T$ must be of the form

$$x_i(\lambda)^T = \begin{bmatrix} u_i(\lambda)^T & w_i(\lambda)^T \end{bmatrix} \quad (i = 1, \dots, t),$$

for some basis $\{u_i(\lambda)^T\}_{i=1}^t$ of the left nullspace of $R(\lambda)$. We notice that the vectors $u_i(\lambda)$ must be polynomial vectors because the $x_i(\lambda)$ vectors are polynomial. Our

final goals are, first, to show that $\{u_i(\lambda)^T\}_{i=1}^T$ is a left minimal basis of $R(\lambda)$ and, second, to show that $\deg u_i(\lambda) = \eta_i$, for $i = 1, \ldots, t$.

We begin by noticing that if we combine Lemmas 2.4.5 and 6.4.2, we get that the vector $z_i(\lambda)^T$ must be of the form

$$z_i(\lambda)^T = \begin{bmatrix} \alpha_{1i}(\lambda)^T & \alpha_{2i}(\lambda)^T & u_i(\lambda)^T & w_i(\lambda)^T \end{bmatrix} \quad (i = 1, \dots, t),$$

for some vectors $\alpha_{ji}(\lambda)$, with j = 1, 2, conformable with the partition of $\mathcal{L}(\lambda)$. We claim that $\{ [\alpha_{1i}(\lambda)^T \quad u_i(\lambda)^T] \}_{i=1}^t$ is a left minimal basis of the polynomial system matrix $P(\lambda) = \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix}$ with left minimal indices $\eta_1 \leq \cdots \leq \eta_t$. To see this, first, from $z_i(\lambda)^T \mathcal{L}(\lambda) = 0$, we obtain

$$\begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix} \begin{bmatrix} M_A(\lambda) & M_B(\lambda) \\ -M_C(\lambda) & M_D(\lambda) \end{bmatrix} + \begin{bmatrix} \alpha_{2i}(\lambda)^T & w_i(\lambda)^T \end{bmatrix} \begin{bmatrix} K_A(\lambda) & 0 \\ 0 & K_D(\lambda) \end{bmatrix} = 0.$$
(B.1)

Multiplying (B.1) on the right by diag $(N_A(\lambda)^T, N_D(\lambda)^T)$ yields

$$\begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix} \begin{bmatrix} A(\lambda) & B(\lambda) \\ -C(\lambda) & D(\lambda) \end{bmatrix} = 0.$$
(B.2)

Hence, the polynomial vector $\begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix} \in \mathcal{N}_{\ell}(P).$

Next, let us consider the polynomial matrix

$$\widehat{U}(\lambda) := \begin{bmatrix} \alpha_{11}(\lambda)^T & u_1(\lambda)^T \\ \vdots & \vdots \\ \alpha_{1t}(\lambda)^T & u_t(\lambda)^T \end{bmatrix} =: \begin{bmatrix} A_1(\lambda) & U(\lambda) \end{bmatrix}$$

From (B.2), $\widehat{U}(\lambda)P(\lambda) = 0$. Let us show that $\widehat{U}(\lambda_0)$ has full row rank for every $\lambda_0 \in \mathbb{F}$. From $Z(\lambda_0)\mathcal{L}(\lambda_0) = 0$, we obtain

$$\begin{bmatrix} \widehat{U}(\lambda_0) & * \end{bmatrix} \begin{bmatrix} M_A(\lambda_0) & M_B(\lambda_0) \\ -M_C(\lambda_0) & M_D(\lambda_0) \\ \hline K_A(\lambda_0) & 0 \\ 0 & K_D(\lambda_0) \end{bmatrix} = 0.$$

Since diag $(K_A(\lambda_0), K_D(\lambda_0))$ has full row rank for every $\lambda_0 \in \mathbb{F}$ (because $K_A(\lambda)$ and $K_D(\lambda)$ are both minimal basis), we conclude, by Lemma 6.4.8, that the matrix $\widehat{U}(\lambda_0)$ has full row rank for every $\lambda_0 \in \mathbb{F}$.

Let us denote by \widehat{U}_{hrd} the highest row degree coefficient matrix of U_{hrd} , and show that \widehat{U}_{hrd} has full row rank. Since $K_A(\lambda)$ and $K_D(\lambda)$ are minimal bases with constant row degrees (equal to 1), by [38, Main Theorem, part 5], we get

$$\deg \begin{bmatrix} \alpha_{2i}(\lambda)^T & w_i(\lambda)^T \end{bmatrix} \begin{bmatrix} K_A(\lambda) & 0\\ 0 & K_D(\lambda) \end{bmatrix} = 1 + \deg \begin{bmatrix} \alpha_{2i}(\lambda)^T & w_i(\lambda)^T \end{bmatrix}.$$

In addition, by (B.1), we have that

$$\begin{bmatrix} \alpha_{2i}(\lambda)^T & w_i(\lambda)^T \end{bmatrix} \begin{bmatrix} K_A(\lambda) & 0\\ 0 & K_D(\lambda) \end{bmatrix} = -\begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix} \begin{bmatrix} M_A(\lambda) & M_B(\lambda)\\ -M_C(\lambda) & M_D(\lambda) \end{bmatrix}$$

and, therefore,

$$1 + \deg \begin{bmatrix} \alpha_{2i}(\lambda)^T & w_i(\lambda)^T \end{bmatrix} \le 1 + \deg \begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix}.$$

Hence, we have

$$\deg \begin{bmatrix} \alpha_{2i}(\lambda)^T & w_i(\lambda)^T \end{bmatrix} \le \deg \begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix}.$$
 (B.3)

Thus, from $Z(\lambda)\mathcal{L}(\lambda) = 0$, we obtain

$$\begin{bmatrix} \widehat{U}_{hrd} & * \end{bmatrix} \begin{bmatrix} M_{1A} & M_{1B} \\ -M_{1C} & M_{1D} \\ \hline K_{1A} & 0 \\ 0 & K_{1D} \end{bmatrix} = 0.$$

Since the matrix diag(K_{1A}, K_{1D}) has full row rank (because $K_A(\lambda)$ and $K_D(\lambda)$ are both minimal bases with constant row degrees), by Lemma 6.4.8, we have that the matrix \widehat{U}_{hrd} has also full row rank. Conclusively, since dim $\mathcal{N}_{\ell}(\mathcal{L}) = \dim \mathcal{N}_{\ell}(P)$ (becasue $\mathcal{L}(\lambda)$ is a linearization of $P(\lambda)$ in the classical matrix polynomial sense), Theorem 2.3.3 implies that $\{ [\alpha_{1i}(\lambda)^T \ u_i(\lambda)^T] \}_{i=1}^t$ is a left minimal basis of $P(\lambda)$, as we claimed. Moreover, by (B.3), we have $\eta_i = \deg z_i(\lambda)^T = \deg [\alpha_{1i}(\lambda)^T \ u_i(\lambda)^T]$. Hence, the left minimal indices of $P(\lambda)$ are also equal to $\eta_1 \leq \cdots \leq \eta_t$.

After this small detour, we are ready to prove that $\{u_i(\lambda)^T\}_{i=1}^t$ is a minimal basis of $R(\lambda)$, with minimal indices $\eta_1 \leq \cdots \leq \eta_t$, by using Theorem 2.3.3. To this goal, let us consider the polynomial matrix

$$U(\lambda) := \begin{bmatrix} u_1(\lambda)^T \\ \vdots \\ u_t(\lambda)^T \end{bmatrix}.$$

From $\widehat{U}(\lambda_0)P(\lambda_0) = 0$, we get

$$\begin{bmatrix} A_1(\lambda_0) & U(\lambda_0) \end{bmatrix} \begin{bmatrix} A(\lambda_0) & B(\lambda_0) \\ -C(\lambda_0) & D(\lambda_0) \end{bmatrix} = 0.$$

Since the matrix $\begin{bmatrix} A(\lambda_0) & B(\lambda_0) \end{bmatrix}$ has full row rank for every $\lambda_0 \in \mathbb{F}$, Lemma 6.4.8 implies that $U(\lambda_0)$ has also full row rank for every $\lambda_0 \in \mathbb{F}$.

Let U_{hrd} denote the highest row degree coefficient matrix of $U(\lambda)$. It remains to show that U_{hrd} has full row rank. First, let us prove that

$$\deg \begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix} = \deg u_i(\lambda)^T \quad (i = 1, \dots, t).$$
(B.4)

By contradiction, assume

 $\begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix} = \begin{bmatrix} \alpha_i^T & 0 \end{bmatrix} \lambda^{\eta_i} + \text{ lower degree terms, with } \alpha_i \neq 0.$

Since $\begin{bmatrix} \alpha_{1i}(\lambda)^T & u_i(\lambda)^T \end{bmatrix} P(\lambda) = 0$, we have that $\alpha_i^T \begin{bmatrix} \operatorname{rev}_{\rho_A+1} A(0) & \operatorname{rev}_{\rho_D+1} B(0) \end{bmatrix} = 0$. Then $\alpha_i = 0$ by (6.24), which is a contradiction. Hence \widehat{U}_{hrd} must be of the form $\widehat{U}_{hrd} = \begin{bmatrix} \mathcal{A} & U_{hrd} \end{bmatrix}$, for some matrix \mathcal{A} . Considering again that $\widehat{U}(\lambda)P(\lambda) = 0$, we have

$$\begin{bmatrix} A_1(\lambda) & U(\lambda) \end{bmatrix} \begin{bmatrix} A(\lambda) \\ -C(\lambda) \end{bmatrix} = 0 \text{ and } \begin{bmatrix} A_1(\lambda) & U(\lambda) \end{bmatrix} \begin{bmatrix} B(\lambda) \\ D(\lambda) \end{bmatrix} = 0,$$

and, therefore,

$$\begin{bmatrix} \mathcal{A} & U_{\text{hrd}} \end{bmatrix} \begin{bmatrix} \operatorname{rev}_{\rho_A+1} A(0) \\ -\operatorname{rev}_{\rho_A+1} C(0) \end{bmatrix} = 0 \text{ and } \begin{bmatrix} \mathcal{A} & U_{\text{hrd}} \end{bmatrix} \begin{bmatrix} \operatorname{rev}_{\rho_D+1} B(0) \\ \operatorname{rev}_{\rho_D+1} D(0) \end{bmatrix} = 0.$$

That is,

$$\begin{bmatrix} \mathcal{A} & U_{\text{hrd}} \end{bmatrix} \begin{bmatrix} \operatorname{rev}_{\rho_A+1} A(0) & \operatorname{rev}_{\rho_D+1} B(0) \\ -\operatorname{rev}_{\rho_A+1} C(0) & \operatorname{rev}_{\rho_D+1} D(0) \end{bmatrix} = 0.$$
(B.5)

Taking into account (B.5) and condition (6.24), Lemma 6.4.8 implies that $U_{\rm hrd}$ has full row rank.

Thus, by Theorem 2.3.3, $\{u_i(\lambda)^T\}_{i=1}^t$ is a left minimal basis of $R(\lambda)$. Moreover, by (B.4), the left minimal indices of $R(\lambda)$ are $\eta_1 \leq \cdots \leq \eta_t$, as we wanted to prove.

Appendix C Auxiliary result for Lemma 9.3.4

We prove in this appendix that the matrix

$$\begin{bmatrix} A & B \\ \hline C & D \end{bmatrix} := \begin{bmatrix} E_k^T \otimes I_k & I_{(k+1)} \otimes E_k \\ \hline F_k^T \otimes I_k & I_{(k+1)} \otimes F_k \end{bmatrix}$$

appearing in the proof of Lemma 9.3.4 can be transformed by row and column permutations to the direct sum of the following matrices :

$$M_1 \oplus M_1 \oplus M_3 \oplus M_3 \oplus \cdots \oplus M_{2k-1} \oplus M_{2k-1} \oplus N_{2k}$$

where the blocks M_k and N_k are as defined in (9.18). Let us take for example k = 3, then the matrix looks like

$$\begin{bmatrix} A & B \\ \hline C & D \end{bmatrix} := \begin{bmatrix} I_3 & E_3 & F_3 & F_3$$

There are three submatrices M_1 , M_3 and M_5 that take elements a, b, c and d in the respective blocks A, B, C and D, as indicated below

$$M_1 = \begin{bmatrix} b \end{bmatrix}, \quad M_3 = \begin{bmatrix} b & a \\ c & d \\ b \end{bmatrix}, \quad M_5 = \begin{bmatrix} b & a \\ c & d \\ b & a \\ c & d \\ c & d \\ b & a \end{bmatrix}$$

and they each start with a leading element in one of the E_3 blocks. For instance, $M_1 = [b_{10,13}]$, M_3 starts with the leading element $b_{7,9}$ in the third E_3 block, and M_5

starts with the leading element in the second E_3 block :

$$M_{1} = \begin{bmatrix} b_{10,13} \end{bmatrix},$$

$$M_{3} = \begin{bmatrix} b_{7,9} & a_{7,7} & & \\ & c_{10,7} & d_{10,14} \\ & & & b_{11,14} \end{bmatrix}, \quad M_{5} = \begin{bmatrix} b_{4,5} & a_{4,4} & & & \\ & c_{7,4} & d_{7,10} & & \\ & & & b_{8,10} & a_{8,8} & \\ & & & & c_{11,8} & d_{11,15} \\ & & & & & b_{12,15} \end{bmatrix}.$$

Notice that the $\begin{bmatrix} b & a \end{bmatrix}$ and $\begin{bmatrix} c & d \end{bmatrix}$ pairs have the same row index and that the $\begin{bmatrix} a \\ c \end{bmatrix}$ and $\begin{bmatrix} d \\ b \end{bmatrix}$ pairs have the same column index, which explains the permutation that has to be constructed to extract the matrix. Also the transitions

$$b_{7,9} \to b_{11,14}$$
, and $b_{4,5} \to b_{8,10} \to b_{12,15}$

always go down to the next diagonal element in the next E_3 block. In a similar fashion, one finds another set of submatrices M_1 , M_3 and M_5 that take elements a, b, c and d in the respective blocks A, B, C and D in a different order, as indicated below

$$M_1 = \begin{bmatrix} d \end{bmatrix}, \quad M_3 = \begin{bmatrix} d & c \\ & a & b \\ & & d \end{bmatrix}, \quad M_5 = \begin{bmatrix} d & c \\ & a & b \\ & & d & c \\ & & & a & b \\ & & & & d \end{bmatrix}$$

and they each start with a trailing element in one of the first three F_3 blocks. Finally, the remaining matrix N_6 takes elements in the blocks A, B, C and D in the following order

$$N_{6} = \begin{bmatrix} b & a & & & \\ & c & d & & \\ & & b & a & \\ & & & c & d \\ & & & & b & a \\ & & & & c & d \end{bmatrix}$$

and starts with the leading element in the leading E_3 block, and ends with the trailing element in the trailing F_3 block.

Bibliography

- E.N. Antoniou, S. Vologiannidis, Linearizations of polynomial matrices with symmetries and their applications, Electron. J. Linear Algebra, 15 (2006) 107–114.
- [2] R. Alam, N. Behera, Linearizations for rational matrix functions and Rosenbrock system polynomials, SIAM J. Matrix Anal. Appl., 37(1) (2016) 354–380.
- [3] R. Alam, N. Behera, Recovery of eigenvectors of rational matrix functions from Fiedler-like linearizations, Linear Algebra Appl., 510 (2016) 373–394.
- [4] R. Alam, N. Behera, Generalized Fielder pencils for rational matrix functions, SIAM J. Matrix Anal. Appl., 39(2) (2018) 587–610.
- [5] A. Amiraslani, R. Corless, P. Lancaster, *Linearization of matrix polynomials expressed in polynomial bases*, IMA J. Numer. Anal., 29(1) (2009) 141–157.
- [6] A. Amparan, F. M. Dopico, S. Marcaida, I. Zaballa, Strong linearizations of rational matrices, SIAM J. Matrix Anal. Appl., 39(4) (2018) 1670–1700.
- [7] A. Amparan, F. M. Dopico, S. Marcaida, I. Zaballa, *Strong linearizations of rational matrices*, extended version available as MIMS EPrint 2016.51, Manchester Institute for Mathematical Sciences, The University of Manchester, UK, 2016.
- [8] A. Amparan, F. M. Dopico, S. Marcaida, I. Zaballa, On minimal bases and indices of rational matrices and their linearizations, Linear Algebra Appl., 623 (2021) 14–67.
- [9] A. Amparan, S. Marcaida, I. Zaballa, On the structure invariants of proper rational matrices with prescribed finite poles, Linear and Multilinear Algebra, 61(11) (2013) 1464–1486.

- [10] A. Amparan, S. Marcaida, I. Zaballa, Finite and infinite structures of rational matrices: a local approach, Electron. J. Linear Algebra, 30 (2015) 196–226.
- [11] A. C. Antoulas, Approximation of Large-Scale Dynamical Systems, SIAM, Philadelphia, 2005.
- [12] T. Beelen, P. Van Dooren, An improved algorithm for the computation of Kronecker's canonical form of a singular pencil, Linear Algebra Appl., 105 (1988) 9–65.
- [13] T. Betcke, N. J. Higham, V. Mehrmann, C. Schröder, F. Tisseur, *NLEVP: a collection of nonlinear eigenvalue problems*, ACM Transactions on Mathematical Software, 39(2) (2013) 1–28.
- [14] M. I. Bueno, F. M. Dopico, S. Furtado, L. Medina, A block-symmetric linearization of odd degree matrix polynomials with optimal eigenvalue condition number and backward error, Calcolo, 55 (2018) 32:1–32:43.
- [15] M. I. Bueno, F. M. Dopico, J. Pérez, R. Saavedra, B. Zykoski, A simplified approach to Fiedler-like pencils via block minimal bases pencils, Linear Algebra Appl., 547 (2018) 45–104.
- [16] D. J. Cullen, Local system equivalence, Math. Systems Theory, 19 (1986) 67–78.
- [17] R. Das, R. Alam, Recovery of minimal bases and minimal indices of rational matrices from Fiedler-like pencils, Linear Algebra Appl., 566 (2019) 34–60.
- [18] R. Das, R. Alam, Affine spaces of strong linearizations for rational matrices and the recovery of eigenvectors and minimal bases, Linear Algebra Appl., 569 (2019) 335–368.
- [19] R. K. Das, R. Alam, Structured strong linearizations of structured rational matrices, Linear and Multilinear Algebra, 0 (2021) 1–34.
- [20] G. Demésy, A. Nicolet, B. Gralak, C. Geuzaine, C. Campos, J. E. Roman, Eigenmode computations of frequency-dispersive photonic open structures: A non-linear eigenvalue problem, submitted. Available as arXiv:1802.02363, 2018.
- [21] F. De Terán, F. M. Dopico, D. S. Mackey, *Linearizations of singular ma*trix polynomials and the recovery of minimal indices, Electron. J. Linear Algebra, 18 (2009) 371–402.

- [22] F. De Terán, F. M. Dopico, D. S. Mackey, Spectral equivalence of matrix polynomials and the index sum theorem, Linear Algebra Appl., 459 (2014) 264–333.
- [23] A. Dmytryshyn, B. Kågström, Coupled Sylvester-type matrix equations and block diagonalization, SIAM J. Matrix Anal. Appl., 36(2) (2015) 580–593.
- [24] F. M. Dopico, J. González-Pizarro, A compact rational Krylov method for large-scale rational eigenvalue problems, Numer. Linear Algebra Appl., 26 (2019) e2214 (26pp).
- [25] F. M. Dopico, P. W. Lawrence, J. Pérez, P. Van Dooren, Block Kronecker linearizations of matrix polynomials and their backward errors, Numer. Math., 140 (2018) 373–426. Extended version available as MIMS EPrint 2016.34, Manchester Institute for Mathematical Sciences, The University of Manchester, UK, 2016.
- [26] F. M. Dopico, P. W. Lawrence, J. Pérez, P. Van Dooren, Block Kronecker linearizations of matrix polynomials and their backward errors, Numer. Math., 140 (2018) 373–426.
- [27] F. M. Dopico, S. Marcaida, M. C. Quintana, Strong linearizations of rational matrices with polynomial part expressed in an orthogonal basis, Linear Algebra Appl., 570 (2019) 1–45.
- [28] F. M. Dopico, S. Marcaida, M. C. Quintana, P. Van Dooren, Local linearizations of rational matrices with application to rational approximations of nonlinear eigenvalue problems, Linear Algebra Appl., 604 (2020) 441–475.
- [29] F. M. Dopico, S. Marcaida, M. C. Quintana, P. Van Dooren, Block full rank linearizations of rational matrices, submitted. Available as arXiv:2011.00955v1.
- [30] F. M. Dopico, J. Pérez, P. Van Dooren, Structured backward error analysis of linearized structured polynomial eigenvalue problems, Math. Comp., 88 (2019) 1189–1228.
- [31] F. M. Dopico, J. Pérez, P. Van Dooren, Block minimal bases l-ifications of matrix polynomials, Linear Algebra Appl., 562 (2019) 163–204.
- [32] F. M. Dopico, M. C. Quintana, P. Van Dooren, *Linear system matrices of rational transfer functions*, to appear in "Realization and Model Reduction of Dynamical Systems. A Festschrift to honor the 70th birthday of Thanos Antoulas", Springer-Verlag. Available as arXiv:1903.05016v1.

- [33] F. M. Dopico, M. C. Quintana, P. Van Dooren, Structural backward stability in rational eigenvalue problems solved via block Kronecker linearizations, submitted. Available as arXiv:2103.16395v1.
- [34] F. M. Dopico, M. C. Quintana, P. Van Dooren, Strongly minimal selfconjugate linearizations for polynomial and rational matrices, in preparation.
- [35] M. El-Guide, A. Miedlar, Y. Saad, A rational approximation method for solving acoustic nonlinear eigenvalue problems, Eng. Anal. Bound. Elem., 111 (2020) 44-54.
- [36] H. Faßbender, P. Saltenberger, On vector spaces of linearizations for matrix polynomials in orthogonal bases, Linear Algebra Appl., 525 (2017) 59–83.
- [37] H. Faßbender, J. Pérez, N. Shayanfar. Symmetric and skew-symmetric block Kronecker linearizations, Technical report (2016). Available as arXiv:1606.01766.
- [38] G. D. Forney, Jr., Minimal bases of rational vector spaces, with applications to multivariable linear systems, SIAM J. Control 13(3) (1975) 493–520.
- [39] F. R. Gantmacher, The Theory of Matrices. Vols. 1, 2, Chelsea Publishing Co., New York, 1959.
- [40] Y. Genin, Y. Hachez, Y. Nesterov, R. Stefan, P. Van Dooren, and S. Xu. *Positivity and linear matrix inequalities*, European Journal of Control, 8(3) (2002) 275–298.
- [41] G. M. L. Gladwell, Inverse Problems in Vibration, 2nd edn., Kluwer Academic, Dordrecht, 2004.
- [42] I. Gohberg, M. A. Kaashoek, D. C. Lay, Equivalence, linearization, and decomposition of holomorphic operator functions, J. Funct. Anal., 28 (1978) 102–144.
- [43] I. Gohberg, M. Kaashoek, P. Lancaster, General theory of regular matrix polynomials and band Toeplitz operators, Integral Equations Operator Theory, 11 (1988), 776–882.
- [44] I. Gohberg, P. Lancaster, L. Rodman, *Matrix Polynomials*, SIAM Publications, 2009. Originally published: Academic Press, New York, 1982.
- [45] G. Golub, C.F. Van Loan, *Matrix Computations*, Johns Hopkins University Press, Baltimore, MD, 3rd ed., 1996.
- [46] S. Güttel, F. Tisseur, The nonlinear eigenvalue problem, Acta Numer., 26 (2017) 1–94.
- [47] S. Güttel, R. Van Beeumen, K. Meerbergen, W. Michiels, NLEIGS: A class of fully rational Krylov methods for nonlinear eigenvalue problems, SIAM J. Sci. Comput., 36(6) (2014) A2842–A2864.
- [48] C. Heij, A. Ran, F. van Schagen, Introduction to Mathematical Systems Theory: Linear Systems, Identification and Control, Birkhäuser Verlag, Basel, 2007.
- [49] G. Heinig, K. Rost, Algebraic Methods for Toeplitz-like Matrices and Operators, Operator Theory, Birkhäuser, Vol.13, 2013.
- [50] N. J. Higham, Accuracy and Stability of Numerical Algorithms, 2nd Edition, SIAM Publications, Philadelphia, 2002.
- [51] N. J. Higham, D. S. Mackey, N. Mackey, F. Tisseur, Symmetric linearizations for matrix polynomials, SIAM J. Matrix Anal. Appl., 29(1) (2006) 143–159.
- [52] N. J. Higham, R.-C. Li, F. Tisseur, Backward error of polynomial eigenproblems solved by linearization, SIAM J. Matrix Anal. Appl., 29(4) (2007) 1218–1241.
- [53] N. J. Higham, D. S. Mackey, N. Mackey, F. Tisseur, Symmetric linearizations for matrix polynomials, SIAM J. Matrix Anal. Appl., 29(1) (2006) 143–159.
- [54] R. A. Horn, C. R. Johnson, *Topics in Matrix Analysis*, Cambridge University Press, Cambridge, 1994. Corrected reprint of the 1991 original.
- [55] T. Kailath, *Linear Systems*, Prentice Hall, New Jersey, 1980.
- [56] N. P. Karampetakis, S. Vologiannidis, Infinite elementary divisors structure - preserving transformations for polynomial matrices, International J. Appl. Math. Comput. Sci., 13 (2003) 493–503.
- [57] D. Kressner, J. E. Roman, Memory-efficient Arnoldi algorithms for linearizations of matrix polynomials in Chebyshev basis, Numer. Linear Algebra Appl., 21(4) (2014) 569–588.
- [58] P. Lancaster, Lambda-Matrices and Vibrating Systems, Pergamon Press, Oxford, UK, 1966.
- [59] P. Lancaster. Strongly stable gyroscopic systems, Electron. J. Linear Algebra, 5 (1999) 53–66.

- [60] P. Lietaert, J. Pérez, B. Vandereycken, K. Meerbergen, Automatic rational approximation and linearization of nonlinear eigenvalue problems, published electronically in IMA J. Numer. Anal., (2021) doi:10.1093/imanum/draa098.
- [61] D. Lu, X. Huang, Z. Bai, Y. Su, A Padé approximate linearization algorithm for solving the quadratic eigenvalue problem with low-rank damping, Int. J. Numer. Meth. Engng., 103 (2015) 840–858.
- [62] C. C. MacDuffee, The Theory of Matrices, Chelsea Publishing Company, New York, 1946.
- [63] D. S. Mackey, N. Mackey, C. Mehl, V. Mehrmann, Vector spaces of linearizations for matrix polynomials, SIAM J. Matrix Anal. Appl., 28(4) (2006) 971–1004.
- [64] D.S. Mackey, N. Mackey, C. Mehl, V. Mehrmann, Structured polynomial eigenvalue problems: good vibrations from good linearizations, SIAM J. Matrix Anal. Appl., 28 (2006) 1029–1051.
- [65] D. S. Mackey, N. Mackey, C. Mehl, V. Mehrmann, *Möbius transformations of matrix polynomials*, Linear Algebra Appl., 470 (2015) 120–184.
- [66] V. Markine, A.D. Man, S. Jovanovic, C. Esveld, Optimal design of embedded rail structure for high-speed railway lines, Railway Engineering 2000, 3rd International Conference, London, 2000.
- [67] B. McMillan, Introduction to formal realizability theory I, Bell System Tech. J., 31 (1952) 217–279.
- [68] B. McMillan, Introduction to formal realizability theory II, Bell System Tech. J., 31 (1952) 541–600.
- [69] V. Mehrmann, H. Voss, Nonlinear eigenvalue problems: A challenge for modern eigenvalue methods, GAMM-Mitt., 27 (2004) 121–152.
- [70] V. Mehrmann, D. Watkins. Structure-preserving methods for computing eigenpairs of large sparse skew-Hamiltonian/Hamiltonian pencils, SIAM J. Sci. Comput., 22 (2001) 1905–1925.
- [71] C. B. Moler, G. W. Stewart, An algorithm for generalized matrix eigenvalue problems, SIAM J. Numer. Anal., 10 (1973) 241–256.
- [72] Y. Nakatsukasa, V. Noferini, A. Townsend, Vector spaces of linearizations for matrix polynomials: A bivariate polynomial approach, SIAM J. Matrix Anal. Appl., 38(1) (2017) 1–29.

- [73] Y. Nakatsukasa, O. Sète, L. N. Trefethen, The AAA algorithm for rational approximation, SIAM J. Sci. Comput., 40(3) (2018) A1494–A1522.
- [74] B. N. Parlett, C. Reinsch, Balancing a matrix for calculation of eigenvalues and eigenvectors, Numer. Math., 13 (1969) 293–304.
- [75] J. Pérez, M. C. Quintana, *Linearizations of rational matrices from general representations*, submitted. Available as arXiv:2003.02934v1.
- [76] L. Robol, R. Vandebril, Efficient Ehrlich-Aberth iteration for finding intersections of interpolating polynomials and rational functions, Linear Algebra Appl., 542 (2018), 282–309.
- [77] L. Robol, R. Vandebril, P. Van Dooren, A framework for structured linearizations of matrix polynomials in various bases, SIAM J. Matrix Anal. Appl., 38(1) (2017) 188–216.
- [78] H. H. Rosenbrock, State-space and Multivariable Theory, Thomas Nelson and Sons, London, 1970.
- [79] Y. Su, Z. Bai, Solving rational eigenvalue problems via linearization, SIAM J. Matrix Anal. Appl., 32(1) (2011) 201–216.
- [80] R. Van Beeumen, O. Marques, E. G. Ng, C. Yang, Z. Bai, L. Ge, O. Kononenko, Z. Li, C.-K. Ng, L. Xiao, Computing resonant modes of accelerator cavities by solving nonlinear eigenvalue problems via rational approximation, J. Comput. Phys., 374 (2018) 1031–1043.
- [81] R. Van Beeumen, K. Meerbergen, W. Michiels, Compact rational Krylov methods for nonlinear eigenvalue problems, SIAM J. Matrix Anal. Appl., 36 (2015) 820–838.
- [82] P. Van Dooren, The generalized eigenstructure problem: applications in linear system theory, PhD thesis, Katholieke Universiteit Leuven, Leuven, Belgium, 1979.
- [83] P. Van Dooren, The computation of Kronecker's canonical form of a singular pencil, Linear Algebra Appl., 27 (1979) 103–140.
- [84] P. Van Dooren, The generalized eigenstructure problem in linear system theory, IEEE Trans. Automat. Contr., 26(1) (1981) 111–129.
- [85] P. Van Dooren, A generalized eigenvalue approach for solving Riccati equations, SIAM J. Sci. Statist. Comput., 2(2) (1981) 121–135.

- [86] P. Van Dooren, Reducing subspaces : definitions, properties and algorithms, Matrix Pencils, Lecture Notes in Mathematics, Springer, 973 (1983) 58–73.
- [87] P. Van Dooren, P. Dewilde, The eigenstructure of an arbitrary polynomial matrix: computational aspects. Linear Algebra Appl., 50 (1983) 545–579.
- [88] P. Van Dooren, P. Dewilde, J. Vandewalle, On the determination of the Smith-McMillan form of a rational matrix from its Laurent expansion, IEEE Trans. Circuit Syst., 26(3) (1979) 180–189.
- [89] P. Van Dooren, F. M. Dopico, Robustness and perturbations of minimal bases, Linear Algebra Appl., 542 (2018) 246–281.
- [90] A. I. G. Vardulakis, *Linear Multivariable Control*, John Wiley and Sons, New York, 1991.
- [91] G. Verghese, Comments on 'Properties of the system matrix of a generalized state-space system', Int. J. Control, 31(5) (1980) 1007–1009.
- [92] G. Verghese, P. Van Dooren, T. Kailath, Properties of the system matrix of a generalized state-space system, Int. J. Control, 30(2) (1979) 235–243.